Network for Studies on Pensions, Aging and Retirement



SERIES ACADEMIC N ETS P A R

Essays on Institutional Investors, Portfolio Choice and Asset Prices

Kristy A.E. Jansen



Essays on Institutional Investors, Portfolio Choice, and Asset Prices

KRISTY A.E. JANSEN



Essays on Institutional Investors, Portfolio Choice, and Asset Prices

Kristy A.E. Jansen

ESSAYS ON INSTITUTIONAL INVESTORS, PORTFOLIO CHOICE, AND ASSET PRICES

Proefschrift ter verkrijging van de graad van doctor aan Tilburg University op gezag van de rector magnificus, prof. dr. W.B.H.J. van de Donk, in het openbaar te verdedigen ten overstaan van een door het college voor promoties aangewezen commissie in de Aula van de Universiteit op vrijdag 15 januari 2021 om 13.30 uur

 door

KRISTY ANNEKE ELISABETH JANSEN,

geboren te Eindhoven

Promotores:

prof. dr. B.J.M. Werker (Tilburg University)prof. dr. T.E. Nijman (Tilburg University)prof. dr. D.W.G.A. Broeders (Maastricht University)

leden promotiecommissie:

prof. dr. F.C.J.M. de Jong (Tilburg University)dr. E. Eiling (University of Amsterdam)prof. dr. R.S.J. Koijen (The University of Chicago)prof. dr. M. Brière (Paris Dauphine University)

This thesis was supported by Instituut Gak and Netspar.

©2021 Kristy Anneke Elisabeth Jansen, The Netherlands. All rights reserved. No parts of this thesis may be reproduced, stored in a retrieval system or transmitted in any form or by any means without permission of the author. Alle rechten voorbehouden. Niets uit deze uitgave mag worden vermenigvuldigd, in enige vorm of op enige wijze, zonder voorafgaande schriftelijke toestemming van de auteur.

Acknowledgements

"The important thing is not to stop questioning. Curiosity has its own reason for existence."

— Albert Einstein

This thesis is the tangible output of a very important and rewarding chapter in my professional life. Looking back, at the start of the research master about five years ago, my understanding of economics and finance was limited. Despite the acquired quantitative skills during my bachelors and masters, I did not have a great sense of the bigger picture questions. Why is it so valuable to have all these quantitative tools for the real world? How can I apply the acquired quantitative skills to answer questions that we as a society care about? By now, I think its fair to say that I am capable of producing these questions. Next to the search for big picture questions, another important lesson I learned over the past years is that asking tough questions is so much easier than coming up with satisfactory solutions. Yet, trying to come up with good solutions is what I started to like a lot about research.

All these important lessons have shaped the person who I am today: curious, critical, and with a clear direction of my future career goals in mind. Of course, I would not have become this person without the support and guidance of many people. This is the opportunity to thank all of them.

First and foremost, I would like to thank my supervisors Dirk Broeders, Theo Nijman, and Bas Werker. Dirk is a great co-author, improved my writing substantially, and has pushed me to expose myself to the outside world early on. Theo was always willing to discuss my research and showed me that there is more to life than academia. Bas is a great co-author and has always forced me to ensure that my intuition, mathematics, and codes line up. I will forever keep in mind when arriving at that stage of research where my results seem so obvious to be even interesting to others, I have truly learned something.

Second, I would like to express my gratitude to Jules van Binsbergen and Ralph Koijen. Their research papers have been a great inspiration for my own work. Jules invited me to spend two intellectually rewarding semesters at Wharton, which opened many new horizons for me. Jules pushed me to think hard about the important questions and his passion has inspired me to take on the challenge. Ralph was always available to discuss the progress of my job market paper, has been critical about my work from the very start, and forced me to go back to the bigger picture question before jumping in the nitty-gritty details of my work.

I would also like to thank to Marie Brière, Esther Eiling, and Frank de Jong for accepting to take part in my dissertation committee. I appreciate the time and effort they have spent reading my chapters and their comments have been very valuable for my current and future work. Frank has been engaged in my research early on and I very much enjoyed the productive cooperation during the course that we taught together.

Furthermore, I would like to thank all the faculty members of Tilburg University, and Martijn Boons and Jac Kragt in particular, for always having their doors open to discuss ideas. I am grateful for the financial support that I received to present my work at many places. A big thank you also goes to our amazing secretaries Loes de Groot, Marie-Cécile Kwinten, and Helma Tigchelaar for their administrative support and lending listening ears.

I would also like to express my gratitude to Instituut Gak and Netspar for funding my PhD project, as well as several international conferences and the semesters I spent abroad. In particular, I would like to thank Sylvia Hanhart for always providing reliable administrative support. On the same note, I thank Inquire Europe for providing additional financial support for two of my projects.

My gratitude also goes to Janko Gorter and Jakob de Haan. After my very first and successful job interview with Dirk, Janko gave me the opportunity to write my master thesis at the Dutch Central Bank (DNB) as an intern of the Strategy department. After finishing my masters, Jakob gave me the opportunity to spend another four years at DNB as a visiting researcher at the Research department. I have greatly benefited from the access to proprietary data, the seminars, and all the opportunities I got to present my work at various conferences. I also owe a thank you to all the colleagues that provided help in the data gathering process, and to Jasper de Boer and Jeroen Meringa in particular. A big thank you also goes to the EBO division for the great conversations we have had on a variety of topics. Furthermore, I would like to thank Iman van Lelyveld for giving me the opportunity to be part of the Data Science Hub.

I would also like to thank my co-authors Patrick Tuijp, Stefano Cassella, and Ricardo Barahona. Patrick showed me important glimpses into the industry world of which my research has greatly benefited. Stefano and Ricardo, I learned a lot from our discussions and I am very much looking forward to finalizing the first version of our paper, and hopefully many more to come.

Over the years, I also met many great fellow PhD students, both in Tilburg and during my research visits at Wharton. I specifically would like to thank Alexandr Beliakov, Pranav Desai, Maaike Diepstraten, Roberto Gómez Cram, Marco Grotteria, Emiel Jerphanion, Joren Koëter, Alejandro Lopez Lira, and Martijn de Vries. Sharing achievements, struggles, and personal experiences have motivated me to keep going and to always stay positive.

Last but not least, I would like to thank my family and friends for their unconditional love and support. A special thanks goes to my mom, José Peters, who has always pushed me to aim high, believed in me (more than I did in myself sometimes), and supports all of my endeavors. Another one to my little brother, Stef Jansen, for just being the best brother one could wish for. To my dad, Thieu Jansen: I am incredibly proud how you managed to improve your life over the past three years. To my aunt, Ellie Megens, for inspiring me to pursue a career in finance from a very young age. To my love, Sander, for your interest in everything I do, your patience when I am frustrated or even unbearable at times, and most importantly, for teaching me to believe in myself.

Thank you all!

Kristy Jansen Amsterdam, November 2020

Contents

In	trod	uction		xiii
1	Lon	ig-term	Investors and the Yield Curve	1
	1.1	Introd	uction	. 1
	1.2	Institu	tional setting - Ultimate Forward Rate (UFR)	. 6
		1.2.1	Regulatory discount curve without UFR	. 6
		1.2.2	Regulatory discount curve with UFR	. 7
		1.2.3	Impact of the UFR on the liability value	. 8
	1.3	Model		. 11
		1.3.1	The financial market	. 11
		1.3.2	Long-term investors	. 11
		1.3.3	Testable predictions holdings	. 16
		1.3.4	Model implied impact on yields	. 17
		1.3.5	Testable predictions future excess returns and yields	. 20
	1.4	Data		. 20
		1.4.1	SHS database	. 20
		1.4.2	CSDB database	. 21
		1.4.3	Supervision database	. 21
		1.4.4	The sample	. 22
		1.4.5	Summary statistics	. 23
	1.5	Empir	ical methodology changes in holdings	. 23
		1.5.1	Long-term bond holdings and the regulatory discount curve	. 23
		1.5.2	Long-term bond holdings and constraints	. 26
		1.5.3	Stock allocation and the regulatory discount curve	. 26
		1.5.4	Derivative positions	. 28
	1.6	Bond	yields and future excess returns	. 31

		1.6.1 Aggregate underfunding and future excess returns
		1.6.2 The connection between portfolio holdings and yields
		1.6.3 Demand curves P&Is including characteristics
	1.7	Conclusion
	1.8	Appendix
	А	Further details on UFR
	В	Model derivation
	С	Tables 48
2	Pen	sion Funds and Drivers of Heterogeneous Investment Strategies 51
	2.1	Introduction
	2.2	Motivating model
		2.2.1 Testable implications
	2.3	Data
		2.3.1 Pension fund returns
		2.3.2 Factor returns
	2.4	Factor exposures 68
		2.4.1 Factor exposures $\ldots \ldots \ldots$
		2.4.2 Heterogeneity in average excess returns
		2.4.3 Variance decomposition
	2.5	Drivers of factor exposures
		2.5.1 Pension funds' characteristics
		2.5.2 Institutional factors
	2.6	Heterogeneity in beliefs
		2.6.1 Remaining heterogeneity in factor exposures
		2.6.2 Implied beliefs on expected factor returns
		2.6.3 The effect via the choice of asset management firms
	2.7	Effect of the liability discount rate on factor exposures
	2.8	Conclusion
2.9 Appendix		Appendix
		A Model derivation
		B Fixed income factors
		C Total AUM by pension funds in the euro area
		D Random-Coefficients Model
		E Additional tables

3	The	Liqui	dity Premium Across Asset Classes	111	
	3.1	3.1 Introduction			
	3.2	3.2 Optimal consumption and investment with illiquid assets			
		3.2.1	Financial market	115	
		3.2.2	The investors' consumption and investment problem	116	
		3.2.3	Optimal strategies	118	
		3.2.4	Willingness to pay for liquidity: theory	120	
		3.2.5	Willingness to pay for liquidity: comparative statics	121	
	3.3	ity premium in four asset classes	128		
		3.3.1	Implications equilibrium	128	
		3.3.2	Private equity	130	
		3.3.3	Real estate	131	
		3.3.4	Corporate bonds	133	
		3.3.5	Stocks	135	
	3.4	Conclu	asion	137	
	3.5	3.5 Appendix			
		А	Proof of optimal consumption and investment strategies	139	
		В	Numerical implementation	142	
		С	Optimal consumption and asset allocation baseline model	143	
		D	Market Indexes	147	
1	Pon	sion F	und's Illiquid Assots Allocation under Liquidity and Capit	പ	
4	Reo	uirem	ents	aı 149	
	4.1	Introd	uction \ldots	149	
	4.2	Illiquie	d assets allocation: theory	152	
		4.2.1	Liquidity requirements	153	
		4.2.2	Capital requirements	156	
		4.2.3	Overall liquidity and capital requirements	158	
		4.2.4	Summary of the hypotheses	160	
	4.3	Data		161	
	4.4	Alloca	tion to illiquid assets: empirical results	168	
	-• *	4.4.1	Empirical results separate illiquid asset classes	177	
	4.5	Robus	tness check	180	
	4.6	Illiquid assets allocation other regulatory frameworks			
	4.7	Conclu	ision	183	
			· · · · · · · · · · · · · · · · · · ·		

5	A Survey of Institutional Investors' Investment and Management Decisions			ıs		
	on Illiquid Assets					
	5.1	Introd	uction	185		
	5.2	Investi	ng in illiquid assets	186		
	5.3	Pensio	n fund regulations	188		
		5.3.1	Regulations in the Netherlands	188		
		5.3.2	Regulations in Canada	190		
	5.4	Survey	voutcomes	191		
		5.4.1	Dutch fund outcomes	191		
		5.4.2	Canadian fund outcomes \ldots	197		
		5.4.3	Comparison of Dutch and Canadian outcomes	201		
	5.5	Best p	ractices	204		
	5.6	Conclu	sions	206		
	5.7	Appendix				
		А	Survey questions illiquid assets	208		
		В	Survey participants	209		
Bi	bliog	raphy		211		

Introduction

This dissertation is a collection of five independent chapters that aim to better understand the investment decisions of institutional investors. The share of institutional investors that holds and trades securities has substantially increased over the past decades. Since the eighties, the share of US equities held by institutional investors has grown by more than half, from 52 percent to approximately 80 percent (Stambaugh, 2014). This increase raises three important questions. First, what are the drivers behind the investment decisions of institutional investors? Second, how do their investment decisions subsequently affect asset prices? Third, and perhaps most importantly, how do their decisions affect the investors where we ultimately care about, namely, the households?

Regulation is one key channel that shapes the investment decisions of institutional investors (e.g. Ellul et al., 2011; Koijen and Yogo, 2015; Andonov et al., 2017). Understanding how regulation shapes investment decisions is important, because it allows us to better foresee how institutional investors will react to policy changes in the future. We can furthermore use this information to improve particular aspects of regulation, such as the design of risk-based capital requirements. For instance, if regulation creates incentives to take on additional risks, we could alter the regulation in such a way to reduce these incentives. Because regulation often creates similar incentives across institutional investors, simultaneous changes in asset demand can directly affect the prices of these assets. Therefore, in the first chapter, I study how regulation shapes investment decisions of pension funds and insurance companies, along with evidence how these decisions subsequently affect asset prices.

The investment decisions of institutional investors may also have real consequences for households. In the case of pension funds, the investment decisions have a direct impact on the retirement income of the beneficiaries. For instance, if a pension fund invests heavily in stocks, this may result in pension cuts during economic downturns. Hence, in the second chapter, I study determinants of pension funds' investment decisions, such as the liability structure and beliefs, and how their decisions ultimately affect the retirement income of the beneficiaries.

INTRODUCTION

Another prominent trend in financial markets is the increased share of illiquid assets in institutional investors' portfolios. For instance, a report by Willis Towers Watson (2019) shows that pension funds have expanded their illiquid asset allocation by on average 20 percentage points over the past decades. The liquidity of a portfolio, i.e. the ease at which assets can be traded, is crucial because it determines the extent to which investors are able to fulfill their immediate, liquid obligations (e.g. Scholes, 2000; Duffie and Ziegler, 2003). As a result, investing in illiquid assets makes liquidity management highly complex. In the third chapter, I therefore analyse how investors should optimally invest in illiquid assets when faced with liquidity shocks, and the compensation investors require for holding these illiquid assets. In the fourth chapter, I study the actual investment decisions of pension funds towards illiquid assets and relate these decisions to both liquidity and capital requirements. To better understand institutional investors' own views about illiquid assets, the last chapter studies how pension funds deal with illiquid assets by means of a survey study. I will turn to a detailed summary of each of the chapters next.

In the first chapter, 'Long-term Investors and the Yield Curve', I show that pension funds and insurance companies (henceforth: P&Is) change their bond holdings following a shift in regulation and find that, at the same time, these changes substantially affected yields. In particular, I exploit a change in the regulatory discount curve at which liabilities are evaluated that made the long-end of the discount curve less dependent on market interest rates. Following the regulatory change, I show that P&Is decreased long-term bond holdings by 42 percent on average. The decline is stronger for constrained than unconstrained P&Is. Furthermore, I show that the aggregate decline in demand resulted in an increase of long-term bond yields by 24 basis points on average. These findings have important implications for the evaluation of other regulatory or monetary policy measures, such as quantitative easing, because they show how demand shifts, and subsequently the impact on yields, depend on the regulatory framework and time-varying characteristics of investors.

In the second chapter, 'Pension Funds and Heterogeneous Investment Strategies', which is joint work with Dirk Broeders, we study the heterogeneity in investment decisions across defined-benefit pension funds. Globally, 50 percent of all occupational retirement savings is in defined-benefit pension funds (Willis Towers Watson, 2019). Yet, so far only a few studies have analysed the investment strategies of pension funds. The lack of access to comprehensive and detailed data on pension funds is the main reason for the limited number of studies. We show that pension funds make very distinct investment decisions, despite having similar objectives. Consistent with a model of mean-variance preferences over asset minus liabilities, we first show that the funding ratio, risk aversion, and liability duration affect investment strategies of pension funds. However, together these characteristics only explain 36 percent of the heterogeneity in the average returns across pension funds. The heterogeneity that remains reflects an economically important difference in expected retirement income of 16-32 percent over a 40-year accrual phase. We show that these remaining differences reflect heterogeneity in beliefs across pension funds. Pension funds reveal these differences in beliefs through their choices of asset management firms to which they delegate the implementation of their investment strategies. Our findings are important because beneficiaries are typically not free to choose their own pension fund as it comes with the job.

In the third chapter, 'The Liquidity Premium Across Asset Classes', which is joint work with Bas Werker, we study how liquidity affects the portfolio choice of investors and we compute the costs of holding these illiquid assets. We solve a flexible model that captures transactions costs and infrequencies of trading opportunities. The latter implies that investors are unable to trade the illiquid asset for an uncertain period of time. We show that illiquidity that results in suboptimal asset allocations carries low costs for the investor, whereas these costs are high when illiquidity results in suboptimal consumption. As a result, the costs of holding illiquid assets are high for short-term investors, investors who face substantial liquidity shocks, and investors who desire to allocate a substantial fraction of their wealth to illiquid assets. Looking at separate asset classes, back-of-the-envelope calculations suggest that the holding costs for private equity and real estate asset classes are low: varying between 5 and 30 basis points, whereas the ranges are higher for stocks and corporate bonds: varying between 20 and 45 basis points. Our results give important guidance to the opposing findings of empirically estimated liquidity premiums across asset classes.

In the fourth chapter, 'Pension Fund's Illiquid Asset Allocation under Liquidity and Capital Requirements', which is joint work with Dirk Broeders and Bas Werker, we study how pension funds decide to allocate towards illiquid assets. We show that liquidity and capital requirements are two key channels through which illiquid asset allocations are affected. Liquidity requirements consist of short-term pension payments and margin calls on derivative positions. Capital requirements result from regulation that requires pension funds to have sufficient capital to manage the risks they are exposed to, such as interest rate risk. Liquidity and capital requirements interact through the liability duration as a measure of a pension fund's average investment horizon. On the one hand, a pension fund with a high liability duration is less liquidity constrained as it will have to pay less pensions in the short-term.

INTRODUCTION

This allows for a higher allocation to illiquid assets. On the other hand, a high liability duration also implies more exposure to interest rate risk through the present value of its liabilities. This restricts the opportunity to invest in illiquid assets as more of the available capital is required for interest rate risk. Consistent with this intuition, we theoretically and empirically find that these two requirements create a hump-shaped impact of liability duration of the fraction of risky assets invested in illiquid assets. These findings are important to better understand the decisions that drive the allocation to illiquid assets.

In the fifth chapter, 'A Survey of Institutional Investors' Investment and Management Decisions on Illiquid Assets', which is joint work with Patrick Tuijp, we interviewed Dutch and Canadian pension funds and fiduciary managers on the investment and management decisions regarding illiquid assets to achieve better understanding of investors' own attitudes towards illiquid assets. Survey data are important to understand investors' decisions, because the process to get to an investment decision are often not fully observed by the econometrician. We find that both Dutch and Canadian survey participants indicated the risk-return tradeoff and diversification benefits as their main reasons to invest in illiquid assets. However, Dutch pension funds in our sample invest 15 percent of their portfolio in illiquid assets, while for the Canadian pension funds this is 34 percent. We put forward three reasons for the stark difference in the average illiquid asset allocations. First, the strong focus of Dutch survey participants on investment costs, which are made available to the public and higher for illiquid assets. Second, supervisory requirements that are risk-based and hence more strict for Dutch pension funds. Third, the division of Dutch pension fund assets into a liability matching portfolio and a return portfolio, which may lead to liquid assets crowding out illiquid assets. Our findings shed important light on how illiquid assets are perceived by large institutional investors that operate in different regulatory frameworks.

Summarizing, this dissertation has three main messages. First, regulation drives a large part of the investment decisions of institutional investors. Second, their investment decisions matter for asset prices. And third, their investment decisions have important welfare implications for the investors where we ultimately care about, namely, the households.

Chapter 1

Long-term Investors and the Yield Curve¹

1.1 Introduction

Over recent years academics have uncovered evidence that long-term investors, such as pension funds and insurance companies, affect yields. For instance, Domanski et al. (2017) argue that 'hunt-for-duration' behavior by German insurance companies might have amplified the decline in euro area bond yields. If liabilities are linked to market interest rates, a reduction in interest rates increases the duration gap between assets and liabilities and solvency positions decrease. To reduce their exposure to this risk, long-term investors have an incentive to buy long-term bonds and thereby reducing yields at the long-end of the curve.² Additionally, Greenwood and Vayanos (2010) and Greenwood and Vissing-Jorgensen (2018) provide suggestive evidence for regulatory effects on yields. They show that changes in regulation which increase the incentive to buy (sell) long-term bonds result in a decline (increase) in long-term yields around policy announcement days. All these findings are consistent with the preferred habitat view: clientele demand creates price pressure in bond

¹First version: May 2020. First and foremost, I would like to thank my supervisors Theo Nijman and Bas Werker for their invaluable support and guidance. I also thank Aleksandar Andonov, Jules van Binsbergen, Martijn Boons, Marie Brière, Dirk Broeders, Maria Chaderina, Stefano Cassella, Joost Driessen, Esther Eiling, Frank de Jong, Ralph Koijen, Jac Kragt, Martin van der Schans, and conference and seminar participants at the ECB Young Economist Competition, EEA, PhD Mentoring Day (Nova), Tilburg University, Netspar Pension Day, De Nederlandsche Bank, FMA doctoral consortium, FMA Special PhD session, and AFA PhD Poster session for very useful comments. I am grateful to the Dutch Central Bank for providing me with the data and acknowledge financial support from Inquire Europe. Views expressed are those of the author and do not necessarily reflect official positions of De Nederlandsche Bank.

²Similarly, Klinger and Sundaresan (2018) explain the negative 30-year US swap spreads as a result of underfunded pension plans optimally using swaps for duration hedging rather than long-term bonds.

markets.

Because of data limitations, the literature so far uses price data or aggregate holdings data alone to study the implications of the preferred habitat theory. As a result, there are two questions largely left unanswered. First, event studies based on price data alone do not reveal the *shift* in demand that is causing the price effect. In other words, how large are the shifts in demand that cause the yields to change? Second, and more importantly, price data or aggregate holdings data alone do not reveal the *motives* of long-term investors to change their bond holdings. For instance, why do investors react to regulatory changes in the first place? Which assets do they buy or sell in return? Do investors close to their solvency constraint react differently to regulatory changes compared to non-constrained ones?

I aim to answer these questions by providing direct evidence how long-term investors shifted their bond holdings following a regulatory change and how this behavior subsequently affected yields. First, I find a substantial shift in demand that is long-lasting and with large heterogeneity across institutions. Second, the aggregate shift in demand created substantial price effects on long-term yields. These findings have important implications for the evaluation of other regulatory shocks, such as QE. In preferred habitat models, the demand for long-term bonds is assumed to be exogenous. My findings show how demand shifts, and subsequently the impact on yields, depend on the regulatory framework and time-varying characteristics of investors.

My study combines holdings data and price data for institutional investors domiciled in the Netherlands. In particular, I exploit a positive shock to the regulatory discount curve at which insurers and pension funds (henceforth: P&Is) have to value their liabilities that was introduced in June 2012. The introduced regulatory discount curve made the long-end of the discount curve less dependent on market interest rates. The discount curve uses market interest rates for maturities up to 20 years, whereas interest rates for maturities exceeding 20 years are set equal to a weighted average between market interest rates and a fixed rate, the Ultimate Forward Rate (UFR). The UFR is substantially higher than market interest rates and as a result, the new regulatory discount curve reduces the value of the liabilities and its sensitivity towards changes in market interest rates.

Why does a change in the regulatory discount curve affect demand for long-term bonds? The demand for long-term bonds arises mainly in two parts: economic and regulatory hedging incentives. The long-term nature of P&Is' liabilities creates a natural preference for long-term bonds from a liability hedge perspective (Sharpe and Tint, 1990a; Campbell and Viceira, 2002). Regulatory hedging incentives are particularly important when the regulatory

framework does not fully reflect the economic state in which investors are operating. For instance, the regulatory discount curve is important in solvency assessments, as its used to estimate the solvency position and hence the financial position of P&Is. As a result, incentives to hedge the regulatory discount curve may increase if the regulatory discount curve diverges from the economic discount curve. The introduction of the UFR thus decreased regulatory hedging demand whereas economic hedging demand was unaffected. The extent to which regulatory hedging demand decreases, as I show, depends on the liability structure and solvency positions of P&Is.

I report the following key results. First, I find that P&Is that are more exposed to the regulatory change, i.e. the ones with long liability durations, decrease long-term bond holdings to a larger extent than less exposed ones. The decrease in long-term bond holdings is economically significant: The total decline in long-term bond holdings ($T \ge 30$) due to the regulatory change equals 15.30 billion, which is equivalent to a relative decrease in demand of 42 percent. Besides, to give additional support for the economic effects, P&Is in my sample invest 20 percent of their bond portfolio in Dutch government bonds, which corresponds to a decline in holdings equivalent to 22 percent of its long-term bonds outstanding. Second, I find that P&Is close to their solvency constraint decrease long-term holdings to a larger extent than non-constrained ones. Third, P&Is increase their allocation to equities following the regulatory change, with a stronger effect for more constrained P&Is. These findings are consistent with a mean-variance optimization problem in an asset liability context with regulatory constraints.

Second, I estimate the effect of the aggregate decline in long-term bond holdings on yields. To cleanly identify the effects of demand on yields, I apply an instrumental variable approach that uses the weights assigned to the UFR for different maturity buckets as instrument for changes in demand. Even though investors may have anticipated the UFR, the determinants of the shape of the UFR such as its level and the slope were unknown. I show that the change in the regulatory discount curve results in an increase of Dutch long-term bond yields by 24 basis points on average, with larger magnitudes for longer maturity bonds. My estimates are larger than the estimates found by typical event studies because P&Is did not sell everything at once, but spread out the decrease in long-term bond holdings over three to four quarters.

Finally, I connect the changes in yields to responses in the demand curves of the different investor types by using the framework of Koijen and Yogo (2019). The estimates show substantial heterogeneity in demand curves across investor types, with P&Is having strong upward sloping demand curves in prices, whereas demand curves are strongly downward

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

sloping for banks and the foreign sector. Moreover, the steepness of the upward sloping demand curves depend, as I show, on the liability duration and solvency positions of P&Is. These findings add to further understand the drivers behind the estimated price effect. Additionally, they are important for the recent intermediary asset pricing literature which directly models intermediaries and how they matter for asset prices, e.g. He and Krishnamurthy (2013). Typically, these models use only one class of intermediaries. My results highlight the importance of incorporating heterogeneity across investor types in order to understand the effects of intermediaries on asset prices, adding to findings of Greenwood et al. (2018), Timmer (2018), and Koijen and Yogo (2019).

My findings also have important policy implications. In recent years, government bond yields have not always reacted in a predictable way to macroeconomic or monetary policy announcements. For instance, long-term yields in the US remained low even as the FED initiated a series of interest rate increases away from zero starting in 2015. My findings show that regulation plays an important role in understanding these patterns. The demand for bonds by long-term investors increases when interest rates decline, reinforcing the decline in long-term yields. My results show that this reinforcing effect decreases if the valuation of assets and liabilities becomes less dependent on market interest rates. Policy makers should take the regulatory framework of long-term investors into account when analyzing the impact of conventional and unconventional monetary policies.

Related Literature

This paper contributes to the preferred-habitat theory proposed by Culbertson (1957) and Modigliani and Sutch (1966), who argue that there are investor clienteles with preferences for specific maturities, and the interest rate for a given maturity is influenced by the demand of the corresponding clientele and the supply of bonds with that maturity. Vayanos and Vila (2009) and Greenwood and Vayanos (2014) study supply effects on yields. Supply shocks positively affect yields, because supply shocks change the amount of bonds arbitrageurs are holdings and thus the duration risk they carry. Similarly, Guibaud et al. (2013) model a clientele-based yield curve. They show that an increase in the relative importance of the clientele with the longer investment horizon, i.e. the young, has two related effects: it renders long-term bonds more expensive, and it increases their optimal supply by the government. I contribute to this literature by showing direct empirical evidence in favor of the preferred habitat theory by using data on holdings and yields simultaneously.

My paper also links to the recent demand based asset pricing literature. For instance,

Koijen and Yogo (2019) propose an asset pricing model with flexible heterogeneity in asset demand across investors. Koijen et al. (2017) and Koijen et al. (2020) apply this model to study the effects of quantitative easing in the euro area. I contribute to this work by relating institutional investor specific characteristics to their demand curves. For instance, I show that the steepness of the upward sloping demand curves of P&Is depends on their liability durations and solvency positions.

This paper also links to Sen (2019), who studies interest rate risk hedging activities by US life insurers after a shift in regulation. The regulatory change leads to distorted hedging incentives due to different treatments of interest rate risk for products with similar economic exposures. Insurers underwriting products that are regulatory sensitive to interest rates do increase hedging activities, and vice versa. I contribute to this paper in two ways: (1) studying spillovers to other asset classes following a shift in hedging incentives and (2) linking changes in hedging incentives to changes in long-term yields.

Finally, this study contributes to the literature on fire sales: forced selling of assets that result in a significant price drop followed by price reversals. Coval and Stafford (2007) study securities commonly held by mutual funds that experience large outflows. Mitchell et al. (2007) study convertible bond prices around hedge fund redemptions. Pulvino (1998) considers commercial aircraft transactions for capital constrained versus unconstrained airlines. Campbell et al. (2011) study house prices for houses sold after foreclosures, or shortly after the death or bankruptcy of a seller. Finally, Ellul et al. (2011) show fire sales as a result of regulatory constraints imposed on insurance companies for corporate bond markets. I also show evidence in favor of regulatory effects on asset prices. Different from the fire sale literature, but consistent with stock additions and deletions to market indexes in e.g. Chang et al. (2015), I detect a permanent effect on prices.

The remainder of the paper is organized as follows. I start with describing the introduction of the regulatory discount curve based on the UFR in Section 1.2. Section 1.3 provides a simple model to derive testable implications of the effect of the change in the regulatory discount curve on long-term bond holdings and yields. A description of the data is given in Section 1.4. In Section 1.5, I test the empirical predictions that follow from the model by using a difference-in-difference specification, and in Section 1.6, I connect changes in holdings directly to changes in yields by using an instrumental variable approach. Section 2.8 concludes.

1.2 Institutional setting - Ultimate Forward Rate (UFR)

The regulatory discount curve is important in solvency assessments, as its used to estimate the liability value and hence the financial position of P&Is. As such, it shows whether P&Is are expected to meet nominal obligations. Important decisions are made based on the solvency positions, such as the amount of dividends paid to the shareholders or the ability to index pension rights. In the Netherlands, the Dutch Central Bank (DNB) constructs and publishes the regulatory discount curve on a monthly basis.

1.2.1 Regulatory discount curve without UFR

Prior to the end of June 2012, the regulatory discount curve was based on the euro swap curve for all maturities. Market interest rates were used until a maturity of 50 years, and interest rates with maturities beyond 50 years equaled the last observed forward rate. Because the regulatory discount curve was based on market interest rates only, the regulatory discount curve approximately equaled the economic discount curve. Formally, the regulatory discount curve was constructed as follows:³

- 1. European semi-annual swap rates from Bloomberg for the time to maturities 1 to 10, 12, 15, 25, 30, 40, and 50 years.⁴
- 2. Zero-coupon interest rates are derived from the swap rates by using bootstrapping (Veronesi, 2010).
- 3. Interest rates for which swap rates are non-observable are estimated by assuming constant forward rates, thereby iterating the following relationship:

$$(1+y^{(h)})^h = (1+y^{(h-1)})^{h-1}(1+f^h_{h-1})$$
(1.1)

where $y^{(h-1)}$ equals the interest rate for time to maturity h-1 and f_{h-1}^h the forward rate for time-to-maturity h-1 to h. For instance, the forward rate from 24 to 25 years (f_{24}^{25}) is used to derive the zero-coupon interest rates for maturities 26-29 years.

³See https://www.toezicht.dnb.nl/binaries/50-212329.pdf.

⁴Bloomberg also offers swap rates for all maturities from 1 to 30 years, and for 35 and 45 year time to maturities. However, the regulator refrained from using some of these interest rates because of less liquid markets for these maturities.

1.2.2 Regulatory discount curve with UFR

DNB announced a change in the regulatory discount curve to anticipate on the new regulatory framework of Solvency II for insurance companies on July 2, 2012. The introduced regulatory discount curve is similar to the regulatory discount curve applicable to all European insurers when Solvency II was introduced in 2016. DNB announced a similar regulatory discount curve for pension funds on September 24, 2012.

The new regulatory discount curve uses an extrapolation method based on the UFR, where the UFR is the convergence of long interest rates to a stable level. In essence, the regulatory discount curve with UFR uses market interest rates up to a maturity of 20 years, and interest rates with maturities longer than 20 years are determined by using a weighted average between market interest rates and a fixed rate, the UFR. The main argument that was used to justify the implementation of the UFR is that the market for long durations (>20 years) is fairly illiquid and only few securities with such long durations exists. As a result, the implied market interest rates were regarded unreliable: a discount curve purely based on market data is highly sensitive to supply and demand shocks, and therefore also the solvency positions of P&Is. A regulatory discount curve based on the UFR solves this issue by making long-term interest rates less dependent on market interest rates.

Formally, the discount curve with UFR is constructed as follows:

- 1. For maturities 1 to 20, zero-coupon interest rates are still derived as before.
- 2. For pension funds, forward rates exceeding maturities of 20 years are a weighted average between the market forward rate and the UFR. The weights are constant over time. As of a maturity of 60 years, forward rates equal the UFR:

$$f_{h-1}^{h,*} \begin{cases} f_{h-1}^h & 1 \le h \le 20\\ (1-w_h) \times f_{h-1}^h + w_h \times \text{UFR} & 21 \le h \le 60\\ \text{UFR} & h \ge 60 \end{cases}$$
(1.2)

For insurance companies, the regulatory discount curve is slightly different and uses the forward rate $f_{h-1}^h = f_{19}^{20}$ for all maturities $21 \le h \le 60$.

3. The weights are equal to

$$w_h = \frac{f_{h-1}^{h,SW} - f_{19}^{20}}{f_{60}^{61,SW} - f_{19}^{20}} \qquad \text{for} \qquad h = 21, ..., 60,$$
(1.3)

7

where $f_{h-1}^{h,SW}$ are the one year forward rates that follow from the Smith-Wilson method.⁵ The Smith-Wilson technique uses the following parameter values: the last liquid point which defines the start of the UFR equals 20 years, the full convergence to the UFR equals 60 years, the UFR level equals 4.2%, and the convergence parameter that defines how quickly the discount curve converges to the UFR equals $\alpha = 0.1$. Details on the parameter values for the Smith-Wilson technique are described in Appendix A. Table 1.14 shows the corresponding weights, where the weights are fixed and increase in h.

4. Zero-coupon interest rates $y^{(h)}$ are computed as follows:

$$(1+y^{(h)})^h = \prod_{n=1}^h (1+f_{n-1}^{n,*})$$
 for $h = 1, 2, \dots 120$ (1.4)

The regulatory discount curve with UFR has two important effects. First, it decreases current liability values as liabilities are discounted against higher rates. Second, it decreases the impact of interest rate changes. Figure 1.1 shows both effects. The red solid line shows the economic discount curve and the blue solid line the economic discount curve after a parallel shock in interest rates of -1%. The dashed green line and the dotted black line show the same discount curves including the UFR.

1.2.3 Impact of the UFR on the liability value

In order to show the economic effects of the UFR, I compute the liability value using both the economic and the regulatory discount curve. Figure 1.2 shows the cash flow pattern of a (fictitious) pension fund. The cash flows are the average cash flow patterns across the Dutch pension funds in my sample. The peak of the cash flow distribution is at a maturity of 20 years, reflecting the importance of the UFR as half of the cash flows materialize at maturities beyond 20 years. The cash flows allow me to compute the value of the liabilities both under the economic and regulatory discount curve. Formally, I compute:

$$L_t = \sum_{n=1}^{N} \frac{CF_n}{(1+y_t^{(n)})^n}$$
(1.5)

⁵The Smith-Wilson technique is described EIOPA 'QIS 5Riskinan paper: eiopa.europa.eu/Publications/QIS/ free interest rates Extrapolation method': ceiops-paper-extrapolation-risk-free-ratesen-20100802.pdf.

Figure 1.1. Regulatory discount curve

This graph shows the economic discount curves (solid red line) and the regulatory discount curve (dashed green line) at implementation of UFR on September 30, 2012. The graph also shows the economic (solid blue line) and regulatory (dotted black line) discount curve after a parallel shock in market interest rates of $\Delta y = -1\%$.



where CF_n are the average projected pension payments for maturity n, where $y_t^{(n)} = y_{E,t}^{(n)}$ under the economic discount curve and $y_t^{(n)} = y_{R,t}^{(n)}$ under the regulatory discount curve.

In Table 1.1, I compute the liability values for the projected pension payments for an average pension fund in my sample using the discount curve with and without UFR on September 30, 2012. Moreover, the table shows the change in the liability value after a parallel decrease in the economic discount curve of 1%. The liability value at implementation of the UFR decreases with 664 million for the average pension fund, or a decrease of 4.23%. This reflects the first effect of the UFR: a direct decrease in the liability values. The average liability value after a negative 1% parallel shift in interest rates equals 19,878 million using the economic discount curve, whereas using the regulatory discount curve this value equals 18,433 million. In other words, following the table, the increase in the value of liabilities equals 3,518 million after the negative interest rate shock pre UFR, and only increases with 2,737 million at implementation of the UFR, or a relative decrease of 28.5%. This reflects the second effect of the UFR: a dampening impact of changes in interest rates on liability values, which is much larger in magnitude than the first effect. Obviously, this effect is particularly visible looking at cash flows that materialize after 20 years in isolation. A negative 1%parallel shock in interest rates increases the value of the liabilities with 1,680 million after implementation of the UFR, which is 46.5% less than the increase in the liability value prior to the implementation of the UFR.

Figure 1.2. Cash flows of pension payments

This graph shows the cash flows of pension payments (*not* discounted) for an average pension fund in million euros.



Table 1.1. Value liabilities with and without UFR: This table shows the value of the liabilities with and without UFR for an average pension fund in my sample on September 30, 2012. The table also shows the liability value after a parallel shock in interest rates of -1%. The liability values are computed for all projected cash flows and for cash flows with maturities longer than 20 years only. The relative change computes the percentage drop in the liability value due to implementation of the UFR. The values are in million euros.

Cash flows all maturities	without UFR	with UFR	relative change
Discounted value liabilities	16360	15696	-4.23%
Discounted value liabilities $\Delta r = -1\%$	19878	18433	-7.84%
Change value liabilities	3518	2737	-28.52%
Cash flows maturities $T > 20$	without UFR	with UFR	relative change
Discounted value liabilities	6694	6030	-11.01%
Discounted value liabilities $\Delta r = -1\%$	9155	7711	-18.74%
Change value liabilities	2461	1680	-46.5%

1.3 Model

I derive my main testable predictions from a mean-variance optimization framework with liabilities, where P&Is care about both their economic and regulatory solvency constraints. Prior to the UFR, economic and regulatory hedging demand were identical, whereas after implementation of the UFR regulatory hedging demand deviated from economic hedging demand. The extent to which economic and regulatory hedging demand deviate depends on the liability structure and solvency positions of P&Is. This results in heterogeneity in the effect of the regulatory change on long-term bond holdings across P&Is.

To derive testable asset pricing implications of P&I demand for long-term bonds, I close the section by solving a simple equilibrium framework where arbitrageurs, or myopic investors, determine yields in equilibrium.

1.3.1 The financial market

The financial market consists of an equity index and a set of bonds. The equity index is denoted by S_t and its corresponding return by r_{t+1}^S . The set of bonds is denoted by B_t , where each bond is characterized by its maturity h and corresponding yield $y_t^{(h)}$.

The return on each bond is defined as:

$$r_{t+1}^{(h-1)} = \ln(\frac{P_{t+1}^{(h-1)}}{P_t^{(h)}}) = y_t^{(h)} - (h-1)[y_{t+1}^{(h-1)} - y_t^{(h)}].$$
(1.6)

The vector of bond returns is denoted by r_{t+1}^B , the return expectations by $\mathbb{E}_t[r_{t+1}^B]$, and the variance-covariance matrix by $\operatorname{Var}_t[r_{t+1}^B]$. I assume that the bond returns are imperfectly correlated, whereas the equity index and the set of bonds are uncorrelated. Furthermore, I assume throughout that the yield curve can be determined using this set of bonds.

1.3.2 Long-term investors

The wealth of the long-term investor evolves as follows:

$$A_{t+1} = \left(R_f + w_t^S(r_{t+1}^S - r_f) + w_t^{B'}(r_{t+1}^B - r_f)\right)A_t,$$
(1.7)

where R_f equals the gross risk-free interest rate, w_t^S the portfolio weight to stocks and w_t^B the vector of portfolio weights to the bonds.

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

For the liabilities, I assume P&Is have to pay out a fixed set of cash flows CF_t , where each cash flow is characterized by its maturity h. I also assume that P&Is have a large enough number of participants such that idiosyncratic longevity risk is fully diversified. Large projected cash flows for long maturities h implies that liabilities have to be paid out in the more distant future.

Finance theory implies that risk-free market interest rates are the applicable discount for guaranteed benefits to exclude arbitrage (e.g. Brown and Wilcox, 2009; Novy-Marx and Rauh, 2009). Hence, the economic value of the liabilities at time t equals:

$$L_t^E = \sum_h CF_t^{(h)} \exp(-hy_t^{(h)}).$$
(1.8)

A first-order Taylor expansion in $hy_t^{(h)}$ results in the following economic value of the liabilities at time t + 1:

$$L_{t+1}^{E} = \sum_{h} CF_{t}^{(h)} \exp(-hy_{t}^{(h)}) \left(1 + y_{t}^{(h)} - (h-1)(y_{t+1}^{(h-1)} - y_{t}^{(h)})\right)$$

= $a_{t}^{\prime} R_{t+1}^{B} L_{t}^{E},$ (1.9)

where

$$a_t^{(h)} = \frac{CF_t^{(h)} \exp(-hy_t^{(h)})}{L_t^E}.$$
(1.10)

Notice that the discounted cash flows are P&I specific, so that $a_t^{(h)}$ is high for long maturities h if the corresponding cash flow $CF_t^{(h)}$ is large relative to L_t^E .

The regulatory value of the liabilities is similar to its economic counterpart, except that for long maturities the regulatory discount curve is less sensitive to market interest rates. The sensitivity of the regulatory discount curve to market interest rates is defined by ξ_L , where ξ_L has the same length as the set of bonds and $0 \leq \xi_L^{(h)} \leq 1$ for all h. This means that the economic and regulatory value of the liabilities are identical if $\xi_L^{(h)} = 1$ for all h, as was the case prior to implementation of the UFR. If on the other hand $\xi_L^{(h)} = 0$ for all h, the regulatory value of the liabilities is insensitive to changes in interest rates. An example is a regulatory discount curve that uses a fixed rate for all maturities. In case of the specific event I look at here, the introduction of the UFR, we have that $\xi_L^{(h)} = 1$ for $h \leq 20$ and $\xi_L^{(h)} < 1$ for h > 20. The regulatory value of the liabilities thus evolves as:

$$L_{t+1}^R = (\xi_L \circ a_t)' R_{t+1}^B L_t^R, \tag{1.11}$$

where (\circ) is the Hadamard product.

I furthermore assume P&Is have mean-variance preferences over the assets minus liabilities, or the surplus, similar to Sharpe and Tint (1990a) and Hoevenaars et al. (2008). Following Koijen and Yogo (2015), I also assume P&Is care about the volatility in the regulatory funding ratio. Important decisions are made based on funding positions of P&Is, such as the amount of dividends paid to shareholders or the ability to index pension rights. The optimization problem of P&Is equals:

$$\max_{w_{t}} \quad \mathbb{E}\left[u\left(\frac{A_{t+1}}{A_{t}} - \frac{L_{t+1}}{A_{t}}\right)\right] \\ = \arg\max_{w_{t}} \mathbb{E}\left[\frac{A_{t+1}}{A_{t}}\right] - \frac{\gamma}{2} \operatorname{Var}\left[\frac{A_{t+1}}{A_{t}} - \frac{L_{t+1}^{E}}{A_{t}}\right] - \frac{\lambda(F_{t}^{R})}{2} \operatorname{Var}\left[\frac{A_{t+1}}{A_{t}} - \frac{L_{t+1}^{R}}{A_{t}}\right], \quad (1.12)$$

subject to

$$w_t'\iota = w_t^S + w_t^B\iota \le 1, \tag{1.13}$$

$$w_t^S, w_{h,t}^B \ge 0 \qquad \forall h, \tag{1.14}$$

where γ equals the risk-aversion parameter, $F_t^R = \frac{A_t}{L_t^R}$, and $\lambda(F_t^R)$ defines the importance of the regulatory funding ratio. As in Sen (2019), I assume that the variance of the regulatory funding ratio is proportional to $\lambda(F_t^R)$, where $\lambda'(F_t^R) < 0$, or in other words P&Is care more about the regulatory funding ratios when the regulatory funding ratio is low. I additionally assume that $\lambda(F_t^R)$ is convex: P&Is care more about a decline in the regulatory funding ratio when the regulatory funding ratio is already low than in case its high.⁶

⁶To keep the model tractable, the functional form of $\lambda(F_t^R)$ is a reduced form of the strict constraint that the funding ratio should be higher than a certain threshold.

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

As I show in Appendix B, solving for w_t results in:

$$w_t^{S*} = \underbrace{\frac{\mathbb{E}_t[r_{t+1}^S - r_f] + \nu_t + \delta_t^S}{(\gamma + \lambda(F_t^R)) \operatorname{Var}_t[r_{t+1}^S]}}_{\text{speculative portfolio}},$$
(1.15)

$$w_t^{B*} = \underbrace{\mathbb{E}_t[r_{t+1}^B - r_f] + \nu_t \iota + \delta_t^B}_{\text{speculative portfolio}} + \underbrace{\frac{\gamma}{\gamma + \lambda(F_t^R)} a_t \frac{1}{F_t^E}}_{\text{economic hedging portfolio}} + \underbrace{\frac{\lambda(F_t^R)}{\gamma + \lambda(F_t^R)} (\xi_L \circ a_t) \frac{1}{F_t^R}}_{\text{regulatory hedging portfolio}},$$
(1.16)

with

$$\begin{split} w_t^{*S}, w_{h,t}^{*B} &\geq 0, \\ \delta_t^S, \delta_{h,t}^B &\geq 0, \\ \delta_t^S w_t^{S*} &= 0, \delta_{h,t}^B w_{h,t}^{B*} &= 0 \ \forall h, \end{split}$$

where ν_t equals the Lagrange multiplier for the restriction that $w'_t \iota = 1$, and δ_t the Kuhn-Tucker multipliers for the restrictions that the portfolio weights are nonnegative.

The optimal demand for stocks consists of speculative demand only, because the liabilities are valued using the yield curve and yields are assumed to be independent of the stocks. The liability hedge portfolio consists of three components: the speculative demand, the economic hedging demand, and the regulatory hedging demand. The economic (regulatory) hedging demand equals the desired hedge against changes in the economic (regulatory) liability value. The heterogeneity in demand for long-term bonds across P&Is depends on two main factors. First, the relative weights of the liabilities in the different maturity buckets. Second, the weight assigned to the economic versus regulatory hedging demand depends on the relative magnitudes of $\lambda(F_t^R)$ and γ , which is driven by solvency positions.

These results together lead to three important model implications. Throughout I indicate variables at implementation of the UFR with a plus sign $(^+)$.

Implication 1 - bond holdings

Prior to the UFR, the regulatory funding ratio exactly equals the economic funding ratio,

i.e. $F_t^E = F_t^R$, and the optimal weights are defined as:

$$w_t^{B*} = \frac{\mathbb{E}_t[r_{t+1}^B - r_f] - \nu_t \iota + \delta_t^B}{(\gamma + \lambda(F_t^E)) \operatorname{Var}_t[r_{t+1}^B]} + a_t \frac{1}{F_t^E}$$
(1.17)

Right after implementation of the UFR, the optimal holdings equal:

$$w_t^{B*+} = \frac{\mathbb{E}_t[r_{t+1}^B - r_f] - \nu_t \iota + \delta_t^B}{(\gamma + \lambda(F_t^R)) \operatorname{Var}_t[r_{t+1}^B]} + \frac{\gamma}{\gamma + \lambda(F_t^R)} a_t \frac{1}{F_t^E} + \frac{\lambda(F_t^R)}{\gamma + \lambda(F_t^R)} (\xi_L \circ a_t) \frac{1}{F_t^R}$$
(1.18)

Subtracting (1.17) from (1.18), we get the change in holdings due to the implementation of the UFR:

$$c_{t} = w_{t}^{B*+} - w_{t}^{B*} = \underbrace{\frac{\mathbb{E}_{t}[r_{t+1}^{B} - r_{f}] - \nu_{t}\iota + \delta_{t}^{B}}{\operatorname{Var}_{t}[r_{t+1}^{B}]} \left(\frac{1}{\gamma + \lambda(F_{t}^{R})} - \frac{1}{\gamma + \lambda(F_{t}^{E})}\right)}_{\text{change in speculative demand} > 0} + \underbrace{\frac{\lambda(F_{t}^{R})}{\gamma + \lambda(F_{t}^{R})} \left((\xi_{L} \circ a_{t})\frac{1}{F_{t}^{R}} - a_{t}\frac{1}{F_{t}^{E}}\right)}_{\text{change in liability hedge demand} < 0} \leq 0$$
(1.19)

Because $F_t^E < F_t^R$ we have that speculative demand increases, whereas liability hedge demand decreases. The UFR substantially decreases the sensitivity towards interest rate changes, but only leads to a small decline in current solvency positions. As a result, the decline in the liability hedge demand is stronger than the increase in the speculative demand.⁷

Furthermore, we have that $\xi_L^{(h)}$ converges to zero for long maturities, and hence P&Is with large projected cash flows $a_t^{(h)}$ in the distant future decrease long-term bond holdings to a larger extent than the ones with projected cash flows in the near future. Formally, for the change in liability hedge demand we have:

$$\lim_{h \to \infty} c_t^{(h)} = -\frac{\lambda(F_t^R)}{\gamma + \lambda(F_t^R)} a_t^{(h)} \frac{1}{F_t^E} < \lim_{h \to 0} c_t^{(h)} = -\frac{\lambda(F_t^R)}{\gamma + \lambda(F_t^R)} a_t^{(h)} \left(\frac{1}{F_t^E} - \frac{1}{F_t^R}\right)$$
(1.20)

In other words, my model predicts that P&Is with long liability durations decrease long-term bond holdings more than the ones with short liability durations.

⁷The average increase in the regulatory funding ratio is from 0.99 to 1.03, which equals the 4% difference in the change in the liability value. On the other hand, the sensitivity towards interest rates decline with 29% (see Table 1.1).

Implication 2 - bond holdings

Constrained P&Is put a larger weight on the regulatory hedging demand relative to the economic hedging demand compared to non-constrained ones, and only regulatory hedging demand is affected by the UFR. Formally, for non-constrained investors we have:

$$\lim_{\lambda(F_t^R) \to 0} c_t = 0.$$
 (1.21)

For constrained investors we have $(\xi_L^{(h)} < 1 \text{ and } F_t^E < F_t^R)$:

$$\lim_{\lambda(F_t^R) \to \infty} c_t = \left((\xi_L \circ a_t) \frac{1}{F_t^R} - a_t \frac{1}{F_t^E} \right) < 0.$$
 (1.22)

In the limit unconstrained investors do not decrease long-term bond holdings, whereas constrained P&Is do.

Implication 3 - stock holdings

My model also predicts a positive change in the stock holdings due to implementation of the UFR $(\lambda(F_t^R) < \lambda(F_t^E))$:

$$w_t^{S*+} - w_t^{S*} = \underbrace{\frac{\mathbb{E}_t[r_{t+1}^S - r_f] - \nu_t \iota + \delta_t^S}{\operatorname{Var}_t[r_{t+1}^S]} \left(\frac{1}{\gamma + \lambda(F_t^R)} - \frac{1}{\gamma + \lambda(F_t^E)}\right)}_{\text{change in speculative demand}} > 0 \tag{1.23}$$

Because the weight that is assigned to the regulatory hedge demand, $\lambda(F_t^R)$, decreases more sharply for P&Is with low funding ratios than for the ones with high funding ratios at implementation of the UFR (convexity), the speculative demand for stocks increases more for P&Is with low funding ratios (similar limits as in (1.21)-(1.22)).

1.3.3 Testable predictions holdings

The first prediction of the model is that the decrease in long-term bond holdings is stronger for P&Is with long liability durations at implementation of the UFR:

Prediction 1 - bond holdings P&Is with long liability durations decrease long-term holdings more compared to P&Is with short liability durations.

Second, the model shows that more capital constrained P&Is decrease long-term holdings to a larger extent compared to non-constrained ones because they have a stronger incentive to hedge the regulatory value of the liabilities. This leads to the second prediction:

Prediction 2 - bond holdings P&Is close to their solvency constraint decrease long-term holdings more compared to non-constrained P&Is.

Third, the model predicts that more capital constrained P&Is increase their stock holdings to a larger extent than less constrained ones:

Prediction 3 - stock holdings P&Is close to their solvency constraint increase stock holdings more compared to non-constrained P&Is.

1.3.4 Model implied impact on yields

Why would the change in long-term bond holdings have such a significant effect on yields? The regulatory change introduces a shock to the demand for long-term bonds. However, the supply of long-term bonds was not affected by the regulatory change. This implies that the interest rate risk, or duration risk, carried by other investors in the market increases. In the preferred habitat models of Vayanos and Vila (2009) and Greenwood and Vayanos (2014), these other investors are defined as the arbitrageurs.⁸ In order to carry the increased duration risk, arbitrageurs (or myopic investors) require a higher return on the long-term bond. The level, however, depends on the risk-bearing capacity of the myopic investors.

I start with describing the optimization problem of the myopic investor. The wealth of the myopic investor evolves as follows:

$$B_{t+1} = \left(R_f + \alpha'_t (r^B_{t+1} - r_f) \right) B_t.$$
(1.24)

The myopic investor has mean-variance preferences over excess returns:

$$\max_{\alpha_{t}} \mathbb{E}_{t}[\frac{B_{t+1}}{B_{t}}] - \frac{\gamma}{2} \operatorname{Var}_{t}[\frac{B_{t+1}}{B_{t}}] = R_{f} + \alpha_{t}' \mathbb{E}_{t}[r_{t+1}^{B} - r_{f}] - \frac{\gamma}{2} \alpha_{t}' \operatorname{Var}_{t}[r_{t+1}^{B}] \alpha_{t}, \quad (1.25)$$

⁸Similarly, to explain the bond pricing implications of the prepayment option in MBS in the US, Hanson (2014) assumes arbitrageurs price interest rate risk in the market.

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

Solving for α_t^* , the optimal solution to the mean-variance investors equals:

$$\alpha_t^* = \frac{\mathbb{E}_t [r_{t+1}^B - r_f]}{\gamma \text{Var}_t [r_{t+1}^B]},$$
(1.26)

First, the two set of investors in the market have to clear. This implies that the yields are determined endogenously depending on the preferences of the long-term and myopic investors. Therefore, the market clearing condition implies:

$$\alpha_t^{(h)} B_t + w_t^{(h)} A_t = Q_t^{(h)} \quad \text{for all } h.$$
 (1.27)

Plugging in the optimal solution of the myopic investor (1.26) for $\alpha_t^{(h)}$, solving for $y_t^{(h)}$ and using (1.6) results in:

$$y_t^{(h)} - r_f = \frac{(h-1)(\mathbb{E}_t[y_{t+1}^{(h-1)}] - r_f)}{h} + \frac{Q_t^{(h)} - w_t^{(h)}A_t}{B_t} \frac{\gamma(h-1)^2 \operatorname{Var}_t[y_{t+1}^{(h-1)}]}{h}$$
(1.28)

where $Q_t^{(h)} - w_t^{(h)} A_t$ is equal to the wealth of the myopic investor in maturity bucket h, i.e. $B_t^{(h)} = Q_t^{(h)} - w_t^{(h)} A_t$. Moreover, this implies that the excess return equals:

$$\mathbb{E}_t[r_{t+1}^{(h-1)}] - r_f = \frac{Q_t^{(h)} - w_t^{(h)} A_t}{B_t} \gamma(h-1)^2 \operatorname{Var}_t[y_{t+1}^{(h-1)}].$$
(1.29)

In other words, the excess return decreases in the holdings of long-term investors: the larger the holdings of the long-term investors, the less interest rate risk has to be carried by the myopic investors, which decreases expected returns.

The market clearing condition should still hold after implementation of the UFR⁹, which implies an excess return that equals:

$$\mathbb{E}_{t}^{+}[r_{t+1}^{(h-1)}] - r_{f} = \frac{Q_{t}^{(h)} - w_{t}^{(h)+}A_{t}}{B_{t}}\gamma(h-1)^{2}\operatorname{Var}_{t}^{+}[y_{t+1}^{(h-1)}]$$
(1.30)

Under the assumption that the conditional variance of future yields does not change because of the introduction of the UFR, i.e. $\operatorname{Var}_t[y_{t+1}^{(h-1)}] = \operatorname{Var}_t^+[y_{t+1}^{(h-1)}]$, subtracting (1.30)

⁹i.e.
$$\alpha_t^{(h)+} B_t + w_t^{(h)+} A_t = Q_t^{(h)}$$
from (1.29) results in:

$$\mathbb{E}_{t}^{+}[r_{t+1}^{(h-1)}] - \mathbb{E}_{t}[r_{t+1}^{(h-1)}] = \frac{(w_{t}^{(h)} - w_{t}^{(h)+})A_{t}}{B_{t}}\gamma(h-1)^{2}\operatorname{Var}_{t}[y_{t+1}^{(h-1)}]$$
$$= \frac{c_{t}^{(h)}A_{t}}{B_{t}}\gamma(h-1)^{2}\operatorname{Var}_{t}[y_{t+1}^{(h-1)}]$$
(1.31)

For the changes in yields that result from the implementation of the UFR we get:

$$y_{t}^{(h)+} - y_{t}^{(h)} = \underbrace{\frac{(h-1)(\mathbb{E}_{t}^{+}[y_{t+1}^{(h-1)}] - \mathbb{E}_{t}[y_{t+1}^{(h-1)}])}{h}}_{\text{change expectations}} + \underbrace{\frac{c_{t}^{(h)}A_{t}}{B_{t}} \frac{\gamma(h-1)^{2} \text{Var}_{t}[y_{t+1}^{(h-1)}]}{h}}_{\text{change risk-bearing capacity}} (1.32)$$

This result shows that the price impact of the regulatory change depends on changes in expectations about future yields, $\mathbb{E}_t^+[y_{t+1}^{(h-1)}] - \mathbb{E}_t[y_{t+1}^{(h-1)}]$, the demand shock, $c_t^{(h)}$, the risk-aversion parameter γ , and the wealth of the arbitrageurs, B_t , relative to the wealth of the long-term investors, A_t . Equation (1.29) and (1.32) result in the following two additional model implications.

Implication 3 - future excess returns

Future excess returns decrease if the aggregate holdings of P&Is increase:

$$\lim_{w_t^{(h)}A_t \to Q_t^{(h)}} \mathbb{E}_t[r_{t+1}^{(h-1)}] - r_f = 0,$$
(1.33)

and

$$\lim_{w_t^{(h)}A_t \to 0} \mathbb{E}_t[r_{t+1}^{(h-1)}] - r_f = \frac{Q_t^{(h)}}{B_t} \gamma(h-1)^2 \operatorname{Var}_t[y_{t+1}^{(h-1)}] > 0.$$
(1.34)

Implication 4 - yields

Under the assumption that expectations about future yields did not change or increased, i.e. $\mathbb{E}_t^+[y_{t+1}^{(h-1)}] - \mathbb{E}_t[y_{t+1}^{(h-1)}] \ge 0$, the change in yields is positive and its magnitude depends on the risk-bearing capacity of myopic investors:

$$y_t^{(h)+} - y_t^{(h)} \ge \frac{c_t^{(h)} A_t}{B_t} \frac{\gamma(h-1)^2 \operatorname{Var}_t[y_{t+1}^{(h-1)}]}{h} > 0$$
(1.35)

19

1.3.5 Testable predictions future excess returns and yields

The model predicts that future excess return increase if long-term investors hold a smaller share of long-term bond holdings, which according to my model occurs if the funding ratio is high:

Prediction 3 - future excess returns Yields and future excess bond returns are negatively related to the aggregate funding positions of P&Is.

Second, the UFR leads to a structural decrease in P&Is long-term bond holdings, which, in turn, increase the yields:

Prediction 4 - yields Yields increase due to the implementation of the UFR.

1.4 Data

In this section I describe the three data sources that I use for my analysis: the SHS database (Subsection 1.4.1), the CSDB database (Subsection 1.4.2), and the supervision database (Subsection 1.4.3).

1.4.1 SHS database

I use Dutch security holdings (SHS) data for four types of institutional investors: banks, insurance companies, investment funds, and pension funds. The investment funds mainly consist of mutual funds. All institutions that report are domiciled in the Netherlands and the regulator decides which institutions have to report, with the aim to have sufficient coverage in terms of AUM for every sector. Institutions have to report their holdings of all securities, both foreign and domestic, to the regulator on a quarterly basis.¹⁰

DNB gathers holdings data to setup, among others, the Dutch balance of payments, the international investment position, and the financial accounts, and subsequently reports the holdings data to the ECB for the setup of the aforementioned statistics at an euro area level. The data that I use is therefore also available at the euro area level. I have three main reasons to use the Dutch data instead. First, the European data starts at the end of

¹⁰All institutions report their foreign holdings on a monthly basis, whereas this is not the case for domestic holdings. However, since Dutch institutions hold significant fixed income holdings in the Netherlands, I use quarterly data to ensure data consistency.

2013 only, whereas introductions of the ultimate forward rates (UFR) in several European countries already happened in 2011 and 2012. Moreover, the European data aggregates over all sectors, whereas the Dutch data is at the institutional level. This allows me to make use of the cross-sectional variation in institutions. For instance, measuring effects of the solvency positions on holdings is only possible when there is data availability at the institutional level. Third, looking at the total AUM of pension funds alone in my database, I already cover 53 percent of the assets of pension funds in the euro area, OECD (2019).

The data provide bond and stock holdings at the International Securities Identification Number (ISIN) level. Institutions report their positions at the start of the corresponding quarter, the total purchases and sales of each position, and the positions at the end of the quarter, all in euros. For both stocks and bonds, purchases and sales are in market values. For stocks, start and end holdings are available in number of shares and market values. For bonds, start and end holdings are available in both nominal and market values.

1.4.2 CSDB database

The SHS database is linked to the Centralised Securities Database (CSDB). The aim of the CSDB database is to hold accurate information on all individual securities relevant for the statistical purposes of the European System of Central Banks, ECB (2010). From the CSDB database I obtain market relevant information: debt type, maturity dates, coupon rates, coupon frequencies, coupon type (e.g. fixed, floating or zero-coupon), last coupon payment date, yield-to-maturity, and closing price. The data from the CSDB database allows me to assign bonds in maturity buckets and compute bond durations.

1.4.3 Supervision database

The supervision database is from mandatory annual and quarterly statements that P&Is report to DNB. P&Is have to report, among others, solvency positions, the value of the assets and liabilities, liability durations, asset allocations, and derivative positions. I describe the solvency requirements for both pension funds and insurance companies in the next two subsections.

Solvency requirements pension funds

A pension fund's solvency position is assessed by computing its funding ratio, or its assets divided by its liabilities. The minimum funding requirement is a flat rate equal to a funding

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

ratio of about 104.2%. In contrast, the required funding ratio is based on a pension fund's risk profile and is calculated such that the probability that the funding ratio falls below 100 percent on a one-year horizon equals 2.5 percent. For a median pension fund this ratio amounts to a required funding ratio of 116 percent.

Solvency requirements insurance companies

Instead of funding ratios, insurance companies compute solvency ratios to assess the solvency position. Solvency ratios equal the available capital divided by the required capital. Prior to the introduction of Solvency II in 2016, solvency ratios of insurance companies were not risk-based. The required eligible capital prior to 2016 equaled 4 percent of the value of the liabilities. At the introduction of Solvency II, the required capital is computed in a similar way as for pension funds, except that the required capital is calculated such that the probability that the funding ratio falls below 100 percent on a one-year horizon equals 0.5 percent, rather than the 2.5 percent for pension funds.

Solvency ratios can be converted into funding ratios and vice versa. Because the model makes predictions based on funding ratios, I convert insurers' solvency ratios to funding ratios. Formally, prior to Solvency II regulation solvency ratios equal $SR = \frac{A-L}{0.04L}$, which implies that the funding ratio equals $\frac{A}{L} = 0.04 * SR + 1$. The solvency ratios under Solvency II are more complex and hence I collect data on the assets and the liabilities for each insurer to compute the funding ratios manually.

1.4.4 The sample

The total sample covers 20 banks, 42 pension funds, 12 life insurers, 27 non-life insurers and 160 non-equity mutual funds. This group of institutional investors represents on average 80-90 percent in terms of AUM for all institutional investors domiciled in the Netherlands.¹¹

I only analyze investors' direct holdings, that is, investments that are not made via other investor types such as mutual funds. The data, unfortunately, does not allow for a linkage between the indirect holdings of the investor to its direct holdings, except for the two largest pension funds and the two largest insurance companies. For these P&Is I know their shares in mutual funds which allows me to use both their direct and indirect holdings.

¹¹See for details on reporting requirements https://statistiek.dnb.nl/statistiek/index.aspx.

1.4.5 Summary statistics

This section briefly describes the summary statistics. Banks are the largest in terms of AUM, followed by life insurers and pension funds.¹² Life insurers and pension funds also have the highest bond duration of their fixed income portfolios. The average bond duration equals 11.3 and 10.6 years for life insurers and pension funds respectively, whereas this equals 4.3 years for banks, 7.3 for mutual funds and 6.3 for non-life insurers.

Life insurers and pension funds have the longest liability durations, equal to 11.0 and 17.8 years respectively. The liability duration of non-life insurers is much shorter and equals 4.2 years. The average funding ratio of pension funds equals 109% and 110% for insurers. The heterogeneity in the funding ratios for pension funds is substantially larger than for insurers. Insurers generally hedge their liabilities more closely than pension funds because they face costs of financial distress, whereas pension funds cannot default.¹³ Life insures and non-life insurers do not substantially deviate in terms of their solvency ratios.

1.5 Empirical methodology changes in holdings

I now turn to the testing of the empirical predictions from my theoretical framework. As the regulatory change affected the pension and insurance sector only, I focus in this section on P&Is and come back to the other investor types in Section 1.6. For bond holdings, I use the *notional* values in all my analyses such that market prices are not driving the results. A change in notional values reflects active choices by investors, which is exactly the focus of this paper.

1.5.1 Long-term bond holdings and the regulatory discount curve

I start with the main regression specification. Even though the regulatory discount curve already affected interest rates as of maturities of 21 years, I focus here on long-term bonds with maturities of 30 years or longer. The weight assigned to the UFR is relatively small for the first affected maturities and these interest rates are closely linked to the 20 year interest

¹²Note that these summary statistics are based on the *direct* holdings only.

¹³In case a pension fund is not compliant with funding requirements, it files a recovery plan to the supervisor. Recovery measures may include an increase in contributions, a reduction of the future benefit accrual rate or, as a measure of last resort, a reduction of accrued benefits.

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

Table 1.2. **Summary statistics**: This table shows the following summary statistics: total AUM of directly reported assets (AUM), AUM in bonds (AUM bonds), AUM in stocks (AUM stocks), bond duration (Bond duration), liability duration (Liability duration), and the solvency positions (Funding ratio). The funding ratio is in percentage points, AUM in million euro, bond and liability duration in years. Equity mutual funds are excluded from the sample.

AUM	mean	std.dev.	p50	AUM bonds	mean	std.dev.	p50
Banke	20 605	26 558	8 803	Banke	19 501	25 185	8 803
Life incurrent	10,035	17,605	0,035	Life incurrence	14 557	12,100	12 504
Life insurers	19,917	17,095	21,004	Life insurers	14,007	15,700	15,594
Non-life insurers	1,383	1,332	831	Non-life insurers	1,112	1,237	650
Mutual funds	763	1,422	432	Mutual funds	622	853	371
Pension funds	14,760	39,731	4,322	Pension funds	$7,\!876$	20,143	$2,\!450$
AUM stocks	mean	std.dev.	p50	Bond duration	mean	std.dev.	p50
Banks	1,104	3,291	9	Banks	4.3	3.5	3.7
Life insurers	5,360	4,644	3,743	Life insurers	11.3	3.5	11.4
Non-life insurers	271	417	117	Non-life insurers	6.3	4.1	6.2
Mutual funds	141	668	145	Mutual funds	7.3	4.8	6.8
Pension funds	6,884	$20,\!197$	$1,\!851$	Pension funds	10.6	3.7	10.4
Liability duration	mean	std.dev.	p50	Funding ratio	mean	std.dev.	p50
Life insurers	11.0	3.5	11.7	Life insurers	110	4	109
Non-life insurers	4.1	2.8	3.5	Non-life insurers	110	5	108
Pension funds	17.8	2.9	17.6	Pension funds	109	12	108

rates and hence a good substitute to hedge the 20 year interest rate.

Figure 1.3 shows the average fraction of long-term bonds in the bond portfolio for P&Is over time. Following the two quarters after the UFR was implemented, there is a sharp decline in long-term bond holdings for both life insurers and pension funds. Long-term bond holdings slightly increase again towards the end of 2014, but remain substantially lower than the pre-UFR levels.

To bring the predictions of the model to the data, I use a difference-in-difference specification which compares long-term bond holdings before and after implementation of the UFR. I exploit the heterogeneity in exposure towards the UFR, which depends on the liability duration of P&Is. I conjecture that investors with long liability durations decrease longEven though investors may have anticipated the UFR, the determinants of the shape of the UFR such as its level and the slope were unknown.¹⁴

Figure 1.3. Long-term bond holdings by institutional investor type This graph shows the average fraction of bonds with a maturity of 30 years or longer for life insurers, non-life insurers, and pension funds over the period 2009q1-2018q1.



term bond holdings more compared to investors with short liability durations:

$$w_{it}^{B} = \alpha + \beta_0 \text{UFR}_t + \beta_1 \text{UFR}_t \times D_{2012q1,i}^{L} + \beta_2 \text{FR}_{it-1} + \beta_3 D_{it-1}^{L} + \nu_i + \epsilon_{it}, \quad (1.36)$$

where UFR_t equals one after implementation of the UFR and zero otherwise, $D_{2012q1,i}^{L}$ is the liability duration as of 2012q1, FR_{it-1} the lagged funding ratio, D_{it-1}^{L} the lagged liability duration, and ν_i are fund fixed effects.

Table 1.3 shows the results. P&Is with long liability durations decrease long-term holdings to a larger extent than the ones with short liability durations, supporting the first prediction of my theoretical framework in Section 1.3. The effect is also economically significant. The average liability duration in the cross-section of P&Is equals 14 years, which implies long-term bond holdings decreased by approximately $14 \times 0.0018 = 2.5$ percent. The total decline in long-term bond holdings due to the regulatory change equals:

$$\sum_{i=1}^{N} \hat{\beta}_{1} \times D_{2012q1,i}^{L} \times AUM_{i}^{B} = 15.30 \text{ billion}, \qquad (1.37)$$

where AUM_i^B is the total AUM in bonds. The total AUM in long-term bonds prior to the regulatory change equals 36.10 billion, so this means a relative decrease of 42 percent. Besides, to give additional support for the economic effects, P&Is invest 20 percent of the total bond portfolio in Dutch government bonds on average, which means a decline in these holdings of $25\% \times 15.30 = 3.06$ billion. The average amount outstanding of 30-year Dutch government bonds equals 14 billion, and hence, the total decline corresponds to 22 percent of its amount outstanding.¹⁵

Table 1.3 also shows the changes in bond holdings with maturities varying between 15 and 25 years, and maturities less than 15 years. P&Is increased their holdings towards bonds with maturities varying between 15 and 25 years, whereas they did not change their holdings to bonds with maturities less than 15 years. These results show that P&Is moved their long-term bonds with maturities of 30 years or longer to bonds with maturities around 20 years.

1.5.2 Long-term bond holdings and constraints

My model also predicts that P&Is closer to their solvency constraint decrease long-term bond holdings to a larger extent than funded P&Is. I use a triple difference-in-difference estimation technique to test this hypothesis:

$$w_{it}^{B} = \alpha + \beta_0 \text{UFR}_t + \beta_1 \text{UFR}_t \times D_{2012q1,i}^{L} \times \text{FR}_{2012q1,i}^{-1} + \beta_2 \text{FR}_{it-1}^{-1} + \beta_3 D_{it-1}^{L} + \nu_i + \epsilon_{it}, (1.38)$$

where $FR_{2012a1,i}^{-1}$ is the inverse of the funding ratio as of 2012q1.

Table 1.4 summarizes the results. P&Is that are more constraint, i.e. have a larger inverse of their funding ratio, decrease long-term bond holdings to a larger extent: A one standard deviation increase in the inverse of the funding ratio (0.08), increases the decline in long-term bond holdings by 25 basis points, which is equivalent to 1.89 billion, or 5.26% of the longterm bond holdings. P&Is that are more constrained also increase their holdings towards bonds with maturities between 15 and 25 years to a larger extent than non-constrained ones.

1.5.3 Stock allocation and the regulatory discount curve

The final model implication for the changes in holdings is that P&Is closer to their solvency constraint allocate more of their assets to stocks at implementation of the UFR. I

 $^{^{15}}$ As a robustness check, I have also added mutual funds to estimate (1.36): Mutual funds do not have liabilities and therefore their liability durations essentially equal zero. Including mutual funds to the sample with a liability duration forced to zero does not affect the sign and the magnitude of the coefficients.

Table 1.3. Long-term bond holdings and the regulatory discount curve: This table presents the results of the main regression described in Equation (1.36): $w_{it}^B = \alpha + \beta_0 \text{UFR}_t + \beta_1 \text{UFR}_t \times D_{2012q1,i}^L + \beta_2 \text{FR}_{it-1} + \beta_3 D_{it-1}^L + \nu_i + \epsilon_{it}$, with UFR equal to 1 as of 2012Q1 and zero otherwise, $D_{2012q1,i}^L$ the duration of the liabilities as of 2012Q1, and controls include the lagged inverse of the funding ratio and the liability duration. Column (1) and (2) show the results for bond holdings with a maturity of 30 years or longer, column (3) and (4) for maturities between 15 and 25 years, and column (5) and (6) for maturities shorter than 15 years. ICL is a dummy that equals one for life insurers and zero otherwise and PF equals one for pension funds and zero otherwise. Robust standard errors are in parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	$\begin{array}{c} \text{Holdings} \\ T \geq 30 \end{array}$		Hold $15 \le T$	$\frac{1}{2} \leq 25$	Holdings $T \le 15$		
UFR	0.0115		-0.0361^{**}		0.0514		
	(0.0080)		(0.0160)		(0.0317)		
$\mathrm{UFR} \times D^L_{2012a1}$	-0.0023^{***}	-0.0018^{***}	0.0033***	0.0034^{***}	-0.0014	-0.0028	
201241	(0.0007)	(0.0007)	(0.0012)	(0.0012)	(0.0021)	(0.0022)	
1/Funding ratio	0.0418	0.0431	-0.1371^{**}	-0.1324	0.0335	0.0471	
, –	(0.0359)	(0.0480)	(0.0684)	(0.0939)	(0.1054)	(0.1447)	
Liability duration	0.0030*	0.0064***	0.0026	0.0029	-0.0085	-0.0219^{***}	
	(0.0017)	(0.0023)	(0.0027)	(0.0039)	(0.0055)	(0.0065)	
ICL	0.0370^{*}		0.0324		-0.0934		
	(0.0199)		(0.0337)		(0.0724)		
PF	0.0043		0.0187		-0.0448		
	(0.0273)		(0.0432)		(0.0899)		
Fund FE	No	Yes	No	Yes	No	Yes	
Time FE	No	Yes	No	Yes	No	Yes	
Ν	2,376	2,376	2,376	2,376	2,376	2,376	
R^2	0.11	0.61	0.15	0.66	0.16	0.73	

use the following difference-in-difference specification to test this hypothesis formally:

$$w_{it}^{S} = \alpha + \beta_0 \text{UFR}_t + \beta_1 \text{UFR}_t \times \text{FR}_{2012q1,i}^{-1} + \beta_2 \text{FR}_{it-1} + \nu_i + \epsilon_{it}.$$
 (1.39)

Table 1.5 shows that more constrained P&Is increase their stock allocation to a larger extent. A one standard deviation increase in the inverse of the funding ratio (0.08) increases the stock allocation with 2.43%, which is equivalent to an average increase of 225 million that is allocated to stocks.

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

Table 1.4. Long-term bond holdings and constraints: This table present the results of the regression described in Equation (1.38): $w_{it}^B = \alpha + \beta_0 \text{UFR}_t + \beta_1 \text{UFR}_t \times D_{2012q1,i}^L \times \text{FR}_{2012q1,i}^{-1} + \beta_3 D_{it-1}^L + \nu_i + \epsilon_{it}$, with UFR equal to 1 as of 2012Q1 and zero otherwise, $D_{2012q1,i}^L$ the duration of the liabilities as of 2012Q1, and controls include the lagged inverse of the funding ratio and the liability duration. Column (1) and (2) show the results for bond holdings with a maturity of 30 years or longer, column (3) and (4) for maturities between 15 and 25 years, and column (5) and (6) for maturities shorter than 15 years. ICL is a dummy that equals one for life insurers and zero otherwise and PF equals one for pension funds and zero otherwise. Robust standard errors are in parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	Holdings $T \ge 30$		Hold $15 \le T$	$\frac{1}{25}$	Holdings $T \le 15$		
UFR	0.0128 (0.0088)		-0.0378^{**} (0.0160)		0.0599^{*} (0.0327)		
$\mathrm{UFR} \times D^L_{2012q1} \times$	-0.0027^{***}	-0.0021^{**}	0.0039***	0.0040^{***}	-0.0023	-0.0037	
$FR_{2012q1,i}^{-1}$	(0.0009)	(0.0009)	(0.0012)	(0.0013)	(0.0025)	(0.0025)	
1 /							
1/Funding ratio	0.0341	0.0317	-0.1266^{*}	-0.1090	0.0199	0.0137	
	(0.0332)	(0.0446)	(0.0680)	(0.0948)	(0.1009)	(0.1394)	
Liability duration	0.0032^{*}	0.0065***	0.0025	0.0028	-0.0079	-0.0210^{***}	
	(0.0019)	(0.0023)	(0.0027)	(0.0038)	(0.0054)	(0.0064)	
ICL	0.0360^{*}	· /	0.0325	× ,	-0.0944	· · · ·	
	(0.0200)		(0.0336)		(0.0720)		
PF	0.0028		0.0192		-0.0493		
	(0.0271)		(0.0429)		(0.0895)		
Fund FE	No	Yes	No	Yes	No	Yes	
Time FE	No	Yes	No	Yes	No	Yes	
N	2,349	2,349	2,349	2,349	2,349	2,349	
R^2	0.11	0.61	0.15	0.67	0.15	0.72	

1.5.4 Derivative positions

The empirical analysis so far uses long-term bond holdings only. However, interest rate risk is also managed by using derivative contracts, especially for very long-term maturities. As of the start of 2012, pension funds have to report interest rate derivative positions on an aggregate level: They report the market value of interest rate derivative contracts broken down in different types of derivative contracts. Moreover, they report the values of these Table 1.5. Stock allocation and the regulatory discount curve: This table present the results of the regression described in Equation (1.39): $w_{it}^S = \alpha + \beta_0 \text{UFR}_t + \beta_1 \text{UFR}_t \times \text{FR}_{2012q1,i}^{-1} + \beta_2 \text{FR}_{it-1} + \nu_i + \epsilon_{it}$, with UFR equal to 1 as of 2012Q1 and zero otherwise, $FR_{2012q1,i}^{-1}$ the inverse of the funding ratio as of 2012Q1, and the control includes the lagged inverse of the funding ratio. Column (1) and (2) show the results for bond holdings with a maturity of 30 years or longer, column (3) and (4) for maturities between 15 and 25 years, and column (5) and (6) for maturities shorter than 15 years. ICL is a dummy that equals one for life insurers and zero otherwise and PF equals one for pension funds and zero otherwise. Robust standard errors are in parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	Equity a	llocation
UFR	-0.2073	
$\mathrm{UFR} \times \mathrm{FR}_{2012a1.i}^{-1}$	(0.1406) 0.2646^*	0.3042**
1/Funding ratio	(0.1601) -0.0289	(0.1558) 0.0751
	(0.0859)	(0.1160)
ICL	0.0773 (0.0869)	
PF	0.1919***	
	(0.0554)	
Fund FE	No	Yes
Time FE	No	Yes
N	2,407	2,407
R^2	0.16	0.86

positions after a parallel shock in interest rates of +1% (-1%) and +0.5% (-0.5%).¹⁶ This allows me to compute the dollar durations of the derivative positions.¹⁷ Because the data on derivative positions is available for only two quarters prior to the regulatory change, the time series is not long enough to statistically test if pension funds decreased their interest rate risk exposure via derivatives as well. However, using the time series of the cross-sectional average implied duration of the derivative portfolios, I show suggestive evidence that pension

 $^{^{16}}$ Unfortunately, insurance companies only started reporting derivative positions at the start of 2016 when Solvency II was introduced.

 $^{^{17}\}mathrm{As}$ the majority of interest rate derivative positions consist of swaps, and swaps have a linear pay off function, I narrow down the analysis to the swap portfolio only.

funds also substantially decreased their derivative positions after the regulatory change.

Formally, I approximate the dollar duration of the swap position as follows:

$$D_{p,t}^{\$} \approx -\frac{dV_t}{dr} = \frac{V_t^{-dr} - V_t^{+dr}}{2|dr|}$$
(1.40)

where $V_t^{-dr}(V_t^{+dr})$ is the value of the derivative portfolio after a negative (positive) change in interest rates, $D_p^{\$}$ the dollar duration of the portfolio, and dr the change in interest rates.

Figure 1.4 depicts the cross-sectional average duration implied by the swap portfolio over time, where the duration is computed as the dollar duration in (1.40) relative to total AUM. The graph also shows the total balance sheet duration as the sum of the relative implied duration of the swap portfolio and the duration of the fixed income portfolio multiplied by the allocation to fixed income. On average, pension funds have a balance sheet duration equal to 10 years. As the duration of the liabilities equals 18 years on average, this means that pension funds hedge approximately half of their interest rate risk. Importantly, the portfolio duration shows a sharp decline at the implementation of the UFR, consistent with the predictions of the model and the empirical findings for long-term bond holdings.

Figure 1.4. Implied duration of pension funds' portfolios

This graph shows the relative implied duration of the swap portfolio and the duration of the total portfolio of pension funds. The (relative) duration of the swap portfolio is determined as the implied dollar duration of the swaps divided by total pension assets. The duration of the total portfolio equals the sum of relative implied duration of the swap portfolio and the duration of the fixed income portfolio times the allocation to fixed income.



1.6 Bond yields and future excess returns

In this section, I test the empirical predictions of my model for yields and future excess returns. As opposed to the previous section, I use data on Dutch government bonds holdings only, because P&Is hold a substantial fraction of their assets in Dutch bonds and as a result, price effects will be particularly visible for this subset of bonds. I start with the first prediction: future excess returns are high when the P&I sector as a whole is underfunded. Then, I estimate the effect of the UFR on yields by using the construction of the UFR that affects yields at different maturities differently as an exogenous shock to demand. Finally, I use this construction as an instrument to estimate the effect of yields on Dutch government bond holdings for various investor types by using the framework of Koijen and Yogo (2019).

1.6.1 Aggregate underfunding and future excess returns

My model predicts that future excess returns are low if the P&I sector is underfunded. Underfunding means that P&Is have less capital than the minimal required capital. If P&Is are underfunded, their demand for long-term bonds increases and hence the interest rate risk that has to be carried by myopic investors decreases, which in turn lowers expected future returns. To test this hypothesis formally, I determine the aggregate level of underfunding as the fraction of pension funds that are underfunded relative to the total at the end of a given quarter. The data are from the website of DNB and cover the period 2007q1-2019q4.¹⁸

I use data from Bloomberg on the nominal Dutch yield curve to compute future excess returns. The log excess return is defined as in (1.6), minus the risk-free interest rate: $rx_{t+1}^{(h)} = y_t^{(h)} - (h-1)[y_{t+1}^{(h-1)} - y_t^{(h)}] - y_t^{(1)}$. Table 1.6 provides the summary statistics on the annual excess returns $rx_{t+1}^{(h)}$ for bonds with maturities 10, 20 and 30 years, together with summary statistics on the instantaneous forward rates f_t^h for h = 1, 2, 3, 4, 5.

Then, to formally test this model implication, I run the following regression:

$$rx_{t+1}^{(h)} = \alpha + \beta_0 UNF_t + \beta_1' x_t + \epsilon_{t+1}^{(h)}, \qquad (1.41)$$

¹⁸The data can be found here: https://statistiek.dnb.nl/en/dashboards/pensions/index.aspx. Because the solvency positions of insurers only go back till 2009, I use data on pension funds alone. Because the solvency positions of both insurers and pension funds are driven by the same factors, using aggregate pension fund data alone is sufficient. Robustness checks that use a weighted average of the solvency positions of pension funds and insurance companies indeed confirm this.

Table 1.6. Summary statistics Dutch yields: This table presents the means, medians standard deviations, minimum, and maximum of Dutch zero-coupon government bond yields. Panel A shows the summary statistics on Dutch yields, where the data is from Bloomberg and covers the period 2007-2020. Panel B presents the summary statistics on the measures for aggregate underfunding. All reported numbers are in percentage points, except the fraction of underfunded pension plans which is in percent. rx_{t+1}^h is the excess return for a bond with maturity h, y_t^h is the yield for a bond with maturity h, $f_t^{(h)}$ is the instantaneous forward rate for a bond with maturity h.

	mean	median	std.dev.	min	max	N
Panel A: Dutch zero-coupon govern	ament b	ond yields	s (%)			
$rx_{t+1}^{10} rx_{t+1}^{20} rx_{t+1}^{30} rx_{t+1}^{30} u_{t}^{10} - u_{t}^{10}$	$5.31 \\ 8.11 \\ 10.37 \\ 1.43$	$5.16 \\ 8.66 \\ 9.63 \\ 1.25$	$6.08 \\ 13.19 \\ 19.43 \\ 0.84$	-5.73 -14.06 -23.05 0.03	$18.35 \\ 41.56 \\ 61.34 \\ 3.09$	48 48 48 48
$egin{aligned} & g_t & g_t \ f_t^1 (= y_t^1) \ f_t^2 & f_t^3 \ f_t^3 & f_t^4 \ f_t^5 \ \end{aligned}$	$ \begin{array}{c} 1.10 \\ 0.60 \\ 1.27 \\ 1.70 \\ 2.03 \\ 2.34 \end{array} $	$ \begin{array}{c} 1.20\\ 0.01\\ 0.72\\ 1.47\\ 1.92\\ 2.45 \end{array} $	$ 1.56 \\ 1.67 \\ 1.72 \\ 1.68 \\ 1.60 $	-0.78 -0.79 -0.68 -0.55 -0.33	$ \begin{array}{r} 4.65 \\ 4.80 \\ 4.82 \\ 4.55 \\ 4.72 \end{array} $	48 48 48 48 48 48
Panel B: Underfunded pension plan	ıs					
1. Fraction underfunded (percent) 2. Level underfunded (%) 3. Liability return (%) $\rho(1,2)$ $\rho(1,3)$ $\rho(2,3)$	$\begin{array}{c} 37 \\ 2.12 \\ 2.62 \\ 0.84 \\ 0.33 \\ 0.46 \end{array}$	$\begin{array}{c} 31\\ 0\\ 2.61\end{array}$	24 3.26 6.13	$0 \\ 0 \\ -12.37$	83 12.2 20.95	48 48 48

where $rx_{t+1}^{(h)}$ is the annual excess bond return from time t to time t + 1 for a bond with maturity h, UNF_t equals the fraction of pension funds that are underfunded at time t, and x_t includes a set of controls.

Because I only observe the fraction of underfunded pension funds at a quarterly frequency, I estimate the regressions on a quarterly frequency.¹⁹ This implies I forecast returns for every four quarters. I compute standard errors using the Newey and West (1987) standard errors that allow for serial correlation up to 6 lags. The controls include the term spread, following Campbell and Shiller (1991), or the first five forward rates, following Cochrane and Piazzesi (2005).

Table 1.7 shows the results of the forecasting regressions. Figure 1.5 plots the fraction of underfunded pension funds and the one-year ahead excess return on a 10-year bond. As the figure shows and the table confirms, a higher fraction of underfunded pension funds is associated with lower future excess returns. As expected, the effect is stronger for longer maturity bonds.

The economic magnitude of the effects are substantial. A one standard deviation increase in the fraction of underfunded pension funds (0.24) decreases the excess future returns with 8.24 percent. Assuming that the change in underfunded pension funds has no effect on expected returns beyond a year, this corresponds to a decline in contemporaneous 30-year yields of 27 basis points (see Equation (1.28): 8.24%/30).²⁰ This equals 19 basis points for a 20-year bond and 12 basis points for a 10-year bond. As a robustness check, Table 1.15 and 1.16 of the Appendix summarize the results if I use the average percentage points of underfunding and the return on the liabilities as alternative proxies for the degree of underfunding. The effects using these alternative proxies are similar to the results discussed here.²¹

Running a regression of the aggregate allocation to fixed income on the level of underfunding indeed confirms that the fraction of underfunding increases the allocation to bonds: a

¹⁹The literature tends to use monthly data, however results of standard forecasting regressions using the term spread or forward rates do not substantially change when using a quarterly or a monthly frequency.

 $^{^{20}}$ In untabulated regressions, I find that the forecasting power substantially decreases when predicting 5 quarter future excess returns and completely vanishes using 6 quarters or more.

 $^{^{21}}$ For the first proxy, the effect of a one standard deviation increase on the 30-year yield equals 32 basis points, whereas for the second proxy this effect equals 19 basis points. The effect of the second proxy is smaller because the degree of underfunding is largely determined by both the returns on equity and bonds, whereas the return on the liabilities is only picking up the latter part.

one standard deviation increase in the fraction of underfunded pension funds increases the allocation to fixed income by 1.97%.²²

Evidently, running a regression of returns on a proxy for demand does generally not allow to cleanly identify the causality between demand shocks and asset prices, because there could be omitted variables that drive demand for bonds and is correlated with the degree of underfunding. Therefore, I turn to the asset pricing implications of the regulatory change that uses the construction of the UFR as an exogenous shock to demand next.

Figure 1.5. Future excess returns and underfunded pension plans

This graph shows the one year future excess return and the fraction of underfunded pension plans or the liability return $(r_t^L = \frac{L_t - L_{t-1}}{L_{t-1}})$. The upper figure shows the excess returns (percentage points) on the left y-axis and the fraction of underfunded pension plans (percent) on the right y-axis. The lower figure shows both the excess returns and liability return on the left y-axis (percentage points).



²²Formally, I run the following regression: $w_t^{FI} = \alpha + \beta_0 UNF_t + \epsilon_t$. I find a coefficient of $\beta_0 = 0.0821$ with a *t*-stat equal to 3.73 with an $R^2 = 0.25$.

able	=	cess	ping	iller,	
Chis t	$rx_{t^{-}}^{(t)}$	the ex	rerlap	nd Sh	0.01.
ns:]	olans:	cast 1	the ov	ell ar	$> d_{**}$
pla	sion 1	I fore	t for t	ampb	.05, *
nsion	l pen	9q4.	orrec	rd (C	p < 0
l pe	unded	il 201	s to c	spree	10, **
unde	nderf	17 unt	error	term	0 < 0.
derfi	ı of u	m 20(ndard	e the	$5). *_{l}$
of un	action	ta fro	87 stal	nclud	i, 200
ion c	he fr	rly da	st 198	rols i	azzes
fract	on t	uartei	d We	Cont	nd Pi
\mathbf{the}	eturns	vith q	rey an	lags.	ane a
Ising	cess re	ated v	e New	l of 6	Cochı
rns u	er ext	estima	. I us	tota	ates (
retu	quarte	s are (arters	vith a	ard r
ond	ure 4-	ession	5 4 qu	$ses), \tau$	us fow
ess b	e futu	e regre	owing	enthe	aneoi
exc	of the	. The	llol at	(pare	ıstant
sting	sions	$+\epsilon_{t+1}^{(h)}$	for the	ssions	five ir
oreca	egres	- $\beta_1' x_t$	larter	regre	first
7. F	the 1	$NF_t +$	ach qı	of the	d the
ble 1.	sents	- $\beta_0 U$ -	urn ei	ure o)1) ar
Tal	pre	α^+	ret	nat	196

20-year 10-year	(4) (5) (6) (7) (8) (9)	-15.807^{*} -15.937^{*} -33.201^{***} -4.135 -5.657^{*} -10.896^{***}	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(2.511) (0.984)	-12.117*** -4.991***	(2.811) (1.519)	6.441 4.031	(6.920) (3.571)	-3.225 -2.902	(4.843) (2.887)	2.537 3.017	(6.332) (3.400)	5.203 0.904	(7.909) (3.887)	48 48 48 48 48 48 48 48	
	(3)	** -57.786***	(6.639)		-13.397^{***}	(4.506)	3.504	(9.513)	-5.338	(7.709)	5.062	(11.943)	8.823	(13.566)	48	
ar	1) (2)	677*** -34.324*	(10.795) ((3.730)											18 48	0 - - - -
<u>30-y</u> €		UNF_{t} -35.	(10.	$y_t = y_t$	$f_t^{(1)}$	5	$f_t^{(2)}$		$f_t^{(3)}$		$f_t^{(4)}$		$f_t^{(5)}$		N	°, t

1.6.2 The connection between portfolio holdings and yields

The regulatory change had a significant effect on Dutch long-term bond yields, as already shown by Greenwood and Vissing-Jorgensen (2018) using event studies. Figure 1.6 shows the 30-20 government bond spread. The spread increased significantly after the announcement of the UFR, and remained at a higher level thereafter.

Figure 1.6. Government bond yield spreads

This graph shows the spread between 30 year and 20 year Dutch government bonds. The vertical lines are three days before and after the announcement of the UFR.



In this section, I connect investors' portfolio holdings to yields using the asset demand system developed by Koijen and Yogo (2019). The asset demand system allows me to compute demand elasticities with respect to price, which, in turn, allow me to study the importance of various investor types in creating price effects.

Investor *i*'s investment in Dutch government bonds within maturity bucket *m* is denoted by $H_{it}(m)$, and the investment in the outside asset is denoted by O_{it} .²³ Because I cannot observe what investors consider to be the outside asset, I use the 10-year German yield as proxy for the outside asset. German government bonds are important alternative liability hedge assets outside of the Netherlands. The results, however, are robust using other proxies for the outside asset. Furthermore, I assume that holdings of the outside asset move only due to changes in the German yield.

²³The results do not materially change using all bonds outside of the Netherlands as the outside asset.

The portfolio weight in the framework of Koijen and Yogo (2019) is then defined as:

$$w_{it}(m) = \frac{H_{it}(m)}{O_{it} + \sum_{m=1}^{M=7} H_{it}(m)} = \frac{\delta_{it}(m)}{1 + \sum_{m=1}^{M=7} \delta_{it}(m)},$$
(1.42)

where $\delta_{it}(m) = H_{it}(m)O_{it}^{-1}$ and $w_{it}(0) = 1 - \sum_{m=1}^{M=7} w_{it}(m)$ equals the fraction invested in the outside asset. Demand for government bonds with maturity m is a function of bond yields and characteristics (Koijen et al., 2020):

$$\ln H_{it}(m) = \ln \delta_{it}(m) + \ln O_{it}$$

= $\hat{\alpha}_i + \beta_{0i} y_t(m) + \beta'_{1i} x_t(m) + \beta_{2i} \ln H_i^{2009q^2}(m) + \hat{\beta}_{3i} y_t^{DE} + \epsilon_{it}(m), (1.43)$

in which $\hat{\alpha}_i = \alpha_i + \ln O_i$, $\hat{\beta}_{3i} = \beta_{3i} + \psi_i$, $y_t(m)$ is the average yield for maturity bucket m, $x_t(m)$ includes bond characteristics, $H_i^{2009q2}(m)$ equals the initial holdings in each maturity bucket m, and y_t^{DE} is the German yield and captures alternative liability hedge opportunities outside of the Netherlands.

Koijen and Yogo (2019) show that (1.43) is consistent with a model in which investors have mean-variance preferences over returns, assume that returns follow a factor model, and assume that both expected returns and factor loadings are affine in a set of characteristics. The component of demand that is not captured by prices, characteristics, and time-invariant characteristics, $\epsilon_{it}(m)$, is referred to as latent demand.

Safe, (long-term) government bond returns are primarily driven by duration and convexity. The vector of bond characteristics therefore includes the average duration of the bond in every maturity bucket and the average convexity, measured as the duration squared. I also include the average coupons.

In order to obtain consistent estimates of the parameters in (1.43) using OLS one has to assume that characteristics are exogenous to latent demand. However, positive latent demand for Dutch government bonds of a particular maturity may result in lower yields. The demand curves are therefore estimated using an instrumental variable approach. I use the weights assigned to the UFR as an instrument for changes in demand. Even though investors may have anticipated the UFR, the determinants of the shape of the UFR such as its level and the slope were unknown.²⁴

The weights assigned to the UFR for each maturity bucket are summarized in Table 1.14

²⁴https://www.dnb.nl/en/news/news-and-archive/dnbulletin-2012/dnb276012.jsp

Table 1.8. **Instrument for every maturity bucket**: This table shows the instrument for every maturity bucket used for the instrumental variable approach. The instrument is constructed as the average weight assigned to the UFR for each maturity bucket. An overview of the weights for separate maturities is given in Table 1.14.

	[0, 5]	(5, 10]	(10, 15]	(15, 20]	(20, 25]	(25, 30]	$(30,\infty)$
ξ_m	0	0	0	0	0.263	0.579	0.910

of the Appendix. The instrumental variable is defined as $z_t(m) = \xi(m)D_t$, where $\xi(m)$ is the average weight assigned to the UFR for maturity bucket m and D_t equals one after implementation of the UFR and zero otherwise. The first-stage regression of the instrumental variable estimator equals:

$$y_t(m) = \beta_0 + \beta_1 z_t(m) + \beta'_2 x_t(m) + y_t^{DE} + \epsilon_t(m), \qquad (1.44)$$

The first stage is summarized in column (1) of Table 1.9. The *t*-statistic equals 4.95 and is higher than the proposed threshold of 4.05 by Stock and Yogo (2005) for rejecting the null of weak instruments at the 5 percent level, suggesting that the instrument is not weak. The coefficient for the instrument equals 0.397, with a standard error of 0.092. A coefficient of 0.435 implies that government bond yields with time-to-maturity between 21 and 25 years went up with 11 basis points, for time-to-maturities between 26 and 30 years with 25 basis points, and for time-to-maturities longer than 30 years with 39 basis points. This implies an increase in the 30-20 year spread of approximately 24 basis points. These estimates are larger than the estimates found in Greenwood and Vissing-Jorgensen (2018), who find an increase in the Dutch 30-10 year spread of 15 basis points by conducting an event study (Figure 1.6). Event studies only capture the immediate decrease in long-term bond holdings around the announcement of the regulatory change. However, P&Is did not sell everything at once, but instead spread out the decrease in long-term bond holdings over four quarters (Figure 1.3).

I estimate the demand curves for banks, insurance companies, mutual funds, pension funds, and the foreign sector. I first show the results aggregated by investor type as I do not have investor specific characteristics for all types in my sample that would allow me to take advantage of cross-sectional heterogeneity within types. The investments of the foreign sector are defined as the difference between the total amount outstanding minus the holdings by the other sectors. The data on total amounts outstanding is from the Dutch State Treasury Agency.²⁵

Columns 2-6 of Table 1.9 show the estimates of the demand system. P&Is have demand curves that are upward sloping, which is consistent with my model. Because of duration mismatch between the assets and the liabilities, a decrease in interest rates decreases the funding ratio. From (1.16), we observe that a low funding ratio increases the demand for long-term bonds. Foreign investors prefer shorter duration bonds, whereas P&Is prefer longer duration bonds, consistent with Section 1.5. Moreover, P&Is prefer bonds with higher coupons, potentially resulting from the desire to match the cash flows of their liabilities. In all cases, the initial holdings are positive and statistically significant, meaning that unobserved time-invariant characteristics explain a substantial part of the holdings.

I can use the demand system to connect prices to elasticity of demand with respect to price for all investor types (Koijen and Yogo, 2019; Koijen et al., 2020):

$$\frac{\partial q_{it}(m)}{\partial p_{it}(m)} = 1 + 100 \frac{\beta_{0i}}{T_{mt}} (1 - w_{it}(m)), \qquad (1.45)$$

where lowercases are log of variables and T_{mt} is the average maturity for maturity bucket m. To compute $w_{it}(m)$, I use the investment in euro area bonds except Dutch ones as the outside asset.

The demand elasticities with respect to price for each investor type are summarized in Table 1.11. A demand elasticity close to zero implies that demand is inelastic and a large value implies that demand is sensitive to the price. Banks have the highest demand elasticity, followed by mutual funds and the foreign sector. However, the estimate for banks is very imprecise. Banks are not holding the longest maturity bonds and therefore the instrument is weak. Consistent with the findings before, demand elasticities are negative for the P&I sector. The weighted average elasticity equals 2.05 and the weight of each sector is computed as the average weights of the different investor types in each maturity bucket prior to the implementation of the UFR. The weights of each sector are summarized in Table 1.10. As in Koijen et al. (2020), demand elasticities are substantially higher than the estimates for stock markets, e.g. Chang et al. (2015) find an elasticity close to one. However, the average weighted price elasticity of demand is lower than measured in Koijen et al. (2020), where the unit of observation are the holdings of government debt in a particular country. This means

²⁵https://english.dsta.nl

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

Table 1.9. **Demand system**: This table shows the regression results of the demand system described in (1.43): $\ln H_{it}(m) = \hat{\alpha}_i + \beta_{0i} y_t(m) + \beta'_{1i} x_t(m) + \beta_{2i} \ln H_i^{2009q^2}(m) + \hat{\beta}_{3i} y_t^{DE} + \epsilon_{it}(m)$. The first column shows the first stage regression for the foreign investors. The instrument $z_t(m)$ equals the weights assigned to the UFR for each maturity bucket m. The controls $x_t(m)$ include the average bond duration, convexity, coupon, and initial bond holdings. Robust standard errors are in parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	$y_t(m)$	Holdings Banks	Holdings Foreign	Holdings IC	Holdings MF	Holdings PF
$y_t(m)$		2.171	1.156**	-2.639^{**}	0.531^{**}	-1.640^{***}
		(3.493)	(0.559)	(1.205)	(0.258)	(0.632)
$z_t(m)$	0.435^{***}		· · · ·	· · ·		· · · ·
	(0.088)					
Duration	0.168***	0.405	-0.730^{***}	0.910^{***}	-0.289	0.523^{***}
	(0.034)	(0.568)	(0.127)	(0.286)	(0.296)	(0.130)
Convexity	-0.006^{***}	-0.019	0.019***	-0.028^{***}	-0.011	-0.014^{***}
	(0.001)	(0.016)	(0.003)	(0.007)	(0.008)	(0.003)
Coupon	-0.021	-0.079	-0.267^{***}	0.375^{***}	0.013	0.106^{***}
	(0.017)	(0.092)	(0.050)	(0.049)	(0.015)	(0.035)
Initial holdings	-0.192^{***}	0.361^{***}	0.129	1.654^{***}	0.895^{***}	0.773^{***}
	(0.062)	(0.089)	(0.222)	(0.253)	(0.122)	(0.113)
10-year German yield	1.019^{***}	2.079	-1.409	2.795	-1.366	2.361^{**}
	(0.015)	(3.544)	(0.878)	(1.725)	(1.930)	(0.967)
		. ,	. ,	. ,	. ,	
N	243	209	243	243	243	243
R^2	0.98	0.68	0.55	0.28	0.75	0.41

that investors are less price elastic across maturity buckets than they are across countries. Government bonds issued by (some) countries in the euro area may be closer substitutes than bonds of different maturities from a balance sheet perspective.

In order to derive pricing effects from the demand system, I can perform a simple backof-the-envelope calculation. Pension funds and insurers sold 22 percent of the amount outstanding of 30-year Dutch government bonds. The weighted average price elasticity of demand equals 2.05 and thus the price effect equals 22%/2.05 = 10.73%. For a bond with a maturity of 30 years, this implies an increase in long-term yields of 36 basis points, which is close to the price effect found for the first stage regression. Table 1.10. Weights of investor types in each maturity bucket: This table shows the weights of the investor types (banks, insurance companies, foreign investors, mutual funds, and pension funds) at the start of 2012q1. The fraction of foreign investors is determined as the fraction of total amount outstanding that is not held by one of the Dutch institutions. The column tot. (tot. long) defines the total fraction that each investor is holding relative to the total amount outstanding (total amount outstanding maturities exceeding 10 years).

	[0, 5]	(5, 10]	(10, 15]	(15, 20]	(20, 25]	(25, 30]	$(30,\infty)$	tot.	tot. long
Banks	3	16	2	9	3	4	0	9	4
Foreign investors	90	70	75	52	44	42	49	75	52
Insurers	3	6	12	21	29	30	27	8	24
Mutual funds	1	1	1	2	1	1	1	1	1
Pension funds	3	7	10	16	23	23	23	9	19

Table 1.11. **Price elasticity of demand**: This table shows the price elasticity of demand, computed as in Equation (1.45). The average, standard deviation, minimum, and maximum over time are given. Total elasticity is the weighted average elasticity, using the weights of each sector defined in Table 1.10.

	obs.	mean	std.dev.	min	max
Banks Foreign investors Insurance companies Mutual funds Pension funds	209 243 243 243 243 243	$23.93 \\ 4.53 \\ -29.95 \\ 8.30 \\ -18.61$	$25.57 \\ 1.89 \\ 31.68 \\ 6.82 \\ 20.19$	5.67 1.84 -102.44 1 -63.63	$\begin{array}{c} 83.88 \\ 11.28 \\ -6.93 \\ 22.23 \\ -3.97 \end{array}$
Total	2.05				

1.6.3 Demand curves P&Is including characteristics

My model predicts that the demand for bonds depends primarily on the liability structure and solvency positions of P&Is. My model predicts that demand for long-term bonds is higher when liability duration is longer as well as when the funding ratio is low. In this section, I extend the framework of Koijen and Yogo (2019) by also including two key P&Is characteristics: the liability duration and the solvency position. In this setting, demand for government bonds with maturity m becomes a function of bond yields, bond characteristics, and fund characteristics interacted with bond characteristics:

$$\ln H_{it}(m) = \hat{\alpha}_{i} + \beta_{0i} y_{t}(m) + \beta'_{1i} x_{t}(m) + \beta'_{2i} (x_{1t}(m) \times v_{it}) + \beta_{3i} \ln H_{i}^{2009q^{2}}(m) + \hat{\beta}_{4i} y_{t}^{DE} + \epsilon_{it}(m)$$
(1.46)

where v_{it} includes the liability duration and the solvency position for investor *i* at time *t*, $x_t(m)$ includes bond characteristics as defined before, and $x_{1t}(m)$ includes bond duration and convexity, and α_i includes investor fixed effects. Again, I include the initial holdings as well as the 10-year German yield.

Table 1.12 shows the results for insurers. On average, insurance companies with long liability durations prefer bonds with long durations, but relatively low convexity. This means that insurance companies prefer long-term bonds, but not the bonds with the longest maturities. The UFR creates incentives to hedge the duration of the liabilities, but less so the convexity of the liabilities, i.e. the cash flows at the very long-end of the maturity spectrum. I also compute the demand system for insurers with high versus low liability durations and high and low solvency positions, respectively. Interestingly, insurers with long liability durations have much stronger upward sloping demand curves than average, whereas insurers with low liability durations have neither upward nor downward sloping demand curves. Moreover, insurers with low solvency positions have slightly stronger upward sloping demand curves than insurers with high solvency positions, again consistent with the predictions of my model.

Table 1.13 shows the results for pension funds. Pension funds with long liability durations generally have a preference for bonds with long liability durations but lower convexity, as for insurance companies. There is some evidence that pension funds with high solvency positions have a stronger preference for bonds with high convexity, consistent with the finding in Section 1.5 that pension funds with high solvency positions decreased long-term bond holdings to a smaller extent than the ones with low solvency positions. As opposed to insurance companies, pension funds with long versus short liability do not differ as much

Table 1.12. **Demand system with insurer characteristics** This table shows the results of the demand system including insurers characteristics (liability duration and solvency ratio) described in (1.46): $\ln B_{it}(m) = \hat{\alpha}_i + \beta_{0i}y_t(m) + \beta'_{1i}x_t(m) + \beta'_{2i}(x_{1t}(m) \times v_{it}) + \hat{\beta}_{3i}y_t^{DE} + \alpha_i + \epsilon_{it}(m)$. The first column shows the first stage regression. The instrument $z_t(m)$ equals the weights assigned to the UFR for every maturity bucket m, described in Table 1.8. The results are reported for all insurers, as well as insurers that are above the 70th percentile or below the 30th percentile of the liability duration (solvency ratio). The 30th and 70th percentile are determined in each quarter. Controls include the bond characteristics, the 10-year German yield, and the initial holdings. Robust standard errors are in parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	u(m)	Holdings all	Holdings high dur	Holdings low dur	Holdings high FB	Holdings low FB
	$g_t(m)$		ingii aai	ion dui		101111
$y_t(m)$		-2.432^{***}	-3.903***	1.300	-3.138^{***}	-3.714^{**}
		(0.697)	(1.340)	(1.568)	(0.984)	(1.837)
$z_t(m)$	0.435^{***}					
	(0.088)					
bond dur \times liability dur		0.011^{***}	-0.032^{***}	-0.020^{*}	0.020***	0.016^{***}
-		(0.001)	(0.007)	(0.012)	(0.003)	(0.003)
bond convex \times liability dur		-0.001^{***}	0.002***	0.003***	-0.001^{***}	-0.001^{***}
, i i i i i i i i i i i i i i i i i i i		(0.000)	(0.000)	(0.001)	(0.000)	(0.000)
bond dur \times solvency ratio		-0.003	0.002	0.019	-0.008	0.075
		(0.004)	(0.027)	(0.012)	(0.008)	(0.054)
bond convex \times solvency ratio		0.000	-0.001	-0.002^{**}	0.001	-0.002
		(0.000)	(0.002)	(0.001)	(0.001)	(0.003)
Controls	Yes	Yes	Yes	Yes	Yes	Yes
N	243	$3,\!418$	848	1,105	910	1,066
R^2	0.98	0.68	0.44	0.68	0.68	0.63

in the slope of their demand curves. Notice, however, that the heterogeneity in liability durations is much smaller for pension funds than for insurers.

1.7 Conclusion

In this paper, using holdings data and price data simultaneously, I study changes in hedging incentives of long-term investors and its effect on asset prices. My findings suggest that regulation plays an nontrivial role in the demand for long-term bonds which, in turn,

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

Table 1.13. **Demand system with pension fund characteristics** This table shows the results of the demand system including pension fund characteristics (liability duration and funding ratio) described in (1.46): $\ln B_{it}(m) = \hat{\alpha}_i + \beta_{0i}y_t(m) + \beta'_{1i}x_t(m) + \beta'_{2i}(x_{1t}(m) \times v_{it}) + \hat{\beta}_{3i}y_t^{DE} + \alpha_i + \epsilon_{it}(m)$. The first column shows the first stage regression. The instrument $z_t(m)$ equals the weights assigned to the UFR for every maturity bucket m, described in Table 1.8. The results are reported for all pension funds, as well as pension funds that are above the 70th percentile or below the 30th percentile of the liability duration (funding ratio). The 30th and 70th percentile are determined in each quarter. Controls include the bond characteristics, the 10-year German yield, and the initial holdings. Robust standard errors are in parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	u(m)	Holdings	Holdings high dur	Holdings low dur	Holdings	Holdings low FB
	$g_t(m)$		ingii dui	low dui	ingii i it	10 1 11
$y_t(m)$		-0.565^{*}	-1.038	-0.284	-1.844***	-1.685^{*}
		(0.323)	(0.777)	(0.508)	(0.507)	(0.994)
$z_t(m)$	0.456^{***}					
	(0.087)					
bond dur \times liability dur	· /	0.007^{***}	0.013^{***}	0.019^{***}	0.004^{*}	-0.016^{***}
		(0.001)	(0.005)	(0.005)	(0.002)	(0.004)
bond convex \times liability dur		-0.0002^{***}	-0.001^{***}	-0.001^{**}	-0.000	0.001***
-		(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
bond dur \times funding ratio		-0.032	0.141	-0.078	-0.217^{**}	0.080
		(0.043)	(0.099)	(0.076)	(0.102)	(0.222)
bond convex \times funding ratio		0.001	0.008	0.002	0.010**	0.001
_		(0.002)	(0.006)	(0.004)	(0.005)	(0.013)
Controls	Yes	Yes	Yes	Yes	Yes	Yes
N	243	6,085	1,580	2,115	1,980	$1,\!497$
R^2	0.98	0.40	0.53	0.38	0.41	0.46

affect the asset prices of these bonds. This has important policy implications as these findings can be used to design long-term investor regulation in a way that is desirable for the economy as a whole. In particular, my findings show the relevance of incorporating the regulatory framework of long-term investors to analyze effects of conventional and unconventional monetary policy.

1.8 Appendix

A Further details on UFR

The UFR was initially discussed as part of the Long-Term Guarantee Assessment (LTGA) of Solvency II regulation. EIOPA proposed the regulatory discount curve based on the UFR method first in 2010. There are three important decisions policy makers have to make when introducing the UFR: the level of the UFR, the point on the curve at which the UFR method starts, and the interpolation method, or the convergence path. The initial EIOPA proposals are first discussed in detail.

The UFR was initially set at 4.2%, which is based on 2% expected inflation and 2.2% historical average of the real short interest rate. The expected inflation rate aligns with the ECB's target inflation. The real interest rate is based on a study by Dimson et al. (2002). The point of curve at which the UFR method starts was set at 20 years and the convergence period is set at 40 years. The extrapolation method proposed by EIOPA is the Smith-Wilson technique. The Smith-Wilson technique only uses the forward rates at time-to-maturity 19 to 20 years and the UFR to compute the yield curve.

For pension funds a slight modification was used, namely the market interest rates at each maturity in combination with the UFR. For pension funds, the convergence is such that its a weighted average between market interest rates and the UFR. So as opposed to insurers, not only the forward rate from time-to-maturity 19 to 20 years is used, but the implied market forward rate for each maturity and the UFR.

B Model derivation

This appendix solves the optimization problems of P&Is. The mean-variance optimization problem equals:

$$\max_{w_{t}} \quad \mathbb{E}\left[u\left(\frac{A_{t+1}}{A_{t}} - \frac{L_{t+1}}{A_{t}}\right)\right] \\ = \arg\max_{w_{t}} \mathbb{E}\left[\frac{A_{t+1}}{A_{t}}\right] - \frac{\gamma}{2} \operatorname{Var}\left[\frac{A_{t+1}}{A_{t}} - \frac{L_{t+1}^{E}}{A_{t}}\right] - \frac{\lambda(F_{t}^{R})}{2} \operatorname{Var}\left[\frac{A_{t+1}}{A_{t}} - \frac{L_{t+1}^{R}}{A_{t}}\right], \quad (A1)$$

subject to

$$w_t'\iota = w_t^S + w_t^B\iota \le 1,\tag{A2}$$

$$w_t^S, w_{h,t}^B \ge 0 \qquad \forall h,$$
 (A3)

The Lagrange equals:

$$\mathcal{L}(w_{t},\nu_{t},\delta_{t}) = R_{f} + w_{t}'\mathbb{E}_{t}[r_{t+1} - r_{f}] - \frac{\gamma}{2} \Big(w_{t}' \operatorname{Var}_{t}[r_{t+1}]w_{t} + a_{t}' \operatorname{Var}_{t}[r_{t+1}^{B}]a_{t} \frac{1}{F_{t}^{E}} - 2w_{t}' \operatorname{Cov}_{t}[r_{t+1}, r_{t+1}^{B}]a_{t} \frac{1}{F_{t}^{E}} \Big) - \frac{\lambda(F_{t}^{R})}{2} \Big(w_{t}' \operatorname{Var}_{t}[r_{t+1}]w_{t} + (\xi_{L} \circ a_{t})' \operatorname{Var}_{t}[r_{t+1}^{B}](\xi_{L} \circ a_{t}) \frac{1}{F_{t}^{R}} - 2w_{t}' \operatorname{Cov}_{t}[r_{t+1}, r_{t+1}^{B}](\xi_{L} \circ a) \frac{1}{F_{t}^{R}} \Big) + \nu_{t}(w_{t}'\iota - 1) + \delta_{t}'w_{t}.$$
(A4)

Taking the derivative with respect to w_t^S , w_t^B , and ν_t gives:

$$\frac{\partial \mathcal{L}(w_t^S, \nu_t, \delta_t^S)}{\partial w_t^S} = \mathbb{E}_t[r_{t+1}^S - r_f] - (\gamma + \lambda(F_t^R)) \operatorname{Var}_t[r_{t+1}^S] w_t^S + \nu_t + \delta_t^S = 0, \quad (A5)$$

$$\frac{\partial \mathcal{L}(w_t^B, \nu_t, \delta_t^B)}{\partial w_t} = \mathbb{E}_t[r_{t+1}^B - r_f] - (\gamma + \lambda(F_t^R)) \operatorname{Var}_t[r_{t+1}^B] w_t^B - \gamma \operatorname{Var}_t[r_{t+1}^B] a_t \frac{1}{F_t^E} - \lambda(F_t^R) \operatorname{Var}_t[r_{t+1}] (\xi_L \circ a_t) \frac{1}{F_t^R} + \nu_t \iota = 0,$$
(A6)

$$\frac{\partial \mathcal{L}(w_t, \nu_t, \delta_t)}{\partial \nu_t} = w'_t \iota - 1 = 0.$$
(A7)

This results in the optimal weights (1.15) and (1.16):

$$w_t^{S*} = \underbrace{\frac{\mathbb{E}_t[r_{t+1}^S - r_f] + \nu_t + \delta_t^S}{(\gamma + \lambda(F_t^R)) \operatorname{Var}_t[r_{t+1}^S]}}_{(A8)}$$

speculative portfolio

$$w_t^{B*} = \underbrace{\frac{\mathbb{E}_t[r_{t+1}^B - r_f] + \nu_t \iota + \delta_t^B}{(\gamma + \lambda(F_t^R)) \operatorname{Var}_t[r_{t+1}^B]}}_{\text{speculative portfolio}} + \underbrace{\frac{\gamma}{\gamma + \lambda(F_t^R)} a_t \frac{1}{F_t^E}}_{\text{economic hedging portfolio}} + \underbrace{\frac{\lambda(F_t^R)}{\gamma + \lambda(F_t^R)} (\xi_L \circ a_t) \frac{1}{F_t^R}}_{\text{regulatory hedging portfolio}}.$$
(A9)

with ν_t (if the constraint binds):

$$\nu_{t} = \frac{1 - \left(\frac{\mathbb{E}_{t}[r_{t+1}^{S} - r_{f}] + \delta_{t}^{S}}{(\gamma + \lambda(F_{t}^{R}))\operatorname{Var}_{t}[r_{t+1}^{S}]} + \left(\frac{\mathbb{E}_{t}[r_{t+1}^{B} - r_{f}] + \delta_{t}^{B}}{(\gamma + \lambda(F_{t}^{R}))\operatorname{Var}_{t}[r_{t+1}^{R}]}\right)'\iota + \left(\frac{\gamma}{\gamma + \lambda(F_{t}^{R})}a_{t}\frac{1}{F_{t}^{E}}\right)'\iota + \left(\frac{\lambda(F_{t}^{R})}{\gamma + \lambda(F_{t}^{R})}\left(\xi_{L} \circ a_{t}\right)\frac{1}{F_{t}^{R}}\right)'\iota\right)}{\left(\frac{\iota}{(\gamma + \lambda(F_{t}^{R}))\operatorname{Var}_{t}[R_{t+1}]}\right)'\iota}.$$
(A10)

C Tables

Table 1.14. Weights UFR in regulatory discount curve: This table shows the weights assigned to the UFR to compute the regulatory discount curve. The weights are derived using the Smith-Wilson technique.

time-to-maturity	weight	time-to-maturity	weight
21	0.086	41	0.903
22	0.186	42	0.914
23	0.274	43	0.923
24	0.351	44	0.932
25	0.420	45	0.940
26	0.481	46	0.947
27	0.536	47	0.954
28	0.584	48	0.960
29	0.628	49	0.965
30	0.666	50	0.970
31	0.701	51	0.970
32	0.732	52	0.978
33	0.760	53	0.982
34	0.785	54	0.985
35	0.808	55	0.988
36	0.828	56	0.990
37	0.846	57	0.993
38	0.863	58	0.995
39	0.878	59	0.997
40	0.891	60	0.998

	30-year			20-year			10-year		
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)
$LUNF_t$	-2.941^{***}	-2.932^{***}	-3.100^{***}	-1.416^{***}	-1.476^{***}	-1.648	-0.415^{*}	-0.503^{**}	-0.530^{***}
	(0.572)	(0.573)	(0.332)	(0.475)	(0.455)	(0.216)	(0.235)	(0.222)	(0.123)
$y_t^{(10)} - y_t^{(1)}$		-0.140			0.917			1.325	
		(3.459)			(2.524)			(1.070)	
$f_t^{(1)}$			-9.154^{**}			-9.562^{***}			-4.124^{***}
2			(4.379)			(2.885)			(1.508)
$f_t^{(2)}$			10.869			10.777			5.445
2			(12.043)			(8.337)			(3.980)
$f_t^{(3)}$			-7.144			-4.808			-3.456
			(8.784)			(5.522)			(2.917)
$f_t^{(4)}$			-6.943			-4.972			0.533
			(14.578)			(8.983)			(4.446)
$f_t^{(5)}$			13.570			8.956			2.183
			(19.517)			(12.343)			(5.526)
N	48	48	48	48	48	48	48	48	48
R^{2}	0.27	0.27	0.41	0.14	0.15	0.40	0.06	0.09	0.37

C. TABLES

	30-year			20-year			10-year		
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)
r_t^L	-1.001^{***} (0.313)	-0.950^{***} (0.318)	-0.809^{**} (0.332)	-0.582^{**} (0.231)	-0.579^{**} (0.231)	-0.440^{**} (0.217)	-0.231^{**} (0.117)	-0.252^{**} (0.114)	-0.172 (0.106)
$y_t^{(10)} - y_t^{(1)}$		-2.466 (4.007)			-0.123 (2.832)			1.043 (1.169)	
$f_t^{(1)}$			-4.086		~	-6.862^{*}			-3.239^{*}
5			(6.430)			(3.915)			(1.776)
$f_t^{(2)}$			7.756			9.076			4.759
			(13.874)			(9.205)			(4.244)
$f_t^{(3)}$			-11.720			-7.152			-3.947
			(13.488)			(6.901)			(2.544)
$f_t^{(4)}$			-15.353			-9.408			-0.792
1			(18.130)			(10.863)			(5.221)
$f_t^{(5)}$			26.670			15.830			4.128
			(20.781)			(13.300)			(6.223)
N	47	47	47	47	47	47	47	47	47
R^{2}	0.11	0.12	0.22	0.08	0.08	0.28	0.06	0.12	0.32

CHAPTER 1. LONG-TERM INVESTORS AND THE YIELD CURVE

Chapter 2

Pension Funds and Drivers of Heterogeneous Investment Strategies¹

2.1 Introduction

Defined-benefit pension funds play a pivotal role in society as many people depend on them for their retirement savings and investments. Globally, 50 percent of all occupational retirement savings is in defined-benefit pension funds (Willis Towers Watson, 2019). The investments of these pension funds serve a similar objective, namely to finance the future liabilities towards their beneficiaries. Therefore, understanding what drives pension funds to structure their investments in a particular way is important. Even small differences in investment strategies may lead to large divergences in performance across pension funds over time. Consequently, these divergences may have a substantial impact on beneficiaries' purchasing power.² This is particularly relevant because beneficiaries are typically not free

¹This chapter is based on joint work with Dirk Broeders. An earlier version of this paper circulated under the title 'Liability-Driven Investors': First version, September 2019. We thank Aleksandar Andonov, Jules van Binsbergen, Zvi Bodie, Marie Brière, Esther Eiling, Luo Haoyi, Frank de Jong, Ralph Koijen, Jac Kragt, Christoph Meinerding, Olivia Mitchell, Theo Nijman, Sofia Ramos, Laurens Swinkels, Patrick Tuijp, Luis Viceira (EFA discussant), Bas Werker; and seminar participants at the EFA 2020 (Helsinki), Women in Finance Conference 2019 (Goethe University, Frankfurt), AFBC 2019 (Sydney), PFMC 2019 (Paris), FMA 2019 (New Orleans), De Nederlandsche Bank, Maastricht University, Robeco, and Tilburg University for very useful comments. Moreover, we are grateful to Jeroen Meringa for gathering the data. We gratefully acknowledge financial support from Inquire Europe. The expressed views are those of the authors and do not necessarily reflect the official positions of De Nederlandsche Bank.

 $^{^{2}}$ In a defined benefit pension plan, benefits are determined by a formula that takes into account an employee's salary and years of service. Many defined benefit plans, however, also contain elements that depend on the investment returns, such as the cost-of-living adjustment or indexation.

to choose their own pension fund because it "comes with the job". Neither can beneficiaries make individual investment decisions within a defined-benefit pension fund. As a result, these pension funds operate in an environment where they do not have to compete for market share as many other institutional investors do, such as mutual funds.

Despite their pivotal role in society, so far only a few studies have analysed the investment strategies of pension funds. The lack of access to comprehensive and detailed data on this type of investor is the main reason for the limited number of studies. Exceptions are Rauh (2009) and Andonov et al. (2017) who study the effects of regulatory incentives on risky asset allocations for US corporate and public pension plans respectively as well as Anantharaman and Lee (2014) who link risky asset allocations to the compensation incentives of the top management in US corporate pension plans.³ Our approach differs from these studies because we do not focus on regulatory or compensation incentives.

The primary objective of our study is to measure the heterogeneity in investment strategies across pension funds, the drivers of this heterogeneity, and the effects it has on expected retirement income. We start with a model that solves a mean-variance optimization problem of assets minus liabilities and show that the following characteristics affect the investment decisions of pension funds: liability duration, funding ratio, and risk aversion. In addition to the model, we show that institutional factors, in particular the pension fund's size and type, influence the investment strategies. Nonetheless, these characteristics only explain 36 percent of the heterogeneity in the average returns across pension funds. The heterogeneity that remains reflects an economically sizeable difference in average annual returns of 0.70-1.50 percentage points between the pension funds with the highest and those with the lowest factor exposures. This is equivalent to a difference in expected retirement income of 16-32 percent over a 40-year accrual phase or to an increase in contributions of 19-46 percent to receive the same retirement income. We show that these differences reflect heterogeneity in beliefs across pension funds that they partially reveal through their choices of the asset management firms that they hire to execute their investment strategies. Our findings are remarkable, because the pension funds in our sample have similar investment objectives, yet even after controlling for differences in their characteristics they make distinct investment decisions.

The object of our study is occupational defined-benefit pension funds in the Netherlands.

³Lakonishok et al. (1992), Blake et al. (1999), Del Guercio and Tkac (2002), Tonks (2005), Goyal and Wahal (2008), and Blake et al. (2013) also examine the investment decisions of pension assets, but the focus in these studies is on the asset managers hired by the pension funds.

The Dutch occupational pension system is economically important because it is large in terms of total assets under management (AUM). In 2018, the AUM equaled approximately 1.4 trillion euros, and the Dutch system represented 53 percent of the total assets of pension funds in the euro area, (OECD, 2019).⁴ The proprietary data that we use are the quarterly returns of asset classes over the period from 1999 to 2017, and the return computations are based on the Global Investment Performance Standards (GIPS) as of 2010. The reporting requirements are mandatory, and the data are therefore free from self-reporting biases.

We study investment strategies through factor exposures. Traditionally, the investment strategies of pension funds focus on the optimal asset allocations to stocks, bonds, real estate, and alternative assets (e.g. Campbell and Viceira, 2002). However, the rise of the global factor literature enables a more granular study of investment strategies within and across asset classes. This literature shows that factors based on a particular signal perform robustly across countries and asset classes. Prime examples include momentum and value (Asness et al., 2013), low beta (Frazzini and Pedersen, 2014), and carry (Koijen et al., 2018). We use the existing global factors for equities: the market, value, momentum, carry, and low beta. For fixed income, we construct European factors as the pension funds in our sample primarily invest in euro-dominated bonds, which confirms the currency bias in Maggiori et al. (2020). The market factor consists of investment-grade bonds. Next to the market factor and a credit factor for fixed income, we again use value, momentum, carry, and low beta factors. With the exception of the market and the credit factors, we refer to factors as long-short factors.

We analyze the heterogeneity in investment strategies in the following three steps: first, we measure factor exposures and estimate the cross-sectional average and heterogeneity in factor exposures for both equity and fixed income portfolios; second, we identify the drivers of factor exposures; and third, we measure the differences in the implied beliefs on factor returns. Along these lines we report the following key results.

First, we show that the average pension fund has a stock market beta lower than one and a fixed income market beta larger than one. Further, for both equities and fixed income the average pension fund has a positive exposure to low beta but a negative exposure to value and carry. We also find substantial heterogeneity in both equity and fixed income factor exposures across pension funds.

Second, we find that pension fund's characteristics drive the heterogeneity in the factor

⁴See the infographic in the Internet Appendix (Figure 2.2).

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

exposures, which follows our theoretical framework. Consistent with the predictions of our model we find that pension funds with a low funding ratio, high risk aversion, and long liability duration have higher exposures to the investment-grade fixed income market factor but lower exposures to the other factors. The pension funds' characteristics explain 50 percent of the heterogeneity in the return contribution of the fixed income market factors, but only 20 percent in case of the long-short factors. Motivated by the academic literature, we also study the effects of institutional factors, in particular the effects of the pension fund's size and type. Size as measured by AUM results in more global investment portfolios but does not have an effect on the exposures to long-short factors. Corporate pension funds follow market benchmarks more closely as opposed to industry-wide or professional group pension funds. However, correcting for these institutional factors does not further reduce the heterogeneity in average returns across pension funds.

Third, we show that the remaining differences in investment strategies can be attributed to differences in the implied beliefs on expected returns. We infer these implied beliefs by using our theoretical framework and show evidence in favor of this conjecture. We then show that the pension funds reveal these differences in beliefs through their choices of asset management firms, at least to a reasonable degree. We assess the effect of newly hired asset management firms and find that these have a statistically and economically sizeable effect on the factor exposures.

Fourth, we show that regulations and in particular the discount rate, affect investment strategies. In 2007 a fixed discount rate of 4 percent was replaced by risk-free market interest rates to determine the present discounted value of accrued benefit obligations. This change in methodology has led pension funds to increase their exposure to the fixed income market index and to low beta factors.

Literature review

Our study contributes to the literature on investment behavior in a regulated environment. Rauh (2009) shows that underfunded corporate defined-benefit pension funds in the US invest less in equities than do overfunded pension funds. The author states that the incentive of risk management to avoid costly financial distress dominates the shifting of risk to the Pension Benefit Guaranty Corporation (PBGC) in pension fund investing. We extend this study by showing that underfunded pension funds take less risk within an asset class. We find that within their fixed income portfolio, underfunded pension funds invest more in investmentgrade bonds and take less credit risk. This finding confirms the risk management incentive
from Rauh (2009). In the Netherlands, no pension guarantee system exists, but underfunded pension funds may try to shift risks to their sponsors (see Broeders and Chen, 2012). Employer representatives on a pension fund board may therefore push for risk reduction to avoid this shift.

Andonov et al. (2017) and Lu et al. (2019) find that US public pension funds increase their risk-taking in financial markets when interest rates are lower, particularly for underfunded pension funds. This increase is a way that these public pension funds can artificially support their funding ratio because they discount pension liabilities against the expected returns on their assets. This incentive is created through the US GASB guidelines. We add to this work by analysing the investment behavior in a regulatory environment in which the liability discount rate is linked to the term structure of market interest rates. This is similar to the regulatory environment for US and Canadian private pension funds, and some European pension funds. Our results are therefore more in line with the investment behavior of German insurance companies that demand more as opposed to less safe long-term bonds when interest rates are low (Domanski et al., 2017).

Greenwood and Vissing-Jorgensen (2018) show that regulatory changes in the liability discount rate that link to market interest rates affect the yield curve due to a shock in demand for long-term bonds from these investors. Our results also support the view that the liability structure shapes pension funds' investment behavior, in particular within fixed income portfolios. Our results show that pension funds prefer safe long-term bonds as well as securities denominated in euros. More importantly, we extend Greenwood and Vissing-Jorgensen (2018) by showing the existence of a large heterogeneity in the demand for safe long-term bonds whereby this demand is larger for pension funds with a low funding ratio, high risk aversion, or long liability duration.

Our study also contributes to the broad literature that assesses the effect of institutional investors on asset prices. For example, Coval and Stafford (2007), Gutierrez and Kelley (2009), and Dasgupta et al. (2011) present evidence that institutional investors contribute to mispricing. In particular, Edelen et al. (2016) find that institutional investors trade contrary to anomalies. Our results support this finding because we find many factor exposures to be negative on average. We conjecture that regulation with respect to the liability discount rate is one driving force behind the preference for assets in the short leg of the anomaly. For instance, the exposure to the fixed income value and carry factors decreased substantially when the fixed liability discount rate was replaced by risk-free market interest rates in 2007. Dutch and German government bonds resemble the risk-free term structure of interest rates

better as opposed to Italian and Spanish bonds. Yet, at the same time the former have lower value and carry ranks. These two forces may contribute to a negative exposure to the value and carry factors.

The remainder of the study is organized as follows: Section 2.2 provides a model to derive the optimal portfolio weights and factor exposures. A description of the data is given in Section 2.3. In Section 2.4, we estimate the factor exposures and we analyse their drivers in Section 2.5. In Section 2.6 we identify the pension funds' implied beliefs on factor returns. We show how factor exposures changed when the fixed liability discount rate was replaced by market interest rates in Section 2.7. Section 2.8 concludes.

2.2 Motivating model

In this section, we present a model to derive the factor exposures within asset classes and to explain the heterogeneity across pension funds. Our theoretical framework considers the derivation of factor exposures from the perspective of a pension fund.⁵ First, we identify the optimal portfolio weights for a mean-variance investor who optimizes their surplus, that is, the value of assets minus that of the liabilities, subject to borrowing and short-sale constraints. Starting with portfolio weights allows us to closely map the model to the existing mean-variance portfolio theory (Markowitz, 1952) and to include borrowing and short-sale constraints that are typically applicable to pension funds. Second, we show the implication of the portfolio weights for factor exposures.

We start with the liability structure. A pension fund *i* pays benefits $B_{i,t+h}$ to its participants in period t + h. These benefits can in practice take any value, but because we only consider defined-benefit pension funds, we assume that benefits are known at time *t*. This assumption is known as the accumulated benefit obligation (ABO). Dutch pension funds do not guarantee inflation protection and we therefore refrain from indexation policies in our analysis. We also assume that the pension fund has a large enough number of participants such that idiosyncratic longevity risk is fully diversified. The present discounted value of all future benefit payments for pension fund *i* is given by:

$$L_{i,t} = \int_0^\infty B_{i,t+h} \exp\left(-hr_t^h\right) dh, \qquad (2.1)$$

⁵This framework distinguishes itself from the literature that considers the perspective of an individual life-cycle investor as in, for example, Bodie et al. (1992).

in which r_t^h is the regulatory discount rate as observed at time t for time-to-maturity h.⁶ Regulatory discount rates vary widely across jurisdictions. For instance, under the US Government Accounting Standards Board (GASB) guidelines, public pension funds are partially free to discount their liabilities at the expected rate of return on the assets (Andonov et al., 2017).⁷ By contrast, US corporate pension funds use the yield on high-quality corporate bonds. In our case, pension funds in the Netherlands used a fixed discount rate of 4 percent until 2007. However, the regulations introduced in 2007 required Dutch pension funds to use the risk-free term structure of market interest rates based on the euro swap curve as the discount rate. Finance theory argues that risk-free market interest rates are indeed the applicable discount for guaranteed pension benefits to exclude arbitrage (e.g. Brown and Wilcox, 2009; Novy-Marx and Rauh, 2009). The value of the liabilities at time t + 1 is then follows from:

$$L_{i,t+1} = \left(1 + r_{i,t+1}^{L}\right) L_{i,t} \approx \left(1 + \psi_{i,t} r_{t+1}^{b}\right) L_{i,t}, \qquad (2.2)$$

in which $r_{i,t+1}^L$ is the liability return that is approximated by the return on the risk-free bonds traded in the market r_{t+1}^b times $\psi_{i,t}$ that represents the duration of pension liabilities over the duration of those bonds. The value of $\psi_{i,t}$ is typically larger than one because the duration of pension liabilities is larger than the average duration of bonds in the market.⁸

Next, we assume the pension fund has access to M assets, and its wealth evolves as follows:

$$A_{i,t+1} = \left(1 + w'_{i,t}r_{t+1}\right)A_{i,t},$$
(2.3)

in which $w_{i,t}$ is a vector of portfolio weights that pension fund *i* chooses at time *t*, and r_{t+1} is a vector of returns from *t* to t + 1. Following Sharpe and Tint (1990a) and Hoevenaars et al. (2008), we assume that the pension fund has mean-variance preferences over the value of its assets minus the value of its liabilities, or its surplus. We normalize this surplus by

⁶Table 2.12 of the Internet Appendix summarizes the symbols that we use in this study.

⁷New GASB rules distinguish between the discount rate calculations for funded and for unfunded pension funds.

⁸We performed a regression of liability returns on factor exposures and found a regression coefficient of 2.2 on the investment-grade fixed income market returns ($R^2 = 0.76$).

dividing it by the value of assets to get the following optimization problem:

$$\max_{w_{i,t}} \mathbb{E}_{i,t} \left[u \left(\frac{A_{i,t+1} - L_{i,t+1}}{A_{i,t}} \right) \right]$$
$$= \max_{w_{i,t}} \mathbb{E}_{i,t} \left[\frac{A_{i,t+1} - L_{i,t+1}}{A_{i,t}} \right] - \frac{\gamma_i}{2} \operatorname{Var}_t \left[\frac{A_{i,t+1} - L_{i,t+1}}{A_{i,t}} \right], \quad (2.4)$$

subject to

$$w_{i,t}'\iota_M \le c,\tag{2.5}$$

$$w_{i,j,t} \ge 0 \ \forall j, \tag{2.6}$$

in which γ_i captures the pension fund's *i* time invariant risk aversion parameter, ι_M is a vector of ones with length M, c is a constant that defines the constraint on the sum of the weights where typically c = 1 that means the pension fund cannot invest more than its entire wealth, and $w_{i,j,t}$ the weight in asset j where j = 1, ..., M. Solving (2.4) for the portfolio weights $w_{i,t}$ results in (see derivation in Appendix A):

$$w_{i,t}^{*} = \underbrace{\frac{\mathbb{E}_{i,t}[r_{t+1}] + \lambda_{i,t}\iota_{M} + \delta_{i,t}}{\gamma_{i}\operatorname{Var}_{t}[r_{t+1}]}}_{\operatorname{speculative portfolio}} + \underbrace{\frac{\operatorname{Cov}_{t}[r_{t+1}^{b}\iota_{M}, r_{t+1}]\psi_{i,t}\iota_{M}}{\operatorname{Var}_{t}[r_{t+1}]}}_{\operatorname{hedging portfolio}}F_{i,t}^{-1}, \qquad (2.7)$$

with

$$w_{i,j,t}^* \geq 0,$$

$$\delta_{i,j,t} \geq 0,$$

$$\delta_{i,j,t}w_{i,j,t}^* = 0 \quad \forall j.$$
(2.8)

The subjective expectations of pension fund *i* about the expected returns is defined by $\mathbb{E}_{i,t}[r_{t+1}]$. Because second moments can be estimated more accurately than first moments (e.g Merton, 1980), we assume that the variance and covariance of the returns are common knowledge across pension funds. The funding ratio for pension fund *i* is defined as $F_{i,t} = \frac{A_{i,t}}{L_{i,t}}$, $\lambda_{i,t}$ is the Lagrange multiplier for the restriction that $w'_{i,t}\iota_M = c$, and $\delta_{i,t}$ consists of the Kuhn-Tucker multipliers for the restriction that the portfolio weights are nonnegative. If the

Lagrange multiplier is binding, then $\lambda_{i,t}$ equals:

$$\lambda_{i,t} = \frac{c - \left(\frac{\mathbb{E}_{i,t}[r_{t+1}] + \delta_{i,t}}{\gamma_i \operatorname{Var}_t[r_{t+1}]}\right)' \iota_M - \left(\frac{\operatorname{Cov}_t[r_{t+1}^b \iota_M, r_{t+1}]\psi_{i,t}\iota_M}{\operatorname{Var}_t[r_{t+1}]}F_{i,t}^{-1}\right)' \iota_M}{\left(\frac{\iota_M}{\gamma_i \operatorname{Var}_t[r_{t+1}]}\right)' \iota_M}.$$
(2.9)

The solution in (2.7) shows that the optimal portfolio weights consist of the sum of two components: a speculative portfolio and a liability hedge portfolio. The Lagrange multiplier (2.9) ensures that the speculative demand decreases if the hedging demand increases, and vice versa.

Unfortunately, we do not have full access to the portfolio weights of the individual assets. In our empirical analysis, we therefore choose an alternative approach and measure factor exposures. The exposure of the portfolio return r^P to the return on the k^{th} factor r^k is measured as:

$$\beta^k = \frac{\operatorname{Cov}(r^P, r^k)}{\operatorname{Var}(r^k)}.$$
(2.10)

In case the factors are long-short factors, it can be further decomposed to:

$$\beta^{k} = \frac{\operatorname{Cov}(r^{P}, r^{k,L} - r^{k,S})}{\operatorname{Var}(r^{k,L} - r^{k,S})} = \frac{\operatorname{Cov}(r^{P}, r^{k,L})}{\operatorname{Var}(r^{k,L} - r^{k,S})} - \frac{\operatorname{Cov}(r^{P}, r^{k,S})}{\operatorname{Var}(r^{k,L} - r^{k,S})},$$
(2.11)

in which $r^{k,L}$ is the return on the "long leg" of the factor, and $r^{k,S}$ is the return on the "short leg" of the factor. Although the pension fund may be restricted to shorting assets, it can have a positive or a negative exposure to a long-short factor. A positive exposure results from a higher demand for the long leg compared to that for the short leg of the factor, and vice versa. To illustrate this point, assume we have a portfolio consisting of value stocks and growth stocks. The portfolio return equals:

$$r^P = w_V r^V + w_G r^G, (2.12)$$

in which w_V is the portfolio weight of value stocks and r^V is the corresponding return, and w_G is the portfolio weight of growth stocks and r^G is the corresponding return. In this example, let us assume that the portfolio weight of value stocks exceeds the weight of growth stocks, so that $w_V > w_G$. We now explore the exposure of this portfolio return to the long-short factor return. The return correlation between the value and growth stocks is less than one, that is, $\rho_{V,G} < 1$. For a beta neutral factor, we also know that the volatility of the value stock is approximately equal to that of the growth stocks, that is, $\sigma_V \approx \sigma_G$. This condition results in the following factor exposure:

$$\beta^{V-G} = \frac{\operatorname{Cov}(r^P, r^V - r^G)}{\operatorname{Var}(r^V - r^G)} = \frac{(w_V - w_G)\sigma_V^2(1 - \rho_{V,G})}{\operatorname{Var}(r^V - r^G)} > 0$$
(2.13)

In other words, a higher portfolio weight for value stocks compared to growth stocks results in a positive factor exposure to value, and vice versa.

2.2.1 Testable implications

This subsection describes the testable implications that follow from our theoretical framework. To formulate the predictions, we first summarize the data that we use to empirically test the model implications.

In our empirical analysis, we use an investment-grade fixed income market index to represent the return on the set of bonds r_{t+1}^{b} .⁹ Further, we use the following factors for fixed income: credit, value, momentum, carry, and low beta. For equities we use a global market, European market, value, momentum, carry, and low beta factor.

Pension funds report their funding ratio and liability duration. We cannot observe the risk aversion parameter directly, but we conjecture that it will be inversely related to the, so-called "required funding ratio". This required funding ratio is prescribed by law and is comparable to the solvency requirements for banks and insurance companies. If a bank or an insurance company takes more risk, then it will have a higher capital requirement. Similarly, pension funds that have a large mismatch between assets and liabilities have a higher required funding ratio. This higher ratio shows a willingness to accept more risk (Broeders et al., 2020). Thus, we propose the following testable hypotheses:

1. Funding ratio

A low funding ratio increases demand for the investment-grade fixed income market factor and decreases the overall demand for other factors, and vice versa.

2. Risk aversion

Pension funds with a low risk aversion have larger exposures to factors other than the investment-grade fixed income market factor, and vice versa. We approximate risk aversion through the inverse of the required funding ratio.

⁹The empirical analysis is robust to including other proxies, such as a 10-year German government bond.

3. Liability duration

Pension funds with a long liability duration have a high exposure to the investmentgrade fixed income market factor, but lower overall exposure to the other factors, and vice versa.

2.3 Data

2.3.1 Pension fund returns

For the core of our analysis, we use proprietary quarterly return data on Dutch occupational pension funds from 1999Q1 through 2017Q4. The prudential supervisor in the Netherlands collects these data for regulatory purposes. Pension funds report the return on investments as the time-weighted return that takes into account the buying and selling in the asset class during the quarter. As of 2010, pension funds used standardized principles to compute the returns in accordance with the Global Investment Performance Standards (GIPS). Pension funds separately report the overall portfolio return as well as the returns from the equity and the fixed income portfolios. Total returns are in euros net of transaction costs. The returns of the equity and fixed income portfolios exclude the returns from derivative positions. The sample contains 572 distinct pension funds. We correct for pension funds that report the same returns during consecutive periods. Because these are clearly reporting errors, we replace the unvaried returns with missing values.

We distinguish between three different types of pension funds: corporate pension funds, industry-wide pension funds, and professional-group pension funds. Corporate pension funds execute a pension scheme for a particular company. Industry-wide pension funds organize pensions for a specific industry or sector; for example, for civil servants or for the care and welfare sector. These pension funds are typically mandatory, so the collective labor agreement in this sector prescribes that employers must join this pension fund. Professional-group pension funds provide pensions for a specific profession, such as veterinarians or pharmacists. Although corporate and professional-group pension funds are not mandatory, for historical reasons most employers offer a pension scheme to their employees. The fraction of the labor force that participates in a pension scheme exceeds 90 percent. The number of corporate pension funds in the sample is 474, the total number of industry-wide pension funds is 10.

Table 4.1 shows the time series of total AUM for all Dutch pension funds. The AUM

grew by a factor of 2.6 over the sample period. The AUM increases each year with the exceptions of a significant drop during the downturn in the stock market following the burst of the Dot-com bubble in 2002 and following the 2008 financial crisis. A continuous and significant drop in the total number of pension funds occurs during the sample period. In 2000, the total number of pension funds was 676; it lowered to a total of 200 in 2017. This drop is in particular due to a large decrease in the number of small corporate pension funds. Because pension funds cannot go bankrupt, our data does not suffer from a survivorship bias. Instead, for cost-efficiency reasons, small pension funds may decide to discontinue their operations and transfer assets and liabilities to an industry-wide pension fund or an insurance company. Not all pension funds fully report returns each year. The table also shows the AUM of pension funds that fully report returns and are therefore in our sample. These pension funds represent on average 93 percent of the AUM of all Dutch pension funds.

Panel A of Table 2.2 presents the summary statistics for pension funds' equity and fixed income returns and allocations. We measure excess returns against the 3-month Euribor rate that we get from the website of the Dutch Central Bank. The equally weighted average excess return on equities across pension funds and time equals 4.38 percent per year with a standard deviation of 21.28 percent.¹⁰ The negative skewness indicates the equity return series has relatively strong negative values. The mean excess return on fixed income is 3.89 percent per year with a standard deviation of 10.04 percent. The high kurtosis demonstrates fat tails and that is, as we show later, due to the large cross-sectional variation in interest rate hedges. In our analysis, we use equally weighted returns. However, the fact that the Dutch occupational pension fund sector has a few very large industry-wide pension funds is well known. Therefore, for comparison reasons, Table 2.2 also presents the value-weighted statistics for returns. The value-weighted mean excess return for equities equals 4.80 and for fixed income it equals 3.73 percent.

Table 2.2 also presents the strategic allocations to equity and fixed income, the duration of the fixed income portfolio, the funding ratio, the required funding ratio, the liability duration, and the fraction of active participants to total participants (active participants plus retirees). Pension funds invest on average 31 percent in equities and 59 percent in fixed income. The average duration of the fixed income portfolio equals 8.2 years with a substantial standard deviation of 8.7 years that indicates the pension funds vary in the extent to which they hedge interest rate risk with bonds. The funding ratio on average equals 116 percent,

¹⁰We compute the standard deviation by using the law of total variance: $\sigma(r) = \sqrt{\mathbb{E}_i(\operatorname{Var}[r]) + \operatorname{Var}_i(\mathbb{E}[r])}$.

Table 2.1. Total assets under management and number of pension funds: This table shows the total assets under management (AUM) in billion euros and the number of pension funds (N). The left hand columns present all pension funds in the Netherlands and the right hand columns all the pension funds that fully report returns and that are used in our analysis. AUM and N are at the end of each year.

	All		Full repo	\mathbf{rting}
year	AUM	N	AUM	N
1999	463.70	663	418.43	315
2000	480.78	676	453.09	408
2001	471.00	656	445.33	429
2002	429.51	658	405.67	447
2003	489.60	642	463.88	439
2004	529.93	605	510.39	450
2005	610.52	575	576.14	365
2006	657.57	524	604.64	390
2007	683.53	442	665.62	403
2008	576.32	413	557.21	376
2009	663.59	376	632.49	336
2010	746.28	350	729.31	328
2011	802.33	329	784.80	298
2012	897.09	260	753.51	287
2013	937.12	258	845.62	245
2014	$1,\!131.74$	247	984.73	228
2015	$1,\!146.66$	227	$1,\!005.96$	195
2016	$1,\!262.54$	216	$1,\!122.37$	190
2017	$1,\!224.07$	200	$1,\!163.47$	175

and the required funding ratio equals 115 percent. The liability duration on average equals 18.6 years, and the fraction of active participants equals 64 percent. The latter indicates that about a third of the participants is in the retirement phase.

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

Table 2.2. Summary statistics: Panel A reports the summary statistics for pension fund returns, both equally and value weighted. The mean returns and standard deviations of returns are measured across time and pension funds for 1999Q1-2017Q4. We also report the means and standard deviations for equity and fixed income allocations (percent), duration (years), funding ratio (fraction, as of 2007), required funding ratio (fraction, as of 2009), liability duration (years, as of 2007) and the ratio of actives to total participants (percent) that are computed from the quarterly reports. Panel B gives the summary statistics for the factor returns. The returns are annualized and in euros.

Panel A: Pension fund returns and characteristics					
Equally weighted	mean	stdev	skewness	kurtosis	
Excess return equity	4.38	21.28	-0.53	3.51	
Excess return fixed income	3.89	10.04	0.37	5.18	
Value weighted					
Excess return equity	4.80	18.97	-0.45	3.85	
Excess return fixed income	3.73	6.91	0.44	5.48	
Characteristics					
Equity allocation	31.00	9.14			
Fixed income allocation	58.76	11.78			
Duration fixed income portfolio	8.20	8.71			
Funding ratio	1.16	0.16			
Required funding ratio	1.15	0.13			
Liability duration	18.63	5.53			
Fraction of active participants	64.25	24.89			
Panel B: Factor return	ıs				
	mean	stdev	skewness	kurtosis	
Euribor 3-month rate	1.94	0.83	0.22	1.76	
Excess MSCI World Total Return Index	4.99	17.25	-0.70	3.83	
Excess Euro Stoxx 50 Total Return Index	4.07	21.37	-0.32	4.11	
Global value stock	4.00	15.81	0.57	11.51	
Global momentum stock	5.20	16.88	0.26	6.44	
Global carry stock	6.49	6.75	0.17	3.71	
Global low beta stock	11.03	11.93	-0.10	6.81	
Excess Bloomberg Barclays EuroAgg FI Index	2.55	3.66	-0.39	2.76	
Excess Bloomberg Barclays EuroAgg High Yield Index	6.38	14.89	0.42	8.12	
Europe value FI	1.17	5.56	-0.27	5.68	
Europe momentum FI	1.24	4.54	-0.57	7.89	
Europe carry FI	1.84	4.52	0.48	6.46	
Europe low beta FI	0.86	4.41	0.18	3.29	

2.3.2 Factor returns

In this subsection, we turn to the factors that explain the cross-section of returns. To distinguish between market factors and the other factors, we refer to the latter as long-short factors. Although controversy exists regarding whether long-short factor returns are rewards for risk or the result of mispricing, we do not take a stance on the underlying driver of these factor returns. We simply interpret these factors as diversified passive benchmark returns that capture patterns in average returns during the sample period we consider.

For the long-short factors we use the four factors that studies have shown to perform robustly across several asset classes and markets: value, momentum, carry, and low beta. The value factor for equities is a strategy that goes long in value stocks and short in growth stocks. As fixed income generally does not have measures of book value, value bonds are defined as bonds with high positive changes in the 5-year yield or high values for the negative 5-year past returns. Long-term past return measures for value come from de Bondt and Thaler (1985).¹¹ Momentum is defined in exactly the same way for equities and bonds: the past 12-month cumulative return that excludes the most recent month's return (see, e.g. Jegadeesh and Titman, 1993). Carry is defined as an asset's future return that assumes the price remains the same. Equity carry is approximately equal to the expected dividend yield minus the risk-free rate. Bond carry is the return that is earned if the yield curve stays the same over the next time period. Low beta is also similarly defined for stocks and bonds: low exposure to the corresponding market index.

Equity factors

We use the excess market return, value return, momentum return, carry return, and low beta return as factors that explain the pension funds' equity returns. Dutch pension funds have European as well as global equity holdings. The fraction of the equity portfolio they on average allocate to the euro area is 23 percent over the 2007-2017 period; and although we do not have data on the exposure to the euro area prior to 2007, we expect this fraction to be higher.¹² For instance, Berk and van Binsbergen (2015) show that the fraction of mutual funds that invests internationally has significantly increased over the last decade.

¹¹For an extended discussion, see Asness et al. (2013).

 $^{^{12}{\}rm Data}$ on investments in the euro and non-euro areas are published on the website of DNB: https://statistiek.dnb.nl/en/downloads.

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

We therefore include both global and European indices to define the market returns and to account for the currency bias (Maggiori et al., 2020). For the global market factor, we use the quarterly MSCI World Total Return Index in euros; for the European market factor, we use the Euro Stoxx 50 Total Return Index from Bloomberg in euros.¹³

Given that the majority of equity holdings are global, we use global value factors, global momentum factors, global carry factors, and global low beta factors to analyse the equity returns. We take the returns on the value, momentum, and low beta equity factors from the AQR website. The returns on the carry factor are from Ralph Koijen's website. Following the usual factor definitions, the global value and momentum factors are zero-cost, long-short portfolios in individual stocks in the US, the UK, continental Europe, and Japan (Asness et al., 2013). The data for carry and low beta include individual stocks from the following five regions: North America, the UK, continental Europe, Asia, and Australia.

The value, momentum, carry, and low beta returns are all monthly. To match with the pension funds' return cycle, we convert the monthly returns to quarterly returns by means of compounding. We assume pension funds fully hedge currency exposures and convert all dollar returns into euros.¹⁴ The factor returns in euros are the dollar factor returns times the gross return on the exchange rate (Koijen et al., 2018) in which the exchange rate measures the number of euros per dollar. For the summary statistics, we furthermore convert quarterly returns into annual ones.

Panel B of Table 2.2 contains the summary statistics for the factor returns. Within equities, the low beta factor has the highest annualized return (11.03 percent), while value has the lowest (4.00 percent). Next to the market factors, momentum is the most volatile long-short factor over the sample period.

Fixed income factors

Compared to equities, Dutch pension funds invest substantially less globally within their fixed income portfolios. Measured over the period from 2007 through 2017, they invested

¹³The Euro Stoxx 50 Total Return Index represents the 50 largest and most liquid stocks in the euro area. It comprises Austria, Belgium, Finland, France, Germany, Ireland, Italy, Luxembourg, the Netherlands, Portugal, and Spain. The MCSI World index includes the stocks in the Euro Stoxx 50 index.

¹⁴The AQR factors are not currency hedged, while the carry factor is fully hedged. Given that currency only explains a minor part of the returns for equities, our results do not materially change if we assume that the currency exposure is not hedged.

on average 87 percent of the fixed income portfolio in the euro area.¹⁵ Again, we expect this fraction to be even larger prior to 2007. A currency bias for euro fixed income is logical because pension funds' liabilities are also denominated in euros, and fixed income is mainly used to hedge liabilities. We therefore use European factors for fixed income, as opposed to global factors for equities. Because bond returns are largely explained by duration and credit risk, we use the Bloomberg Barclays Euro Aggregate Bond Index and the Bloomberg Barclays Euro High Yield Index in euros as the market and credit factors respectively.¹⁶ Table 2.2 shows that both the equally and value-weighted excess fixed income index. Pension funds have an incentive to invest in bonds with a high duration to match the high duration of their liabilities. The average duration of the fixed income portfolio equals 8.2 (Table 2.2). As such, benchmark durations are typically lower than the portfolio duration of pension funds. An upward-sloping term structure of interest rates therefore (in part) explains the higher pension fund returns.

As opposed to global, European fixed income long-short factors are not available, so we construct the value, momentum, carry, and low beta factors following the methods of Asness et al. (2013), Koijen et al. (2018), and Frazzini and Pedersen (2014). As the purpose of this study is to gain an insight into the factor exposures of institutional investors rather than the construction of factor returns themselves, we use the exact definitions of the aforementioned authors. We include the following European countries in constructing our factors: Austria, Belgium, Denmark, Finland, France, Germany, Italy, Netherlands, Norway, Spain, Sweden, Switzerland, and the UK. All these countries have investment-grade ratings over our sample period. In Appendix B, we describe the exact procedure for how we construct the factors. For all three factors, we assume the investor fully hedges currency exposures against the euro. Again, we convert monthly returns to quarterly returns by means of compounding. In the case of fixed income, carry has the highest annualized return (1.84 percent) followed by momentum (1.24 percent) and value (1.17). Low beta has a relative low average return equal to 0.86 percent, which is consistent with the findings in Frazzini and Pedersen (2014)

¹⁵Data on investments in the euro and non-euro areas are published on the website of DNB: https: //statistiek.dnb.nl/en/downloads.

¹⁶The Bloomberg Barclays Euro Aggregate Bond Index is a benchmark that measures the investmentgrade, euro-denominated fixed-rate bond market that comprises Treasuries, government-related, corporate, and securitized fixed-rate bonds with issuers in Europe. The Bloomberg Barclays Euro High Yield Index measures the market for non-investment grade, fixed-rate corporate bonds with issuers in Europe.

who do not find a significant average return for the global low beta bond factor. Value has the highest standard deviation (5.56 percent) followed by momentum (4.54), carry (4.52), and low beta (4.41). Figure 2.1 shows the evolution of the long-short fixed income factors over time. The correlation matrix in Table 2.14 of the Internet Appendix also confirms the well-known stylized fact in the literature of the strikingly high negative correlation between value and momentum for the European fixed income factors (Asness et al., 2013).

2.4 Factor exposures

In this section, we proceed with the estimation of the (unconditional) factor exposures. We follow a three-step approach to account for measurement errors in the factor exposures. We describe this procedure in subsection 2.4.1. In Subsection 2.4.2 we show the implications of heterogeneity in factor exposures for heterogeneity in average performance across pension funds. Subsection 2.4.3 performs a variance decomposition to quantify how much of the cross-sectional differences in average returns are explained by the factors.

2.4.1 Factor exposures

We follow a three-step approach to account for measurement errors in the factor exposures. These measurement errors stem from the infrequent observations of pension fund returns. First, we estimate the factor exposures for equity and fixed income returns separately by using the arbitrage pricing theory (APT) developed by Stephen Ross (Ross, 1976). We denote equity by a = E and fixed income by a = FI and measure the factor exposures by regressing the excess returns of pension fund i = 1, ..., N for asset class a on the excess factor returns in the following way:

$$r_{i,t}^{a} - r_{t}^{f} = \alpha_{i}^{a} + \beta_{i}^{a'} f_{t}^{a} + \epsilon_{i,t}^{a}, \quad \text{for} \quad i = 1, \dots, N,$$
 (2.14)

in which r_t^f is our proxy for the risk-free rate: the Euribor 3-month rate, f_t^a is a vector of factor returns of length K for asset class a, and $\epsilon_{i,t}^a$ is the idiosyncratic error term with standard deviation σ_i^a . For equities, vector f_t^E contains the following six elements: the global excess market return, the European excess market return, the global value stock return, the global momentum stock return, the global carry stock return, and the global low beta stock return. For fixed income, the vector f_t^{FI} has the following six elements: the European excess investment-grade fixed income market return, the European excess high yield fixed income





Fixed income factors

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

return, the European value fixed income return, the European momentum fixed income return, the European carry fixed income return, and the European low beta fixed income return. In the remainder of the study, we drop the superscript a to simplify the notations. In Table 2.15 of the Internet Appendix, we present the results of the estimated betas using the time-series OLS in detail.

Second, we use a random-coefficient model to estimate the priors for the factor exposures. The estimated factor exposures that use the time-series OLS suffer from measurement error because we only observe quarterly returns (Merton, 1980). The cross-sectional mean and standard deviation may therefore substantially deviate from the true moments. Because the focus is on the cross-sectional mean and standard deviation of factor exposures, we correct for this deviation by using a prior on the mean and the variance in the factor exposures that we derive from a random-coefficients model. Compared to a standard regression model in which the parameters are fixed to a single value, the random-coefficients model allows for cross-sectional variation in the parameters. We specify the random-coefficients model as follows:

$$r_{i,t} - r_t^f = \alpha_i + \beta'_i f_t + \epsilon_{i,t}$$

= $\alpha + \beta' f_t + v'_i f_t + u_i + \epsilon_{i,t},$ (2.15)

in which v_i is a vector of length L that captures all the random-effect coefficients, and $\epsilon_{i,t}$ is the idiosyncratic error term with variance σ_i . Furthermore, we assume that L is equal to the number of factors K; in other words, we allow all factor exposures to vary across pension funds. The exact procedure for estimating the random-coefficients model is in Internet Appendix D. We use the regression coefficients of the random coefficients model as the prior distribution in the analysis. Thus, the prior betas are defined as:

$$\beta_i^k \sim N(\hat{\beta}^k, \hat{\sigma}_{\beta^k}^2) \quad \text{for} \quad k = 1, \dots, K \quad \text{and} \quad i = 1, \dots, N,$$

$$(2.16)$$

in which $\hat{\beta}^k$ is the fixed-effect estimator, and $\hat{\sigma}^2_{\beta^k}$ is the variance in the random effect from Equation (2.15). The variances in the random effects facilitate the testing for the existence of true heterogeneity in the factor exposures. We find significant average factor exposures in both the equity and the fixed income portfolios. Similarly, we also find significant cross-sectional heterogeneity in all factor exposures except for momentum in the fixed income portfolios. The coefficient estimates that include a detailed interpretation appear in the Internet Appendix D.

Third, we derive posterior factor exposures. Following Vasicek (1973), Elton et al. (2003), and Cosemans et al. (2016), we combine the estimated factor exposures from the timeseries OLS regressions with the prior to obtain the posterior betas. These exposures are approximately normally distributed with the following mean and variance:

$$\tilde{\beta}_{i}^{k} = \frac{\hat{\beta}_{i}^{k}/se(\beta_{i}^{k})^{2} + \hat{\beta}^{k}/\hat{\sigma}_{\beta^{k}}^{2}}{1/se(\beta_{i}^{k})^{2} + 1/\hat{\sigma}_{\beta^{k}}^{2}} \quad \text{for} \quad k = 1, \dots, K \quad \text{and} \quad i = 1, \dots, N$$
(2.17)

$$\tilde{\sigma}_{\beta_i^k}^2 = \frac{1}{1/se(\beta_i^k)^2 + 1/\hat{\sigma}_{\beta^k}^2} \quad \text{for} \quad k = 1, \dots, K \quad \text{and} \quad i = 1, \dots, N,$$
(2.18)

in which $\hat{\beta}_i^k$ is the estimated exposure to factor k from the time-series OLS regressions presented in Equation (2.14) for pension fund i, and $se(\beta_i^k)$ is the corresponding standard error. Equation (2.17) shows that the factor exposures of pension funds with less precise sample estimates shrink to the prior. The distribution of posterior factor exposures shows the heterogeneity across pension funds corrected for the measurement error. As a result, the posterior betas are economically interpretable.

For equities, Table 2.3 shows the average exposures to the world and European market factors as equaling 0.67 and 0.27 respectively and with standard deviations equaling 0.18 and 0.15. The sum of the market exposures equals 0.94 that indicates the pension funds, on average, take slightly less systemic risk than the market portfolio. The standard deviations of the posterior market exposures shrink by about one-half compared to the time series regressions that indicates that the substantial variation in the market exposures remains after correcting for the measurement error. The average exposures to value, momentum, carry, and low beta equal -0.05, -0.04, -0.06, and 0.08, respectively. Average negative momentum and value exposures for equities are consistent with recent findings for retail investors in e.g. Luo et al. (2020). The standard deviation in factor exposures for value, momentum, carry, and low beta are 0.07, 0.04, 0.13, and 0.08, respectively. The standard deviation in the posterior factor exposures shrinks by two-thirds for value, five-sixths for momentum, three-fourths for carry, and two-thirds for low beta compared to the time series regressions. A substantial part of the cross-sectional variation in the factor exposures is thus the result of measurement error. Yet, the heterogeneity in the factor exposures remains, especially for value, carry, and low beta.

For fixed income, the average exposure to the investment-grade market factor equals

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

1.11. A fixed income market beta larger than one is consistent with our model, because the duration of the liabilities is much longer compared to the duration of the bond market index. The cross-sectional standard deviation equals 0.31. The standard deviation of the posterior market exposure shrinks by one-half compared to the time-series regressions that indicates the substantial variation in the market exposures remains after correcting for the measurement error. The average exposures to credit risk, value, momentum, carry, and low beta are 0.02, -0.16, 0.07, -0.07, and 0.21, respectively. The cross-sectional standard deviations of credit, value, carry, and low beta equal 0.06, 0.15, 0.09, and 0.18 respectively. Again, substantial variation in factor exposures from the time series regressions is due to measurement error, although the heterogeneity in the factor exposures remains. Because we are not able to detect any variation in the exposure to momentum (Table 2.16), all estimates shrink to the mean; and the standard deviations obtained from the time-series regressions are almost all due to measurement error.

2.4.2 Heterogeneity in average excess returns

The variation in the factor exposures that we observe has consequences for the average excess return differences across pension funds. To determine these differences, we compute the contribution of each of the factors to the average excess returns. We use the posterior betas obtained from Equation (2.17). The contribution of each of the factors is then computed as:

$$\mathbb{E}(r_i^k) = \tilde{\beta}_i^k \lambda^k \quad \text{for} \quad k = 1, \dots, K \quad \text{and} \quad i = 1, \dots, N,$$
(2.19)

in which λ^k is the historical average return for factor k.

The total average excess return for pension fund *i* equals $\mathbb{E}(r_i^e) = \sum_{k=1}^K \tilde{\beta}_i^k \lambda^k$. We rank pension funds based on the total average excess returns from highest to lowest and split them in five equally weighted groups. We then disentangle the contribution of each factor to the average excess returns using (2.19). Table 2.4 summarizes the results for equities, fixed income, and the total portfolio level.

For equities, we find that taking all factors together, the contribution of the factor exposures to average excess returns varies between 2.23 and 6.40 percentage points. Pension funds with the highest average excess returns have a return contribution of the market that is equal to 4.60 percentage points, while this return contribution for pension funds with the lowest average excess returns equals 4.11 percentage points. For the long-short factors,

Table 2.3. Factor exposures: This table displays the cross-sectional means and standard deviations of the OLS betas from Equation (2.14), the prior betas from Equation (2.15), and the posterior betas from Equation (2.17). M,W indicates the excess global market index for equities, M,EU indicates the excess European market index for the corresponding asset class, HY-EU indicates the excess high yield index for fixed income, VAL, MOM, CARRY, and BAB indicate the value, momentum, carry, and low beta factor for the corresponding asset class.

			Panel A	A: Equity		
	C	DLS	P	rior	Pos	terior
	mean	std.dev.	mean	std.dev.	mean	$\operatorname{std.dev.}$
$\hat{eta}_i^{M,W}$	0.656	0.297	0.649	0.184	0.668	0.179
$\hat{\beta}_i^{M,EU}$	0.270	0.311	0.299	0.160	0.273	0.153
$\hat{\beta}_{i}^{VAL}$	-0.060	0.230	-0.043	0.085	-0.048	0.066
$\hat{\beta}_i^{MOM}$	-0.056	0.244	-0.041	0.048	-0.044	0.041
$\hat{\beta}_i^{CARRY}$	-0.106	0.549	-0.054	0.148	-0.057	0.126
$\hat{\beta}_i^{BAB}$	0.088	0.240	0.087	0.107	0.075	0.082
		Pa	anel B: F	Fixed incor	ne	
	OLS					
	С	DLS	P	rior	\mathbf{Pos}	terior
	C mean	DLS std.dev.	P: mean	rior std.dev.	Pos mean	terior std.dev.
$\hat{eta}^{M,EU}_i$	C mean 1.139	DLS std.dev. 0.564	P mean 1.126	rior std.dev. 0.485	Pos mean 1.107	terior std.dev. 0.306
$\hat{\beta}_{i}^{M,EU}$ $\hat{\beta}_{i}^{HY,EU}$	C mean 1.139 0.019	DLS std.dev. 0.564 0.111	P mean 1.126 0.024	rior std.dev. 0.485 0.086	Pos mean 1.107 0.023	terior std.dev. 0.306 0.061
$ \hat{\beta}_{i}^{M,EU} \\ \hat{\beta}_{i}^{HY,EU} \\ \hat{\beta}_{i}^{VAL} $	C mean 1.139 0.019 -0.146	DLS std.dev. 0.564 0.111 0.402	P mean 1.126 0.024 -0.208	rior std.dev. 0.485 0.086 0.155	Pos mean 1.107 0.023 -0.158	terior std.dev. 0.306 0.061 0.147
$ \hat{\beta}_{i}^{M,EU} \\ \hat{\beta}_{i}^{HY,EU} \\ \hat{\beta}_{i}^{VAL} \\ \hat{\beta}_{i}^{MOM} $	C mean 1.139 0.019 -0.146 0.024	DLS std.dev. 0.564 0.111 0.402 0.623	P mean 1.126 0.024 -0.208 0.071	rior std.dev. 0.485 0.086 0.155 0.000	Pos mean 1.107 0.023 -0.158 0.070	terior std.dev. 0.306 0.061 0.147 0.007
$ \hat{\beta}_{i}^{M,EU} \\ \hat{\beta}_{i}^{HY,EU} \\ \hat{\beta}_{i}^{VAL} \\ \hat{\beta}_{i}^{MOM} \\ \hat{\beta}_{i}^{CARRY} $	C mean 1.139 0.019 -0.146 0.024 -0.037	DLS std.dev. 0.564 0.111 0.402 0.623 0.552	P mean 1.126 0.024 -0.208 0.071 -0.079	rior std.dev. 0.485 0.086 0.155 0.000 0.092	Pos mean 1.107 0.023 -0.158 0.070 -0.067	terior std.dev. 0.306 0.061 0.147 0.007 0.087
$ \hat{\beta}_{i}^{M,EU} \\ \hat{\beta}_{i}^{HY,EU} \\ \hat{\beta}_{i}^{VAL} \\ \hat{\beta}_{i}^{MOM} \\ \hat{\beta}_{i}^{CARRY} \\ \hat{\beta}_{i}^{BAB} $	C mean 1.139 0.019 -0.146 0.024 -0.037 0.253	DLS std.dev. 0.564 0.111 0.402 0.623 0.552 0.508	P mean 1.126 0.024 -0.208 0.071 -0.079 0.271	rior std.dev. 0.485 0.086 0.155 0.000 0.092 0.194	Pos mean 1.107 0.023 -0.158 0.070 -0.067 0.205	terior std.dev. 0.306 0.061 0.147 0.007 0.087 0.176

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

the dispersion is much larger for carry and low beta compared to the market. The average return contribution of the carry factor equals 0.44 percentage points at the highest and -1.25 percentage points at the lowest percentiles. For low beta, the return contribution varies between 1.66 and -0.05.

For fixed income, taking both market and long-short factors together, the contribution of the factor exposures vary between 1.91 and 3.95 percentage points. The variation in the contributions of market exposures is larger than for equities and varies between 1.99 and 3.82 percentage points. The long-short factors play a subordinate role. The negative contribution of the long-short factor exposures is due to the typically negative exposure to value and carry factors.

We now turn to the total portfolio level. The average excess return for the total portfolio is computed as the sum of the equity average excess return times the equity weight and the fixed income average excess return times the fixed income weight. All factors taken together, the contribution to the average returns differs by 2.35 percentage points. In other words, pension funds with the highest factor exposures versus pension funds with the lowest factor exposures have a 2.35 percentage point higher average return on the entire portfolio. The contribution of the market factor has values that vary between 2.74 and 4.04 percentage points, and the contribution of the long-short factor exposures has values that vary between -0.53 and 0.51 percentage points.

2.4.3 Variance decomposition

Next, we perform a variance decomposition to quantify how much of the cross-sectional differences in average excess returns are explained by the factor exposures. We first calculate the average excess return of each pension fund per asset class using Equation (2.17):

$$\widetilde{\mu}_i = \widetilde{\alpha}_i + \widetilde{\beta}'_i \lambda \quad \text{for} \quad i = 1, \dots, N,$$
(2.20)

in which λ is a vector of historical average factor returns.

Second, we take the cross-sectional covariance of each side with $\tilde{\mu}$ that is the vector of average excess returns with a length that is equal to N. Because $\text{Cov}(\tilde{\mu}, \tilde{\mu}) = \text{Var}(\tilde{\mu})$, we can divide it by the variance of $\tilde{\mu}$ to get:

$$1 = \frac{\operatorname{Cov}(\tilde{\beta}'\lambda,\tilde{\mu}) + \operatorname{Cov}(\tilde{\alpha},\tilde{\mu})}{\operatorname{Var}(\tilde{\mu})} = \frac{\sum_{k=1}^{K} \operatorname{Cov}(\tilde{\beta}^{k'}\lambda^{k},\tilde{\mu}) + \operatorname{Cov}(\tilde{\alpha},\tilde{\mu})}{\operatorname{Var}(\tilde{\mu})}, \quad (2.21)$$

Table 2.4. Heterogeneity of average excess returns: This table shows the distribution of the average excess return contributions of the market factors, long-short factors, and all factors, to the total equity returns (Panel A), fixed income returns (Panel B), and overall portfolio returns (Panel C). The overall portfolio contribution of the market factors (longshort factors) (all factors) is calculated as the equity weight times the contribution of market factors (long-short factors) (all factors) for equity, plus the fixed income weight times the contribution of the market factor (long-short factors) (all factors) for fixed income. We report the averages within the 0-20th, 20-40th, 40-60th, 60-80th, and 80-100th percentiles. The last column shows the difference between the 100th-80th and the 0-20th percentile. All values are percentage points and annualized.

Panel A: Equity							
	0-20th	20-40th	40-60th	60-80th	80-100th	diff	
All factors	2.23	3.92	4.66	5.26	6.40	4.17	
Global market	3.15	3.49	3.31	3.38	3.32	0.17	
EU market	0.96	0.96	1.20	1.17	1.28	0.32	
Value	-0.31	-0.23	-0.16	-0.15	-0.10	0.21	
Momentum	-0.28	-0.23	-0.22	-0.20	-0.20	0.07	
Carry	-1.25	-0.61	-0.33	-0.08	0.44	1.69	
Low beta	-0.05	0.55	0.87	1.14	1.66	1.71	
Panel B: Fixed inco	ome						
	0-20th	20-40th	40-60th	60-80th	80-100th	diff	
All factors	1.91	2.61	2.93	3.29	3.95	2.04	
Market	1.99	2.53	2.76	3.10	3.82	1.83	
High yield	0.07	0.11	0.19	0.23	0.14	0.07	
Value	-0.18	-0.12	-0.17	-0.22	-0.24	-0.06	
Momentum	0.09	0.09	0.09	0.09	0.09	0.00	
Carry	-0.17	-0.11	-0.12	-0.12	-0.09	0.08	
Low beta	0.12	0.11	0.18	0.21	0.24	0.13	
Panel C: Overall po	ortfolio						
	0-20th	20-40th	40-60th	60-80th	80-100th	diff	
All factors	2.21	3.02	3.51	3.93	4.56	2.35	
Market factors	2.74	3.24	3.46	3.78	4.04	1.31	
Long-short factors	-0.53	-0.21	0.05	0.15	0.51	1.04	

in which $\tilde{\mu}$ and $\tilde{\beta}^{k'} \lambda^k$ are both vectors of length N.

Table 2.5 shows the results for both equity and fixed income returns. The exposures to the global and European market returns explain 14.05 and 5.41 percent of the variation in average equity excess returns, respectively. For the long-short factors, the ones with the most explanatory power are carry and low beta, and they respectively explain 40.15 and 41.22 percent of the variation in average excess returns. Value explains 3.67 of the variation in average excess returns and momentum 2.58 percent. This finding is consistent with the highest return contribution heterogeneity found for the carry and low beta factors. Alpha has negative explanatory power for average excess returns, which means that the pension funds with a high alpha have slightly lower average excess returns.

For fixed income, the European investment-grade market return explains 71.32 percent of the variation in average excess returns, and the high yield return explains 2.79 percent. Low beta, value, and carry explain 4.83, 1.53, and 6.48 of the variation in average excess returns. Consistent with the absence of true heterogeneity across momentum exposures, momentum has negligible explanation power. Alpha has positive explanatory power for average excess returns equal to 13.28 percent.

Table 2.5. Variance decomposition: This table shows how much of the variance in estimated average excess returns $\tilde{\mu}$ is explained by the alpha and the factor exposures for equities and fixed income presented in Equation (2.21). We calculate the average return per asset class of each pension fund using $\tilde{\mu}_i = \tilde{\alpha}_i + \tilde{\beta}'_i \lambda$ in which λ is the average factor return. All values are percentages.

Variance contribution						
Equity		Fixed income				
α Global market EU market Value Momentum Carry Low beta	-7.08 14.05 5.41 3.67 2.58 40.15 41.22	α Market High yield Value Momentum Carry Low beta	$13.28 \\ 71.32 \\ 2.79 \\ 1.53 \\ -0.22 \\ 6.48 \\ 4.83$			

2.5 Drivers of factor exposures

The previous section shows that there is substantial heterogeneity in the factor exposures across pension funds. In this section, we analyse the drivers behind these factor exposures. We have two groups of drivers. The first group are the pension funds' characteristics from our theoretical framework in Section 2.2. This group includes the funding ratio, risk aversion, and liability duration.¹⁷ The second group consists of institutional factors, in particular the pension fund's size and type.

We perform a panel data regression of the funding ratio, the risk aversion, the liability duration, size, and type that we interact with the factor returns:

$$r_{i,t}^{e} = \delta_{0}'f_{t} + \delta_{1}'f_{t} \times F_{i,t-1} + \delta_{2}'f_{t} \times \gamma_{i,t-1} + \delta_{3}'f_{t} \times D_{i,t-1} + \delta_{4}'f_{t} \times AUM_{i,t-1} + \delta_{5}'f_{t} \times Type_{i} + \epsilon_{i,t}, \qquad (2.22)$$

in which $F_{i,t-1}$ is the funding ratio of pension fund *i* at time t-1, γ_{it-1} is the risk aversion of pension fund *i* at time t-1, $D_{i,t-1}$ is the liability duration for pension fund *i* at time t-1, AUM_{it-1} is the AUM for pension fund *i* at time t-1 for the corresponding asset class, and Type_i is the pension fund type. We demean FR_{it-1} , γ_{it-1} , D_{it-1} , and AUM_{it-1} such that δ_0 can be interpreted as the average (industry-wide) pension fund.

2.5.1 Pension funds' characteristics

Funding ratio

Our theoretical framework predicts that pension funds with a low funding ratio should have a high exposure to the fixed income market factor, and vice versa. Moreover, the lower the funding ratio, the less room for the speculative portfolio if the borrowing constraint is binding. Hence, we predict that on average, lower exposures will exist to factors other than the fixed income market factor for pension funds with low funding ratios, and vice versa. For equities, we find that pension funds with a high funding ratio do not have different equity factor exposures (Table 2.6). For fixed income, we find that pension funds with a high funding ratio have less exposure to the market factor and more exposure to the credit

¹⁷In this study, we focus on factor exposures. In the Internet Appendix we show that the predictions from our theoretical framework also empirically hold for the allocation to equity and fixed income (Table 2.17).

and carry factor. A one standard deviation increase in the funding ratio (0.16) decreases the exposure to the market factor by 0.18 and increases the exposure to the credit factor by 0.02 and the carry factor by 0.07. Overall, these findings are consistent with our theoretical framework: pension funds with a low funding ratio invest more in investment-grade fixed income securities that correlate positively with their liabilities, while they have a lower aggregate exposure to the other factors.

Andonov et al. (2017) find the opposite result for US public pension funds. This difference is driven by regulation: The discount rate of US public pension links to the expected returns on assets, while Dutch pension funds have to use a discount rate that links to market interest rates. The incentive to invest in risky assets to artificially improve the funding status of US public pension funds therefore does not apply to Dutch pension funds. A discount rate based on market interest rates is widely used and is the standard for US private pension funds, Canadian, and other European pension funds.

Risk aversion

We use the inverse of the required funding ratio as an implicit measure of the risk aversion in which $\gamma \propto 1/\text{RFR}$, as described in Section 2.2. Our theoretical framework predicts that pension funds with a higher risk aversion coefficient should have a higher exposure to the fixed income market factor and lower exposure to the other factors. For equities, an increase of one standard deviation in the proxy for risk aversion (0.04) slightly decreases the exposure to the global market factor by 0.01. For fixed income, a higher risk aversion coefficient increases the exposure to the market factor substantially and the exposure to momentum slightly. An increase of one standard deviation in the implicit risk aversion coefficient increases the exposure to the market factor by 0.39 and to momentum by 0.04. On the other hand, a higher implicit risk aversion coefficient decreases the exposure to the credit factor, value, carry, and low beta. An increase of one standard deviation in the implicit measure of the risk aversion coefficient decreases the exposure to the credit factor, value, carry by 0.15, and low beta by 0.07. Overall, these findings are consistent with our theoretical framework: pension funds with a higher risk aversion coefficient have a higher exposure to safe assets and less exposure to assets that are uncorrelated with their liabilities.¹⁸

¹⁸The relation between the required funding ratio and asset allocation decisions is mechanical: a higher allocation to equities increases the required funding ratio. However, *within* asset classes there is no such relation. Pension funds that for instance invest more in risky equities (i.e., high beta stocks) do not experience a higher required funding ratio.

Liability duration

Our theoretical framework predicts that pension funds with a long liability duration should have a higher exposure to the fixed income market factor and lower exposures to the other factors, and vice versa. For equities, pension funds with a long liability duration have a higher exposure to the global market index that is approximately offset by a lower exposure to the European market index. Moreover, a one standard deviation increase in the liability duration decreases the exposure to low beta by 0.02. For fixed income, pension funds with a long liability duration have a larger exposure to the market factor and a lower exposure to the credit factor. An increase of one standard deviation in the liability duration increases the exposure to the market factor by 0.40 and decreases the exposure to credit by 0.03. Pension funds with a long liability duration also have lower exposure to value and carry. An increase of one standard deviation in the liability duration funds are consistent with our theoretical framework. We find similar results when using the ratio of active participants relative to the retirees as a proxy for the liability duration (Table 2.18 in the Internet Appendix).

2.5.2 Institutional factors

Size

Size might affect market and long-short factor exposures for two competing reasons. First, large pension funds generally have economies of scale and therefore can bring more expertise to their investment process. As a result, we predict that large pension funds will invest in a more globally and sophisticated manner. Second, due to the price effect of large trades – see, for example, Easley and O'Hara (1987) – pension funds with a substantial AUM in a specific asset class are constrained and might choose to implement factor investing on a low scale relative to pension funds with a lower AUM. The results are in Table 2.7. For equities, size has a positive and significant effect on the exposure to the excess global market return, and a negative and significant effect on the exposure to the excess European market return. A pension fund that is 10 times larger has a 0.06 higher exposure to the global market and a 0.05 lower exposure to the European market. This finding confirms earlier conjectures that large pension funds. Size does not affect the other factor exposures. These results thus do not confirm the conjecture that large pension funds might be constrained in implementing factor strategies.

Table 2.6. Effect of pension fund's characteristics on factor exposures: This table shows the coefficient estimates of Equation (2.22): We regress the pension funds' excess returns on the factor returns and the factor returns interacted with the funding ratio, the risk aversion coefficient that we represent with the inverse of the required funding ratio, and the liability duration during the period from 2009Q1-2017Q4. We control in both specifications for the model parameters included in Table 2.7. Standard errors are in parentheses and clustered at the pension fund level; *p < 0.10, **p < 0.05, and ***p < 0.01.

Panel A: Equity							
	average	funding ratio	risk aversion	liability duration			
$\beta^{M,W}$	0.646^{***}	-0.0012	-0.302*	0.0042^{***}			
	[0.0144]	[0.0538]	[0.1757]	[0.0013]			
$\beta^{M,EU}$	0.289^{***}	0.0445	0.0052	-0.0032**			
	[0.0209]	[0.0459]	[0.1806]	[0.0013]			
β^{VAL}	0.041	-0.107	0.2520	0.0017			
	[0.0299]	[0.0977]	[0.3615]	[0.0025]			
β^{MOM}	-0.0668***	-0.0363	0.1300	-0.0012			
	[0.0231]	[0.0645]	[0.2218]	[0.0015]			
β^{CARRY}	0.0193	0.0836	-0.3960	0.001			
	[0.0219]	[0.1253]	[0.3019]	[0.0020]			
β^{BAB}	0.132^{***}	0.0451	0.0068	-0.0041**			
	[0.0203]	[0.0603]	[0.2308]	[0.0018]			
obs.	8,774	adj. R-sq.	0.863				
		Panel B: Fixe	d income				
		c 1: .:		1. 1.1. 1			
$\circ M EU$	average	funding ratio	risk aversion	hability duration			
$\beta^{m,EC}$	2.193^{***}	-1.123***	9.755***	0.0736^{***}			
$\cap HY EU$	[0.1579]	[0.2414]	[1.4087]				
$\beta^{III,LC}$	-0.0272^{*}	0.107^{***}	-0.749^{***}	-0.0058***			
oVAL	[0.0157]	[0.0345]	[0.1570]	[0.0013]			
β , m	-0.121	-0.0036	-0.965*	-0.0084*			
$_{OMOM}$	[0.0404]	[0.0919]	[0.5134]	[0.0048]			
β^{mom}	0.0636*	-0.0685	1.010	-0.0002			
$\circ CABBY$	[0.0338]	[0.0617]	[0.3225]	[0.0031]			
β°	$-0.00(^{++})$	0.4270^{++}	$-3.(32^{++})$	-0.0255 ^{***}			
OBAB	[0.1024]	[0.1(0)]	[U.9043] 1.coo**	[0.0108]			
β^{DIID}	$-0.1/0^{++}$	U.108U	-1.099 ^{**}	-0.0129			
	[0.0763]	[0.1306]	[0.7453]	[0.0084]			
obs.	8,856	adj. R-sq.	0.574				

80

Pension fund type

Differences in the organizational structure across pension funds might also affect factor exposures. One important difference is that listed companies have to report on the status of their pension funds in their own financial disclosures. Also the risk of the pension fund may be reflected in the risk profile of the company (Jin et al., 2006). This is not the case for the sponsors of an industry-wide pension fund or a professional-group pension fund. Corporate pension funds may therefore be less willing to take risk or to deviate from market benchmarks. In Table 2.7 we use industry-wide pension funds as a reference group. We observe that the only notable difference between corporate pension funds and industry-wide pension funds is that they have a higher exposure to the global market equity index. This is consistent with corporate pension funds following benchmarks more closely as opposed to the industry-wide pension funds.

2.6 Heterogeneity in beliefs

So far, we have seen that pension fund's characteristics and institutional factors drive factor exposures. In this section, we first show that these do not fully explain investment strategies and that there still remains significant heterogeneity in the factor exposures. We then explain the remaining heterogeneity from implied beliefs in the factor returns. We also show that these beliefs (partially) reveal themselves via the choice of the asset management firm.

2.6.1 Remaining heterogeneity in factor exposures

Consistent with our theoretical framework, the previous section shows that part of the heterogeneity results from differences in the pension fund characteristics. The theoretical framework shows, and the empirical analysis confirms, that the relative weight of the liability hedge portfolio increases if the funding ratio is low, when the risk aversion is high and when the liability duration is long. In this section, we adjust the posterior betas for each pension fund such that the liability hedge demand is equal across pension funds and compute the heterogeneity in performance with the adjusted exposures. Table 2.7. Effect of institutional factors on factor exposures: We regress the pension funds' excess returns on the factor returns and the factor returns interacted with the log AUM (size) and the pension fund type (base group: industry pension funds, other groups: corporate and professional group pension funds) during the period from 2009Q1-2017Q4. We control in both specifications for the model parameters included in Table 2.6. Panel A shows the results for equities and Panel B for fixed income. Standard errors are in parentheses and clustered at the pension fund level; *p < 0.10, **p < 0.05, and ***p < 0.01.

Panel A: Equity							
	average	size	corporate	professional			
$\beta^{M,W}$	0.646***	0.0558^{***}	0.0590***	0.0285			
1	[0.0144]	[0.0085]	[0.0156]	[0.0236]			
$\beta^{M,EU}$	0.289***	-0.0506***	-0.0301	-0.0161			
1	[0.0209]	[0.0084]	[0.0210]	[0.0261]			
β^{VAL}	0.041	-0.0127	-0.0098	0.0271			
1	[0.0299]	[0.0154]	[0.0310]	[0.0433]			
β^{MOM}	-0.0668***	-0.0084	0.0199	0.0283			
1	[0.0231]	[0.0111]	[0.0230]	[0.0361]			
β^{CARRY}	0.0193	0.0004	-0.0196	-0.0626*			
1	[0.0219]	[0.0130]	[0.0229]	[0.0361]			
β^{BAB}	0.132***	-0.0057	-0.0326*	0.0052			
,	[0.0203]	[0.0110]	[0.0177]	[0.0324]			
	. ,	Ľ	. ,	L J			
obs.	8,774	adj. R-sq.	0.863				
	,	0 1					
Panel B: Fixed income							
	Par	nel B: Fixed i	income				
	Par	nel B: Fixed i	corporate	professional			
$\beta^{M,EU}$	Par average 2.193***	nel B: Fixed i size 0.1170	corporate -0.0422	professional -0.3200			
$\beta^{M,EU}$	Par average 2.193*** [0.1579]	nel B: Fixed i size 0.1170 [0.0960]	corporate -0.0422 [0.1828]	professional -0.3200 [0.2920]			
$\beta^{M,EU}$ $\beta^{HY,EU}$	Par average 2.193*** [0.1579] -0.0272*	nel B: Fixed i size 0.1170 [0.0960] 0.0125	corporate -0.0422 [0.1828] -0.0067	professional -0.3200 [0.2920] 0.001			
$\beta^{M,EU}$ $\beta^{HY,EU}$	Par average 2.193*** [0.1579] -0.0272* [0.0157]	nel B: Fixed i size 0.1170 [0.0960] 0.0125 [0.0099]	corporate -0.0422 [0.1828] -0.0067 [0.0176]	professional -0.3200 [0.2920] 0.001 [0.0317]			
$\beta^{M,EU}$ $\beta^{HY,EU}$ β^{VAL}	Par average 2.193*** [0.1579] -0.0272* [0.0157] -0.121***	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366	corporate -0.0422 [0.1828] -0.0067 [0.0176] -0.0644	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914			
$eta^{M,EU}$ $eta^{HY,EU}$ eta^{VAL}	Par average 2.193*** [0.1579] -0.0272* [0.0157] -0.121*** [0.0464]	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366 [0.0295]	corporate -0.0422 [0.1828] -0.0067 [0.0176] -0.0644 [0.0543]	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914 [0.1125]			
$eta^{M,EU}$ $eta^{HY,EU}$ eta^{VAL} eta^{MOM}	Par average 2.193*** [0.1579] -0.0272* [0.0157] -0.121*** [0.0464] 0.0636*	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366 [0.0295] -0.0162	$\begin{array}{c} \text{corporate} \\ -0.0422 \\ [0.1828] \\ -0.0067 \\ [0.0176] \\ -0.0644 \\ [0.0543] \\ -0.0675^{*} \end{array}$	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914 [0.1125] -0.0019			
$eta^{M,EU}$ $eta^{HY,EU}$ eta^{VAL} eta^{MOM}	Par average 2.193*** [0.1579] -0.0272* [0.0157] -0.121*** [0.0464] 0.0636* [0.0338]	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366 [0.0295] -0.0162 [0.0194]	$\begin{array}{c} \text{corporate} \\ -0.0422 \\ [0.1828] \\ -0.0067 \\ [0.0176] \\ -0.0644 \\ [0.0543] \\ -0.0675^* \\ [0.0384] \end{array}$	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914 [0.1125] -0.0019 [0.0724]			
$eta^{M,EU}$ $eta^{HY,EU}$ eta^{VAL} eta^{MOM} eta^{CARRY}	Par average 2.193^{***} [0.1579] -0.0272^{*} [0.0157] -0.121^{***} [0.0464] 0.0636^{*} [0.0338] -0.667^{***}	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366 [0.0295] -0.0162 [0.0194] -0.0566	corporate -0.0422 [0.1828] -0.0067 [0.0176] -0.0644 [0.0543] -0.0675* [0.0384] 0.1810	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914 [0.1125] -0.0019 [0.0724] 0.1180			
$eta^{M,EU}$ $eta^{HY,EU}$ eta^{VAL} eta^{MOM} eta^{CARRY}	Par average 2.193^{***} [0.1579] -0.0272^{*} [0.0157] -0.121^{***} [0.0464] 0.0636^{*} [0.0338] -0.667^{***} [0.1024]	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366 [0.0295] -0.0162 [0.0194] -0.0566 [0.0635]	$\begin{array}{c} \text{corporate} \\ -0.0422 \\ [0.1828] \\ -0.0067 \\ [0.0176] \\ -0.0644 \\ [0.0543] \\ -0.0675^* \\ [0.0384] \\ 0.1810 \\ [0.1204] \end{array}$	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914 [0.1125] -0.0019 [0.0724] 0.1180 [0.1955]			
$eta^{M,EU}$ $eta^{HY,EU}$ eta^{VAL} eta^{MOM} eta^{CARRY} eta^{BAB}	Par average 2.193^{***} [0.1579] -0.0272^{*} [0.0157] -0.121^{***} [0.0464] 0.0636^{*} [0.0338] -0.667^{***} [0.1024] -0.170^{**}	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366 [0.0295] -0.0162 [0.0194] -0.0566 [0.0635] 0.0138	corporate -0.0422 [0.1828] -0.0067 [0.0176] -0.0644 [0.0543] -0.0675* [0.0384] 0.1810 [0.1204] 0.1350	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914 [0.1125] -0.0019 [0.0724] 0.1180 [0.1955] 0.1420			
$eta^{M,EU}$ $eta^{HY,EU}$ eta^{VAL} eta^{MOM} eta^{CARRY} eta^{BAB}	Par average 2.193^{***} [0.1579] -0.0272^{*} [0.0157] -0.121^{***} [0.0464] 0.0636^{*} [0.0338] -0.667^{***} [0.1024] -0.170^{**} [0.0763]	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366 [0.0295] -0.0162 [0.0194] -0.0566 [0.0635] 0.0138 [0.0469]	$\begin{array}{c} \text{corporate} \\ -0.0422 \\ [0.1828] \\ -0.0067 \\ [0.0176] \\ -0.0644 \\ [0.0543] \\ -0.0675^* \\ [0.0384] \\ 0.1810 \\ [0.1204] \\ 0.1350 \\ [0.0894] \end{array}$	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914 [0.1125] -0.0019 [0.0724] 0.1180 [0.1955] 0.1420 [0.1629]			
$eta^{M,EU}$ $eta^{HY,EU}$ eta^{VAL} eta^{MOM} eta^{CARRY} eta^{BAB}	Par average 2.193^{***} [0.1579] -0.0272^{*} [0.0157] -0.121^{***} [0.0464] 0.0636^{*} [0.0338] -0.667^{***} [0.1024] -0.170^{**} [0.0763]	size 0.1170 [0.0960] 0.0125 [0.0099] -0.0366 [0.0295] -0.0162 [0.0194] -0.0566 [0.0635] 0.0138 [0.0469]	$\begin{array}{c} \text{corporate} \\ -0.0422 \\ [0.1828] \\ -0.0067 \\ [0.0176] \\ -0.0644 \\ [0.0543] \\ -0.0675^* \\ [0.0384] \\ 0.1810 \\ [0.1204] \\ 0.1350 \\ [0.0894] \end{array}$	professional -0.3200 [0.2920] 0.001 [0.0317] 0.0914 [0.1125] -0.0019 [0.0724] 0.1180 [0.1955] 0.1420 [0.1629]			

Formally, we adjust the posterior betas of each pension fund as follows:

$$\tilde{\beta}_{adj,i}^{k} = \tilde{\beta}_{i}^{k} - \hat{\delta}_{1}^{k} \times \overline{F}_{i} - \hat{\delta}_{2}^{k} \times \overline{\gamma}_{i} - \hat{\delta}_{3}^{k} \times \overline{D}_{i} \quad \text{for} \quad k = 1, \dots, K \quad \text{and} \quad i = 1, \dots, N.$$
(2.23)

Because the time series averages \overline{F}_i , $\overline{\gamma}_i$, and \overline{D}_i are defined relative to the cross-sectional sample average, the adjusted factor exposures decrease for a positive coefficient (i.e. $\hat{\delta}_1^k, \hat{\delta}_2^k, \hat{\delta}_3^k > 0$) when the funding ratio, risk aversion, or liability duration is higher than average, and vice versa. We do the same for the institutional factors. The heterogeneity that remains is unexplained by the pension fund characteristics or institutional drivers.

We redo the analysis from Section 2.4 (subsection 2.4.2) and Table 2.8 summarizes the results. In Panel A, we present the unadjusted excess returns. Pension funds with the highest factor exposures versus pension funds with the lowest factor exposures have a higher average return by 2.35 percentage points on the entire portfolio. Of this difference, 1.30 percentage points are driven by market factors and 1.05 by long-short factors. In Panel B, we correct the excess returns for the pension fund characteristics. The return difference between the top and bottom percentiles is 1.50 percentage points. The contribution of the market factors varies from 3.22 to 3.91 percent, and the contribution of long-short factors varies between -0.25 and 0.56 percent. A total return difference of 1.50 percentage points that cannot be explained by the pension fund characteristics is an economically sizable effect. A lower annual return by 1.50 percentage points decreases the expected retirement income by 32 percent over a 40-year accrual phase or increases contributions by 46 percent to get the same income.

The heterogeneity in the average excess returns remains the same in Panel C if we also adjust the factor exposures for institutional factors in the same way as in (2.23). Because pension funds do not differ substantially in their aggregate equity market exposure, Panels B and C show that the differences in the pension fund characteristics account for roughly 50 percent of the heterogeneity in the return contribution of the fixed income market factors. The pension fund characteristics only explain 20 percent of the heterogeneity in the return contribution of the long-short factors.

Finally, in Panel D we redo the analysis for a subsample of pension funds that are in the sample for at least 24 quarters to further reduce the effect of the measurement error. For this group of pension funds, we find a difference in average excess returns of 1.16 percentage points, and this is equivalent to a difference in expected retirement income of 24 percent.

Controversy about the performance of long-short factors exists. However, excluding longshort factors all together, we still find an unexplained difference in the average annual returns of roughly 70 basis points, which is equivalent to a difference in expected retirement income of 16 percent.

2.6.2 Implied beliefs on expected factor returns

The substantial heterogeneity in the average excess returns that is left after the correction for differences in the pension fund characteristics and institutional drivers may also indicate that pension funds differ in their beliefs about factor returns, particularly so for equities. To show this heterogeneity we identify the pension funds' unconditional implied beliefs about expected (excess) factor returns. To do so, we apply the method as described in Shumway et al. (2011). In their work, they assume that fund managers choose portfolio weights such that they maximize their expected returns over a benchmark while minimizing the tracking error volatility. They find true beliefs to be:

$$\mu_i \approx \gamma_i \delta_i \Sigma_i (w_i - q_i) - \lambda \mathbf{1} \qquad \text{for} \qquad i = 1, \dots, N, \tag{2.24}$$

in which Σ_i is the variance-covariance matrix of returns that is estimated with historical return data and is therefore similar across managers ($\Sigma_i = \Sigma$), w_i are the portfolio weights, q_i are the benchmark portfolio weights, γ_i is the risk aversion parameter of fund manager i, δ_i is the total precision of fund manager i, and λ is the Lagrange multiplier of the borrowing constraint. The total precision parameter measures the informedness of the fund manager about future returns and is the sum of two parts $\delta_i = \tau^{-1} + \tau_i^{-1}$, in which τ^{-1} is the precision of the prior on expected returns, and τ_i^{-1} is the precision of a signal about the expected returns of fund manager i.

The true beliefs are an affine function of the implied beliefs in which the *i*th fund manager's implied beliefs about the expected returns, $\hat{\mu}_i$, are derived in Shumway et al. (2011) as follows:

$$\hat{\mu}_i = \Sigma_i (w_i - q_i) \qquad \text{for} \qquad i = 1, \dots, N.$$
(2.25)

We can apply this framework to our model in Section 2.2. Because we cannot observe all the parameters required to identify the true beliefs, we assume reasonable parameter values to get estimates of the implied beliefs on the expected factor returns. The results that Table 2.8. Remaining heterogeneity of average excess returns: This table shows the distribution of the average excess return contributions of market factors, long-short factors, and all factors to the overall portfolio returns for unadjusted returns (Panel A), returns corrected for the pension fund characteristics (Panel B), and corrected for both the pension fund characteristics and the institutional factors (Panel C). Panel D uses the same specification as Panel C but for pension funds that are at least 24 quarters in the sample. The overall portfolio contribution of the market factors (long-short factors) (all factors) is calculated as the equity weight times the contribution of market factors (long-short factors) (all factors) for equity, plus the fixed income weight times the contribution of the market factor (long-short factors) (all factors) for fixed income. We report the averages within the 0-20th, 20-40th, 40-60th, 60-80th, and 80-100th percentiles. The last column shows the difference between the 80th-100th and the 0-20th percentile. All values are percentage points and annualized.

Panel A: Unadjusted returns							
	0-20th	20 - 40th	40 - 60th	60-80th	80-100th	diff	
All factors	2.21	3.02	3.51	3 93	4 56	2.35	
Market factors	2.74	3.24	3.46	3.78	4.04	1.30	
Long-short factors	-0.53	-0.21	0.05	0.15	0.51	1.04	
C C							
Panel B: Returns c	orrected	for pensio	n fund ch	aracteristi	CS		
	0.001	22.40.1	10 00 1		00.1001	1.0	
	0-20th	20-40th	40-60th	60-80th	80-100th	diff.	
All factors	2.97	3.36	3.65	4.05	4.47	1.50	
Market factors	3.22	3.57	3.54	3.82	3.91	0.69	
Long-short factors	-0.25	-0.20	0.10	0.23	0.56	0.81	
Panel C: Returns c	orrected	for pensio	n fund ch	aracteristi	cs and inst	itutional factors	
	0-20th	20-40th	40-60th	60-80th	80-100th	diff	
All factors	3 15	3 58	3.86	4 30	4.65	1 50	
Market factors	3 30	3.00	3.00	4.05	4.05	0.68	
	0.04	0.10	0.10	4.00	4.07	0.00	
Long-short factors	-0.24	-0.21	0.13	0.25	0.58	0.82	
Panel D: Subsampl	Panel D: Subsample of pension funds						
	0.001	20 401		60.001 ¹	00.100/1	1.0	
	0-20th	20-40th	40-60th	60-80th	80-100th	diff.	
All factors	2.95	3.42	3.55	3.85	4.11	1.16	
Market factors	3.21	3.48	3.60	3.81	3.94	0.73	
Long-short factors	-0.26	-0.07	-0.04	0.04	0.17	0.43	

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

follow should therefore be interpreted as approximations of the true beliefs in which we are particularly interested in the magnitude of differences in the expected returns across pension funds.

As in Shumway et al. (2011), to measure implied beliefs, we refrain from private signals and set $\tau_i^{-1} = 0$. We also assume that pension funds have the same overall precision in the prior equal to $\tau = 1$. Together with the assumption of no private signals ($\tau_i^{-1} = 0$), we have $\delta_i = 1$. A precision in the prior equal to $\tau = 1$ means that pension funds have a prior $p(\mu_0)$ that is normally distributed with a mean μ and a variance-covariance Σ , that are, for instance, based on historical returns:

$$p(\mu_0) \sim N(\mu, \Sigma). \tag{2.26}$$

For the benchmark factor exposures q_i , we assume an exposure of one to the global market factor and a zero exposure for all the other factors for equities. For fixed income, we assume an exposure of one to the investment-grade fixed income market factor and zero to all other factors. These weights corresponds to a passive investor who follows the benchmark exactly.

Using our model in Section 2.2 in an unconditional setting and applying the above assumptions to (2.24), we can derive the implied beliefs about the expected factor returns for pension fund *i* as:

$$\hat{\mathbb{E}}_{i}[r_{t+1}] = \gamma_{i} \operatorname{Var}[r_{t+1}](\beta_{i} - q_{i}) - \gamma_{i} \operatorname{Cov}[r_{t+1}^{b} \iota_{N}, r_{t+1}] \psi_{i} \iota_{N} F_{i}^{-1} \quad \text{for} \quad i = 1, \dots, N.$$
(2.27)

As opposed to Shumway et al. (2011), we do not get rid of γ_i when estimating the implied beliefs as we do have information about the risk aversion coefficient of pension funds. Additionally, compared to (2.25), we correct implied beliefs for the liability hedge demand of pension funds.

As we are interested in the unconditional expectation of returns, we take the average funding ratio over the sample period as the estimate for F_i . We apply the same method for the liability duration and divide it by the typical duration of the fixed income market index of seven years to compute ψ_i . We represent γ_i with $\gamma_i \approx 6 \times \frac{1}{RFR_i}$ in which RFR_i indicates the average required funding ratio for pension fund *i*. As the average required funding ratio equals 1.15, $\gamma_i = 6 \times \frac{1}{1.15} = 5$ means that there is a risk aversion parameter of five for the average pension fund. Because we are particularly interested in the crosssectional heterogeneity in implied beliefs across pension funds, the precise magnitude of the average risk aversion coefficient is of less importance.

Table 2.9 shows the results for the annualized implied beliefs (2.27) on the expected factor returns and conditional on all pension funds having the same informedness. For equities, a median pension fund has positive implied beliefs about the European market factor (1.10 percentage points), while they are slightly negative for the global market factor (-0.12 percentage points). This is consistent with a home/currency bias to European countries. For value and low beta, pension funds have positive implied beliefs, while they are negative for momentum and carry. The median implied belief for the value factor equals 0.41 percentage points and equals -0.40 percentage points for momentum, -0.21 for carry, and 0.39 for low beta. This belief means that pension funds on average expect a higher return on value and on low beta of 0.80 percentage points compared to momentum. There is substantial heterogeneity in the implied beliefs about the expected factor returns. For instance, the pension funds with the most pessimistic views on value expect a negative return of 0.34 percentage points on top of the benchmark return, while pension funds with the most optimistic views expect a positive return of 1.30 percentage points.

For fixed income, the median implied beliefs on the investment-grade market factor equals -0.03 percentage points. The heterogeneity in the implied beliefs for the market factor is limited and indicates that when correcting for the hedging demand, pension funds disagree far less about the expected returns on the market factor. For the credit factor, the implied beliefs equal on average -0.19 and have substantial heterogeneity across pension funds. They range from -1.18 to 0.49 percentage points. For the value factor, the implied beliefs equal -0.46, 0.28 for momentum, -0.27 for carry, and 0.45 for low beta. The greatest heterogeneity in the implied beliefs on the expected long-short factor returns exists for low beta in which pension funds with the most pessimistic views on low beta expect a return of zero percentage points, while pension funds with the most optimistic views expect a positive return of 0.81 percentage points, both are on top of the benchmark return.

Other choices for the benchmark exposures q_i may also come to mind, such as the average factor exposures across pension funds. However, the different choices of the benchmark factor exposures result in a shifted distribution to either the right or left from the one in Table 2.9 but does not affect the cross-sectional heterogeneity across pension funds.

2.6.3 The effect via the choice of asset management firms

We next address the question of whether the differences in implied beliefs are intentional choices by the pension funds. We hypothesize that if beliefs are intentional, then they will Table 2.9. Implied beliefs on expected factor returns: Panel A gives the statistics of the implied beliefs on the expected factor returns for equities, and Panel B shows the results for fixed income. Column 1 shows the historical mean of the factor returns over our sample period, and columns 2-6 show the implied beliefs on top of the benchmark return. The results are derived from Equation (2.27). Panel A shows the results for equities and Panel B for fixed income. We report the 10th, 25th, 50th, 75th, and 90th percentiles. All values are percentage points and annualized.

Panel A: Equity						
	mean	$10 { m th}$	$25 \mathrm{th}$	$50 { m th}$	$75 \mathrm{th}$	90th
Benchmark return	4.99					
Global market index	4.99	-1.53	-0.79	-0.12	0.12	0.71
European market index	4.07	-0.18	0.00	1.10	1.86	2.70
Value	4.00	-0.34	0.00	0.41	0.85	1.30
Momentum	5.20	-1.01	-0.70	-0.40	0.00	0.06
Carry	6.49	-0.55	-0.40	-0.21	0.00	0.00
Low beta	11.03	-0.15	0.00	0.39	0.75	1.11
Panel B: Fixed income						
	mean	10th	$25 \mathrm{th}$	50th	$75 \mathrm{th}$	90th
Benchmark return	2.55					
Global market index	2.55	-0.33	-0.19	-0.03	0.02	0.17
High yield	6.38	-1.18	-0.78	-0.19	0.08	0.49
Value	1.17	-0.71	-0.59	-0.46	-0.16	0.00
Momentum	1.24	0.00	0.11	0.28	0.39	0.47
Carry	1.84	-0.41	-0.34	-0.27	-0.14	0.00
Low beta	0.86	0.00	0.11	0.45	0.66	0.81

show up in the mandates that pension funds give to external asset management firms.

Pension funds do not necessarily manage assets themselves. In fact, most Dutch pension funds delegate the implementation of their investment strategy to for-profit asset management firms through asset management mandates (e.g. Binsbergen et al., 2008; Blake et al., 2013). Although the information on these mandates is scarce, pension funds do report each quarter the name of the asset management firm that executes at least 30 percent of the total AUM on behalf of the pension fund.¹⁹ These firm names are available for the period from 2009 through

¹⁹For confidentiality reasons, we cannot disclose the names of the asset management firms.

2017 and facilitate the analyzation of the effect that the choice of the asset management firm has on factor exposures. Furthermore, to extend the supervision data, we manually check the asset management firm as reported in pension funds' annual reports. Some pension funds report multiple asset management firms (roughly 15 percent of the sample). Because we do not observe in either the supervision data or the annual reports the fraction of assets managed by each of those asset management firms, we are not able to clearly identify the changes due to these firms. Thus, apart from the pension funds that have multiple asset management firms, we can now identify whether pension funds switch from one asset management firm to another and in which quarter.

We have two important reasons to look at changes in asset management firms as opposed to those contracted by the pension fund at a specific point in time. First of all, looking at these changes rules out the possibility that we will find effects simply because the asset management firm correlates with unobservable time-invariant pension fund characteristics. Second, pension funds are likely to hire a new asset management firm if they want to implement a change in their beliefs. In the process of contracting a new asset management firm, pension funds typically do a "search" that is supported by specialised consultants (e.g. Del Guercio and Tkac, 2002; Goyal and Wahal, 2008). Once an asset management firm is selected the mandate is agreed upon according to the preferences of the pension fund.

We focus on asset management firms that were newly hired by at least two pension funds during 2009Q2-2017Q4. A new hire means that we observe a different asset management firm from one quarter to the other. These observations result in a total of 10 asset management firms that gained new business from 59 Dutch pension funds of the 350 pension funds that are in our sample over the period from 2009Q2-2017Q4, which is equivalent to 17 percent of the pension funds. We subsequently run the following regression:²⁰

$$r_{i,t}^{e} = \delta'_{0}f_{t} + \delta'_{1}(f_{t} \times \mathrm{AM}'_{i,t-1})\iota_{10} + \epsilon_{i,t}, \qquad (2.28)$$

in which $AM_{i,t-1}$ is a vector of length 10 and equals 1 for each quarter that the corresponding asset management firm is hired by pension fund *i* after 2009Q1 and zero otherwise; ι_{10} is a vector of ones with length 10.

Table 2.10 shows the results for this regression and indicates that changing asset management firms has an effect on factor exposures in a substantial amount of cases. For equities, pension

²⁰For confidentiality reasons, we cannot disclose the names of the asset management firms.

funds that contract asset management firms 3, 4, 5, and 6 get a significantly higher exposure to the global market index. Therefore, it is likely that these asset management firms are deliberately hired to implement a more global allocation to equities. Other pension funds hire asset management firm 9 to lower their global equity allocation and to increase their allocation of European equities. Pension funds switching to asset management firm 1 result in higher low beta exposures. Pension funds that hire asset management firm 8 obtain negative carry exposures, while the ones that hire asset management firms 2, 6, and 8 obtain negative low beta exposures. The economic magnitudes of these changes are substantial. For instance, pension funds that hire asset management firm 1 have an exposure of 0.17 to low beta compared to an average exposure across all pension funds of 0.06. For fixed income, we find that hiring asset management firms 5, 6, and 7 result in lower exposures to the market index. For high yield, asset management firms 5 and 6 result in a substantially higher credit exposure: 0.26 and 0.37 compared to the average of -0.04. Pension funds that hire asset management firms 3, 7, or 8 have higher value exposures. Asset management firms 3, 6, and 8 result in lower carry exposures.

During the sample period, 59 pension funds switch to one of the ten asset management firms, so we do not have a large power for statistical significance. Despite this, we still find statistical significant, and economically sizeable, effects of asset management firms on some of the factor exposures. We therefore argue that these findings support the idea that factor exposures are at least to a reasonable degree driven by choices about beliefs made by pension funds. Pension funds may change asset management firms for multiple reasons, such as low (excess) returns delivered by the old asset management firm, a change in strategic asset allocation, or a change in beliefs. In most cases pension funds that switch to a new asset management firm will select one that can execute the pension funds' investment policy. And at least a fraction of pension funds will change asset management firms because they have a change in beliefs. This is indeed supported by our analysis. This is furthermore consistent with the findings in Del Guercio and Tkac (2002) who show that quantitative performance variables have a much lower explanatory power in explaining flows for pension fund managers as opposed to mutual fund managers. Importantly, they argue that pension funds regularly change asset management firms because they fail to stay within the guidelines of the investment mandate, regardless of their performance.
Table 2.10. Effect of asset management firm changes on factor exposures: This table shows the coefficient estimates of Equation (2.28): We regress the pension funds' excess returns on the factor returns and the factor returns interacted with the changes in asset management firms (AM1-AM10) during the period from 2009Q2-2017Q4. Panel A shows the results for equities and Panel B for fixed income. Standard errors are in parentheses and clustered at the pension fund level; *p < 0.10, **p < 0.05, and ***p < 0.01.

Panel A: Equity						
$eta^{M,W}$	average 0.711***	AM1 -0.055	AM2 -0.168	AM3 0.0890**	AM4 0.122**	AM5 0.101**
$\beta^{M,EU}$	[0.0063] 0.242^{***}	[0.0492] -0.0075	[0.1094] 0.330^{***}	[0.0352] -0.0367	[0.0546] -0.169***	[0.0478] -0.126***
β^{VAL}	[0.0057] 0.0311** [0.0122]	[0.0377] -0.0474 [0.0728]	[0.1176] - 0.260^{*} [0.1547]	[0.0321] -0.0095 [0.0515]	[0.0458] -0.218** [0.0052]	[0.0436] 0.0871 [0.0001]
β^{MOM}	[0.0125] -0.0552*** [0.0076]	[0.0728] -0.0112 [0.0506]	[0.1347] 0.139 [0.1320]	[0.0515] 0.0168 [0.0419]	[0.0952] - 0.222^{***} [0.0708]	[0.0991] 0.0047 [0.0690]
β^{CARRY}	0.0271***	-0.0298 [0.0689]	[0.1020] 0.218 [0.1615]	-0.0388 [0.0352]	[0.0100] 0.0092 [0.0803]	-0.144 [0.1003]
β^{BAB}	0.0551^{***} [0.0144]	0.110** [0.0550]	-0.432^{***} [0.1655]	-0.0336 [0.0330]	0.0029 [0.0530]	0.0687 [0.0965]
		AM6	AM7	AM8	AM9	AM10
$\beta^{M,W}$		0.132^{***}	-0.0465	0.0585	-0.201*** [0.0448]	-0.0727
$\beta^{M,EU}$		[0.0427] - 0.141^{***} [0.0351]	[0.0002] -0.0734 [0.0617]	-0.0098	0.196***	0.0901
β^{VAL}		[0.0351] -0.0247 [0.0650]	-0.175	-0.0069	[0.0401] 0.00053 [0.0053]	-0.1120
β^{MOM}		-0.0801* [0.0418]	-0.101	[0.0023] -0.00486 [0.0420]	-0.0185	[0.0995] -0.001 [0.0823]
β^{CARRY}		[0.0410] -0.0263 [0.0544]	0.0037	-0.0948** [0.0460]	0.0208	[0.0025] 0.126 [0.1010]
β^{BAB}		[0.0344] -0.0917** [0.0383]	[0.1210] 0.137 [0.1159]	[0.0400] -0.109^{**} [0.0510]	[0.0038] 0.0607 [0.0711]	[0.1010] -0.0452 [0.1030]
obs.	9,319	adj. R-sq.	0.867			

		Panel	B: Fixed in	come		
		A N / 1	4 1 4 6	4 1 4 9	Δ λ <i>π</i> 4	4 N / F
$\circ M EU$	average	AMI	AM2	AM3	AM4	AM5
$\beta^{m,LC}$	2.183***	-0.21	-0.787	0.702	0.778	-0.795***
$\sim UV EU$	[0.0550]	[0.3366]	[2.2588]	[0.5028]	[0.7524]	[0.1811]
$\beta^{\mu \mu, EU}$	-0.0444***	-0.0351	0.506	-0.124	-0.111	0.306***
- 17 4 7	[0.0064]	[0.0813]	[0.7004]	[0.1872]	[0.2772]	[0.0778]
β^{VAL}	-0.175***	0.0537	-0.764	0.273**	0.39	0.104
14014	[0.0197]	[0.1203]	[1.1188]	[0.1146]	[0.3567]	[0.0893]
β^{MOM}	0.0285^{**}	-0.0906	-0.081	0.250^{*}	-0.0806	0.118
	[0.0120]	[0.0722]	[0.3436]	[0.1394]	[0.1747]	[0.1375]
β^{CARRY}	-0.486***	0.125	0.828	-0.948***	-0.236	-0.126
	[0.0374]	[0.1967]	[1.2438]	[0.2721]	[0.4979]	[0.1491]
β^{BAB}	-0.0585**	0.303	0.761	-0.378*	0.644	0.117
	[0.0278]	[0.2233]	[0.8373]	[0.2253]	[0.4971]	[0.1055]
		AM6	AM7	AM8	AM9	AM10
$\beta^{M,EU}$		-0.816**	-1.055***	0.983	1.102	1.084
		[0.3786]	[0.3360]	[0.7955]	[0.7147]	[0.9558]
$\beta^{HY,EU}$		0.416^{***}	0.12	-0.236	-0.197	-0.144
		[0.1514]	[0.0967]	[0.2169]	[0.2181]	[0.2972]
β^{VAL}		0.125	0.200***	0.486**	0.203	0.00876
		[0.0970]	[0.0727]	[0.2178]	[0.2409]	[0.3613]
β^{MOM}		0.0168	-0.0704	0.0525	0.195	0.112
		[0.0838]	[0.0795]	[0.1485]	[0.1746]	[0.2104]
β^{CARRY}		-0.356*	0.229	-0.870*	-0.695	-0.263
		[0.1984]	[0.1834]	[0.4476]	[0.4285]	[0.5508]
β^{BAB}		-0.32	-0.182	-0.453	-0.136	-0.268
		[0.1958]	[0.1712]	[0.3198]	[0.3482]	[0.4591]
obs.	9,435	adj. R-sq.	0.534			

2.7 Effect of the liability discount rate on factor exposures

Pension funds operate in a highly regulated environment; therefore, we find that regulations affect pension fund investment strategies. Halfway through our sample period, pension funds in the Netherlands experienced an important change with the introduction of riskbased regulation in 2007. The main elements of this framework include the marked-tomarket valuation of assets and liabilities and a risk-based minimum required funding ratio. For pension funds, one of the key channels through which regulations affect investment strategies is the liability discount rate (Andonov et al., 2017). Before 2007, pension funds in the Netherlands used a fixed rate of 4 percent to discount liabilities. Under such a discount rule, liabilities artificially contain no interest rate risk. The fixed discount rate was replaced by the full term structure of market interest rates in 2007. Consequently, the present value of liabilities fluctuates significantly with changes in the market interest rates. Furthermore, before 2007 there were no risk-based minimum funding requirements. This lack changed in 2007 and since then the risk-based funding requirement has been implemented. This requirement is derived from a well-known value-at-risk (VaR) concept, whereby risk is measured as the mismatch between the assets and liabilities. Pension funds' investment strategies affect the funding requirement. Pension funds that better match assets and liabilities through investing more in long-term bonds, have a lower required funding ratio, and vice versa.

To show the effect of these regulatory changes, we split the sample period in two and identify the factor exposures prior to and after 2007. Table 2.11 shows the results. We observe a large change in the exposure to the Euro fixed income market index. Prior to 2007, pension funds had an average exposure to the market of 1.02. After 2007 this average exposure increased to 1.23 that reflected that they allocated more to long-term bonds. This finding is consistent with our theoretical framework, because liability hedge demand was nonexistent from a regulatory perspective prior to 2007 but became apparent after 2007. We also observe a decrease in the exposure to the value and carry factors for fixed income. Dutch and German government bonds resemble the risk-free term structure of interest rates better as opposed to Italian and Spanish bonds. Yet, at the same time the former have lower value and carry ranks.²¹ These two forces may contribute to a negative exposure to the value and carry factors. For equities, we observe a small shift from global to European stocks.

²¹We obtain these ranks from the construction of the European fixed income factors.

This way pension funds may reduce the exchange rate risk and lower the risk-based funding requirement. Another striking consequence of the change in regulation is the increase to the low beta factor for both equities and fixed income. Because of risk-based regulations, pension funds may aim to decrease the downside risk of their portfolios by investing more in low beta assets. Changes in long-short factor exposures may also result from developments and insights in the literature on factors. We however leave those channels for future research.

2.8 Conclusion

In this study, we provide detailed insight into the investment strategies of defined-benefit pension funds that represent a large fraction of the European market for pension funds' assets. We measure investment strategies through factor exposures within equity and fixed income portfolios. Factor exposures are key to understanding the heterogeneity in the performance and investment strategies of liability-driven investors. We analyse two groups of drivers that influence factor exposures: pension funds' characteristics and institutional factors. These drivers only partially explain investment strategies across pension funds. We attribute the remaining heterogeneity to differences in implied beliefs about factor returns. These differences partially appear through the pension funds' choice of asset management firms to execute their investment policy.

Our results have important policy implications. We suggest that liability-driven investors can use the approach in this study for strategic investment decision-making. Our approach makes a distinction between a liability hedge demand and a speculative demand, that we measure through factor exposures. While the liabilities of a defined benefit pension fund can be measured objectively, a crucial part in the investment strategy is to form beliefs that drive the speculative demand. These are subjective in nature and require careful consideration and decision-making by a pension fund's board of trustees. Further, liability-driven investors should explain this strategy in a clear and transparent way to their stakeholders. This is particularly important because beneficiaries are typically not free to choose their own pension fund as it comes with the job. The exit costs to leave a pension fund if a beneficiary is dissatisfied with the investment strategy are prohibitively high. Employees would need to change jobs to change pension funds and retirees cannot change pension funds whatsoever. Therefore an import fiduciary duty rests on liability-driven investors to invest in the best interest of their beneficiaries.

Table 2.11. Factor exposures before and after a change in regulations: This table displays the cross-sectional means and standard deviations of the posterior betas from Equation (2.17) estimated for the period prior to 2007 and the period thereafter. The last column shows the difference between the average posterior betas in the two subsamples and the significance of the difference is based on a t-test; *p < 0.10, **p < 0.05, and ***p < 0.01. Panel A shows the results for equities and Panel B for fixed income. M,W indicates the excess global market index for equities, M,EU indicates the excess European market index for the corresponding asset class, HY-EU indicates the excess high yield index for fixed income, VAL, MOM, CARRY, and BAB indicate the value, momentum, carry, and low beta factor for the corresponding asset class.

				Panel A	A: Equity	7	
	f	full	prior	r 2007	after	r 2007	
	mean	std.dev.	mean	std.dev.	mean	std.dev.	diff. after - prior
$\hat{\beta}_i^{M,W}$	0.668	0.179	0.676	0.208	0.645	0.207	-0.031
$\hat{\beta}_i^{M,EU}$	0.273	0.153	0.228	0.165	0.296	0.170	0.068***
$\hat{\beta}_i^{VAL}$	-0.048	0.066	-0.064	0.071	-0.036	0.093	0.029***
$\hat{\beta}_i^{MOM}$	-0.044	0.041	-0.057	0.060	-0.041	0.056	0.016***
$\hat{\beta}_i^{CARRY}$	-0.057	0.126	-0.193	0.400	0.010	0.073	0.203***
$\hat{\beta}_i^{BAB}$	0.075	0.082	0.031	0.071	0.132	0.113	0.101^{***}
	Panel B: Fixed income						
	f	full	prior 2007		after	r 2007	
	mean	std.dev.	mean	std.dev.	mean	std.dev.	diff. after - prior
$\hat{eta}^{M,EU}_i$	1.107	0.306	1.021	0.115	1.232	0.377	0.211^{***}
$\hat{\beta}_{i}^{HY,EU}$	0.023	0.061	0.017	0.008	0.023	0.096	0.006
$\hat{\beta}_i^{VAL}$	-0.158	0.147	-0.013	0.008	-0.242	0.181	-0.229***
$\hat{\beta}_{i}^{MOM}$	0.070	0.007	-0.029	0.004	0.070	0.004	0.099***
$\hat{\beta}_i^{CARRY}$	-0.067	0.087	0.033	0.024	-0.060	0.128	-0.092***
$\hat{\beta}_i^{BAB}$	0.205	0.176	0.033	0.034	0.299	0.213	0.266^{***}

2.9 Appendix

A Model derivation

The mean-variance optimization problem of pension fund i equals:

$$\max_{w_{i,t}} = \max_{w_{i,t}} \mathbb{E}_t \left[\frac{A_{i,t+1} - L_{i,t+1}}{A_{i,t}} \right] - \frac{\gamma_i}{2} \operatorname{Var}_{i,t} \left[\frac{A_{i,t+1} - L_{i,t+1}}{A_{i,t}} \right],$$
(A1)

subject to

$$w_{i,t}'\iota_M \le c,\tag{A2}$$

$$w_{i,j,t} \ge 0 \ \forall j, \tag{A3}$$

where the assets equal $A_{i,t+1} = (1 + w'_{i,t}r_{t+1})A_{i,t}$, the liabilities equal $L_{i,t+1} = (1 + \psi_{i,t}r^b_{t+1})L_{i,t}$, and the funding ratio equals $F_{i,t} = A_{i,t}/L_{i,t}$.

The Lagrange of this optimization problem equals:

$$\mathcal{L}(w_{i,t},\lambda_{i,t}) = 1 + w'_{i,t}\mathbb{E}_{i,t}[r_{t+1}] - \left(1 + \psi_{i,t}\mathbb{E}_{i,t}[r_{t+1}]\right)F_{i,t}^{-1} - \frac{\gamma_i}{2}\left(w'_{i,t}\operatorname{Var}_t[r_{t+1}]w_{i,t} + \psi_{i,t}^2\operatorname{Var}_t[r_{t+1}]F_{i,t}^{-2} - 2w'_{i,t}\operatorname{Cov}_t[r_{t+1}^b\iota_M, r_{t+1}]\psi_{i,t}\iota_MF_{i,t}^{-1}\right) + \lambda_{i,t}(w'_{i,t}\iota_M - c) + \delta'_{i,t}w_{i,t}.$$
(A4)

Taking the derivative with respect to $w_{i,t}$ and $\lambda_{i,t}$ gives:

$$\frac{\partial \mathcal{L}(w_{i,t},\lambda_{i,t})}{\partial w_{i,t}} = \mathbb{E}_{i,t}[r_{t+1}] - \gamma_i \operatorname{Var}_{i,t}[r_{t+1}]w_{i,t} + \gamma_i \operatorname{Cov}_t[r_{t+1}^b \iota_M, r_{t+1}]\psi_{i,t}\iota_M F_{i,t}^{-1} + \lambda_{i,t}\iota_M + \delta_{i,t} = 0,$$
(A5)

$$\frac{\partial \mathcal{L}(w_{i,t},\lambda_{i,t})}{\partial \lambda_{i,t}} = w'_{i,t}\iota_M - c = 0.$$
(A6)

This results in the optimal weights (2.7):

$$w_{i,t}^{*} = \underbrace{\frac{\mathbb{E}_{i,t}[r_{t+1}] + \lambda_{i,t}\iota_{M} + \delta_{i,t}}{\gamma_{i}\operatorname{Var}_{t}[r_{t+1}]}}_{\operatorname{speculative portfolio}} + \underbrace{\frac{\operatorname{Cov}_{t}[r_{t+1}^{b}\iota_{M}, r_{t+1}]\psi_{i,t}\iota_{M}}{\operatorname{Var}_{t}[r_{t+1}]}}_{\operatorname{hedging portfolio}}F_{i,t}^{-1}$$

with $\lambda_{i,t}$:

$$\lambda_{i,t} = \frac{c - \left(\frac{\mathbb{E}_{i,t}[r_{t+1}] + \delta_{i,t}}{\gamma_i \operatorname{Var}_t[r_{t+1}]}\right)' \iota_M - \left(\frac{\operatorname{Cov}_t[r_{t+1}^b \iota_M, r_{t+1}]\psi_{i,t}\iota_M}{\operatorname{Var}_t[r_{t+1}]}F_{i,t}^{-1}\right)' \iota_M}{\left(\frac{\iota_M}{\gamma_i \operatorname{Var}_t[r_{t+1}]}\right)' \iota_M}.$$

B Fixed income factors

Fixed income returns

The universe of European government bond securities that we analyze consists of Austria, Belgium, Denmark, Finland, France Germany, Italy, Netherlands, Norway, Spain, Sweden, Switzerland, and the UK. We use constant maturity, zero-coupon bond yields from Bloomberg for all countries on a monthly basis from 1994 to 2017. We complement the missing data points prior to 1998 with zero coupon bond yields from Jonathan Wright's webpage for Norway, Sweden, Switzerland, and the UK. We use the Libor counterpart in each country as a proxy for the risk-free rate. The corresponding Bloomberg ticker numbers are listed in Table 2.13 in the Internet Appendix E. All included countries had investment-grade credit ratings over the entire sample period by Fitch, Moody's, and Standard & Poor's.

We start by deriving the bond returns. Following Koijen et al. (2018), we calculate the price of synthetic $\tau = 1$ -month futures on a T = 10-year zero-coupon bond each month from the no-arbitrage relation:

$$P_{i,t}^{\tau,syn} = \frac{1}{1 + r_{i,t}^f} \frac{1}{(1 + y_{i,t})^T},\tag{A7}$$

in which $y_{i,t}$ is the T = 10-year zero-coupon bond for country i = 1, ..., J, and r_{it}^{f} is the corresponding risk-free rate. At expiration, the price of the $\tau = 1$ -month futures contract equals:

$$P_{i,t+1}^{\tau-1,syn} = \frac{1}{(1+y_{i,t+\tau})^{T-\tau}},\tag{A8}$$

where we find $y_{i,t+\tau}$ by linear interpolation. The return on a fully-collateralized, currencyhedged, one-month futures contract equals:

$$r_{i,t}^{syn} = \left(\frac{(1+r_{i,t}^f)(1+y_{i,t})^T}{(1+y_{i,t+\tau})^{T-\tau}} - 1\right) \times \left(1 + \frac{e_{i,t+1} - e_{i,t}}{e_{i,t}}\right)$$
(A9)

in which $e_{i,t}$ is the time t exchange rate in euros per unit of foreign currency i. Furthermore,

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

the correction term for the exchange rate equals one for all countries in the euro area (Austria, Belgium, Finland, France, Germany, Italy, Netherlands, and Spain).

Factors

We construct value, momentum, carry, and low beta factors for the fixed income portfolios which are zero-cost long-short portfolios that use all the government bonds specified before. For any security i = 1, ..., J at time t with signal S_{it} (value, momentum, carry, or low beta), we weight securities in proportion to their cross-sectional rank based on the signal minus the cross-sectional average rank of that signal:

$$w_{it}^{S} = c_t(\operatorname{rank}(S_{it}) - \sum_{i=1}^{J} \operatorname{rank}(S_{it})/J), \quad \text{where } S \in (\text{value, momentum, carry, low beta}).$$
(A10)

The weights across all securities sum to zero and represent a dollar-neutral long-short portfolio. The scalar c_t ensures the overall portfolio is scaled one-dollar long and one-dollar short.

The signals are as follows. As in Asness et al. (2013), we define value as the 5-year change in the 10-year yield (5-year Δy). For momentum, we use the standard measure, namely, the return over the past 12 months but skip the most recent month. The signal for carry is defined as in Koijen et al. (2018):

$$C_{it} = \frac{(1+y_{i,t}^T)^T}{(1+r_{i,t}^f)(1+y_{i,t}^{T-\tau})^{T-\tau}}.$$
(A11)

To construct the low beta factor, we estimate the betas as in Frazzini and Pedersen (2014). The estimated beta for country i is:

$$\hat{\beta}_i = \hat{\rho} \frac{\hat{\sigma}_i}{\hat{\sigma}_m},\tag{A12}$$

in which $\hat{\sigma}_i$ and $\hat{\sigma}_m$ are the estimated volatilities for the bond and the market, and $\hat{\rho}$ is their correlation. We estimate the volatilities and correlations with 1- and 5-year windows respectively. The market is defined as the average return of all bonds in our sample. To reduce the effect of outliers, we follow Frazzini and Pedersen (2014) and shrink the time series estimate of beta to one: $\tilde{\beta}_i = 0.6 \times \hat{\beta}_i + 0.4 \times 1$. The factor returns for value, momentum, and carry are now constructed as:

$$r_t^S = \sum_{i=1}^J w_{it-1}^S r_{it}^{syn}, \quad \text{where } S \in (\text{value, momentum, carry}).$$
(A13)

The factor return for low beta is constructed as:

$$r_t^{\rm S} = \frac{1}{\beta_{t-1}^L} (r_t^L - r_t^f) - \frac{1}{\beta_{t-1}^H} (r_t^H - r_t^f), \quad \text{where } S \in (\text{low beta}), \tag{A14}$$

and $\beta_{t-1}^L = w'_{Lt-1}\hat{\beta}_{t-1}, \ \beta_{t-1}^H = w'_{Ht-1}\hat{\beta}_{t-1}, \ r_t^L = w'_{Lt-1}r_t^{syn}$, and $r_t^H = w'_{Ht-1}r_t^{syn}$. The weights w_{Lt-1} (w_{Ht-1}) equal the absolute weights of the long portfolio (short portfolio).

C Total AUM by pension funds in the euro area



Figure 2.2. Total assets in million EUR in funded and private pension plans (OECD, 2019).

D Random-Coefficients Model

We make the following assumptions when estimating the regression in Equation (2.15):

1.
$$\alpha_i = \alpha + u_i$$
 and $u_i \sim N(0, \sigma_{\alpha}^2)$

2. $\beta_i = \beta + v_i$ and $v_i \sim N(0, G)$, where

$$G = \mathbb{E}(v_k v'_j) = \begin{cases} \sigma_{\beta^k}^2 & \text{for } j = k\\ \sigma_{\beta^k \beta^j} & \text{for } j \neq k \end{cases}$$
(A15)

3. $\{\epsilon_{it}\}_{i,t=1}^{N,T} \perp \{u_i\}_{i=1}^N \perp \{v_i\}_{i=1}^N$.

99

In almost all cases, we assume independence across the random effects of the factor exposures, that is, $\sigma_{\beta^k\beta^j} = 0$, except for the two market factors for equities. Because the Euro Stoxx 50 index is a subset of the MSCI World Index, a higher exposure to the Euro Stoxx 50 Index directly indicates a lower exposure to the MSCI World Index, and vice versa.²²

The random-coefficients model is estimated using maximum likelihood. We show the derivation here for equities. The procedure works in the same way for fixed income, except that we allow for no correlations between the random coefficients.

To derive the likelihood, we start with writing Equation (2.15) in vector notation:²³

$$r_i^e = \alpha \iota_T + \beta' f + v_i' f + u_i + \epsilon_i, \tag{A16}$$

in which r_i^e is the $T \times 1$ vector of excess returns for fund *i*, *f* is the $T \times k$ matrix of factor returns for the fixed effects $\beta = \begin{bmatrix} \beta^1 \\ \vdots \\ \beta^K \end{bmatrix}$ and the random effect $v_i = \begin{bmatrix} v_i^1 \\ \vdots \\ v_i^K \end{bmatrix}$, and u_i is the

random intercept.

The $T \times 1$ vector of errors ϵ_i is assumed to be multivariate normal with a mean zero and variance matrix $\sigma_{\epsilon}^2 \mathbf{I}_T$. We have:

$$\operatorname{Var} \begin{bmatrix} \alpha_{i} \\ v_{i}^{1} \\ .. \\ v_{i}^{K} \\ \epsilon_{i} \end{bmatrix} = \begin{bmatrix} \sigma_{\alpha}^{2} \iota_{T} \, \iota_{T}' & 0 & 0 & 0 & 0 \\ 0 & \sigma_{\beta}^{2} \iota_{T} \iota_{T}' & \sigma_{\beta} \iota_{\beta}^{2} \iota_{T} \iota_{T}' & 0 & 0 \\ 0 & \sigma_{\beta}^{2} \rho_{1} \iota_{T} \iota_{T}' & \ddots & 0 & 0 \\ 0 & 0 & 0 & \sigma_{\beta}^{2} \kappa \iota_{T} \iota_{T}' & 0 \\ 0 & 0 & 0 & 0 & \sigma_{\epsilon}^{2} \mathbf{I}_{T} \end{bmatrix}.$$
(A17)

The error term: $v_i^1 f^1 + \ldots + v_i^K f^K + u_i + \epsilon_i$ has a $T \times T$ variance-covariance matrix

$$V = \operatorname{Var}[r_i^e|f] = \sigma_{\alpha}^2 \iota_T \, \iota_T' + \sigma_{\beta^1}^2 f^1 f^{1'} + 2\sigma_{\beta^1\beta^2} f^1 f^{2'} + \sigma_{\beta^2}^2 f^2 f^{2'} + \dots + \sigma_{\beta^K}^2 f^K f^{K'} + \sigma_{\epsilon}^2 \mathbf{I}_T.$$
(A18)

 $^{^{22}}$ We perform a simulation test to ensure the high correlation between the MSCI World Index and the Euro Stoxx 50 Index does not cause multicollinearity problems. We simulate returns consisting of a mix between the MSCI World Index, the Euro Stoxx 50 Index, and an error term. We then regress the simulated returns on the MSCI World Index and the Euro Stoxx 50 index. We find the exact coefficients with high precision (i.e., low standard errors) that we imposed for the simulated returns.

²³Here we assume all pension funds have the same T. For pension funds with different T, the T should be replaced by T_i .

The log-likelihood for fund i can now be written as:

$$L_{i}(\alpha,\beta,\sigma_{\alpha}^{2},\sigma_{\beta^{1}}^{2},...,\sigma_{\beta^{K}}^{2},\sigma_{\epsilon}^{2}|r_{i}^{e}) = -\frac{1}{2}\{T\log(2\pi) + \log|V| + (r_{i}^{e} - \alpha\iota_{T} - \beta'f)'V^{-1}(r_{i}^{e} - \alpha\iota_{T} - \beta'f)\}.$$
(A19)

Then, the total log-likelihood equals:

$$L(\alpha, \beta, \sigma_{\alpha}^{2}, \sigma_{\beta^{1}}^{2}, ..., \sigma_{\beta^{K}}^{2}, \sigma_{\epsilon}^{2} | r^{e}) = -\frac{1}{2} \{ NT \log(2\pi) + N \log |V| + \sum_{i=1}^{N} (r_{i}^{e} - \alpha \iota_{T} - \beta' f)' V^{-1} (r_{i}^{e} - \alpha \iota_{T} - \beta' f) \}.$$
(A20)

We now turn to a detailed description of the estimation results described in Table 2.16. We begin by analyzing the results for equities. The exposure to the global market factor equals 0.65, and the exposure to the European factor equals 0.30. Both are statistically significant. The positive and significant exposure to the excess European market return displays the existence of a currency bias; that is, Dutch pension funds on average tend to invest more in Europe relative to the global market portfolio. Additionally, sizable crosssectional variation exists in pension funds' market betas. The exposure to the global market factor varies between 0.28 and 1.02, and the exposure to the European market factor varies between -0.02 and 0.62. Pension funds on average have significantly negative exposures to value (-0.04), momentum (-0.04), and carry (-0.05). Significant cross-sectional variation exists in all three factor exposures. The highest cross-sectional standard deviation equals 0.15 for the carry factor that indicates the range of factor exposures is between -0.35 and 0.24. The exposure to value varies between -0.21 and 0.13, and between -0.14 and 0.05for momentum. Pension funds on average have a significantly positive exposure to the low beta factor that is equal to 0.09. Again, we find significant and substantial cross-sectional variation in the low beta exposure that ranges from -0.13 to 0.30.

In case of fixed income, pension funds have an average (significant) exposure to the investment-grade market factor that is equal to 1.13. The cross-sectional variation ranges from 0.16 to 2.10. For the fixed income factors we find that pension funds, on average, have a negative exposure to value (-0.21) and carry (-0.08), a positive exposure to momentum (0.07), and a strong positive exposure to low beta (0.27). The exposure to value varies between -0.52 and 0.10, between -0.27 and 0.11 for carry, and between -0.12 and 0.66 for low beta. The cross-sectional heterogeneity is significant at the 1 percent level for the market factors, value, and low beta, and at the 5 percent level for carry. We are unable to

statistically detect significant cross-sectional variation in momentum exposures based on the random-coefficients model.

For equities, we also find cross-sectional variation in alphas, or the part of the return that is not explained by the factors. The standard deviation equals 0.0028, and the alphas vary between -0.0064 and 0.0048 on a quarterly basis. For fixed income we do not observe statistically significant variation in the alphas. This finding indicates that pension funds are unable to outperform each other consistently. However, even if pension funds slightly vary in their alphas, our sample might not have enough observations to say something statistically meaningful about the alphas. This finding is expected, because first moments can be estimated less accurately than second moments (Merton, 1980).

E Additional tables

Symbol	Description
4	
A	Asset value
AM	Vector of asset management firms
AUM	Total asset under management for the corresponding asset class
B_{-}	Pension benefits
D	Liability duration
F	Funding ratio
K	Total number of factors
L	Present discounted value of future pension benefits
M	Total number of assets
N	Total number of pension funds
RFR	Required funding ratio
Type	Pension fund type
f_t^a	Vector of factor returns for asset class a
r	Vector of asset returns
r^a	Pension fund return for asset class a
r^b	Return on the risk-free bonds traded in the market
r^e	Pension fund excess return (relative to short-term risk-free rate)
r^{f}	Short-term risk-free rate
r^h	Regulatory discount rate for time-to-maturity h
r^L	Liability return
w	Vector of portfolio weights
q	Benchmark factor exposures
$se(\beta_i^k)$	Standard error of the time-series OLS factor exposures for factor k
v	Vector of random-effect coefficients
β^a	Vector of factor exposures for asset class a
$\hat{eta}^{m k}$	Fixed-effect estimator for factor k (prior mean)
$ ilde{eta}^{m k}$	Posterior factor exposure for factor k
$\hat{eta}^{m k}_{m i}$	Time-series OLS factor exposures for factor k
$\tilde{\beta}^k_{adi}$	Posterior factor exposures adjusted for pension fund characteristics
γ	Risk aversion coefficient
δ	Kuhn-Tucker multipliers for the short-sale constraints
ι	Vector of ones
λ	Lagrange multiplier for the borrowing constraint
λ^k	Historical average return for factor k
μ	Expected (excess) returns
Σ	Variance-covariance matrix of returns
$\hat{\sigma}^2_{\beta^k}$	Variance estimator of the random effects for factor k (prior variance)
$ ilde{\sigma}_{eta^k}^2$	Posterior variance for factor k
ψ	Duration of the liabilities over the duration of the risk-free bonds

Table 2.12. Glossary of symbols: This table summarizes the main symbols in this study.

_

Table 2.13. Bloomberg ticker list: The Bloomberg ticker numbers used to construct the European fixed income factors described in Appendix B. The x in each ticker number should be replaced by the corresponding maturity: x=10 years, x=09 years, and x=03 months; and y by the corresponding unit of time: y=y for years and y=m for months.

Table 2.14. MSCI-W is t the global ve stocks, and I Return Bonc fixed income fixed income	Correlat the excess due factor 3AB-S is t 3AB-S is t 1 Index, F , MOM-F , and BAI	tion tak MSCI V for stoc the globs HY-EU is T is the B-FI is t	ble of f : Vorld To ks, MON al low be al low be the Blc Europea he Euroj	actor ret tal Return <i>M</i> -S is the ta factor f oomberg B n moment pean low l	urns: Thi 1 Index, EU global mor or stocks. arclays Eu uum factor beta factor	s table] J-50 is tl mentum FI-EU is ro High for fixed for fixed	provides he exces factor fa the exc Yield Ir l income 1 income	the corr s Euro St or stocks, ess Bloot dex, VA dex, VA . CARR' 3. All ret	celation r toxx 50 T Carry-S Carry-S nberg Ba nberg Ba L-FI is th Y-FI is th urns are	natrix of 1 otal Retun is the glot rclays Eur re Europe: ne Europe: converted	he factor re n Index, VA val carry fact o Aggregate an value fact an carry fact into euro re	turns. L-S is for for Total or for or for turns.
Correlation 1	natrix											
	MSCI-W	EU-50	VAL-S	MOM-S	CARRY-S	BAB-S	FI-EU	HY-EU	VAL-FI	MOM-FI	CARRY-FI	BAB-FI
MSCI-W	-											
EU-50	0.87											
VAL-S	-0.22	-0.11	Ц									
MOM-S	-0.18	-0.24	-0.68	1								
CARRY-S	-0.15	-0.26	0.03	-0.01	1							
BAB-S	-0.32	-0.30	0.25	0.13	0.14	1						
FI-EU	-0.15	-0.13	0.11	-0.07	0.08	0.04	1					
HY-EU	0.64	0.63	0.04	-0.41	0.15	-0.10	0.09	1				
VAL-FI	0.18	0.26	0.17	-0.19	-0.03	0.13	0.06	0.37	1			
MOM-FI	-0.12	-0.11	-0.08	0.20	-0.05	-0.06	0.05	-0.34	-0.51	1		
CARRY-FI	0.16	0.27	0.12	-0.17	-0.03	0.10	0.33	0.30	0.66	-0.34	1	
BAB-FI	-0.29	-0.34	0.21	-0.01	0.00	0.10	0.37	-0.25	-0.29	0.31	-0.29	1

CHAPTER 2. DRIVERS OF INVESTMENT STRATEGIES

Table 2.15. **OLS factor exposures**: This table displays the cross-sectional mean and standard deviation of the estimated betas from the time-series regression presented in Equation (2.14). The cross-sectional mean and standard deviation of the *R*-squared from the time-series regressions are also provided. 10%-level and 5%-level sign. indicate the number of pension funds for which the corresponding factor is statistically different from zero at the 5% and 10% significance level, respectively, by using the Newey-West adjusted standard errors. M,W indicates the excess global market index for equities, M,EU indicates the excess European market index for the corresponding asset class, HY-EU indicates the excess high yield index for fixed income, VAL, MOM, CARRY, and BAB indicate the value, momentum, carry, and low beta factor for the corresponding asset class.

Equity returns						
	mean	std.dev.	5%-level sign.	10%-level sign.		
$\hat{\beta}_i^{M,W}$	0.656	0.297	531	537		
$\hat{\beta}_i^{M,EU}$	0.270	0.311	429	455		
$\hat{\beta}_i^{VAL}$	-0.060	0.230	131	182		
$\hat{\beta}_i^{MOM}$	-0.056	0.244	143	192		
$\hat{\beta}_i^{CARRY}$	-0.106	0.549	139	196		
$\hat{\beta}_i^{BAB}$	0.088	0.240	221	269		
R^2	0.928	0.092				
Fixed income returns						
	mean	std.dev.	5%-level sign.	10%-level sign.		
$\hat{\beta}_i^{M,EU}$	1.139	0.564	553	559		
$\hat{\beta}_{i}^{HY,EU}$	0.019	0.111	206	256		
$\hat{\beta}_i^{VAL}$	-0.146	0.402	218	274		
$\hat{\beta}_i^{MOM}$	0.024	0.623	93	119		
$\hat{\beta}_i^{CARRY}$	-0.037	0.552	101	132		
$\hat{\beta}_{i}^{BAB}$	0.253	0.508	249	310		
$\dot{R^2}$	0 760	0 185				

Table 2.16. **Prior factor exposures**: This table shows the coefficient estimates and corresponding standard errors for the random-coefficients model in Equation (2.15) that is used as a prior to compute the posterior betas. The estimates $\hat{\alpha}$ and $\hat{\beta}^k$ indicate the fixed effects, and $\hat{\sigma}^2_{\alpha}$, and $\hat{\sigma}^2_k$ indicate the random effects of the random-coefficients model. M,W indicates the excess global market index for equities, M,EU indicates the excess European market index for the corresponding asset class, HY-EU indicates the excess high yield index for fixed income, VAL, MOM, CARRY, and BAB indicate the value, momentum, carry, and low beta factor for the corresponding asset class. Standard errors are clustered at the pension fund level; *p < 0.10, **p < 0.05, and ***p < 0.01. The significance for each random coefficients model with a random-coefficients model that assumes the factor exposure of interest to be fixed.

Equity returns	5	F	Fixed income ret	turns	
	Coefficient	std. error		Coefficient	std. error
$\hat{lpha} _{\hat{eta}M,W}$	-0.001^{**}	0.0003	$\hat{lpha} _{\hat{arrho}M,EU}$	0.001***	0.0002
$\hat{\beta}^{M,EU}$	0.049 0.299^{***}	0.0090	$\hat{eta}^{HY,EU}$	0.024^{***}	0.00220
$\hat{\beta}^{VAL}$ $\hat{\beta}^{MOM}$	-0.043^{***} -0.041^{***}	$0.0051 \\ 0.0042$	β^{VAL} $\hat{\beta}^{MOM}$	-0.208^{***} 0.071^{***}	0.0093 0.0081
\hat{eta}^{CARRY} \hat{eta}^{BAB}	-0.054^{***}	0.0102	$\hat{\beta}^{CARRY}$ $\hat{\beta}^{BAB}$	-0.079^{***}	0.0122
ρ	0.087	0.0005	ρ	0.271	0.0117
$\sigma^2_{lpha} \hat{\sigma}^2_{M,W}$	0.00001^{*} 0.0338^{***}	0.0000 0.0049	$\sigma^2_{lpha} \hat{\sigma}^2_{M,EU}$	0.0000005 0.235***	0.0000 0.0904
$\hat{\sigma}^2_{M,EU} \ \hat{\sigma}^2_{VAL}$	0.0256^{***} 0.0073^{***}	$0.0043 \\ 0.0026$	$\hat{\sigma}^2_{HY,EU} \ \hat{\sigma}^2_{VAL}$	0.007^{***} 0.024^{***}	$0.0012 \\ 0.0052$
$\hat{\sigma}^2_{MOM}$ $\hat{\sigma}^2_{CABBY}$	0.0023*** 0.0218***	$0.0011 \\ 0.0057$	$\hat{\sigma}^2_{MOM} \\ \hat{\sigma}^2_{CABBV}$	0.001 0.009**	$0.0028 \\ 0.0068$
$\hat{\sigma}^2_{BAB}$	0.0115***	0.0029	$\hat{\sigma}^2_{BAB}$	0.038***	0.0204
$\hat{\sigma}_{M,WM,EU}$	-0.0259^{***}	0.0042			
Wald $chi2(6)$ obs.	47,345 25,434		Wald $chi2(6)$ obs.	$4,192 \\ 25,839$	

Table 2.17. Effect of model parameters on the equity allocation: This table shows the coefficient estimates of a regression of the equity allocation on pension fund characteristics from 2009Q1-2017Q4. The equity allocation is computed as the total AUM in equities divided by the total AUM in fixed income and equities. The pension fund characteristics are the funding ratio, the risk aversion coefficient that we represent with the inverse of the required funding ratio, and the liability duration. Standard errors are in parentheses and clustered at the pension fund level; *p < 0.10, **p < 0.05, and ***p < 0.01.

	Equity allocation				
Funding ratio	0.0756***	0.0748***	0.0772***		
Risk aversion	[0.0106] -1.865***	[0.0106] -1.866***	[0.0106] -1.853***		
Liability duration	[0.0401] -0.0015***	[0.0391] -0.0015***	[0.0397] -0.0015***		
Size	[0.0003]	[0.0003]	[0.0003] 0.0038^{*}		
Corporate			[0.0021] -0.0081** [0.0025]		
Professional			-0.0200*** [0.0063]		
Constant	0.360^{***} [0.0013]	0.360^{***} [0.0013]	$[0.0003] \\ 0.367^{***} \\ [0.0030]$		
time FE	No	Yes	Yes		
obs.	8,648	8,648	8,648		
adj. R-squared	0.246	0.275	0.277		

Table 2.18. Impact of pension fund's characteristics on factor exposures - proxy: We regress the pension funds' excess returns on the factor returns and the factor returns interacted with the funding ratio, the risk aversion coefficient that we represent with the inverse of the required funding ratio, and the ratio of actives relative to total participants during the period from 2009Q1-2017Q4, where the total equals the active participants and the retirees. Standard errors are in parentheses and clustered at the pension fund level; *p < 0.10, **p < 0.05, and ***p < 0.01.

	average	funding ratio	risk aversion	% active participants
$\beta^{M,W}$	0.713^{***}	-0.0166	-0.482***	0.0243
	[0.0061]	[0.0513]	[0.1683]	[0.0266]
$\beta^{M,EU}$	0.253***	0.0518	0.208	-0.0027
	[0.0063]	[0.0437]	[0.1644]	[0.0256]
β^{VAL}	0.0263**	-0.0583	0.477	0.0616
	[0.0116]	[0.0929]	[0.3415]	[0.0482]
β^{MOM}	-0.0487***	-0.0064	0.274	0.001
	[0.0080]	[0.0599]	[0.2059]	[0.0330]
β^{CARRY}	0.0165^{*}	0.0794	-0.385	0.0053
	[0.0091]	[0.1146]	[0.2909]	[0.0427]
β^{BAB}	0.0838***	0.0275	-0.0616	-0.0744**
	[0.0153]	[0.0569]	[0.2126]	[0.0330]
obs.	8,851	adj. R-sq.	0.860	

Panel A: Equity returns

Panel B: Fixed income returns

$\beta^{M,EU}$	average 2.204***	funding ratio -0.977***	risk-aversion 9.394***	$\%$ active participants 1.561^{***}
	[0.0505]	[0.2358]	[1.3144]	[0.2362]
$\beta^{HY,EU}$	-0.0371***	0.0714*	-0.776***	-0.130***
	[0.0065]	[0.0424]	[0.1504]	[0.0267]
β^{VAL}	-0.157***	0.0261	-0.682	-0.0426
	[0.0187]	[0.0893]	[0.4933]	[0.0789]
β^{MOM}	0.015	-0.0351	1.081***	0.0749
	[0.0118]	[0.0606]	[0.3158]	[0.0518]
β^{CARRY}	-0.557***	0.360**	-3.770***	-0.733***
	[0.0341]	[0.1709]	[0.9128]	[0.1582]
β^{BAB}	-0.0821***	0.142	-1.964***	-0.406***
	[0.0260]	[0.1220]	[0.7088]	[0.1205]
obs.	8,954	adj. R-sq.	0.558	

Chapter 3

The Liquidity Premium Across Asset Classes¹

3.1 Introduction

Illiquid assets increasingly have a role in investors' portfolios. The illiquid assets of large US endowment funds comprised 55% of the total portfolios in 2015 (Dimmock et al., 2019). Moreover, the seven largest pension funds in the world have increased their average allocations of illiquid assets from 4% in 1997 to 25% in 2017 (Willis Towers Watson, 2019). One potential reason for investing in illiquid assets is to capture liquidity premiums (OECD, 2014; Watson, 2019). In other words, investing in illiquid assets might compensate the investor for bearing this cost. Yet, there is no consensus in the empirical literature that justifies which asset classes should have first-order liquidity premiums. In this study, we aim to give more guidance to its opposing findings by modeling illiquid assets and heterogeneous investor types in four different asset classes: private equity, real estate, corporate bonds, and stocks.

Studies have either modeled illiquidity as proportional transactions costs, for example, Constantinides (1986); Vayanos (1998), or as the inability to trade illiquid assets for random time periods, for example, Ang et al. (2014). We combine these two dimensions of illiquidity here for two reasons. First, several asset classes show both aspects of illiquidity simultaneously. For instance, traders of corporate bonds face transaction costs, yet sometimes specific bonds

¹This chapter is based on joint work with Bas Werker. We thank Jules van Binsbergen, Marie Brière, Dirk Broeders, Joost Driessen, Esther Eiling, Frank de Jong, Ralph Koijen, Olivia Mitchell, Theo Nijman, Antoon Pelsser, Patrick Tuijp, seminar participants at Tilburg University, the Wharton School, Maastricht University, Netspar Pension Day, Ortec Finance Canada, Ortec Finance Rotterdam, Achmea Investment Management, Robeco, and conference participants of the Market Microstructure & High Frequency Data Stevanovich Center University of Chicago, SoFiE, and the International Pension Workshop for very useful comments. Kristy gratefully acknowledges financial support from Instituut Gak.

may not trade for several days. Similarly, selling real estate may take several years, and it obviously carries transaction costs as well. Second, combining both sources of illiquidity allows us to capture the heterogeneity across asset classes by adjusting the prominence of both effects.

We solve for liquidity premiums in a partial equilibrium power utility framework. We propose that the cost of illiquidity involves two aspects: suboptimal asset allocation and suboptimal consumption. If the cost of liquidating the illiquid asset is too high and the investor prefers not to trade the illiquid asset, or the illiquid asset cannot be traded at all; then the illiquidity leads to suboptimal portfolio allocations. Yet, deviating from the optimal asset allocation generally only leads to small utility costs (Constantinides, 1986). At the same time, illiquidity may hamper the possibility of smoothing consumption after negative wealth shocks. If the investor is unable to sell the illiquid asset or only at high costs, they may face an even stronger negative consumption shock compared to the case where the illiquid asset can always be traded without costs. Shocks to consumption generally carry a high utility cost and hence illiquidity does generate first-order liquidity premiums.

Our model's flexibility allows us to quantify the magnitude of the liquidity premium in four markets. This flexibility is important as the effects of liquidity highly depend on the characteristics of illiquid assets such as the frequencies of trade and potential income returns on illiquid assets as well as the characteristics of the investor types that operate in those markets. For each asset classes we calculate the premium different investor types are willing to pay for the illiquid asset to become liquid. We find that this willingness-to-pay is larger for short-term investors, investors who face substantial liquidity shocks, and investors who desire to allocate over 60% of their wealth to illiquid assets if that same asset would otherwise be liquid. Thus, in equilibrium, the liquidity premiums depend on the relative prevalence of different investor types. For instance, Amihud and Mendelson (1986), Longstaff (2003), and Chen et al. (2020) find investor heterogeneity in which short-term investors hold liquid assets and long-term investors hold illiquid assets. The heterogeneity in investor types across asset classes enables us to relate to the empirical evidence about liquidity premiums for each market.

We make several contributions. Even though private equity is the most illiquid asset that we analyze, we find an average annual liquidity premium no larger than 5-15 basis points. Private equity investors have to lockup their money for long periods of time; and mainly for that reason, only long-term investors are present in this market: approximately 20% is held by endowments and 60% by pension funds (Harris et al., 2014). For these investors illiquidity is unlikely to substantially harm investors' consumption patterns. Moreover, the non-trading period is typically fixed at 10 years, so by and large investors know when the position will be liquidated. For direct real estate, we find an average annual liquidity premium equal to 15-35 basis points. Direct real estate can often not be traded for a substantial amount of time, the timing of the trading opportunities are uncertain, and the transaction costs are high. Yet, the threat of illiquidity is dampened because of the liquid return component (rents) of real estate investments and the large share of long-term investors in this market. We find the largest liquidity premiums for stocks and corporate bonds; the average annual liquidity premium for corporate bonds equals 30-50 basis points. For corporate bonds, the transaction costs are small, but uncertainty in trading opportunities and its high price of risk amplify liquidity premiums. For stocks we find an average annual liquidity premium equal to 20-45 basis points. Even though stocks trade very often, liquidity premiums do not disappear: short-term investors are the largest group of investors in these markets (84%), and they demand first-order liquidity premiums even at relatively low transaction costs.

A potential shortcoming of our approach is that we are not able to model investors' preferences perfectly. However, we make two assumptions that are more likely to overestimate rather than to underestimate the liquidity premium. First, we assume that the investors cannot borrow against the illiquid assets. This caveat may be a realistic assumption for some asset classes, but not in others. For instance, real estate investors are typically able to borrow a substantial amount using the property as collateral. However, taking this borrowing into account decreases liquidity premiums because the investors can partially undo the illiquidity of the asset. Second, we allow for liquidity shocks as large as 50% of the investors' total wealth. Even though larger wealth shocks are in practice possible, our model shows that investors avoid risky assets all together if faced with such shocks. As a result, large liquidity shocks do not necessarily amplify liquidity premiums.

Our study contributes to the empirical literature on liquidity premiums. The early empirical literature found significant effects of illiquidity on stock prices. For instance, Amihud and Mendelson (1986) and Brennan and Subrahmanyam (1996) show that a 1% higher transaction cost means a 1.5% to 2% higher expected return for stocks. Yet, some recent studies have challenged the empirical evidence for liquidity premiums in stocks.² For instance, Ben-Rephael et al. (2015) show that liquidity premiums have become insignificant

²Note that we refer here to the *level* of the liquidity premium. There are also studies on the liquidity risk premium, for example, Pastor and Stambaugh (2003).

in recent decades for public US equities, except for very small stocks. Our findings show that a 1% higher transaction cost means a 0.20-0.45% higher expected return for stocks, which is substantially lower than the estimates from the early literature and more in favor with the recent evidence of a small liquidity premium for stocks. First-order liquidity premiums also exist for corporate bonds (e.g. Chen et al., 2007; Bao et al., 2011; Bongaerts et al., 2017). In particular, Bongaerts et al. (2017) find an average (level) liquidity premium equal to 0.54% for corporate bonds that carry 0.52% transaction costs. Yet, Palhares and Richardson (2019) find only limited evidence for liquidity premiums for corporate bonds after using illiquidityfactor portfolios, that is, a strategy that goes long in illiquid bonds and short in the liquid ones. Our model indicates that liquidity premiums are below, yet fairly close to the estimates in Bongaerts et al. (2017). However, in our model the liquidity premium is not only driven by transaction costs but also by the inability to trade corporate bonds for several weeks. This may mean that empirical estimates indirectly incorporate the inability to trade as part of the liquidity premium.

Similarly, for private equity there is no clear consensus regarding the existence of a liquidity premium, although the evidence is more indirect. Franzoni et al. (2012) report no out-performance of private equity relative to public equity, while Harris et al. (2014) finds a substantial out-performance of 3% annually. Our findings point in the direction of second-order liquidity premiums for private equity. Finally, opposing indirect evidence also exists for real estate investments. Qian and Liu (2012) find a somewhat higher expected return for direct compared to indirect real estate, while Ang et al. (2013a) find comparable performance for direct and indirect real estate investments. Our range of model implied liquidity premiums for real estate is consistent with the dearth of empirical evidence for liquidity premiums in real estate markets.

Our study also contributes to the theoretical literature on liquidity premiums. The early theoretical literature did not find evidence for the existence of liquidity premiums. Constantinides (1986) and Vayanos (1998) show that transaction costs only have a second-order effect on prices, that is, a 1% higher transaction cost increases the liquidity premiums by only roughly 10 basis points per year. After their work, a great deal of literature was developed that studies illiquidity or transactions costs by using assumptions more in line with real-world investment problems. This work finds that illiquidity can have first-order effects on prices. For instance, theoretical work from Huang (2003) and Gârleanu (2009) show that illiquidity may arise when investors face borrowing constraints. Jang et al. (2007) add return predictability to the investor's problem in a market with transactions costs and find

a slight increase in liquidity premiums. Lynch and Tan (2011) solve a model that comprises labor income, wealth shocks, return predictability, and transaction costs and are able to generate liquidity premiums for stocks in the same order of magnitude as the early empirical literature. The main goal of these studies is to understand the mechanism behind the effect of illiquidity in equilibrium. By contrast, we focus on explaining the observed variation in the magnitude of the liquidity premium for different asset classes using heterogeneity in the illiquidity source and investors' preferences.

The remainder of the study is organized as follows: Section 3.2 shows the theoretical framework of the model and describes the corresponding optimal strategies and the partial equilibrium implications for the liquidity premiums. We link our theoretical findings to the empirical literature on liquidity premiums in Section 3.3. Section 3.4 concludes.

3.2 Optimal consumption and investment with illiquid assets

In this section, we model illiquidity as the inability to trade an asset frequently and by the cost that occurs when trading. This is formalized in Section 3.2.1. In Section 3.2.2 we describe the optimization problem of the investor and the solution of this optimization problem is presented in Section 3.2.3. Section 3.2.4 shows how we derive model implied liquidity premiums and Section 3.2.5 reveals magnitudes of model implied liquidity premiums depending on different parameter values.

3.2.1 Financial market

The financial market consists of three assets: a risk-free asset B, a liquid risky asset S, and an illiquid risky asset denoted by X. The risk-free asset has a constant (annual) rate of return r_f . The liquid risky asset earns a nominal return r_t^S over the period (t - 1, t], while we denote the nominal return on the illiquid asset over the same period by r_t^X . All returns are continuously compounded. Further, we assume that the price of the illiquid asset is observed, even though it cannot be traded every period.

The prices of risk of the liquid and illiquid assets are denoted by λ_S and λ_X , respectively. Their volatilities are similarly denoted by σ_S and σ_X ; their correlation by ρ_{SX} . The returns r_t^S and r_t^X are jointly normally distributed:

$$\begin{bmatrix} r_t^S \\ r_t^X \end{bmatrix} \sim \mathcal{N}\left(\begin{bmatrix} r_f + \lambda_S \sigma_S - \frac{1}{2}\sigma_S^2 \\ r_f + \lambda_X \sigma_X - \frac{1}{2}\sigma_X^2 \end{bmatrix}, \begin{bmatrix} \sigma_S^2 & \rho_{SX}\sigma_S\sigma_X \\ \rho_{SX}\sigma_S\sigma_X & \sigma_X^2 \end{bmatrix}\right).$$
(3.1)

The differences between the liquid and the illiquid risky asset are the trading opportunities and transaction costs. While the investor can always trade the liquid risky asset S at no cost, the illiquid asset X can only be traded at random points in time and at a cost. We denote the trading indicator for time t with $\mathbf{1}_t^p$ and interpret $\mathbf{1}_t^p = 1$ as the trading opportunity that arises for illiquid asset X, while $\mathbf{1}_t^p = 0$ indicates that the illiquid asset cannot be traded that period. We assume throughout that the trading indicators are i.i.d. Bernoulli random variables with trading probability $p = \mathbb{P}\{\mathbf{1}_t^p = 1\}$. If a trading opportunity occurs $(\mathbf{1}_t^p = 1)$ and the investor decides to trade, then the proportional transaction costs ϕ must be paid, $0 \le \phi \le 1$. We allow for two types of return components on the illiquid asset: income return and capital gains. That is, we separate r_t^X into a liquid part d_t (income return) and an illiquid part $r_t^X - d_t$ (capital gains).

3.2.2 The investors' consumption and investment problem

The investor has an investment horizon equal to T. We assume that the illiquid asset can be traded (and thus liquidated) against transaction costs ϕ at the final date T. Notice that under the assumption that the illiquid asset might not be liquidated at T, the liquidity premiums are amplified. We do not consider this because, in that case, the investor is better off postponing the liquidation of the illiquid asset until a trading opportunity arises. We introduce the following notation to distinguish liquid and illiquid wealth. We denote liquid wealth as available at time t by W_t . This wealth consists of investments in both the risk-free asset B and the risky liquid asset S. The value of the investment in the illiquid asset at time t is denoted by X_t . Therefore, total wealth equals $W_t + X_t$.

Furthermore, we assume that the investor may face a liquidity shock L_t that is assumed to be fraction l_t of the previous period's total wealth $W_{t-1} + X_{t-1}$. The liquidity shock indicator is denoted by $\mathbf{1}_t^s$ and follows an i.i.d. Bernoulli process with probability $q = \mathbb{P}\{\mathbf{1}_t^s = 1\}$. Various causes of such a liquidity shock arise. For individual investors, examples are health care costs, unforeseen expenditures, or extreme weather events. For institutional investors a possible cause for liquidity shocks are margin calls on derivative positions. For long-term investors, such as pension funds and life insurers, shocks to the mortality rate can increase the temporary need for liquidity.

We denote the fraction of liquid wealth W_t that is invested in the liquid risky asset S by

 θ_t , and $1 - \theta_t$ is invested in the risk-free asset *B*. Consumption at time *t* is denoted by C_t and must be financed with liquid wealth W_t . Preferences are represented by a standard constant relative risk aversion (CRRA) expected utility function with risk-aversion parameter γ .

Illiquid wealth X_t can only be converted into liquid wealth (and, if desired, immediately consumed) if a trading opportunity arises, that is, if $\mathbf{1}_t^p = 1$. We denote the transfer from liquid to illiquid wealth by ΔX_t . Thus, $\Delta X_t > 0$ means that at time t, an additional amount ΔX_t of the illiquid asset is bought, and thus liquid wealth W_t decreases by $\Delta X_t + \phi |\Delta X|$. If no trading opportunity arises, then $\mathbf{1}_t^p = 0$; and we automatically have $\Delta X_t = 0$.

The investor optimizes its utility of a stream of consumption levels C_t over a horizon $t = 0, \ldots, T$. Thus, the criterion function is:

$$\mathbb{E}_0\left[\sum_{t=0}^T \beta^t \frac{C_t^{1-\gamma}}{1-\gamma}\right],\tag{3.2}$$

where β denotes the time-preference discount factor, and $\gamma > 1$ is the risk-aversion parameter.

The investor faces two budget constraints: one for liquid wealth W_t and one for illiquid wealth X_t . Formally, we have:

$$W_t = (W_{t-1} - \triangle X_{t-1} - \phi | \triangle X_{t-1} | - C_{t-1}) \left(\exp(r_f) + \theta_{t-1} \left[\exp(r_t^S) - \exp(r_f) \right] \right)$$

+
$$(X_{t-1} + \Delta X_{t-1}) \exp(d_t) - L_t$$
 (3.3)

$$X_t = (X_{t-1} + \Delta X_{t-1}) \left(\exp(r_t^X) - \exp(d_t) \right).$$
(3.4)

We assume that the investor cannot borrow against the illiquid investments. The effect of illiquidity would be strongly reduced if this borrowing was possible, as the investor could always undo the illiquidity by borrowing against the illiquid asset if needed. Thus, we impose:

$$C_t \le W_t, \quad t = 0, \dots, T. \tag{3.5}$$

The optimal consumption problem can now be stated as follows:

Problem 3.2.1. The investor maximizes

$$\max_{\{\theta_t, \triangle X_t, C_t\}_{t=0}^{T-1}} \mathbb{E}_0\left[\sum_{t=0}^T \beta^t \frac{C_t^{1-\gamma}}{1-\gamma}\right]$$
(3.6)

subject to the budget constraints (3.3) and (3.4) and the borrowing constraint (3.5). Moreover, when $\mathbf{1}_t^p = 0$, we must have $\Delta X_t = 0$.

117

The decision variables θ_t , ΔX_t , and C_t are non-anticipative. Formally, $\{\theta_t, \Delta X_t, C_t\}$ are adapted to the filtration $\mathbb{F} = \{\mathcal{F}_t\}_{t=0}^T$, where $\{\mathcal{F}_t\}$ is the filtration generated by $\{r_t^S, r_t^X, \mathbf{1}_t^p, \mathbf{1}_t^s\}$.

To summarize, illiquidity limits the investor's consumption and investment decisions in three ways compared to the case where the illiquid asset is fully liquid, that is, the two risky asset Merton case (Merton, 1969): the inability to trade the illiquid assets for uncertain periods of time; transaction costs of the illiquid asset when a trading opportunity arises; and the investor cannot borrow against the illiquid assets. All three assumptions are important characteristics of (most) illiquid assets.

3.2.3 Optimal strategies

Because of incomplete markets, the optimization in Problem 3.2.1 cannot be solved analytically, so we resort to numerical methods. For these, the numerical complexity is well-known to strongly increase with the dimension of the endogenous state variables. In the formulation of Problem 3.2.1 there are two: W_t and X_t . Yet, in line with Ang et al. (2014), a simple transformation leads to a partly analytical result due to the homogeneity of the CRRA utility function we consider, see Theorem 3.2.2 below.

More precisely, we consider as endogenous state variables the total wealth $W_t + X_t$ and the fraction of total wealth invested in the illiquid asset, that is,

$$\xi_t = \frac{X_t}{W_t + X_t}.\tag{3.7}$$

With this reparametrization we define the value function using the Bellman principle as:

$$V_t (W_t + X_t, \xi_t) = \max_{\theta_t, \Delta X_t, C_t} \beta^t \frac{C_t^{1-\gamma}}{1-\gamma} + \mathbb{E}_t V_{t+1} (W_{t+1} + X_{t+1}, \xi_{t+1}),$$
(3.8)

with the boundary condition at time T given by:

$$V_T (W_T + X_T, \xi_T) = \beta^T \frac{(W_T + (1 - \phi)X_T)^{1 - \gamma}}{1 - \gamma}.$$
(3.9)

The boundary condition means that we assume that all assets can be traded (and thus liquidated) at time T against transaction costs ϕ . With the above introduced change of variables, that is, the pair (W_t, X_t) is replaced by the pair $(W_t + X_t, \xi_t)$, then the solution to the investor's problem satisfies the following theorem.

Theorem 3.2.2. There are time-dependent (deterministic) functions α_t , θ_t , and H_t such that the optimal solution $\{C_t^*, \Delta X_t^*, \theta_t^*\}$ to Problem 3.2.1 can be written as:

$$V_t (W_t + X_t, \xi_t) = \beta^t \frac{(W_t + X_t)^{1-\gamma}}{1-\gamma} H_t (\xi_t), \qquad (3.10)$$

$$C_t^* = \alpha_t(\xi_t) (W_t + X_t), \qquad (3.11)$$

$$\theta_t^* = \theta_t(\xi_t), \qquad (3.12)$$

$$\xi_t^* = \underset{\xi_t}{\operatorname{arg\,min}} H_t(\xi_t). \tag{3.13}$$

The proof of Theorem 3.2.2 is provided in Appendix A. Moreover, the optimal investment decision concerning illiquid wealth, that is, ΔX_t^* , is determined by the choice ξ_t^* ; but only when trading is allowed, that is, when $\mathbf{1}_t^p = 1$:

$$\Delta X_t^* = \begin{cases} (\xi_t^* - \xi_t)(W_t + X_t) & \text{if } \mathbf{1}_t^p = 1\\ 0 & \text{if } \mathbf{1}_t^p = 0, \end{cases}$$
(3.14)

where $\xi_t^* = \xi_t$ if the transaction costs do not outweigh the benefit of rebalancing back to the optimal illiquid asset allocation ξ_t^* .

The advantage of Theorem 3.2.2 is that the dependence of the value function on total wealth $W_t + X_t$ is known analytically. The fact that the value function is proportional to $(W_t + X_t)^{1-\gamma}$ simplifies the numerical optimization to a one dimensional grid search over ξ_t only. Details are provided in Appendix B.

We posit that the function $H_t(\xi_t)$ can be viewed as a penalty function which is minimized at the optimal fraction of total wealth invested in the illiquid asset ξ_t^* . If the investor is able to trade the illiquid asset at time t, then they will rebalance their portfolio towards the optimal ratio of illiquid wealth to total wealth ξ_t^* , if the decrease in the penalty function is sufficient to outweigh the transaction cost ϕ . Thus, in line with Constantinides (1986), there is a no trading region where the investor will not rebalance their portfolio. We depict the no trading region in Appendix C. As opposed to Constantinides (1986), the no trading region in our model is not symmetric and the upper bound is more stringent than the lower bound; an over-investment in the illiquid asset relative to the optimal amount may prevent the investor from smoothing consumption due to the borrowing constraints and/or potential liquidity shocks. In order to avoid these states of the world, they rebalance back to the optimal illiquid asset allocation more quickly as opposed to under investing in the illiquid asset.

Theorem 3.2.2 furthermore indicates that the optimal consumption choice and the optimal investment strategy in the liquid risky asset depend on the fraction of total wealth invested in the illiquid asset ξ_t . If illiquid wealth is substantial relative to liquid wealth, for instance after a liquidity shock L_t occurs; then the investor might have to cut their consumption relative to the case where the illiquid asset can always be traded. Moreover, to compensate for the increased risk exposure that results from the high fraction invested in illiquid wealth, the investor reduces their allocation to the liquid risky asset. The optimal consumption and asset allocation decisions are depicted in Appendix C.

3.2.4 Willingness to pay for liquidity: theory

To understand why liquidity premiums exist in some asset classes but not in others, we analyze the willingness to pay for liquidity. We define the investor's willingness to pay δ_t as the decrease in the expected return on the illiquid asset over period t that they are willing to pay to convert the illiquid asset into a liquid one. In other words, δ_t is the compensation the investor demands for holding the illiquid asset. To formalize the willingness to pay, denote the value function for Problem 3.2.1 by assuming that the asset X is actually also liquid by $V_t^{LIQ}(W_t + X_t)$. In other words, we solve Problem 3.2.1 subject to the budget constraints (3.3) and (3.4), where p = 1, $\phi = 0$. This value function factorizes as:

$$V_t^{LIQ}(W_t + X_t) = \beta^t \frac{(W_t + X_t)^{1-\gamma}}{1-\gamma} H_t^{LIQ}, \qquad (3.15)$$

for a deterministic constant H_t^{LIQ} , where H_t^{LIQ} no longer depends on ξ_t as the illiquid asset is tradeable as well.

The value function V_t^{LIQ} depends on the expected return $r_f + \lambda_X \sigma_X - 0.5\sigma_X^2$ of asset X. Subtracting δ_t from this expected return leads to a (lower) value function that we denote by $V_t^{LIQ}(W_t + X_t | \delta_t)$. We then define the willingness to pay as that value of δ_t that solves:

$$V_t^{LIQ}(W_t + X_t | \delta_t) = V_t(W_t + X_t, \xi_t).$$
(3.16)

Given (3.10) and (3.15), we can find δ_t by solving:

$$H_t^{LIQ}(\delta_t) = H_t(\xi_t), \tag{3.17}$$

where $H_t^{LIQ}(\delta_t)$ denotes the value function when the illiquid asset is actually liquid at a risk premium reduced by δ_t . This willingness to pay depends on the actual allocation to the illiquid asset: ξ_t . To determine the liquidity premium, we assume that an investor with horizon T chooses the optimal allocation to the illiquid asset when entering the investment, so the actual allocation at time t = 0 equals $\xi_{t=0}^T = \xi_{t=0}^{T,*}$.

3.2.5 Willingness to pay for liquidity: comparative statics

In this subsection we show how the model implied liquidity premiums depend on different parameter values. We use a range of realistic parameter values and show how the results change depending on the assumptions.³ With respect to the investor's preferences we assume the investor faces a liquidity shock with probability q = 10% equal to l = 20% of the previous period's total wealth, (i.e., the liquidity shock occurs on average once in 10 years), has a riskaversion parameter equal to $\gamma = 5$ and the time-preference discount factor equals $\beta = 0.91.^4$ With respect to the financial market, we assume the liquid asset has a price of risk $\lambda_S = 38\%$ and volatility $\sigma_S = 18.5\%$, and the risk-free rate is $r_f = 2\%$. These parameter values result in an optimal risky asset allocation of approximately 40% and a risk-free bond allocation of 60%.

The parameter values of the illiquid asset are set equal to the parameter values of the liquid risky asset: $\lambda_X = 38\%$ and $\sigma_X = 18.5\%$. In this way, we isolate the effect of illiquidity instead of relying on a higher Sharpe ratio for the illiquid asset. In line with this reasoning, we also assume no correlation between the liquid and illiquid risky assets in the baseline model; $\rho_{SX} = 0$. We also assume no income return for the illiquid assets, that is, $d_t = 0$. For the illiquidity parameters, we assume that the investor can trade the illiquid asset on average once in two years, or in other words p = 50%. If the investor decides to trade the illiquid asset when a trading opportunity arises, then the proportional transactions costs equal $\phi = 1\%$. At the final date, the investor has to pay transaction costs $\phi = 1\%$ in all states of the world. Table 3.1 summarizes the parameter values.

³In the next section, when linking our model to empirical evidence on liquidity premiums in different asset classes, we motivate the choice of the parameter values more thoroughly.

⁴A lower value for the time-preference discount factor has a negligible effect on liquidity premiums.

Parameters	Symbol	Value
Liquidity shock	l	20%
Probability liquidity shock	q	10%
Risk aversion parameter	γ	5
Time-preference discount factor	β	0.91
Risk-free rate	r_{f}	2%
Price of risk liquid risky asset	$\dot{\lambda_S}$	38%
Volatility liquid risky asset	σ_S	18.5%
Price of risk illiquid asset	λ_X	38%
Volatility illiquid asset	σ_X	18.5%
Correlation coefficient	ρ	0
Income return	d	0
Trading probability illiquid asset	p	50%
Transaction costs illiquid asset	ϕ	1%

Table 3.1. Parameter values

This table summarizes the parameter values of our baseline model.

Liquidity premium and investment horizon

Perhaps the most straightforward, but nevertheless important result, is that the willingness to pay depends on the investment horizon. Figure 3.1 shows that the shorter the investor's investment horizon amplifies the liquidity premium. Illiquidity is typically a bigger threat for short-term investors as they want to consume or payout a large part of their total wealth compared to long-term investors. Under the baseline parameters, the liquidity premium demanded by the short-term investors (T = 1) equals 42 basis points, while this premium converges to only a few basis points as the horizon T becomes large.

Liquidity premium and trading opportunities

Higher trading opportunities decrease the willingness to pay for illiquidity. Figure 3.2 shows that for the short-term investors (T = 2) the liquidity premium equals 32 basis points if the investor is unable to trade the illiquid asset before the final date and decreases to 20 basis points when the probability of trading is high.⁵ For long-term investors (T = 10), the liquidity premium equals 13 basis points if the investor is unable to trade the illiquid

⁵Because the transaction costs of investors with a horizon equal to T = 1 are the only component of illiquidity that affects liquidity premiums, we typically refer in the comparative statistics to the short-term investor as the investor with a horizon equal to T = 2 instead of T = 1 years.

Figure 3.1. The liquidity premium as a function of the investment horizon

This graph shows the liquidity premium as a function of the investment horizon T that assumes the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the income return d = 0, the trading probability of the illiquid asset p = 50%, and the transactions costs $\phi = 1\%$.



asset before the final date and converges to zero if the probability of trading is high. The relation between the trading probability and the liquidity premium is approximately linear for the short-term investors but decreases exponentially for the long-term investor. The short-term investor knows for sure that they are able to trade in two periods from now on, so the trading probability only affects trading one period from now. However, the inability to trade lengthens with a lower trading probability for the long-term investor. As a result, the probability of scenarios where illiquid wealth grows too fast relative to liquid wealth increases more rapidly at low trading probabilities for the long-term investor compared to the short-term investor.

Liquidity premium and transaction costs

Figure 3.3 shows that rising transaction costs increase the willingness to pay for both shortterm (T = 2) and long-term (T = 10) investors. Nevertheless, the increase is much more substantial for the short-term investor, such that a 1 percentage point increase in transaction costs increases the liquidity premium demanded by approximately 18 basis points. Short-

Figure 3.2. The liquidity premium as a function of the trading probability

This graph shows the liquidity premium as a function of the trading probability p for the investor at horizons T = 2 and T = 10 that assumes the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the income return d = 0, and the transactions costs $\phi = 1\%$.



term investors always liquidate their illiquid wealth two periods from now, and they can only do so at cost ϕ . Long-term investors only liquidate their illiquid wealth at proportional cost ϕ if the realized illiquid asset allocation is too far from the optimal level. As a result, the liquidity premium increases by 2 basis points when transaction costs increase by 1 percentage points. This increase confirms earlier results from Constantinides (1986), where the investor has an infinite horizon and transactions costs endogenously decrease their trading frequency. The larger the transaction costs, the larger the investor's no-trading region. As the investor's value function is insensitive to some deviations from the optimal (non-transaction) portfolio allocation, the transaction costs lead to second-order effects on prices.

Liquidity premium and liquidity shocks

Figure 3.4 shows that the liquidity premium is hump-shaped in the level of the liquidity shock. This hump-shaped relation means that the liquidity premium is amplified when the level of the liquidity shock increases up to a shock of l = 70% for the short-term investors

Figure 3.3. The liquidity premium as a function of the transaction costs

This graph shows the liquidity premium as a function of the transaction costs ϕ for the investor at horizons T = 2 and T = 10 that assumes the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the income return d = 0, and the trading probability of the illiquid asset p = 50%.



and l = 50% for the long-term investors, but decreases again for larger shocks.⁶ Because a liquidity shock can only be financed out of liquid wealth, a shock increases the probability that liquid wealth becomes insufficient to fulfill consumption needs. In order to prevent these states of the world, the investor reduces their optimal allocation to the illiquid asset substantially as compared to the liquid case. However, when the liquidity shock gets too severe, the allocation to the illiquid asset if it where fully liquid also gets closer to zero. Such large liquidity shocks make risky assets unattractive also in the fully liquid case, and as a result, liquidity premiums drop.

Liquidity premium and income return

The liquidity premium is also hump-shaped in the level of income return. On the one hand, the income return on the illiquid asset may diverge the optimal fraction allocated to liquid

⁶We find a similar hump-shaped relation between the liquidity premium and the probability of a liquidity shock. Also, the turning point of the hump-shaped pattern depends on the choice of the other parameter values.

Figure 3.4. The liquidity premium as a function of the level liquidity shock

This graph shows the liquidity premium as a function of the level liquidity shock l for the investor at horizon T = 2 and T = 10 that assumes the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a probability liquidity shock q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the income return d = 0, the trading probability of the illiquid asset p = 50%, and the transactions costs $\phi = 1\%$.



versus illiquid wealth. On the other hand, the income return on the illiquid asset can be used to smooth consumption and dampen the effect of liquidity shocks. Figure 3.5 shows both effects. The liquidity premium increases when the level of the income return increases up to d = 3% for the short-term investors and d = 4% for the long-term investors, but decreases again thereafter.⁷

Liquidity premium and price of risk

Figure 3.6 shows that a higher price of risk strongly amplifies liquidity premiums. A higher price of risk increases the illiquid asset's attractiveness so that the optimal fraction of wealth allocated to it increases in case the illiquid asset becomes fully liquid. The higher the fraction of total wealth that the investor optimally wants to invest in the illiquid asset, the stronger the threat of illiquidity. Because the investor cannot borrow against the illiquid asset and to remain able to smooth consumption, the gap between the optimal allocation to the illiquid

⁷The turning point of the hump-shaped pattern depends on the choice of the other parameter values.
Figure 3.5. The liquidity premium as a function of income return

This graph shows the liquidity premium as a function of the income return d for the investor at horizon T = 2 and T = 10 that assumes the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the trading probability of the illiquid asset p = 50%, and the transactions costs $\phi = 1\%$.



asset compared to when it is fully liquid widens as the illiquid asset becomes more attractive. These findings are consistent with Kahl et al. (2003) and Longstaff (2003) who show that the welfare effects of illiquidity are much larger when more wealth is tied up in the illiquid asset. For the long-term investor, illiquidity starts to pose a challenge for the investor's portfolio when the price of risk exceeds 0.55, which is equivalent to a 60% allocation to the illiquid asset.

Further, the threat of illiquidity gets more severe for the long-term investors as opposed to short-term ones at very high prices of risk. The reason is that the short-term investors know for sure that they will be able to liquidate illiquid wealth in two years and can therefore allocate a substantial amount to the illiquid asset. However, for the long-term investors there is uncertainty in the next trading opportunity. A substantial allocation to the illiquid asset therefore increases the likelihood of scenarios of insufficient liquid wealth to smooth consumption due to liquidity shocks or the poor performance of liquid relative to illiquid wealth.

Figure 3.6. The liquidity premium as a function of the price of risk

This graph shows the liquidity premium as a function of the price of risk of the illiquid asset for the investor at horizon T = 2 and T = 10 that assumes the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid risky asset $\mu_S = 9\%$, the volatility of the liquid risky asset $\sigma_S = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the income return d = 0, the trading probability of the illiquid asset p = 50%, and the transactions costs $\phi = 1\%$.



3.3 Liquidity premium in four asset classes

Next, we link our theoretical findings to the empirical evidence for liquidity premiums in several asset classes. Section 3.3.1 describes the implications of our theoretical findings in an equilibrium setting. In subsequent sections we relate these findings to the empirical evidence for liquidity premiums in four asset classes. The asset classes we consider are: private equity, real estate, corporate bonds, and stocks. We focus on US markets.

3.3.1 Implications equilibrium

So far we have analyzed the willingness to pay for a single investor. The next question to ask is: what does the willingness to pay mean for the liquidity premium in a general equilibrium setting? Under market clearing conditions the total amount of illiquid assets in the market should equal the sum of all current investors' holdings of the illiquid asset:

$$X_{mt} = \sum_{i=1}^{N} X_{it}$$
 (3.18)

The N individual investors will demand illiquid assets depending on the illiquidity source, their preferences, and their total wealth. The liquidity premium in each asset class will subsequently depend on the composition of their investors. Suppose all investors in a certain asset class have low liquidity needs, then a positive liquidity premium would generate excessive demand for the illiquid asset and therefore push down the liquidity premium. In that case, the liquidity premium will converge to zero in equilibrium. On the other hand, if liquidity matters for a substantial portion of the investors, the demand for those who are less liquidity constrained might not be enough to push down the liquidity premium.

To quantify the liquidity premium demanded by different types of investors for each asset class, we choose parameter values representative of that asset class. Throughout, we assume that the liquid risky asset S represents a liquid stock index. We use the annualized mean and the standard deviation of the S&P500 Index to model the diversified liquid stock index. Calibrated over the last 25 years, the average return is $\mu_S = 11.3\%$ and the standard deviation $\sigma_S = 17.8\%$. Moreover, we use as the risk-free rate the annualized 1-year Treasury yield over the last 25 years that gives us $r_f = 2.8\%$. Appendix D provides details on the indexes.

The preferences of investors in each market are less well-known, as researchers only have a very rough idea about investors' investment horizons and their liquidity needs that are usually represented as holdings periods (e.g. Atkins and Dyl, 1997) or investors' funding constraints (e.g. Chen et al., 2020). These measures are generally incomplete as these proxies do not measure other liquidity risks such as margin calls on derivative positions or rare disasters that investors potentially face. For this reason, we provide qualitative indicators of investors' preferences for each asset class. The main goal of this section is therefore not to model investors perfectly but to give a rough indication of the average preferences of the investors in each market.

Despite this drawback, we argue that our findings can be interpreted as upper bounds on the liquidity premiums. We make two assumptions that are more likely to overestimate rather than to underestimate the liquidity premium. First, we assume that the investors cannot borrow against the illiquid assets. This constraint may be a realistic assumption for some asset classes but not for others. For instance, real estate investors are typically able to borrow a substantial amount using the property as collateral. However, taking this borrowing into account decreases liquidity premiums because the investors can partially undo the illiquidity of the asset. Second, we allow for liquidity shocks as large as 50% of the investors' total wealth. Even though larger wealth shocks are in practice possible, our model shows that investors avoid risky assets all together if faced with such shocks. As a result, large liquidity shocks do not necessarily have a positive effect on liquidity premiums (Figure 3.4).

3.3.2 Private equity

The evidence on liquidity premiums in private equity markets is mixed, mostly because there is only indirect evidence of the potential existence or non-existence of liquidity premiums. For instance, Franzoni et al. (2012) show that adding a liquidity risk factor to the traditional three-factor model of Fama and French leads to an alpha that is statistically insignificant from zero. This is indirect evidence of no liquidity (level) premium. By contrast, Harris et al. (2014) find an out-performance of private equity versus the S&P500 Index of 3% annually. They use the compensation for illiquidity as a potential explanation for this out-performance. Here we find an average liquidity premium of around 5-15 basis points on an annual basis when using both calibrated parameters values and realistic assumptions about investors' preferences.

To assess the liquidity premium for private equity, we specify model parameters such that the illiquid asset has properties similar to private equity investments. We use the mean and standard deviation of the S&P500 Index to model the liquid counterpart of the illiquid private equity investment in our model, as the S&P500 Index is generally taken as the benchmark for private equity, see, for example, Franzoni et al. (2012) and Harris et al. (2014). This benchmark means that $\mu_X = 11.3\%$ and $\sigma_X = 17.8\%$. We do not take a stance on the correlation between private equity and the S&P500 Index. The performance of private equity varies substantially across private equity investments, as noted by Phalippou and Gottschalg (2009) for example, and as a result the correlation coefficient varies across the specifications as well. Nevertheless, Ang et al. (2014) show a correlation between the S&P500 and private equity of 0.63, S&P500 and buyouts of 0.27, and the S&P500 and venture capital of 0.57. We therefore analyze the results for the correlation coefficients of $\rho_{SX} = 0.25$ and $\rho_{SX} = 0.60$.

Private equity investments have two distinctive features. First, private equity contracts generally run for 10 years, and trading is unusual before a contract expires (Metrick and Yasuda, 2010). Therefore, we set p = 0 over the first 10 years of the investment horizon.

Second, private equity usually involves capital commitment agreements. The investor agrees to provide a preset amount of capital over the first three to five years of the project. Yet, the capital commitment is preset, so we treat it as an upfront investment in our model. We furthermore assume that the transaction cost at exiting the contract is $\phi = 1\%$. A study by Dechert and Preqin (2011) shows that transaction fees at completion of a private equity contract vary between 0.84%-1.25% depending on the size of the private equity investment.

Turning to investors' preferences in the private equity market, we posit that they likely have a low demand for liquidity, as the lock-up period of private equity is long and known beforehand. Unfortunately, as opposed to stocks and bonds, there is no public holdings data available for private equity. However, Harris et al. (2014) report that for the Burgiss database that covers \$1 trillion of committed capital to private equity over 20% is held by endowment funds and 60% by pension funds. Phalippou and Gottschalg (2009) also report endowments and pension funds as the main investors in private equity for the Thomson Venture Economics database. These findings support the assumption that only long-term investors are present in private equity markets. We then assess the liquidity premium for long-term investors who face different liquidity shocks.

Table 3.2 shows that liquidity premiums are lower for a positive correlation between the liquid and illiquid assets. The reasoning is that the effect of illiquidity is smaller because of lower diversification benefits when the correlation coefficient is larger. The liquidity shock, as shown before, amplifies liquidity premiums. Nevertheless, taking the results of Table 3.2 together indicates an average liquidity premium for private equity of 5-15 basis points only. Our findings therefore point in the direction of second-order liquidity premiums for private equity.

3.3.3 Real estate

Similar mixed evidence exists for liquidity premiums in real estate markets. Qian and Liu (2012) show that a 10% increase in their illiquidity measure decreases the contemporaneous real estate prices by 0.3 - 0.5% and increases future returns by 25 basis points over the next quarter. On the other hand, Ang et al. (2013a) show comparable performance of direct and indirect real estate investments that indicates no liquidity premium. We find an average liquidity premium of 15-35 basis points on an annual basis by using both calibrated parameters values and realistic assumptions about the investors' preferences.

To model a direct investment in real estate, we use the first two moments of the S&P US REIT Index to model the liquid counterpart of a direct real estate investment. We calibrate

Table 3.2. Liquidity premiums for private equity

This table shows the liquidity premium in basis points for investors who face liquidity shock l and the correlation coefficient equal to ρ_{SX} . Further, we use the following parameter values: investment horizon T = 10, risk-aversion parameter $\gamma = 5$, time-preference discount factor $\beta = 0.91$, return on the risk-free rate $r_f = 2.8\%$, average and standard deviation of the return on the liquid risky asset $\mu_S = 11.3\%$ and $\sigma_S = 17.8\%$, average and standard deviation of the return on the illiquid asset $\mu_X = 11.3\%$ and $\sigma_X = 17.8\%$, the income return d = 0, the trading probability of the illiquid asset p = 0, and the transaction costs equal to $\phi = 1\%$.

	Correlation coefficient ρ_{SX}		
	0.25	0.6	
Liquidity shock	l		
0.0	6	5	
0.2	8	7	
0.5	17	9	

the annualized average return on the REIT index as $\mu_X = 12.22\%$ and the corresponding standard deviation as $\sigma_X = 18.31\%$. The correlation coefficient between the liquid and illiquid assets is calibrated as the correlation between the S&P500 Index and S&P US REIT Index, which equals $\rho_{SX} = 0.4$.

The return on real estate includes both the income return and capital gains. The income return refers to the rent on properties, so the real estate returns are partially liquid. Following Hardin III et al. (2002) we assume that the income return or rent payments explain the majority of the total investment returns for REITs. In the model, we therefore set the income return equal to $d = \mu_X - r_f$. The volatility is instead largely defined by the volatility in the capital gains, and we therefore assume that the volatility relates to volatility in capital gains only.⁸

To describe the illiquidity parameters for real estate, we point out that the transaction costs consist of various main components: registration costs, legal fees, and sales and transfer taxes. On average, the total transaction costs lie in the range of 5%-8%, and we therefore set transaction costs equal to $\phi = 6\%$ (Ommeren, 2008). The typical time between transactions for residential housing is 4-5 years and 8-11 years for institutional real estate, see, for example, Hansen (1998) and Miller et al. (2011). We thus assume trading probabilities p in the range

 $^{^{8}}$ The last two assumptions lead to similar results if we slightly alter them, that is, the findings are robust to a 1% higher or lower income return and/or a positive volatility for the income return.

of 10% - 20%. Unfortunately, we are not aware of public holdings data on real estate. However, pension funds are large investors in real estate markets worldwide (Willis Towers Watson, 2019). We therefore conjecture that the majority of investors are either medium term (T = 5) or long term (T = 10). We again assess the liquidity premium for various liquidity shocks.

Table 3.3 shows the model implied liquidity premiums for real estate. The liquidity premium for real estate is diminished mainly by the large share of long-term investors and the large income return component of the real estate returns (Figure 3.1 and 3.5). The substantial transaction costs increase the no trading region resulting in similar liquidity premiums demanded for various magnitudes of the liquidity shock. Taking the average of Table 3.3, we find a liquidity premium equal to 15-35 basis points over the past 25 years. Our range of model implied liquidity premiums is consistent with the dearth of empirical evidence for liquidity premiums in real estate markets.

Table 3.3. Liquidity premiums for real estate

This table shows the liquidity premium in basis points demanded by investors at horizon T that face liquidity shock l and trading probability p. Further, we use the following parameter values: risk-aversion parameter $\gamma = 5$, time-preference discount factor $\beta = 0.91$, return on the risk-free rate $r_f = 2.8\%$, average and standard deviation of the return on the liquid risky asset $\mu_S = 11.3\%$ and $\sigma_S = 17.8\%$, average and standard deviation of the return on the illiquid asset $\mu_X = 12.2\%$ and $\sigma_X = 18.3\%$, the correlation coefficient $\rho_{SX} = 0.4$, the income return d = 9.4%, and the transaction costs $\phi = 6\%$.

	Investment	horizon $T = 5$	Investment h	norizon $T = 10$
	Trading probability p			
	10%	20%	10%	20%
Liquidity shock l				
0.0	33	31	14	13
0.2	34	32	15	14
0.5	38	35	19	18

3.3.4 Corporate bonds

In the corporate bond market, several studies have found empirical evidence for a firstorder level liquidity premium (e.g. Chen et al., 2007; Bao et al., 2011; Bongaerts et al., 2017). In particular, Bongaerts et al. (2017) show that transaction costs of 0.54% lead to a 0.56% liquidity level premium. Yet, Palhares and Richardson (2019) find only limited evidence for liquidity premiums of corporate bonds using illiquidity-factor portfolios. We find an average annual liquidity premium around 30-50 basis points by using both calibrated parameter values and realistic assumptions about the investors' preferences.

To assess the liquidity premiums of corporate bonds we again choose parameter values representative of the US corporate bond market. We use the first two moments of the Bloomberg Barclays US Corporate Bond Index to model the liquid counterpart of corporate bonds. We calibrate the annualized average return as $\mu_X = 7.0\%$ and the corresponding standard deviation as $\sigma_X = 6.6\%$. We calibrate the correlation coefficient as the correlation between the Bloomberg Barclays US Corporate Bond Index and the S&P500 Index, which equals $\rho_{SX} = 0.35$. Finally, we assume fixed coupon payments, and as for real estate, we assume that the income return equals $d = \mu_X - r_f$.

Turning to the illiquidity parameters, we see that Bongaerts et al. (2017) find average transaction costs for corporate bonds of 0.52% that range between 0.46% and 0.58% from the most liquid to the least liquid corporate bond. As the spread in transaction costs is rather low, we analyze the liquidity premium for transaction costs equal to $\phi = 0.46\%$ and $\phi = 0.58\%$. Corporate bonds are not traded as often as public equities, but the period between trades differs across bonds. For instance, Bao et al. (2011) show an annualized turnover of corporate bonds that varies between 25%-35% and Bongaerts et al. (2017) show that 15% to 25% of corporate bonds are not traded in a given week. We translate these no trading periods in trading probabilities that vary between 92% and 95% that means trades occur on average once a month to once every two weeks, so we set $p = 0.935.^9$

Table 3.4 reveals that the liquidity premium demanded by short- and long-term investors are relatively close to each other. This result is driven by the high price of risk of corporate bonds ($\lambda_X = 0.64$): Figure 3.6 shows that for this price of risk the liquidity premium demanded by short and long-term investors gets close to each other. A high price of risk also leads to a liquidity premium that is decreasing in the liquidity shock: a liquidity shock decreases the optimal allocation to the illiquid asset if it where fully liquid, diminishing the threat of illiquidity (Figure 3.4).

The Financial Accounts of the United States (Fed, 2020) reports the holdings of several asset classes within the US by investor type. We define long-term investors as the insurance companies, pension funds, and the government. The short-term investors are the banks,

⁹The results are in the same order of magnitude using p = 0.92 or p = 0.95 instead.

broker-dealers, households, and mutual funds. The Fed (2020) reports that for 2019, 38% of corporate bonds were held by long-term investors, (26% by insurance companies, 11% by pension funds, and 1% by the government), and 34% by short-term investors (7% by households, 5% by banks, 1% by broker-dealers, and 21% by mutual funds that include ETFs). The remaining 28% of the holdings are unspecified. Assuming that the distribution over long- and short-term investors is similar for the unspecified holdings, this assumption means that 53% of the corporate bonds are held by long-term investors and 47% by short-term investors. Taking the weighted average between short- and long-term investors, we expect a liquidity premium equal to 30-50 basis points over the past 25 years. This finding is below, yet fairly close to the empirical estimates found in Bongaerts et al. (2017). This may mean that empirical estimates indirectly incorporate the inability to trade as part of the liquidity premium.

Table 3.4. Liquidity premiums for corporate bonds

This table shows the liquidity premium in basis points demanded by investors at horizon T that face liquidity shock l and transaction costs ϕ . Further, we use the following parameter values: risk-aversion parameter $\gamma = 5$, time-preference discount factor $\beta = 0.91$, return on the risk-free rate $r_f = 2.8\%$, average and standard deviation of the return on the liquid risky asset $\mu_S = 11.3\%$ and $\sigma_S = 17.8\%$, average and standard deviation of the return on the illiquid asset $\mu_X = 7.0\%$ and $\sigma_X = 6.6\%$, the correlation coefficient $\rho_{SX} = 0.35$, the income return d = 4.2%, and the trading probability of the illiquid asset p = 0.935.

	Investment h	norizon $T = 1$	Investment h	orizon $T = 10$
	Transaction costs ϕ			
	0.46%	0.58%	0.46%	0.58%
Liquidity shock l				
0.0	51	56	46	51
0.2	40	46	29	33
0.5	29	32	25	26

3.3.5 Stocks

Several early studies have reported significant liquidity premiums for US equities. For instance, Amihud and Mendelson (1986) find that a 1% increase in the bid-ask spread increased the expected return by 2.4% per year. Acharya and Pedersen (2005) also show a liquidity premium of 4.5% for the most illiquid stocks versus the most liquid stocks. Ben-

Rephael et al. (2015) have challenged the existence of the liquidity premium in US stocks. They show that the liquidity premium almost disappeared over the 1997-2008 period due to lower transaction costs. For instance, for stocks listed on the NYSE, the liquidity premium for an average liquid stock was 16 basis points per month in the period from 1964-1974 but was in fact zero for the period from 1997-2008. The liquidity premium only survived for very small stocks. We find an average liquidity premium of 20-45 basis points on an annual basis using both calibrated parameters values and realistic assumptions about the investors' preferences.

For our model we again use the first two moments of the S&P500 Index to model the liquid counterpart of the illiquid stock, so $\mu_X = 11.3\%$ and $\sigma_X = 17.8\%$. Equity asset classes, such as small-growth stocks, value stocks, and mid cap stocks are highly correlated with the US stock market. For instance, the correlation between small-growth stocks and the S&P500 Index is approximately 0.9. We therefore set the correlation coefficient equal to $\rho_{SX} = 0.9$. Given the improved liquidity of US equities, we assume that the trading probability equals p = 100%; in other words, the investor can always trade the illiquid stock. The transaction costs for stocks range from 0.25% for the most liquid stocks to 8% for the least liquid stocks (Beber et al., 2020). The average transaction costs across 25 liquidity sorted portfolios equals 1.14\%. We therefore assess the liquidity premium for transaction costs equal to 0.30\%, 1%, 4\%, and 8\%.

Table 3.5 shows that the short-term investors demand substantially higher liquidity premiums than long-term investors, consistent with Figure 3.3. The liquidity premium demanded by short-term investors is bounded, because the short-term investors do not invest in the illiquid asset at all when transaction costs exceed 2%.¹⁰

The Fed (2020) reports that for 2019, 13% of US corporate equity is held by long-term investors (2% by insurance companies and 11% by pension funds), and 70% by short-term investors (38% by households, 4% by non-financial corporations, and 28% by mutual funds (including ETFs)). The remaining 17% of the holdings are unspecified. Assuming that the distribution over long- and short-term investors is similar for the unspecified holdings, this assumption means that 16% of the equities are held by long-term investors and 84% by the short-term investors. Given the average transaction costs equal to 1.14% and taking the weighted average between short and long-term investors, we expect liquidity premiums for stocks to be 20-45 basis points on average over the past 25 years. Our findings are therefore

¹⁰Recall that our model does not allow the investors to short the illiquid asset.

substantially lower than the estimates from the early literature and more in favor with the recent evidence of a small liquidity premium for stocks.

Table 3.5. Liquidity premiums for stocks

This table shows the liquidity premium in basis points demanded by investors at horizon T that face liquidity shock l, and transaction costs ϕ . Further, we use the following parameter values: risk-aversion parameter $\gamma = 5$, time-preference discount factor $\beta = 0.91$, return on the risk-free rate $r_f = 2.8\%$, average and standard deviation of the return on the liquid risky asset $\mu_S = 11.3\%$ and $\sigma_S = 17.8\%$, average and standard deviation of the return on the illiquid asset $\mu_X = 11.3\%$ and $\sigma_X = 17.8\%$, the correlation coefficient $\rho_{SX} = 0.9$, the income return d = 0, and the trading probability of the illiquid asset p = 1.

Panel A: Investment horizon $T = 1$	= 1 Transaction costs ϕ			
	0.30%	1%	4%	8%
Liquidity shock l				
0.0	15	48	105	105
0.2	11	37	98	98
0.5	6	20	85	85
Panel B: Investment horizon $T = 1$	0 Tra	insaction	costs ϕ	
	0.30%	1%	4%	8%
Liquidity shock l				
0.0	0	1	4	7
0.2	0.5	2	7	13
0.5	0	0.5	1	2

3.4 Conclusion

In this study, we investigate the liquidity premium demanded by different investor types. The cost of illiquidity may be twofold: suboptimal asset allocation and suboptimal consumption smoothing. We show that only the illiquidity that results in suboptimal consumption smoothing is able to generate a substantial liquidity premium, while the illiquidity that leads to suboptimal asset allocations does not. From this we conclude that, in equilibrium, the liquidity premiums depend on the source of illiquidity and the relative prevalence of investor types in the market. This heterogeneity across asset classes enables us to relate to the empirical evidence about the liquidity premiums in those markets.

We make several contributions. First, although private equity is the most illiquid asset

CHAPTER 3. THE LIQUIDITY PREMIUM ACROSS ASSET CLASSES

that we analyze, we only find a liquidity premium of 5-15 basis points that point in the direction of second-order liquidity premiums for this market. Private equity investors have to lockup their money for a long period of time, and mainly for that reason only long-term investors are present in this market. For these investors illiquidity is unlikely to substantially harm their consumption patterns. Moreover, the non-trading period is fixed so that investors know by and large when the position is liquidated. Second, for direct real estate we find an average liquidity premium of 15-35 basis points. Direct real estate can often not be traded for a substantial amount of time, and the timing of the trading opportunities are uncertain. Yet, the liquidity premium substantially dampens because of the large share of long-term investors and the liquid return component (rents) in real estate markets. This finding is consistent with the scare empirical evidence on liquidity premiums for real estate markets. Third, the average liquidity premium for corporate bonds equals 30-50 basis points. This finding gets close to the majority of the empirical literature on liquidity premiums for corporate bonds. Finally, for stocks we find an average liquidity premium of 20-45 basis points. This finding is in line with the recent literature that shows a diminishing liquidity premium in US equity markets.

3.5 Appendix

A Proof of optimal consumption and investment strategies

Proof of Theorem 3.2.2. Instead of the endogenous variables (W_t, X_t) , we use the pair $(W_t + X_t, \xi_t)$ as endogenous state variables. That is, we write:

$$C_t = \alpha_t (W_t + X_t, \xi_t) (W_t + X_t),$$
$$\theta_t = \theta_t (W_t + X_t, \xi_t).$$

Now, rewrite the evolution of total wealth $W_t + X_t$ using budget constraints (3.3)-(3.4) as:

$$W_{t} + X_{t} = (W_{t-1} + X_{t-1}) \\ \times [(1 - \xi_{t-1} - \alpha_{t-1} - \phi | \Delta \xi_{t-1} |) \times (\exp(r_{f}) + \theta_{t-1}(\exp(r_{t}^{S}) - \exp(r_{f}))) \\ + \xi_{t-1} \exp(d_{t}) - l_{t} + \xi_{t-1} (\exp(r_{t}^{X}) - \exp(d_{t}))],$$
(A1)

$$\xi_{t} = \frac{\xi_{t-1} \left(\exp(r_{t}^{X}) - \exp(d_{t}) \right)}{\left(1 - \xi_{t-1} - \alpha_{t-1} - \phi | \triangle \xi_{t-1} | \right) \times \left(\exp(r_{f}) + \theta_{t-1} (\exp(r_{t}^{S}) - \exp(r_{f})) \right)} + \xi_{t-1} \exp(d_{t}) - l_{t} + \xi_{t-1} \left(\exp(r_{t}^{X}) - \exp(d_{t}) \right)}$$
(A2)

where $\Delta \xi_{t-1} = \xi_{t-1}^* - \xi_{t-1}$.

The proof is by backward induction. At the final horizon t = T, the claim is obviously correct with $\alpha_T \equiv 1$ and $H_T \equiv (1 - \phi \xi_T)^{1-\gamma}$. At time T, θ_T is irrelevant. Now, for the induction argument, assume that (3.10)-(3.13) holds at time t. Then, we need to show that (3.10)-(3.13) also holds at time t - 1. From the value function (3.8), evaluated at time t - 1 and substituting (3.10), we find:

$$V_{t-1}(W_{t-1} + X_{t-1}, \xi_{t-1}) = \max_{\alpha_{t-1}, \theta_{t-1}, \xi_{t-1}} \beta^{t-1} \frac{C_{t-1}^{1-\gamma}}{1-\gamma} + \mathbb{E}_{t-1} V_t(W_t + X_t, \xi_t)$$

$$= \max_{\alpha_{t-1}, \theta_{t-1}, \xi_{t-1}} \beta^{t-1} \frac{(\alpha_{t-1}(W_{t-1} + X_{t-1}))^{1-\gamma}}{1-\gamma}$$

$$+ \mathbb{E}_{t-1} [\beta^t \frac{(W_t + X_t)^{1-\gamma}}{1-\gamma} H_t(\xi_t)]$$

$$= \max_{\alpha_{t-1}, \theta_{t-1}, \xi_{t-1}} \beta^{t-1} \frac{(W_{t-1} + X_{t-1})^{1-\gamma}}{1-\gamma}$$

$$\times \left(\alpha_{t-1}^{1-\gamma} + \beta \mathbb{E}_{t-1} [\{(1 - \xi_{t-1} - \alpha_{t-1} - \phi | \Delta \xi_{t-1}|) \times (\exp(r_f) + \theta_{t-1}(\exp(r_t^X) - \exp(d_t))\}^{1-\gamma} H_t(\xi_t)] \right)$$

$$= \max_{\alpha_{t-1}, \theta_{t-1}, \xi_{t-1}} \beta^{t-1} \frac{(W_{t-1} + X_{t-1})^{1-\gamma}}{1-\gamma} H_t(\xi_t)$$
(A3)

At time t - 1, the penalty function $H_{t-1}(\xi_{t-1})$ equals

$$H_{t-1}(\xi_{t-1}) = \alpha_{t-1}^{1-\gamma} + \beta \mathbb{E}_{t-1}[\{(1 - \xi_{t-1} - \alpha_{t-1} - \phi | \Delta \xi_{t-1} |) \\ \times (\exp(r_f) + \theta_{t-1}(\exp(r_t^S) - \exp(r_f))) \\ + \xi_{t-1} \exp(d_t) - l_t + \xi_{t-1} \left(\exp(r_t^X) - \exp(d_t)\right)\}^{1-\gamma} H_t(\xi_t)]$$
(A4)

Therefore, the function $H_{t-1}(\xi_{t-1})$ is a function of t-1 and ξ_{t-1} only and hence (3.10) from (3.10)-(3.13) hold for all t. We continue with proving (3.11)-(3.12) at time t-1. The first-order conditions of the decision variables α_{t-1} and θ_{t-1} equal:

$$\alpha_{t-1}^{UC*} = \underset{\alpha_{t-1}}{\arg\max} \frac{(\alpha_{t-1}(W_{t-1} + X_{t-1}))^{1-\gamma}}{1-\gamma} + \mathbb{E}_{t-1}V_t(W_t + X_t, \xi_t),$$
(A5)

$$\theta_{t-1}^* \underset{\theta_{t-1}}{\arg\max} \frac{(\alpha_{t-1}(W_{t-1} + X_{t-1}))^{1-\gamma}}{1-\gamma} + \mathbb{E}_{t-1}V_t(W_t + X_t, \xi_t), \tag{A6}$$

where α_{t-1}^{UC*} is the solution if the investor were unconstrained, i.e. when constraint (3.5) does not bind. Because we assume that the investor cannot borrow against the illiquid asset

the constrained solution becomes:

$$\alpha_{t-1}^{C*} = \begin{cases} \alpha_{t-1}^{UC*} & \text{if } \alpha_{t-1}^{UC*} \le 1 - \xi_{t-1} \\ 1 - \xi_{t-1} & \text{if } \alpha_{t-1}^{UC*} > 1 - \xi_{t-1} \end{cases}$$
(A7)

We can now rewrite (A5) and (A6) as:

$$\frac{\partial V_{t-1}}{\partial \alpha_{t-1}} = \beta^{t-1} (\alpha_{t-1}(W_{t-1} + X_{t-1}))^{-\gamma} \\
- \mathbb{E}_{t-1} \left[\frac{\partial V_t}{\partial W_t + X_t} \Big(\exp(r_f) + \theta_{t-1}(\exp(r_t^S) - \exp(r_f)) \Big) \right] \\
+ \mathbb{E}_{t-1} \left[\frac{\partial V_t}{\partial \xi_t} \xi_t \frac{1}{W_t + X_t} \Big(\exp(r_f) + \theta_{t-1}(\exp(r_t^S) - \exp(r_f)) \Big) \right] = 0, \quad (A8) \\
\frac{\partial V_{t-1}}{\partial \theta_{t-1}} = \mathbb{E}_{t-1} \left[\frac{\partial V_t}{\partial W_t + X_t} \Big(\exp(r_t^S) - \exp(r_f) \Big) \right] \\
+ \mathbb{E}_{t-1} \left[\frac{\partial V_t}{\partial \xi_t} \xi_t \frac{1}{W_t + X_t} \Big(\exp(r_t^S) - \exp(r_f) \Big) \right] = 0. \quad (A9)$$

To see that both α_{t-1}^* and θ_{t-1}^* depend only on ξ_{t-1} , we solve (A8) and (A9) and substitute (A1) into (A8) and (A9), we get:

$$\frac{\partial V_{t-1}}{\partial \alpha_{t-1}} = \alpha_{t-1}^{-\gamma} + \beta \mathbb{E}_{t-1} [\{ (1 - \xi_{t-1} - \alpha_{t-1} - \phi | \Delta \xi_{t-1} |) (\exp(r_f) + \theta_{t-1} (\exp(r_t^S) - \exp(r_f))) \\
+ \xi_{t-1} \exp(d_t) - l_t + \xi_{t-1} \left(\exp(r_t^X) - \exp(d_t) \right) \}^{-\gamma} \times \left(\frac{H'_t(\xi_t)}{1 - \gamma} \xi_t - H_t(\xi_t) \right) \exp(r_f)], \\
= 0 \tag{A10}$$

$$\frac{\partial V_{t-1}}{\partial \theta_{t-1}} = \mathbb{E}_{t-1} [\{ (1 - \xi_{t-1} - \alpha_{t-1} - \phi | \Delta \xi_{t-1} |) (\exp(r_f) + \theta_{t-1} (\exp(r_t^S) - \exp(r_f))) + \xi_{t-1} \exp(d_t) \\
- l_t + \xi_{t-1} \left(\exp(r_t^X) - \exp(d_t) \right) \}^{-\gamma} \times \left(H_t(\xi_t) - \frac{H'_t(\xi_t)}{1 - \gamma} \xi_t \right) \left(\exp(r_t^S) - \exp(r_f) \right)] \\
= 0. \tag{A11}$$

The first-order conditions (A10) and (A11) depend only on time t - 1 and the fraction invested in the illiquid asset ξ_{t-1} . In this way, the optimal consumption and the fraction invested in the liquid risky assets can indeed be written as in (3.11) and (3.12), so (3.11)-(3.12) from (3.10)-(3.13) holds for all t. We finish the proof by showing that (3.13) also holds at time t - 1. When a trading opportunity arises at t - 1, the investor chooses ξ_{t-1} such that the value function at t-1 is optimized:

$$\xi_{t-1}^{*} = \arg \max_{\xi_{t-1}} V_{t-1}(W_{t-1} + X_{t-1}, \xi_{t-1}) = \arg \max_{\xi_{t-1}} \beta^{t-1} \frac{(W_{t-1} + X_{t-1})^{1-\gamma}}{1-\gamma} H_{t-1}(\xi_{t-1})$$

=
$$\arg \min_{\xi_{t-1}} H_{t-1}(\xi_{t-1}).$$
 (A12)

B Numerical implementation

This appendix provides an outline of the numerical method to solve the baseline model. First, we describe the sequence of making decisions. Second, we explain the numerical solution technique to solve for the decision variables.

Sequence of decision making

Figure B depicts the sequence of making decisions. The endogenous variables, liquid wealth W_t and illiquid wealth X_t , are defined as total wealth before consumption, liquidity shocks, and returns earned in period (t, t + 1]. Based on the actual fraction allocated to the illiquid asset ξ_t , the investor chooses the optimal fraction of total wealth to be consumed in period (t, t + 1], $\alpha_t(\xi_t)$, and the optimal allocation towards the liquid risky asset, $\theta_t(\xi_t)$. If a trading opportunity arises at time t, the investor chooses simultaneously ξ_t^* , $\alpha_t(\xi_t^*)$ and $\theta_t(\xi_t^*)$. Further, by the assumption $X_t \ge 0$ and the inability to borrow against the illiquid asset, the possible values for ξ_t are restricted to the interval [0, 1].

Numerical solution technique

The baseline model is solved by means of backward induction, where we start solving the problem at the final date t = T and solve the model backwards for each period until arriving at time t = 0. At the final horizon t = T, we have $\alpha_T \equiv 1$ and $H_T \equiv (1 - \phi \xi_T)^{1-\gamma}$. To solve for ξ_{T-1}^* , $\alpha_t(\xi_{T-1})$ and $\theta_t(\xi_{T-1})$, we construct a grid for $\xi_{T-1} \in [0, 1]$. We simulate M = 100,000 trajectories for the exogenous state variables, the returns on the liquid and illiquid risky asset in the periods (T - 1, T], R_T^S , and R_T^X from a multi-normal distribution with means and variance-covariance matrix, as described in Section 3.2.1. We also simulate M = 100,000 trajectories for the liquidity shock L_T from a Bernoulli distribution as described in Section 3.2.2.

For each grid point by using a nonlinear least squares, we solve the first order conditions with respect to consumption (A10) and the allocation towards the liquid risky asset (A11) by using $H_T(\xi_T) \equiv (1 - \phi \xi_T)^{1-\gamma}$, R_T^S , R_T^X , and l_T to find $\alpha_{T-1}(\xi_{T-1})$ and $\theta_{T-1}(\xi_{T-1})$. Then we are able to compute $H_{T-1}(\xi_{T-1})$ and solve for $\xi_{T-1}^* = \arg \min_{\xi_{T-1}} H_{T-1}(\xi_{T-1})$ with the corresponding consumption level $\alpha_{T-1}(\xi_{T-1}^*)$ and the allocation to the liquid risky asset $\theta_{T-1}(\xi_{T-1}^*)$. This gives us the optimal solution at time T-1.

We then solve for the optimal solution at time T-2 in the same way, except that we also simulate M = 100,000 trajectories for the trading probability p_{T-1} from a Bernoulli distribution as described in Section 3.2.2. In the scenarios the investor is able to trade $(\mathbf{1}_{T-1}^{p} = 1)$, we use $H_{T-1}(\xi_{T-1}^{*})$ and in the scenarios the investor is unable to trade $(\mathbf{1}_{T-1}^{p} = 0)$, we use $H_{T-1}(\xi_{T-1})$. Together with R_{T-1}^{S} , R_{T-1}^{X} , and l_{T-1} we find $\alpha_{T-2}(\xi_{T-2})$ and $\theta_{T-2}(\xi_{T-2})$. We again compute $H_{T-2}(\xi_{T-2})$ and solve for $\xi_{T-2}^{*} = \arg\min_{\xi_{T-2}} H_{T-2}(\xi_{T-2})$ with the corresponding consumption level $\alpha_{T-2}(\xi_{T-2}^{*})$ and the allocation to the liquid risky asset $\theta_{T-2}(\xi_{T-2}^{*})$. We can continue this approach until we arrive at t = 0.

C Optimal consumption and asset allocation baseline model

This appendix shows the optimal consumption pattern and asset allocation decisions for the baseline model described in Section 3.2. Both consumption (Figure 3.7) and the allocation to the liquid risky assets (Figure 3.8) are functions of the investment horizon and the fraction invested in the illiquid asset, as derived in Theorem 3.2.2. These graphs show that the larger the actual allocation to illiquid assets, the lower the consumption and allocation to the liquid risky asset. The optimal allocation to the illiquid asset is in Figure 3.9 and found by minimizing the penalty function $H_t(\xi_t)$ (Theorem 3.2.2). We compare this allocation to the case where the illiquid asset is fully liquid, that is, the two risky asset Merton case (Merton, 1969). The shorter the investment horizon, the lower the optimal allocation to the illiquid asset relative to the Merton case.

The no-trade region induced by transaction costs is in Figure 3.10. As long as the illiquid asset allocation is within the no-trade region, that is, the area within the two dashed lines, the investor does not trade if a trading opportunity arises, while the investor re-balances back to the optimal allocation when outside of the no-trade region. Compared to Constantinides

(1986), the upper bound of the no-trading region is lower in our model; an over-investment in the illiquid asset relative to the optimal amount may prevent the investor from smoothing consumption due to the borrowing constraints and/or potential liquidity shocks. In order to avoid these states of the world, they re-balance back to the optimal illiquid asset allocation more quickly as opposed to under-investment in the illiquid asset.

Figures

Figure 3.7. Optimal consumption

This graph shows the optimal consumption as a function of the investment horizon T and fraction invested in the illiquid asset that uses the following parameter values: the risk-aversion parameter $\gamma = 5$, the timepreference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, income return d = 0, the trading probability of the illiquid asset p = 50%, and the transactions costs $\phi = 1\%$.



Figure 3.8. Optimal liquid risky asset allocation

This graph shows the optimal liquid risky asset allocation as a function of the investment horizon T and fraction invested in the illiquid asset that uses the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the income return d = 0, the trading probability of the illiquid asset p = 50%, and the transactions costs $\phi = 1\%$.



Figure 3.9. Optimal illiquid asset allocation

This graph shows the optimal illiquid asset allocation as a function of the investment horizon T that uses the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the income return d = 0, the trading probability of the illiquid asset p = 50%, and the transactions costs $\phi = 1\%$.



Figure 3.10. No-trading region

This graph shows the no-trading region of the investor with an investment horizon T = 10 that uses the following parameter values: the risk-aversion parameter $\gamma = 5$, the time-preference discount factor $\beta = 0.91$, a liquidity shock l = 20% with probability q = 10%, the return on the risk-free rate $r_f = 2\%$, the average return on the liquid and illiquid risky asset $\mu_S = \mu_X = 9\%$, the volatility of the liquid and illiquid risky asset $\sigma_S = \sigma_X = 18.5\%$, the correlation coefficient $\rho_{SX} = 0$, the income d = 0, the trading probability of the illiquid asset p = 50%, and the transactions costs $\phi = 1\%$.



D Market Indexes

Here we list the market indexes used to calibrate the four different asset classes: private equity, real estate, corporate bonds, and stocks.

Table 3.6. Market Indexes calibration

This table lists the indexes used to calibrate the four different asset classes described in Section 3.3 as well as the source for the risk-free rate.

Risk-free rate	1-year Treasury yield
Stocks	$S \& P500 \ Index$ The index includes 500 leading companies publicly traded in the US stock market.
Corporate bonds	Bloomberg Barclays US Corporate Bond Index The index measures the investment grade, fixed-rate taxable corporate bond market. It includes USD denominated securities publicly issued by US and non-US industrial, utility and financial issuers.
Real estate	S&P United States REIT Index The index includes the invest-able universe of publicly traded real estate investment trusts domiciled in the US.
Private Equity	S&P500 Index

Chapter 4

Pension Fund's Illiquid Assets Allocation under Liquidity and Capital Requirements¹

4.1 Introduction

Pension funds are important investors in illiquid asset classes such as real estate, mortgages, private equity, hedge funds, and infrastructure. The annual OECD survey of large pension funds reveals an average illiquid assets allocation of 15 percent in 34 countries.² According to the 2018 Willis Towers Watson Global Pension Asset Study the 7 largest pension markets in the world (Australia, Canada, Japan, the Netherlands, Switzerland, UK, and US) have increased their average illiquid assets allocation from 4 percent in 1997 to 25 percent in 2017.³ Illiquid assets may offer pension plans benefits in terms of a liquidity premium e.g., Sadka (2010), Franzoni et al. (2012), and Qian and Liu (2012), portfolio diversification (Jacobs et al., 2014) and liability hedging (Hoevenaars et al., 2008). Their long investment horizon

¹This chapter is based on joint work with Dirk Broeders and Bas Werker and published in the Journal of Pension Economics and Finance. We thank two anonymous referees, editor Stephen Dimmock, Jules van Binsbergen, Bart Bos, Lisanne Cock, Pieter van Foreest, Rob van den Goorbergh, Janko Gorter, Leo de Haan, Frank de Jong, Theo Nijman, Arco van Oord, Marco Teunissen, Marno Verbeek, Manon ten Voorde, Sweder van Wijnbergen, participants of seminars at Achmea Investment Management, APG, Basel University, Cardano, DNB, EPFIF, Fidelity, ICPM, Maastricht University, Ortec Finance, Robeco and conference participants of the INPARR Seminar Consistency Amid Complextity: Navigating the Future of Pensions (OECD, Paris) and the International Pension Workshop (Netspar, Leiden) for very useful comments. Moreover, we are grateful to David Keijzer and Jeroen Meringa for the data gathering process. Kristy gratefully acknowledges financial support from Instituut Gak.

²www.oecd.org/daf/fin/private-pensions/2015-Large-Pension-Funds-Survey.pdf

³https://www.willistowerswatson.com/-/.../Global-Pension-Asset-Study-2018-Japan.pdf

makes pension funds well placed to invest in illiquid asset classes (Attig et al., 2012).

Several papers study investment risk-taking by pension funds, e.g. Sharpe (1976), Jin et al. (2006), Rauh (2009), An et al. (2013), and Andonov et al. (2017). These papers typically focus on the allocation to risk-free assets (such as government bonds) on the one hand versus risky assets (such as equities) on the other hand. In this paper we address the investment policy of pension funds regarding illiquid asset classes. So far, little is known about the factors that influence how much they invest in illiquid asset classes. The main purpose of this paper is to show how liquidity and capital requirements affect the appetite of pension funds towards investing in illiquid assets. Several papers provide indeed evidence that regulations have a significant impact on pension fund's investment decisions, e.g., Sias (2004), An et al. (2013), and Andonov et al. (2017). We, therefore, build our paper on a theoretical model of liquidity and capital requirements. Based on that model we link pension funds' characteristics to their investments in illiquid assets. We test this in an empirical study using unique and unbiased data on Dutch pension funds.

First, we assess the aggregate illiquid assets allocation as a fraction of the risky assets allocation, following the typical top-down decision-making process of a pension fund. Second, we study the implications for the following illiquid asset classes separately: real estate, mortgages, private equity, and hedge funds. While we do the empirical analysis on defined benefit pension plans in this paper, our theoretical framework also applies to different types of institutional investors. For instance, insurers with guaranteed products take into account both liquidity and capital requirements in choosing their optimal asset allocation (Niedrig, 2015). Also banks are exposed to liquidity and capital requirements in their asset allocation decision (Khan et al., 2017).

The *liquidity requirements* of pension funds consist of two components: short-term pension payments and collateral requirements following margin calls on derivative contracts. The cash required for pension payments over the next year is well predictable. However, the cash needed for collateral requirements is much less predictable. If the market value of a derivative contract declines, a pension fund must transfer cash or highly liquid short-term bonds to a margin account in order to limit the risk the counterparty faces. Margining on derivatives can become quite substantial, especially during financial crises.

We also consider *capital requirements* as a source of variation in illiquid assets holdings by pension funds. Regulations often require defined benefit plans to have sufficient capital to manage the risks they are exposed to, such as financial market risks and longevity risk.⁴ In choosing its strategic asset allocation, a pension fund optimizes the trade-offs between different risk factors for a given level of required capital. As a result, a pension fund may be constrained to increase their exposure to illiquid assets. If a certain risk factor consumes a larger part of the available capital, less capital remains to be allocated to illiquid assets.

Liquidity and capital requirements interact in two ways. First, they interact through the liability duration as a measure of a pension fund's average investment horizon. On the one hand, a pension fund with a high liability duration is less liquidity constrained as it will have to pay less pensions in the short-term. This allows for a higher allocation to illiquid assets. On the other hand, a high liability duration also implies more exposure to interest rate risk through the present value of its liabilities. This restricts the opportunity to invest in illiquid assets as more of the available capital is required for interest rate risk. Second, liquidity and capital requirements also interact through derivatives. For instance, hedging interest rate risk increases the liquidity requirement as a result of higher collateral requirements. However, by hedging interest rate risk, a pension fund frees up capital to invest more in illiquid assets. Our theoretical model in Section 4.2 precisely disentangles these effects.

We empirically test the predictions from our theoretical model using the investment decisions of Dutch occupational pension funds from 2012 to 2017. The Dutch pension system is particularly well suited to study the effect of liquidity and capital requirements on the illiquid assets allocation for multiple reasons. First, Dutch pension funds have mandatory reporting requirements to the prudential supervisor, De Nederlandsche Bank (DNB). This gives us unbiased data. Second, the Dutch pension system is large in terms of size. The total assets under management (AUM) of Dutch pension funds' is approximately 1.3 trillion euro or 1.5 times the GDP of the Netherlands. Third, Dutch pension funds do not face quantitative investment restrictions. Regulation allows them to invest in any asset class in any country, as long as a pension fund complies with the capital requirements and the well-known prudent person principle. Dutch pension funds invest in a broad range of asset classes. In fact, over two-thirds of them invest in illiquid assets. Fourth, Dutch pension funds mainly have defined benefit pension liabilities. This means that they have a clear asset-liability perspective in making strategic investment decisions. Liabilities are valued marked-to-market by discounting accrued benefits against the prevailing term structure of

 $^{^{4}}$ In Section 4.2.2 we explain that a pension fund's capital is the difference between the market value of assets and liabilities. The term capital requirements is synonymous to funding requirements.

market interest rates. As a result, we can analyze the impact of liability interest rate risk on the illiquid assets allocation.

In short, these are our key findings. First, in line with our theoretical framework, we find a hump-shaped impact of liability duration on the fraction of risky assets invested in illiquid assets. Up to 18 years, liability duration positively affects this allocation. For higher durations the effect reverses. Second, we do not find evidence that interest rate risk hedging impacts the fraction of risky assets allocated to illiquid assets. Currency risk hedging, by contrast, creates the opportunity to allocate more to illiquid assets. A one standard deviation increase in currency risk hedging leads to an increase in the fraction of risky assets allocated to illiquid assets of approximately 0.71 percentage points. This implies a relative increase in the fraction of risky assets allocated to illiquid assets of 5.5 percent. Third, we find that, next to liability duration, also other pension fund characteristics impact strategic asset allocation decisions. Size positively affects the fraction of risky assets allocated to illiquid assets, which is in line with Andonov (2014) and Dyck and Pomorski (2016). A pension fund that is ten times larger in terms of assets under management has a 7.4 percentage points higher fraction of risky assets allocated to illiquid assets. Furthermore, corporate pension funds invest 7.6 percentage points less in illiquid assets as fraction of risky assets compared to industry-wide and professional group pension funds. Finally, pension funds with a lower funding ratio invest a larger fraction of risky assets in illiquid assets, supporting the results found in Basak and Shapiro (2001). A one standard deviation decrease in the funding ratio increases the fraction of risky assets allocated to illiquid assets by 0.89 percentage points.

The remainder of this paper is organized as follows. Section 4.2 explains the liquidity and capital requirements of pension funds and in which way these requirements affect the illiquid assets allocation in a theoretical framework. The data description is given in Section 4.3. The model and results are discussed in Section 4.4. The robustness check is in Section 4.5. Section 4.6 describes the implications of our theoretical framework for pension funds in different regulatory frameworks and Section 4.7 concludes.

4.2 Illiquid assets allocation: theory

A key responsibility of a pension fund is to optimize the asset allocation given its liability structure. This is known as Asset Liability Management (ALM). Here, we focus on a specific part of the ALM process: the strategic allocation to illiquid assets. For our theoretical framework the focus is on the aggregate illiquid assets allocation. In the empirical section we, however, also consider individual illiquid asset classes.

In deriving their strategic allocation to illiquid asset classes, pension funds assess the benefits and costs imposed by these investments. Important drivers of the investment decision are the risk-return trade-offs, the portfolio diversification benefits, and the ability of illiquid assets to hedge liability risk. Other drivers of investment decisions are constraints. We focus on liquidity and capital requirements and conjecture that these impact investment decisions. We, therefore, first formally introduce the liquidity and capital requirements. Next, we show in a theoretical model how the two requirements interact and affect investment decisions.

4.2.1 Liquidity requirements

A pension fund must have sufficient liquidity available to fulfill its immediate obligations. Liquidity requirements consist mainly of two components: short-term pension payments and collateral requirements on interest rate and currency derivatives. The cash required for pension payments in the foreseeable future is generally well predictable. The number of retirees and their benefits are known. However, the cash needed for collateral requirements, is much less predictable. If the market value of a derivative declines, the pension fund is required to transfer cash or highly liquid short-term bonds to a margin account. Margining on derivatives can become quite substantial, especially during financial crises. Pension payments can be seen as the expected liquidity requirement, whereas collateral requirements present liquidity risk. Liquidity problems arise when a pension fund lacks the resources to fulfill its immediate obligations.⁵

An investor facing liquidity risk should reduce the allocation to illiquid assets in order to avoid states of the world in which it would be short liquidity. Theoretical studies from for instance Gârleanu (2009) and Ang et al. (2014) show this formally. From a theoretical perspective we, therefore, predict that higher liquidity requirements restrict investments in illiquid assets. We will now discuss liquidity requirements in more detail.

⁵Salary payments to pension fund's staff, administrative expenses and investment costs are also sources that require short-term liquidity, but are generally small compared to the size of the pension fund and therefore outside the scope of this paper.

Liquidity requirement for pension payments

We start with the short-term pension payments and denote these by LR_P . We define this as the ratio of pension payments in the first year to the present value of total liabilities. To formalise this, suppose we have an homogeneous group of participants with mortality rate λ that receive annual pension payments of A at the beginning of each year. The pension payments in year t are then given by

$$A\exp(-\lambda t).\tag{4.1}$$

Assuming a flat term-structure of interest rates r, the present value of all future pension payments equals

$$V = \int_0^\infty A \exp(-(r+\lambda)t) dt = \frac{A}{r+\lambda}.$$
(4.2)

Then, in relative terms, the short-term pension payments equal

$$LR_P = \frac{A}{A(r+\lambda)^{-1}} = r + \lambda.$$
(4.3)

The duration of the present value of future pension payments V equals

$$D_V = -\frac{1}{V}\frac{dV}{dr} = \frac{1}{r+\lambda}.$$
(4.4)

Using (4.3) and (4.4) we rewrite the liquidity requirement from pension payments as

$$LR_P = \frac{1}{D_V}.\tag{4.5}$$

Equation (4.5) has an intuitive interpretation. The liability duration is the weighted average time to maturity of all pension payments. In other words, it measures a pension fund's average investment horizon. A high liability duration D_V implies less short-term payments as only a small fraction of the liabilities has to be paid out in the near future. In line with the findings in Ang et al. (2014), a high liability duration, therefore, creates opportunities to invest in illiquid assets. The inability to frequently trade illiquid assets is less of a restriction for a pension fund with a high liability duration.

Liquidity requirement for interest rate derivatives

Next to short-term pension payments, liquidity requirements also arise to meet margin requirements on derivatives. We consider margin requirements on interest rate swaps in this section and on currency forwards in the next section. For model tractability we refer to the interest rate derivates portfolio as a position in a single receiver swap. Pension funds mainly use receiver swaps to hedge the interest rate risk embedded in the present value of the pension liabilities. In a receiver swap a pension fund pays a counterparty a floating rate and receives, in return, a fixed rate over a certain notional amount. We denote the notional of the receiver swap by N and the duration of its fixed leg by D_R . The fraction of interest rate risk hedged with the receiver swap, again relative to the total value of liabilities V, is given by the following swap hedge ratio

$$\phi^R = \frac{N}{V} \frac{D_R}{D_V}.$$

We model the liquidity requirement on the swap through a margin call. In case of a receiver swap an increase in the interest rate dr^+ leads to a margin call because it lowers the value of the fixed leg of the swap. As a first-order approximation, the margin calls on the swap then equals

$$MC_R = \frac{ND_R dr^+}{V} = \phi^R D_V dr^+.$$
(4.6)

We could add a second-order, convexity, effect to (4.6). However, this has only a small effect on the quantitative results of the model and is, therefore, excluded here to keep the model simple.

Liquidity requirement for foreign exchange derivatives

The liquidity requirement on foreign exchange derivatives also results from margin calls in case the value of the derivative portfolio decreases. Pension funds generally hedge exchange rate risk using currency forwards. We assume that the pension funds in our model each have a position in a single forward contract. We consider an increase in the foreign exchange rate dFX^+ because this lowers the value of the forward contract and, thus, results in a margin call. The liquidity requirement from margin calls on foreign exchange derivatives then equals

$$MC_{FX} = w^{FX} \phi^{FX} dFX^+, \tag{4.7}$$

155

where w^{FX} is the fraction of V invested in foreign assets and ϕ^{FX} is the foreign exchange hedge ratio relative to V.

We can now determine the total liquidity requirement by adding together (4.5)-(4.7). This gives us the total liquidity requirement LR, relative to the value of the pension liabilities V, as

$$LR = \frac{1}{D_V} + \phi^R D_V dr^+ + w^{FX} \phi^{FX} dFX^+.$$
(4.8)

4.2.2 Capital requirements

Next to liquidity requirements the pension funds in our sample also face capital requirements to be able to absorb financial markets risks and longevity risk. In contrast to a bank or an insurer, a pension fund generally does not have shareholders that provide capital. Instead, a pension fund's capital is the difference between the value of the assets and the value of the liabilities. In the Netherlands, the capital requirement is risk-based and calculated as a Value-at-Risk risk measure (Broeders and Pröpper, 2010). In fact, the level of the required capital is calculated such that the probability that the funding ratio falls below 100 percent on a one-year horizon equals 2.5 percent. This level is pension fund specific. Pension funds determine the capital requirement by applying a method prescribed by law that takes the following risk factors into account: interest rate risk, equity and real estate risk, currency risk, commodity risk, credit risk, and longevity risk. The capital required for each of these risk factors is determined by simulating the impact on a pension fund's capital of a prescribed shock in the risk factor. The total capital requirement follows from aggregating the individual requirements using a correlation matrix. In practice a pension fund does not always have sufficient capital. If that is the case a pension fund gets a 10 year recovery period to become compliant again. The Dutch setting is unique and allows us study the effect of capital requirements on the illiquid assets allocation. For the purpose of our paper we only consider the capital requirement for interest rate risk and for currency risk. Pension funds also have a capital requirement for equity risk. But this requirement is only related to the equity allocation and does not depend on the liability structure, which is the core risk driver in our model.

Capital requirement for interest rate risk

The root cause of interest rate risk is embedded in the present value of the future pension payments. Pension funds can hedge this liability interest rate risk not only with receiver swaps but also with bonds. In the Dutch regulations, the capital requirement is therefore based on the part of the liability interest rate risk that is not hedged with swaps and bonds. We already introduced the swap hedge ratio ϕ^R . The bond hedge ratio, or the fraction of interest rate risk hedged with bonds, is denoted by ϕ^B . If a pension fund invests fraction B/V in bonds with a duration of D_B then ϕ^B is defined as

$$\phi^B = \frac{B}{V} \frac{D_B}{D_V}.$$

Next we consider a decrease in the interest rate dr^- . An interest rate decrease typically lowers the funding ratio because it increases the value of the liabilities more than the value of the swaps and bonds. The first-order approximation of the capital requirement for interest rate risk is given by⁶

$$CR_R = -(1 - \phi^R - \phi^B)D_V dr^-.$$
(4.9)

To keep the capital requirement for interest rate risk in (4.9) positive, we assume that pension funds never over-hedge their interest rate risk exposure, so that $\phi^R + \phi^B \leq 1.^7$

Capital requirement for foreign exchange rate risk

Pension funds hedge foreign exchange rate risk typically with currency forwards. Exchange rate risk occurs when the liabilities are dominated in one currency and the investments in another currency. The capital requirement depends on the part of the exchange rate risk that is not hedged with forwards. We consider a decrease in the foreign exchange rate $dFX^$ because this lowers the value of the assets in the local currency. The capital requirement for

 $^{^{6}\}mathrm{We}$ again exclude the convexity effect here as it only has a small second-order effect on the quantitative results.

⁷This assumption is realistic as the value-weighted average interest rate risk hedging across Dutch pension funds equals 40 percent, meaning that $\phi^R + \phi^B = 0.40$.

exchange rate risk equals

$$CR_{FX} = -w^{FX}(1 - \phi^{FX})dFX^{-}.$$
(4.10)

Adding (4.9) and (4.10) together gives us the total capital requirement CR, again relative to V, as

$$CR = -(1 - \phi^R - \phi^B)D_V dr^- - w^{FX}(1 - \phi^{FX})dFX^-.$$
(4.11)

4.2.3 Overall liquidity and capital requirements

To derive the overall impact of liquidity and capital requirements on investment policy we will combine them. In order to do that we make the following three assumptions. First, we assume that capital and margin requirements are interchangable. If a pension fund hedges risks with derivatives it will have a lower capital requirement but it will be more exposed to margin calls. Second, we assume that a positive and negative shocks in a risk factor are equally likely and that the changes are identical in absolute value. The latter means for interest rate risk that $dr^+ = -dr^- = dr$ and for exchange rate risk that $dFX^+ =$ $-dFX^- = dFX$. This is reasonable as a pension fund cannot ex ante know whether an increase or decrease in the underlying risk factor occurs. Third, we assume that the capital requirements and the margin calls are calibrated with the same probability on the same horizon. We believe this to be reasonable because from a risk management perspective it is logical to measure risk independent of how to manage the risk (via capital or via derivatives).

Under these assumptions we can combine (4.8) and (4.11) to obtain the following total requirement

$$TR = \frac{1}{D_V} + (1 - \phi^B) D_V dr + w^{FX} dFX.$$
(4.12)

Equation (4.12) shows that the total requirement is a non-linear function of the liability duration and a linear function of the foreign exchange risk exposure. The reader may notice that Equation (4.12) does not depend on the swap hedge ratio ϕ^R . The intuition behind this is that in our model a margin call on the swap cancels against the capital requirement. The more a pension fund hedges it's interest rate risk with swaps, the higher the liquidity requirement and equally lower the capital requirement.⁸ For the same reason, ϕ^{FX} also does not appear in the total requirement (4.12).

Based on the analysis above we can derive for which pension funds the total requirement TR is least constraining to invest in illiquid assets. For that we examine how the liability duration of a pension fund impacts the total requirement. We equate the first-order derivative of TR with respect to the liability duration D_V to zero, to get

$$\frac{dTR}{dD_V} = -\frac{1}{D_V^2} + (1 - \phi^B)dr = 0.$$
(4.13)

If we solve this, we find the liability duration for which the total requirement TR is least constraining to be

$$D_V^* = \frac{1}{\sqrt{(1 - \phi^B)dr}}.$$
(4.14)

Based on this we expect pension funds with a liability duration of around D_V^* to have the highest exposure to illiquid assets. Pension funds with a higher or a lower liability duration are more constrained by capital and liquidity requirements to invest in illiquid assets.

We illustrate how liability duration impacts the total requirement TR in a stylized, but representative, example. We take the following shocks to calculate the liquidity and capital requirement to be equal to dr = 0.5% and dFX = 25%. These values are comparable to the ones in Dutch pension funds' regulations and to the calibration in Solvency II regulation. Next we take a continuum of pension funds that are equal except for their liability duration. We use the following parameter values that match with the average Dutch pension fund: $\phi^R = 20\%$, $\phi^B = 20\%$, $\phi^{FX} = 25\%$, and $w^{FX} = 50\%$. Figure 4.1 shows how, all else equal, the liquidity, capital, and total requirement depend on the liability duration. From this figure it follows that the liquidity requirement is a convex decreasing function of D_V , whereas the capital requirement is a linearly increasing function of D_V . The total requirement TR is therefore first increasing and then decreasing depending on a pension fund's liability duration. A low total requirement implies that a pension fund has more opportunities to invest in illiquid assets. Thus, our model predicts that the impact of D_V on the illiquid assets allocation follows the inverse shape of the total requirement in Figure 4.1. In other words, D_V creates a hump-shaped effect on the illiquid assets allocation. We will see in Section 4.4

⁸This is not exactly true in practice, because the total capital requirement takes into account a diversification effect between interest rate risk and other risk factors. This diversification effect however only has a marginal second order effect on the numerical solution of the model.



Figure 4.1. Liquidity requirement, capital requirement, and total requirement as a function of liability duration

We use the following parameter values dr = 0.5%, $\phi^R = 20\%$, $\phi^B = 20\%$, dFX = 25%, $\phi^{FX} = 50\%$, and $w^{FX} = 50\%$.

that our empirical analysis confirms this.

The economic interpretation of Figure 4.1 is as follows. If we compare pension fund A with a liability duration of 10 years, with pension fund B with a liability duration of 15 years, we see that pension fund B has a lower total requirement. The lower liquidity requirement of pension fund B (Δ_{LR}) is not fully compensated by a higher capital requirement (Δ_{CR}) compared to A. The higher liability duration of B thus creates more opportunities to invest in illiquid assets compared to pension fund B, we see the reverse effect. The total requirement of pension fund C is higher in comparison to pension fund B. Although pension fund C has a lower liquidity requirement, this is more than fully compensated by a higher capital requirement relative to pension fund B. In Figure 4.1 the reflection point is at $D_V = 15.8$. This reflection point will be different for other parameter configurations.

4.2.4 Summary of the hypotheses

In this section we summarize the hypotheses that follow from our theoretical framework. We expect a hump-shaped impact of liability duration on the illiquid assets allocation and we assess this by adding the square of the liability duration to the regression model. Based on the total requirement in (4.12) we expect that the swap hedge ratio and the foreign exchange hedge ratio have no effect on the fraction of risky assets invested in illiquid assets. Both the swap and foreign exchange hedge ratios do not appear in the total requirement because the liquidity and capital requirement cancel out against each other. Following (4.12) we do however expect that the bond hedge ratio will positively impact the illiquid assets allocation. If a pension fund hedges more interest rate risk with bonds it will unlock some of its capital. This can be used to invest a larger fraction of risky assets in illiquid assets. Further, we expect that pension funds that invest more in non-euro dominated assets will allocate less to illiquid assets. A higher exposure to currency risk will seize a larger part of the capital that therefore is less available to invest in illiquid assets.

In the empirical section we will focus on three regression specifications. The main regression uses the fraction of risky assets invested in illiquid assets as dependent variable. We explain in detail why we use the fraction of risky assets allocated to illiquid assets as dependent variable in Section 4.4. In addition to that, we will also assess the impact of liability duration and hedging on the allocation to illiquid assets and on the allocation to risky assets.

4.3 Data

We use quarterly data on N=219 Dutch pension funds including their asset allocations, interest rate derivatives, currency derivatives, and other characteristics such as size and pension fund type. The data are free from reporting biases because pension funds report mandatory, following strict reporting requirements, to De Nederlandsche Bank the prudential supervisor in the Netherlands. The sample runs from the beginning of 2012 to the end of 2017, or 24 quarters. Following a change in the reporting requirements in 2015, we carefully merge the data before and after 2015 to ensure consistency.⁹ We only consider defined benefit pension funds because those are subject to capital requirements. Because Dutch pension funds cannot go bankrupt, there is no survivorship bias.¹⁰ We do however exclude pension funds that are "liquidated" during the sample period. A liquidation means that the

 $^{^{9}}$ From 2015 onwards the reporting requirements are more granular in order to improve the knowledge on pension fund investment behavior.

 $^{^{10}\}mathrm{As}$ a measure of last resort to prevent bank ruptcy pension funds in the Netherlands can reduce accrued pension benefits.

pension fund will gradually transfer assets and liabilities to either another pension funds or to an insurance company for cost efficiency reasons. Because such a gradual transfer of assets may result in non-representative asset allocations we exclude these pension funds. Table 4.1 shows that in terms of total AUM we only exclude a minor fraction of pension funds this way, about 2 percent.

Table 4.1. Total assets under management

This table shows the total assets under management in billions of all pension funds (left column) and of the pension funds selected for the analysis (right column) at the end of each year. The total number of pension funds is denoted by N. In the analysis, we leave out pension funds that are liquidated during the sample period.

year	AUM all	AUM selected
2012	904.15	894.33
2013	945.98	936.50
2014	1,131.74	1,124.42
2015	1,146.66	$1,\!118.72$
2016	1,262.54	1,233.41
2017	$1,\!326.07$	$1,\!297.67$
Ν	330	219

A - Assets

Our data distinguishes between the following 12 assets classes: government bonds, stocks in mature markets, credits, stocks in emerging markets, inflation index-linked bonds, cash and short-term receivables, listed indirect real estate, commodities, non-listed real estate, mortgages, private equity, and hedge funds.¹¹ One minus the allocation to government bonds, inflation-index bonds, cash and short-term receivables is defined as the allocation to risky assets (w^{RISKY}) .¹²

To distinguish between illiquid and liquid asset classes we need to define the concept of

¹¹Credits contain all credit related products, e.g. corporate bonds, bank loans, and syndicated loans.

 $^{^{12}}$ Privately issued inflation-index bonds constitute only a small portion of the market. As a result, we assume that all inflation-index bonds are issued by governments.
liquidity. An asset is considered less liquid if the investor cannot quickly sell a significant quantity of the asset at a price near its fundamental value. Asset classes such as private equity require buyers to have significant capital and particular knowledge about the asset class, which are both often limited in supply. Therefore, transactions costs of illiquid assets can become substantial. For some illiquid assets, legal impediments make it impossible to trade for a particular time period at all, such as lock-up periods that some hedge funds and private equity funds impose. Certainly, each asset or asset class has some time-varying degree of liquidity. Trading volume in a corporate bond may, e.g., drop to nil if the company runs into a bankruptcy. During the Great Financial Crisis trading in mortgage backed securities came to a stop. As a result, no clear distinction can be made between liquid and illiquid assets. However, some asset classes are substantially more illiquid than others in terms of the three dimensions (time, quantity and price) mentioned above. As pension funds are long term investors, we use immediacy as the key criterion to distinguish between liquid and illiquid asset classes. We classify the sum of non-listed real estate, private equity, mortgages, and hedge funds allocations as the total allocation to illiquid assets (w^{ILLIQ}) . Private equity includes both listed and non-listed private equity, infrastructure investments and micro finance investments. The allocation to private equity contains only the commitments already made, so future commitments are not included. Mortgages contains mortgage-backed securities and direct mortgage lending.

The sum of allocations to stocks in mature markets, credits, stocks in emerging markets, listed indirect real estate and commodities is defined as total allocation to liquid risky assets (w^{LIQ}) . We define the allocation to risky assets equal to the sum of the allocation to risky illiquid assets and the allocation to risky liquid assets

$$w^{RISKY} = w^{ILLIQ} + w^{LIQ}$$

Pension funds report both strategic and actual asset allocations. We focus on the strategic asset allocations because those better reflect the decisions made by pension funds. The actual asset allocation, by contrast, is less useful for our research question because it is influenced by market fluctuations and imperfect portfolio rebalancing (Bikker et al., 2010).

Table 4.2 presents the summary statistics. Panel A highlights the strategic asset allocations. The averages are equally weighted over pension funds and time. Government bonds, stocks mature markets and credits are the most important asset classes, with average allocations of 33, 29 and 18 percent respectively. Concerning the illiquid asset classes, non-listed real estate is the largest asset class with an average allocation of 4 percent. The 90th percentile

Table 4.2. **Descriptive statistics**

Panel A provides summary statistics of pension funds' strategic asset allocation. The means and standard deviations of the actual asset allocation are shown between brackets. Panel B provides the summary statistics of the variables specified in Section 4.3: allocation to illiquid assets (w^{ILLIQ}), allocation to risky assets (w^{RISKY}), fraction of risky assets allocated to illiquid assets (w^{ILLIQ}/w^{RISKY}), liability duration (D_V), collateral requirements on interest rate derivatives (CRr), collateral requirements on currency derivatives (CRfx), bond hedge ratio (ϕ^B), fraction of investments outside euro area (w^{FX}), log of total AUM (Size), required funding ratio (Rfr), and one period lag of the actual funding ratio (Fr). The summary statistics are computed as the equally weighted average over all pension funds and all quarters in the 2012-2017 period.

Panel A: asset allocations	mean	std. dev.	p10	p50	p90	obs.
Liquid assets						
Government bonds	0.33 [0.31]	0.20 [0.21]	0.05	0.33	0.60	4,956
Stocks mature markets	0.29 [0.30]	0.13 [0.13]	0.16	0.27	0.43	4,956
Credits	0.18[0.19]	0.12[0.11]	0.00	0.17	0.34	4,956
Stocks emerging markets	0.05[0.05]	0.04[0.04]	0.00	0.05	0.10	4,956
Inflation index-linked bonds	0.02[0.02]	0.05[0.05]	0.00	0.00	0.08	4,956
Listed real estate	0.02[0.03]	0.03[0.04]	0.00	0.00	0.05	4,956
Commodities	0.01 [0.01]	0.02 [0.02]	0.00	0.00	0.05	4,956
Cash and short-term receivables	0.01 [0.03]	0.07 $[0.08]$	0.00	0.00	0.03	4,956
Illiquid assets						
Non-listed real estate	$0.04 \ [0.04]$	$0.05 \ [0.05]$	0.00	0.03	0.12	4,956
Mortgages	0.02 [0.02]	0.04 [0.04]	0.00	0.00	0.07	4,956
Private equity	0.01 [0.01]	0.02 [0.02]	0.00	0.00	0.05	4,956
Hedge funds	0.01 [0.01]	$0.02 \ [0.02]$	0.00	0.00	0.04	$4,\!956$
Panel B: variables	mean	std. dev.	p10	p50	p90	obs.
Allocation to illiquid assets	0.08	0.08	0.00	0.06	0.20	4.956
Allocation to risky assets	0.64	0.18	0.40	0.65	0.90	4.956
Fraction illiquid in risky assets	0.13	0.12	0.00	0.11	0.30	4.904
Liability duration	18.90	3.98	14.60	18.60	23.90	4,978
CR on interest rate derivatives	0.05	0.04	0.00	0.04	0.10	4,973
CR on currency derivatives	0.05	0.04	0.00	0.05	0.10	4,991
Bond hedge ratio	0.25	0.14	0.09	0.22	0.45	4,940
Foreign investments	0.22	0.22	0.00	0.20	0.51	3,682
Log of total AUM	5.84	0.81	5.00	5.79	6.86	4,997
Required funding ratio	1.16	0.07	1.10	1.16	1.23	4,992
Funding ratio (t-1)	1.09	0.13	0.96	1.07	1.22	4,992

shows that 10 percent of the pension funds invest more than 12 percent in this asset class. Pension funds on average invest 2 percent in mortgages. The other two illiquid asset classes have a strategic allocation of around 1 percent on average. Panel A also shows the mean and standard deviation of the actual asset allocations between brackets. The differences between the strategic and actual asset allocations are small suggesting that pension funds on average rebalance their portfolios accurately.

Turning to Panel B, we see that the average allocation to the illiquid assets equals 8 percent. Approximately one-third of Dutch pension funds do not invest in illiquid assets at all. The 90th percentile however shows that 10 percent of the pension funds allocate over 20 percent of their total AUM to illiquid assets.

Table 4.3 shows that the strategic illiquid assets allocation is not fixed and varies over the sample period. This holds for both the aggregate illiquid assets allocation and the illiquid asset classes separately. Pension funds generally review their strategic asset allocation every 3 years. This implies that pension funds on average reviewed there illiquid assets allocation twice during the sample period.

Table 4.3. Time variation in the strategic illiquid assets allocation

This table shows the cross-sectional average time variation of pension fund's illiquid assets allocation over the period 2012–2017. The average time variation is computed as the cross-sectional average of the standard deviation of pension funds' strategic illiquid assets allocations over time.

Illiquid asset class	time variation
Total illiquid assets	0.0251
Non-listed real estate	0.0146
Mortgages	0.0153
Private equity	0.0040
Hedge funds	0.0048

B - Liability duration

Pension funds report the modified duration of their liabilities (D_V) . The liability duration is summarized in Panel B of Table 4.2. The average liability duration equals 18.9 years. However, 10 percent of the pension funds have a liability duration below 14.6 years and 10 percent have a liability duration in excess of 23.9 years. This shows there is quite some variation in the average investment horizon of pension funds.

C - Collateral requirements

Unfortunately, pension funds do not report on their swap hedge ratio or their foreign exchange hedge ratio. However, the data do allow us to calculate the collateral requirements in case of an adverse event in the underlying risk factor, which we argue are good proxies for the swap hedge ratio and the foreign exchange hedge ratio.¹³ We will, therefore, explain in detail how we measure the collateral requirements.

Pension funds report the market value of the total interest rate derivatives and currency derivatives that they have entered into. In addition, they also report the expected values of these positions after some predefined shocks in the underlying risk factors. For interest rate risk these shocks are an increase (decrease) in the term structure of market interest rates of 1 percentage points and for foreign exchange risk an appreciation (depreciation) of foreign currencies with respect to the euro of 25 percent.

The collateral requirement on interest rate derivatives (CRr) is the absolute difference between the market value of the portfolio of derivatives after a predetermined shock, MVr_s , minus its current market value, MVr_c . We divide the absolute value of this difference by the total assets under management (AUM)

$$CRr = \frac{|MVr_s - MVr_c|}{AUM}$$

We define the collateral requirements on currency derivatives (CRfx) as the absolute difference between the market value of the portfolio of derivatives after a predetermined shock, $MVfx_s$, minus its current market value, $MVfx_c$. We express the absolute value of the change relative to the total AUM

$$CRfx = \frac{|MVfx_s - MVfx_c|}{AUM}$$

We determine the collateral requirements by using an increase in the interest rate of 1 percentage points and an increase of the foreign currency relative to the euro of 25 percent.

¹³As of 2015, pension funds do report the swap hedge ratio. The correlation between the reported swap hedge ratio and our computation of the collateral requirement on interest rate derivatives equals 0.75. This shows that our approximation of the swap hedge ratio is not far off.

The collateral requirements on interest rate derivatives and on currency derivatives are summarized in Panel B of Table 4.2. The averages of both are of the same order of magnitude and equal 5 percent of AUM. The 10th percentiles are in both cases equal to zero, revealing that some pension funds do not hedge these risks with derivatives at all.

D - Bond hedge ratio and foreign investments

Our theoretical model in (4.12) shows that the bond hedge ratio (ϕ^B) and the fraction invested outside the euro area (w^{FX}) affect the illiquid assets allocation. Table 4.2 shows that pension fund on average hedge 25 percent of their liability interest rate risk with bonds. About 10 percent of the pension funds in the sample have a bond hedge ratio in excess of 45 percent. The fraction investments outside the euro area equals 22 percent on average and 51 percent in the 90th percentile. In our theoretical model we express all results relative to the liabilities. Here we take both the bond hedge ratio and the foreign investments as a fraction of total AUM to make the quantities easier to interpret. Expressing the quantities in either AUM or liabilities does however not materially affect our empirical analysis.

E - Control variables

As control variables we include the log of total assets under management (Size), pension fund type (Type), the required funding ratio (Rfr) and the actual funding ratio at the end of the previous period (Fr). We distinguish between three types of pension funds. There are 55 industry-wide pension funds, 10 professional group pension funds and 154 corporate pension funds in the sample. Industry-wide pension funds are generally mandatory pension funds and organize pensions for a specific industry or sector, e.g., civil servants or hospital staff. Professional group pension funds provide pensions for a single profession such as hairdressers or pharmacists. Corporate pension funds arrange pensions for a particular company.

The level of the required funding ratio to comply with regulations depends on the specific risk profile of a pension fund.¹⁴ Including the required funding ratio in the regression model, therefore, controls for differences in risk appetite. For instance, a "young" pension fund (with a high liability duration) could substantially differ in its risk appetite compared to an "old" pension fund (with a low liability duration), potentially driving the impact of liability duration on the illiquid assets allocation. Furthermore, pension funds that hedge only a

¹⁴The risk profile is measured by the "mismatch" between assets and liabilities.

small part of their interest rate risk or currency risk, could be the ones that take more risk in general, potentially driving the effect of hedging on the illiquid assets allocation.

The actual, previous period, funding ratio is also included in the analysis. From the literature on risk-taking behavior of pension funds, the actual funding ratio could either have a positive or a negative effect on the illiquid assets allocation. Basak and Shapiro (2001) show that, compared to normal circumstances, investors take more risk in worst case scenarios when they are subjected to a VaR requirement. On the other hand, Rauh (2009), using US corporate defined benefit pension funds data, finds that poorly funded pension plans allocate a larger share of their assets to government bonds.

Size, required funding ratio and actual funding ratio are summarized in Panel B, Table 4.2. The 10th and 90th percentile of the log of total AUM reveals that the size distribution is right skewed: the largest pension funds in our sample are considerably larger compared to the mean pension fund. The average required funding ratio equals 116 percent and varies from 110 to 123 percent. The actual funding ratio equals 109 percent on average, which is substantially below the required funding ratio of pension funds.

Table 4.4 splits the sample in pension funds that invest and that do not invest in illiquid assets. The latter group takes less risk in general, having on average a 10 percentage points lower risky assets allocation. The table also reveals that the average size of pension funds that do not invest in illiquid assets is considerably smaller. On other dimensions, however, the subsamples are comparable.

4.4 Allocation to illiquid assets: empirical results

To measure the impact of liquidity and capital requirements on the illiquid assets allocation, we estimate a static random effects Tobit model.¹⁵ The Tobit model controls for leftcensoring of the allocation to illiquid assets at zero. This is necessary as a substantial number of pension funds does not invest in illiquid assets. Asset allocation decisions are typically first made to very broad asset classes such as safe assets (bonds) and risky assets (stocks). In a second step, asset allocations are determined within each broad asset class (Binsbergen et al., 2008). Following this top-down decision-making process, we first analyze the aggregate illiquid assets allocation. Because illiquid asset classes differ in their degree of

 $^{^{15}\}mathrm{A}$ random effects structure corrects for time-invariant pension funds' characteristics that we do not observe but potentially play a role in investment decisions.

Table 4.4. Descriptive statistics - subsamples

This table specifies the summary statistics of the variables specified in Section 4.3 for pension funds that do invest in illiquid assets (Panel A) and pension funds that do not invest in illiquid assets (Panel B): allocation to illiquid assets (w^{ILLIQ}), allocation to risky assets (w^{RISKY}), fraction of risky assets allocated to illiquid assets (w^{ILLIQ}/w^{RISKY}), liability duration (D_V), collateral requirements on interest rate derivatives (CRr), collateral requirements on currency derivatives (CRfx), bond hedge ratio (ϕ^B), fraction of investments outside euro area (w^{FX}), log of total AUM (Size), required funding ratio (Rfr), and the one period lag of the actual funding ratio (Fr). The summary statistics are computed as the equally weighted average over all pension funds and all quarters in the 2012-2017 period.

	mean	std. dev.	p10	p50	p90
Allocation to illiquid agents	0.19	0.08	0.02	0.11	0.99
Anocation to iniquid assets	0.12	0.08	0.05	0.11	0.22
Allocation to risky assets	0.67	0.16	0.47	0.67	0.90
Fraction illiquid in risky assets	0.18	0.10	0.05	0.16	0.32
Liability duration	18.61	3.50	15.00	18.00	23.00
CR on interest rate derivatives	0.05	0.04	0.00	0.04	0.09
CR on currency derivatives	0.06	0.04	0.00	0.05	0.10
Bond hedge ratio	0.24	0.12	0.10	0.22	0.40
Foreign investments	0.23	0.22	0.00	0.23	0.53
Log of total AUM	6.05	0.77	5.19	5.98	7.00
Required funding ratio	1.17	0.07	1.11	1.17	1.24
Funding ratio (t-1)	1.09	0.12	0.96	1.08	1.22

Panel A: pension funds that invest in illiquid assets

Panel B: pension funds that do not invest in illiquid assets

	mean	std. dev.	p10	p50	p90
Allocation to illiquid agents	0.00	0.00	0.00	0.00	0.00
Allocation to risky assets	$0.00 \\ 0.57$	0.00 0.22	0.00	0.00 0.56	0.00
Fraction illiquid in risky assets	0.00	0.22	0.00	0.00	0.00
Liability duration	20.00	4.88	14.00	20.00	26.00
CR on interest rate derivatives	0.05	0.05	0.00	0.05	0.12
CR on currency derivatives	0.04	0.04	0.00	0.03	0.09
Bond hedge ratio	0.28	0.18	0.08	0.23	0.58
Foreign investments	0.18	0.19	0.00	0.12	0.44
Log of total AUM	5.28	0.62	4.65	5.30	6.03
Required funding ratio	1.15	0.06	1.08	1.14	1.22
Funding ratio (t-1)	1.10	0.16	0.95	1.07	1.24

immediacy, we look at individual illiquid asset classes separately in the next section.

We will however not assess the illiquid assets allocation directly. Instead we will take the fraction of risky assets invested in illiquid assets as our key dependent variable. The main reason to take this fraction is to circumvent mechanical effects. The following two examples show these mechanical effects. First, suppose that a pension fund decides to invest more in bonds, this mechanically results in a lower allocation to risky assets and therefore, potentially, also a lower allocation to illiquid assets. Second, Dutch pension funds invest the majority of their bond portfolio in the euro area, whereas the risky assets are generally invested more globally. Currency hedging is thus directly related to the risky assets allocation. A positive effect of currency hedging on the illiquid assets allocation could therefore be mechanical.

In alternative model specifications we nonetheless also use the allocation to illiquid assets (w^{ILLIQ}) and to risky assets (w^{RISKY}) as dependent variables. In the latter case we estimate a standard random effects model as there is no censoring around zero.

The general model specification is

$$\frac{w_{it}^{ILLIQ}}{w_{it}^{RISKY}} = \beta_0 + \beta_1 D_{V,it} + \beta_2 D_{V,it}^2 + \beta_3 CRr_{it} + \beta_4 CRfx_{it} + \beta_5 \phi_{it}^B + \beta_6 w_{it}^{FX}$$
(4.15)
+ $\beta_7 Size_{it} + \beta_8 Type_i + \beta_9 Rfr_{it} + \beta_{10} Fr_{it-1} + \lambda_t + \epsilon_{it}$

where w_i indicates the allocation of pension fund i = 1, ..., N at the end of the quarter t = 2012Q1, ..., 2017Q4. The main explanatory variables are the liability duration (D_V) , the square of the liability duration (D_V^2) , the collateral requirements on interest rate derivatives (CRr), the collateral requirements on currency derivatives (CRfx), the bond hedge ratio ϕ^B , and the fraction of total foreign investments w^{FX} . To link the main regression specification directly to our theoretical framework, we recall from the previous section that CRr proxies for the swap hedge ratio ϕ^R and CRfx proxies for the foreign exchange hedge ratio ϕ^{FX} . The control variables are the log of total AUM (Size), the pension fund type (Type), the required funding ratio (Rfr) and the lag of the actual funding ratio (Fr). We include dummies for the pension fund type, where the reference group consists of industry-wide pension funds. A professional group pension fund is denoted by dummy *Prof* and a corporate pension fund by dummy *Corp*. The fall in market interest rates over our sample period may have caused pension funds to increase their allocation towards illiquid assets in a search for yield (Boubaker et al., 2017). Therefore, we control for time-fixed effects (λ) in all our model specifications.

Another potential concern is that pension funds with a high liability duration are more

likely to hedge interest rate risk using swaps. The reason for this is that there are not sufficient long-term government bonds available to match with the duration of long-term pension liabilities. This issue is more severe for pension funds with a high liability duration. Therefore, liability duration D_V and collateral requirement for interest rate risk CRr may be correlated. Table 4.5 shows that there is indeed a positive correlation between D_V and CRr, however, given a value of 0.40 collinearity should not be an issue. This implies that the results for D_V and CRr can be interpreted separately.¹⁶

Table 4.5. Correlation table of key variables

This table provides the correlation matrix of the key variables: liability duration (D_V) , collateral requirements on interest rate derivatives (CRr), collateral requirements on currency derivatives (CRfx), bond hedge ratio (ϕ^B) , fraction of investments outside euro area (w^{FX}) , log of total AUM (Size), required funding ratio (Rfr), and the one period lag of the actual funding ratio (Fr).

Correla	tion mat	trix						
	D_V	CRr	CRfx	ϕ^B	w^{FX}	Size	Rfr	Fr
D_V	1							
CRr	0.40	1						
CRfx	0.02	0.12	1					
ϕ^B	-0.29	-0.40	-0.24	1				
w^{FX}	0.20	0.01	0.27	-0.17	1			
Size	0.04	0.06	0.21	-0.15	0.28	1		
Rfr	0.12	-0.17	0.13	-0.24	0.29	0.16	1	
Fr	-0.33	-0.19	-0.08	0.06	-0.07	-0.15	-0.02	1

¹⁶We also run our models to test for a possible interaction between liability duration and collateral requirements. The interaction term between both variables is, however, statistically indifferent from zero in all of the model specifications. These results are available upon request. Further, the liability duration (D_v) is negatively correlated with bond hedge ratio (ϕ^B) as well as the lagged funding ratio (Fr). The variance inflation factor for liability duration equals $\frac{1}{1-R^2} = 1.15$, which is far below the threshold for multicollinearity issues (10), and hence multicollinearity is not an issue here.

A - Liability duration

Table 4.6 shows the results for the main model specification in (4.15). The first column shows that liability duration has a positive effect on the fraction of risky assets allocated to illiquid assets, whereas the square of the liability duration has a negative impact. Both coefficients are statistically significant at the 1 percent significance level. This means that there is indeed a hump-shaped relationship. Up to a liability duration of 18 years $\left(\frac{\partial w^{ILLIQ}/w^{RISKY}}{\partial D_V}\right) = \beta_1 + 2\beta_2 D_V = 0 \rightarrow D_V = -\frac{\beta_1}{2\beta_2}\right)$, the effect of liability duration is positive. After this point however, the effect reverses. The fraction of illiquid assets for a pension fund with a liability duration of 11 years is 1.07 percentage points higher compared to a pension fund with a liability duration of 10 years. A pension fund with a liability duration of 26 years allocates 1.02 percentage points less to illiquid assets compared to a pension funds with a 25 year duration.¹⁷ Figure 4.2 shows the estimated hump-shaped impact of liability duration on the fraction of the risky assets allocated to illiquid assets.

The results are in line with our theoretical predictions. Up to a liability duration of 18 years, the liquidity requirement decreases faster than the capital requirement increases. A pension fund with a high liability duration has less short-term liabilities relative to a pension fund with a low liability duration. A high liability duration implies that it is less of a constraint for a pension fund not being able to trade frequently in illiquid assets. Beyond a liability duration of 18 years however, the capital requirement increases faster than the liquidity requirement decreases. This lowers the opportunity to invest in illiquid assets.

As an alternative specification we use the allocation to illiquid risky assets w^{ILLIQ} (Table 4.6, Column (2)). Again, a hump-shaped impact of liability duration on the illiquid assets allocation appears. We also assess the allocation to risky assets w^{RISKY} . Table 4.6, Column (3) shows a negative impact of liability duration on the risky assets allocation, while the of duration squared is positive. Figure 4.3 plots the combined impact on the allocation to risky assets. The increase in the risky asset allocation on the right-hand side of this figure coincides with the findings in Andonov et al. (2017), who find that the investment policy of public and private European and Canadian pension funds and private US pension funds is more aggressive for less mature pension funds. A study by Alestalo and Puttonen (2006)

¹⁷Notice that the coefficients are estimates based on the uncensored latent variable, not the observed outcome. The coefficients are the right interpretation for all observations on the dependent variable, w^{ILLIQ}/w^{RISKY} , above zero. In order to get the effect on the actual observed dependent variable, the coefficient estimates have to be multiplied by the probability of the dependent variable being above zero.

Table 4.6. Main results - total illiquid assets allocation

In this table, we show the random effects estimators based on the regression (4.15). The dependent variable in the first column is the strategic fraction of risky assets allocated to illiquid assets (w^{ILLIQ}/w^{RISKY}) , the dependent variable in the second column is the strategic allocation to illiquid assets (w^{ILLIQ}) and the third column uses the strategic risky assets allocation as dependent variable (w^{RISKY}) . The fourth column uses actual rather than strategic allocations. The independent variables include the liability duration (D_V) , the liability duration squared (D_V^2) , the collateral requirements on interest rate derivatives (CRr), the collateral requirements on currency derivatives (CRfx), bond hedge ratio (ϕ^B) , fraction of foreign investments (w^{FX}) , the log of total AUM (Size), Corp indicates a corporate pension fund, Prof indicates a professional pension fund, Rfr is the required funding ratio, and Fr the one period lag of the actual funding ratio. Standard errors are between parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	(1)	(2)	(3)	(4)
Dependent variable	w^{ILLIQ}/w^{RISKY}	w^{ILLIQ}	w^{RISKY}	w^{ILLIQ}/w^{RISKY}
D_V	.0254***	.0126***	0404***	.0120**
v	(0056)	(.0039)	(.0062)	(.0050)
D_V^2	0007***	0004***	.0009***	0004***
v	(.0001)	(.0001)	(.0002)	(.0001)
CRr	0124	0141	0208	.0355
	(.0576)	(.0405)	(.0713)	(.0517)
CRfx	.1784***	.1371***	.4104***	.0151
	(.0363)	(.0255)	(.0481)	(.0325)
ϕ^B	.0204	0761^{***}	4038^{***}	.0127
	(.0181)	(.0127)	(.0220)	(.0162)
w^{FX}	0125	0013	1342^{***}	$.0424^{***}$
	(.0125)	(.0088)	(.0152)	(.0108)
Size	.0736***	$.0566^{***}$	$.0604^{***}$	$.0625^{***}$
	(.0102)	(.0075)	(.0098)	(.0093)
Corp	0757^{***}	0454^{***}	.0218	0535^{***}
	(.0217)	(.0143)	(.0194)	(.0173)
Prof	.0184	.0068	0234	.0053
	(.0438)	(.0288)	(.0380)	(.0341)
Rfr	0006	.0045	.0192	$.0608^{***}$
	(.0190)	(.0133)	(.0277)	(.0180)
Fr	0686^{**}	0562^{***}	0723^{**}	1418^{**}
	(.0275)	(.0192)	(.0304)	(.0245)
Time FE	Υ	Υ	Υ	Υ
Ν	$3,\!401$	$3,\!401$	$3,\!433$	3,400
Left-censored	877	890	n/a	606
Uncensored	2,524	2,511	n/a	2,794

173



Figure 4.2. The effect of the liability duration on the fraction of risky assets allocated to illiquid assets

The calculations are based on assuming an industry-wide pension fund that has average foreign exchange risk hedging activities, average size, and average lag funding ratio (other variables are set equal to zero as they are not statistically significant). The dashed line represents the 95% confidence interval.

reports similar findings for Finnish pension funds. However, Figure 4.3 also highlights on the left-hand side that mature pension funds invest more in risky assets compared to pension funds with an average liability duration. A possible explanation is that mature pension funds can more easily hedge their liabilities with bonds instead of swaps and can consequently use their capital for risk-taking.

In Column (4) of Table 4.6 we use actual rather than strategic asset allocations. We observe the following. First, the liability duration has a again hump-shaped impact on the fraction of risky assets allocated to illiquid assets. The reflection point is now at a liability duration of 15.0 years. Second, hedging of interest rate risk and currency risk does not affect the fraction of risky assets allocated to illiquid assets. Third, foreign investments has a significant and positive impact. Although these results seem to better support or theoretical predictions, using actual instead of strategic asset allocations might cause effects that are (partly) driven by the performance of the different asset classes or imperfect rebalancing by pension funds.



Figure 4.3. The effect of the liability duration on the risky assets allocation The calculations are based on assuming an industry-wide pension fund that has average foreign exchange risk hedging activities, average bond hedging activities, average foreign investments, average size, and average lag funding ratio (other variables are set equal to zero as they are not statistically significant). The dashed line represents the 95% confidence interval.

B - Collateral requirements

Table 4.6 shows that the collateral requirements on interest rate swaps (CRr) do not have an effect on the allocation to illiquid assets. This is in line with the theoretical predictions of our model: the liquidity and capital requirement cancel out against each other. However, we do find a significant and positive impact of collateral requirements on currency forwards. At first glance, this result seems to conflict the predictions of our model. However, in practice collateral requirements for currency forwards are less strict than for interest rate swaps and in some cases even non-existent. In our theoretical framework, this implies the that the increase in liquidity requirements for currency risk hedging is smaller than the decrease in the capital requirement. This creates opportunities to invest in illiquid assets. A one standard deviation increase in the collateral requirements (an increase of $0.04 \ CRfx$) implies an increase in the fraction of risky assets allocated to illiquid assets of approximately 0.71 percentage points. The average fraction to illiquid assets equals 18 percent. This implies a relative increase in the fraction of risky assets allocated to illiquid assets of approximately 4.0 percent.

CHAPTER 4. ILLIQUID ASSETS: LIQUIDITY AND CAPITAL REQUIREMENTS

The impact of collateral requirements on the allocation to risky assets is also shown in Table 4.6, Column (3). The findings are comparable to the effects of the fraction of risky assets invested in illiquid assets. Notice that we control for the total fraction invested in foreign currencies such that the results we obtain are not mechanical. A one standard deviation increase in the collateral requirement (an increase of $0.04 \ CRfx$) implies an increase in the risky assets allocation of approximately 1.64 percentage points. This implies a relative increase in the risky asset allocation of 2.6 percent. Taken together, a larger foreign exchange hedge ratio increases the allocation to risky assets which in turn increases the fraction of risky assets that is allocated to illiquid assets.

C - Bond hedge ratio and foreign investments

We do not find a statistically significant effect of the bond hedge ratio on the fraction allocated to illiquid assets, although the sign of the coefficient is in line with our theoretical framework. The bond hedge ratio negatively affects the allocation to illiquid assets and to risky assets. The latter however is a mechanical effect because hedging interest rate risk with bonds by construction lowers the allocation to risky assets.

We also do not find an effect of foreign investments on the fraction invested in illiquid assets, although the sign of the coefficient is again in line with our model. The same is true for the impact on the illiquid assets allocation. The fraction invested in foreign assets has a significant negative impact on the risky assets allocation. Investing more in foreign currencies, while keeping hedging fixed, increases the capital requirement and therefore limits opportunities to invest in risky assets. A 10 percentage points increase in the fraction invested in foreign assets decreases the risky assets allocation with 1.35 percentage points.

D - Control variables

We now turn to the control variables. Table 4.6 shows that size has a positive and significant impact on the allocation to illiquid assets. A pension fund that is ten times larger in terms of total assets under management allocates a 7.4 percentage points higher fraction of risky assets allocated to illiquid assets. Illiquid assets are generally complex investment products and therefore the pension fund needs to have sufficient knowledge to manage these products. The larger a pension fund, the better it can afford to pay the costs involved in managing complex investment products. Our finding is consistent with Andonov (2014) and Dyck and Pomorski (2016), who show that the increase in the allocation to illiquid assets is

more pronounced for large institutional investors. Moreover, Stoughton and Zechner (2011) argue that economies of scale in alternative assets exist because only large investors can afford to pay high fixed search costs to identify profitable projects or select skilled external managers. On top of that, large institutional investors have more power to negotiate better fees (Broeders et al., 2016). Table 4.6 also shows that large pension funds invest more in risky assets in general, although the economic impact is somewhat smaller. A pension fund that is ten times larger invests 6.0 percentage points more in risky assets.

We also find that corporate pension funds allocate significantly less to illiquid assets as a fraction of risky assets compared to industry-wide pension funds. We believe that this difference is due to the fact that a corporation is required to report on its pension fund in the annual accounts and is therefore less willing to take risks. Indeed, the riskiness of a corporate pension plan impacts the risk profile of the corporation (Jin et al., 2006). An additional explanation is that corporate pension funds are more exposed to sponsor default risk compared to compulsory and professional group pension funds. Therefore, they are less willing to take risk (Broeders, 2010).

Finally the required funding ratio does not affect asset allocations, while the lag of the actual funding ratio affects all allocations in Table 4.6 negatively. This implies that pension funds with a lower last period funding ratio invest a larger fraction of their risky assets in illiquid assets. This result supports the theoretical finding by Basak and Shapiro (2001) that pension funds for which a VaR requirement is binding take additional investment risks. This finding is also consistent with the empirical study of Peng and Wang (2019), who study alternative investments decisions of US public pension plans, and find that pension plans with lower funding ratios allocate more to alternative assets. A one standard deviation decrease in the previous period funding ratio increases the fraction of risky assets allocated to illiquid assets by 0.89 percentage points. Pension funds also increase the total risky assets allocation when previous period funding ratio decreased. In other words, a lower funding ratio implies a higher total risky assets allocation and a larger fraction of the risky assets allocation is invested in illiquid assets.

4.4.1 Empirical results separate illiquid asset classes

So far, we treated illiquid asset classes as a homogeneous group. However, the degree of immediacy differs across separate illiquid asset classes. For instance, the typical time between transactions for residential housing is 4-5 years, although it can vary from months to decades (Hansen, 1998) and (Miller et al., 2011). The average time between transactions is relatively

high in case of private equity. Private equity investments generally run for 10 years and trading before a contract expires is unusual (Metrick and Yasuda, 2010).

To reflect these difference in immediacy, Table 4.7 presents the results for the following illiquid asset classes separately: real estate, mortgages, private equity and hedge funds. The hump-shaped effect of the liability duration on the fraction of risky assets allocated to illiquid assets is present in all four asset classes. The cut-off point where the marginal benefits (lower liquidity requirement) and the marginal costs (higher capital requirement) of a higher liability duration are equal, is close to the aggregate cut-off point of 18 for the overall fraction of risky assets allocated to illiquid assets.

The results of swap hedge ratios are however mixed across the different asset classes. Consistent with the aggregated results, the swap hedge ratio does not affect the allocation to real estate and mortgages. In case of private equity, the swap hedge ratio positively affects the real estate allocation. Notice that the economic magnitude is however small. A one standard deviation increase in collateral requirement leads to an increase of the fraction of risky assets allocated to private equity of 0.23 percentage points. On the other hand, the swap hedge ratio negatively affects the allocation to hedge funds. The results for hedge funds should however be interpreted with care for mainly two reasons. First, the number of uncensored observations is relatively small compared to the total number of observations. Second, events outside our model may have impacted hedge fund allocation. Some hedge funds received negative publicity during the financial crisis, because investors who tried to withdraw their money to meet liquidity needs where unable to do so. This may play a role in the allocation to hedge funds.

The results for foreign exchange hedge ratio are also mixed across the different asset classes. The foreign exchange hedge ratio does not affect the allocation to hedge funds and negatively affects the allocation to private equity. A one standard deviation increase in collateral requirement leads to an increase in the fraction of risky assets allocated to private equity of 0.80 percentage points.

The required funding ratio also has different effects. Pension funds with a higher required funding ratio invest more in private equity and less in mortgages. This reveals that pension funds with higher tolerance for risk might prefer certain illiquid assets over others.

These differences in findings indicate that strategic decisions made at the top level do not have to be implemented at each individual asset class level. What matters is that the overall investment policy, risk exposures, and liquidity profile are in accordance with the policy set by pension fund.

Table 4.7. Main results - separate illiquid assets classes

In this table, we show the random effects estimators based on the regression (4.15) for separate illiquid asset classes. The dependent variable in the first column is the fraction of risky assets allocated to real estate (w^{RE}/w^{RISKY}) , in the second column the fraction of risky assets allocated to mortgages (w^{MG}/w^{RISKY}) , in the third column the fraction of risky assets allocated to private equity (w^{PE}/w^{RISKY}) , and in the fourth column the fraction of risky assets allocated to hedge funds (w^{HF}/w^{RISKY}) . The independent variables include the liability duration (D_V) , the liability duration squared (D_V^2) , the collateral requirements on interest rate derivatives (CRr), the collateral requirements (w^{FX}) , the log of total AUM (Size), Corp indicates a corporate pension fund, Prof indicates a professional pension fund, Rfr is the required funding ratio, and Fr the one period lag of the actual funding ratio. Standard errors are between parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	(1)	(2)	(3)	(4)
Dependent variable	w^{RE}/w^{RISKY}	w^{MG}/w^{RISKY}	w^{PE}/w^{RISKY}	w^{HF}/w^{RISKY}
D_V	.0138***	.0178**	.0076**	.0753***
	(.0045)	(.0070)	(.0036)	(.0088)
D_V^2	0004^{***}	0004**	0002^{***}	0022
	(.0001)	(.0002)	(.0001)	(.0002)
CRr	.0409	0453	$.0585^{*}$	1474^{*}
	(.0427)	(.0947)	(.0325)	(.0835)
CRfx	$.1085^{***}$.2535***	2068^{***}	0216
	(.0316)	(.0420)	(.0435)	(.0653)
ϕ^B	.0187	.0151	0209^{**}	0855^{***}
	(.0143)	(.0281)	(.0102)	(.0243)
w^{FX}	$.0767^{***}$	0454^{***}	.0041	.0290*
	(.0101)	(.0172)	(.0069)	(.0164)
Size	$.0541^{***}$.0510***	.0608***	.0156
	(.0088)	(.0120)	(.0065)	(.0102)
Corp	0544^{***}	0846^{***}	0218^{**}	.0068
	(.0175)	(.0223)	(.0091)	(.0178)
Prof	0004	0620	$.0300^{*}$.0937***
	(.0377)	(.0444)	(.0161)	(.0330)
Rfr	0001	2514^{***}	.1340***	.0172
	(.0134)	(.0705)	(.0309)	(.0138)
Fr	1206^{***}	.1164***	$.0725^{***}$	1167
	(.0220)	(.0357)	(.0158)	(.0360)
Time FE	Y	Υ	Y	Y
Ν	$3,\!432$	3,432	3,432	$3,\!432$
Left-censored	$1,\!359$	2,186	2,200	2,851
Uncensored	2,073	$1,\!246$	1,232	581

4.5 Robustness check

From the previous sections, it follows that the impact of liability duration on the illiquid assets allocation is strong. As a robustness check we take an alternative measure of liability duration in this section. Liability duration measures the maturity of a pension fund and this is directly related to demographics. Although we have no data on demographics, we do know the pension payments and the value of the liabilities per year. The ratio of pension payments to the value of the liabilities is an alternative measure for maturity (De Haan, 2018). If this ratio is high, a large part of the participants consists of retirees.

In the robustness check we replace the liability duration by the ratio of pension payments to liabilities (*Benefits*).¹⁸ Table 4.8 shows that using this alternative measure also results in a hump-shaped impact on the fraction of risky assets allocated to illiquid assets. Up to a ratio of 4.8 percent, the fraction of risky assets allocated to illiquid assets is positively affected.¹⁹ After this point, the effect is reversed. Column (3) of Table 4.8 shows that the hump-shaped effect is absent for the allocation to risky assets.

In our theoretical framework we saw that the ratio of pension payments to the value of liabilities is given by $\frac{1}{D_V}$ (see Eq. (4.4) in Section 4.2). Figure 4.4 provides evidence that the fitted curve for the observed ratios of pension payments to pension liabilities is indeed a convex and decreasing function of liability duration, in line with the theoretical framework. This confirms that the inverse of the liability duration is a good proxy for the liquidity requirement from pension payments.

4.6 Illiquid assets allocation other regulatory frameworks

Compared to banks and insurance companies, pension fund regulation is much more diverse across countries. Here, we focus on the difference in capital requirements for pension funds in the US and Canada to show our model implications outside the Netherlands. The US and Canada have fixed capital requirements, whereas the Netherlands has risk-based capital requirements.

$${}^{19}\frac{\partial w^{\scriptscriptstyle ILLIQ}/w^{\scriptscriptstyle RISKY}}{\partial Benefits} = \beta_1 + 2\beta_2 Benefits = 0 \rightarrow Benefits = -\frac{\beta_1}{2\beta_2}$$

¹⁸We only include observations at the end of each year as the pension payments are only available on an annual basis. The sample period is 2012-2016, as the pension payments for the year 2017 are not available yet.

Table 4.8. Robustness - alternative measure of liability duration

In this table, we show the random effects estimators based on (4.15) using an alternative measure for the liability duration. The dependent variable in the first column is the fraction of risky assets allocated to illiquid assets (w^{ILLIQ}/w^{RISKY}) , the dependent variable in the second column is the allocation to illiquid assets (w^{ILLIQ}) and the third column uses the risky assets allocation as dependent variable (w^{RISKY}) . The independent variables include the ratio of pension payments to liabilities (*Benefits*), the ratio of pension payments to liabilities squared (*Benefits*²), the collateral requirements on interest rate derivatives (*CRr*), the collateral requirements on currency derivatives (*CRfx*), bond hedge ratio (ϕ^B), fraction of foreign investments (w^{FX}), the log of total AUM (*Size*), *Corp* indicates a corporate pension fund, *Prof* indicates a professional pension fund, *Rfr* is the required funding ratio, and *Fr* the one period lag of the actual funding ratio. Standard errors are between parentheses; *p < 0.10, **p < 0.05, ***p < 0.01.

	(1)	(2)	(3)
			DICKY
Dependent variable	w^{ILLIQ}/w^{RISKY}	w^{ILLIQ}	w ^{RISKY}
Benefits	6.9264^{***}	4.3555^{***}	2.8844^{*}
	(1.5747)	(1.0672)	(1.7548)
$Benefits^2$	-71.5539^{***}	-38.0856^{**}	12.4495
	(22.1661)	(15.0461)	(24.3063)
CRr	.0894	.0708	0512
	(.1408)	(.0964)	(.1830)
CRfx	.1016	.0875	.3857***
	(.0806)	(.0553)	(.1100)
ϕ^B	0767^{*}	1045***	4227***
,	(.0382)	(.0264)	(.0469)
w^{FX}	.0342	.0090	1412***
	(.0271)	(.0186)	(.0345)
Size	.0774***	.0543***	.0368***
	(.0125)	(.0083)	(.0131)
Corp	0760***	0466	.0069
1	(.0216)	(.0142)	(.0227)
Prof	0026	0099	0485
v	(.0415)	(.0272)	(.0431)
Rfr	.2196**	.2517***	.6764***
J.	(.1153)	(.0788)	(.1479)
Fr	0128	0406	1867***
	(.0513)	(.0349)	(.0576)
Time FE	Y	Y	Y
Ν	700	700	707
Left-censored	182	182	n/a
Uncensored	518	518	n/a



Figure 4.4. Benefits as a function of the liability duration

The dots in this figure show the observed ratios of pension payments to pension liabilities (Benefits) and the fitted curve for the observed ratios of pension payments to pension liabilities (red line), both as a function of the liability duration.

Boon et al. (2018) study public, corporate, and industry wide pension funds in the US, Canada, and the Netherlands and find that the regulatory framework matters for the asset allocation decisions. They show that risk-based capital requirements and mark-to-market valuation are associated with a 7 percentage points lower allocation to risky assets, regardless of market conditions. Given this evidence and the setup of our model we conjecture the following implications for the illiquid assets allocations in the US and in Canada. A higher interest rate risk exposure does not increase the capital requirements for pension funds in the US and Canada. Under this condition the impact of the liability duration on the illiquid assets allocation is likely to be increasing. In other words, the higher the liability duration, the higher we expect the illiquid assets allocation to be. Therefore, we expect higher illiquid asset allocations in the US and Canada. This is indeed what we observe empirically. The Willis Towers Watson Global Pension Assets Study (2018) shows that the value weighted average illiquid assets allocation in the US equals 28 percent and Canada equals 31 percent, whereas in the Netherlands it is only 17 percent.²⁰

The predictions of our model are likely to be amplified when considering life insurers. Life insurers have high liability durations and are subject to Solvency II regulation in Europe.

²⁰https://www.willistowerswatson.com/-/.../Global-Pension-Asset-Study-2018-Japan.pdf

Under this solvency regime the capital requirement is calibrated on a 0.5% one year default probability. This implies that the capital requirement is more profound for European life insurers than for Dutch pension funds, for which the capital requirement is calibrated on a 2.5% probability. We therefore predict the turning point of the hump-shaped effect of the liability duration on the illiquid assets allocation to be at a lower liability duration for insurance companies. In other words, the negative effect of having a higher liability duration (higher capital requirements) will outweigh the benefits (lower liquidity requirements) at a faster rate.

4.7 Conclusion

In this paper, we empirically study the impact of liquidity and capital requirements on a pension fund's illiquid assets allocation. Liquidity requirements result from short-term pension payments and collateral requirements on derivatives used for hedging purposes. In addition, the pension funds in our sample are capital constrained. Liquidity and capital requirements interact. By hedging interest rate risk and currency risk, a pension fund is less exposed to these two risk factors and can take additional investment risks. However, hedging strategies using derivatives involve margining. This hampers a pension fund to invest in illiquid assets as they impose a liquidity requirement.

The key conclusions of our empirical analysis are as follows. First, we find a hump-shaped impact of liability duration on the fraction of risky assets allocated to illiquid assets. Up to 18 years, the liability duration positively affects this allocation. However, beyond this point, the effect is reversed and the allocation to illiquid assets decreases. Second, we do not find evidence that swap hedge ratios impact the illiquid assets allocation: the liquidity and capital requirements cancel each other out. In the case of currency risk, however, hedging does impact the illiquid assets allocation positively. Finally, pension fund size, pension fund type and funding ratio also impact the illiquid assets allocation.

These findings offer important policy implications. Based on our model and empirical findings it does not appear obvious for long-term investors to always invest more in illiquid assets. The capital requirement for defined benefit pension funds becomes a binding constraint if the liability duration is substantially high. Although relaxing capital requirements for interest rate risk might seem a reasonable solution to mitigate the impact of this constraint, this is not what we recommend. The interest rate risk is inherent to the nature of the pension liabilities in a defined benefit pension contract. If a pension fund guarantees benefits,

CHAPTER 4. ILLIQUID ASSETS: LIQUIDITY AND CAPITAL REQUIREMENTS

interest rate risk becomes the key risk factors to manage. An alternative approach would be to redefine pension liabilities such that they no longer embed long-term interest rate guarantees. This, however, means that the interest rate risk shifts to the beneficiaries. Beneficiaries are less knowledgeable to understand and less equipped to manage interest rate risk.

A more important policy implication concerns adequate liquidity and collateral management for pension funds. Pension funds increasingly use derivatives to hedge risks. Derivatives involve greater liquidity needs. These liquidity needs can increase exponentially in times of market turbulence. Collateral management is therefore key and becomes even greater once pension funds are integrated in the central clearing of derivatives. Pension funds should prepare adequate contingency planning to be able to manage collateral through periods of market turbulence. More emphasis could, therefore, be placed on liquidity and collateral management in pension regulations.

Chapter 5

A Survey of Institutional Investors' Investment and Management Decisions on Illiquid Assets¹

5.1 Introduction

Investment and management decisions regarding illiquid assets (e.g. real estate, private equity, and infrastructure) are of particular relevance to pension funds since they are important investors in these asset classes. The recent Willis Towers Watson Global Pension Assets Study 2018 shows that the seven largest pension markets in the world,² with total pension assets of USD 37,770 billion at the end of 2017, increased their average illiquid asset allocation from 4% in 1997 to 25% in 2017. While there is a large literature on the benefits and challenges of investing in illiquid assets, to the best of our knowledge the self-reported attitudes and policies of institutional investors with respect to illiquid assets have not been documented extensively before.

In this survey we provide insight into the self-reported attitudes and policies with respect to illiquid assets and present the results of a survey of nine Dutch and five Canadian pension funds and fiduciary managers. We have interviewed these survey participants regarding the role of illiquid assets in the strategic asset allocation, the valuation of illiquid assets, and

¹This chapter is based on joint work with Patrick Tuijp and forthcoming in the Journal of Portfolio Management. We thank the survey participants for their willingness to participate and for their valuable input, and Ortec Finance and Theo Nijman for arranging the interviews. We thank Mark Anson, Martin Bakker, Marie Brière, Dirk Broeders, Chantal de Groot, Bert Kramer, Loranne van Lieshout, Rients Miedema, Theo Nijman, Bas Werker, Pieter Wijnhoven, and the Netspar Editorial Board for their helpful suggestions and comments. Kristy Jansen gratefully acknowledges the financial support from Netspar.

²Australia, Canada, Japan, the Netherlands, Switzerland, the UK, and the US.

liquidity management. We formulate four best practices based on the results of the survey.

The six Dutch pension funds in our survey represent assets under management (AUM) of EUR 342 billion, and the four Canadian pension funds represent AUM of CAD 203 billion.³ We exclude the AUM of the three Dutch fiduciary managers and the one Canadian fiduciary manager from the reported numbers since they also manage assets for investors other than pension funds. The Dutch pension funds in our sample on average invest 15% of their portfolio in illiquid assets, while for Canadian pension funds this is 34%.

The interviews with the survey participants were arranged by Ortec Finance and Theo Nijman and took place during the period from December 2017 to January 2019. While we do not have a random sample, we do cover approximately 26% of total pension assets in the Netherlands and approximately 9% for Canada.⁴ These percentages do not include the fiduciary managers for the same reason why we do not include them in the AUM figures. We specifically asked the fiduciary managers to give answers from the perspective of their pension fund clients. We therefore jointly analyze the answers given by pension funds and fiduciary managers. The participating pension funds focus mainly on defined benefit plans. The interview questions and the list of survey participants are listed in the Appendix. The survey participants were allowed to give more than one answer to each survey question.

In our survey, we define the following asset classes as illiquid: private equity, private debt, direct real estate, infrastructure, hedge funds, and mortgages. We base this classification on the criterion that pension funds might not be able to trade the asset for a substantial period of time. While the classification for particular asset classes may sometimes differ, the survey participants generally indicated that our definition of illiquid asset classes aligns with theirs.

5.2 Investing in illiquid assets

The academic literature shows that illiquid assets may offer benefits to pension funds and other investors in terms of liquidity premiums (Driessen and De Jong, 2015; Vayanos

 $^{^{3}}$ Total AUM reflects the situation at year-end 2017 and is based on the annual reports of the participating pension funds.

⁴Willis Towers Watson (2019) reports total Dutch pension assets at year-end of 2017 of about EUR 1,331 billion and total Canadian pension assets at year-end of 2017 of about CAD 2,225 billion. Exchange rate conversion has been applied from USD to local currency against year-end 2017 foreign exchange rates obtained from Exchange Data International.

and Wang, 2013),⁵ liability hedging (e.g. Sharpe and Tint, 1990b; Lucas and Zeldes, 2009), and portfolio diversification (e.g. Kallberg et al., 1996; Hoesli et al., 2004; Bianchi et al., 2014). In addition to these benefits, the long investment horizon of pension funds makes them well-suited for investing in illiquid assets classes (Attig et al., 2012). The search for yield in a low interest rate environment may provide additional incentives to consider riskier and less liquid investments (e.g. Rajan, 2006; Yellen, 2011).

Investing in illiquid assets also involves challenges. Infrequent trading of illiquid assets may lead to large losses in case of forced selling (Diamond and Rajan, 2011), for instance due to sudden and unpredictable margin requirements on derivative positions. In addition, whether real estate investments are managed internally or externally matters for investment costs (Andonov et al., 2015). Finally, the valuation smoothing that results from appraisal effects may present a challenge when assessing illiquid investments. Using an appraisalbased index could lead to lower volatility and higher autocorrelation estimates relative to a transaction-based index for similar assets (Fisher et al., 1994; Hoesli et al., 2004). Anson (2016) demonstrates how to un-smooth illiquid asset returns for asset allocation, risk management, and performance evaluation purposes.

In addition to these benefits and challenges, the literature has investigated portfolio choice with illiquid assets. Ang et al. (2014) show that an investor should reduce the allocation to illiquid assets in order to avoid situations where it would not be possible to meet, e.g., pension payments.⁶ Given a certain allocation to illiquid assets, investors may still adopt different strategies in response to immediate liquidity needs. For instance, they may sell liquid assets first to minimize transactions costs, as opposed to selling illiquid assets, and keeping a cushion of liquid assets (Duffie and Ziegler, 2003; Scholes, 2000). Empirical evidence indicates that institutional investors tend to follow the former strategy (Ben-David et al., 2012; Manconi et al., 2012; Driessen and Xing, 2017).

⁵However, the literature shows mixed evidence of liquidity premiums in different asset classes. For instance in the case of direct real estate, Qian and Liu (2012) report a positive effect of illiquidity on expected returns, although the effect is relatively small. Benveniste et al. (2001) show that REITs increase the value of underlying real estate assets by 12-22%, which suggests that a direct real estate liquidity premium may indeed exist. Ang et al. (2013b), on the other hand, show comparable performance of REITs and direct real estate, which indicates the absence of a direct real estate liquidity premium.

⁶Beber et al. (2020) go a step further and show that short-term investors may choose not to invest in the least liquid assets at all because the expected transaction costs may be prohibitively high relative to the return over a short horizon.

5.3 Pension fund regulations

Pension fund regulations are highly relevant to understand how pension funds invest. For instance, Andonov et al. (2017) show that U.S. public pension funds have incentives to invest in risky assets as their liability discount rate is linked to the expected return on assets. In addition, Boon et al. (2018) study U.S., Canadian, and Dutch pension funds and show that risk-based capital requirements and mark-to-market valuation of assets and liabilities are associated with a 7 percentage points lower risky asset exposure.

In this section we provide an overview of relevant aspects of the Dutch and Canadian pension fund regulations. First, we summarize the retirement systems for both countries. Second, we focus on the following four aspects of pension fund regulation: investment restrictions, asset and liability valuation methods, funding requirements, and recovery periods.

5.3.1 Regulations in the Netherlands

The Dutch pension system is built on three pillars. The first pillar is a pay-as-you-go (PAYG) system and consists of state pensions that provide a basic income, called AOW (Algemene Ouderdomswet). The second pillar consists of collective pension schemes which are administered by pension funds. The pension schemes are financed from mandatory contributions by employers and employees. The contribution rate amounts on average to 20-25% of total wages.⁷ In most cases the employer contributes 2/3 and the employee 1/3.⁸ The first pillar replaces broadly 30% of average wages and the second pillar 60-65% of average lifetime wages (Gerard, 2019). The third pillar consists of individual pension products, such as an annuity insurance. The pension funds participating in our survey are second pillar funds.

Dutch pension funds do not face any quantitative investment restrictions, but they are required to invest in accordance with the prudent-person rules outlined in the Pension Act.⁹ This implies that pension funds should invest in the best interest of the fund participants. Pension funds use market valuation for both assets and liabilities. Market valuation of defined benefit liabilities implies that they are discounted using a yield curve that is based

⁷Pension funds differ in their contribution rates, to be found in the annual reports of pension funds.

⁸https://www.rijksoverheid.nl/onderwerpen/pensioen/opbouw-pensioenstelsel

⁹https://wetten.overheid.nl/BWBR0020809

on the euro swap curve as set by the Dutch Central Bank (DNB). The minimum funding requirement (Minimaal Vereist Eigen Vermogen, MVEV) is determined by a calculation method described in the Pensions Act. Importantly, the minimum funding ratio is based on nominal valuation of liabilities, and only accrued benefits are taken into account. For most pension funds, the minimum capital requirement corresponds to a funding ratio of about 104.2%.¹⁰

Besides the minimum capital requirement, Dutch pension funds also have risk-based capital requirements. In fact, the required capital is calculated such that the probability that the nominal funding ratio falls below 100% on a one-year horizon equals 2.5%. For a stylized pension fund with equal investment in equity and bonds, the capital requirement (Vereist Eigen Vermogen, VEV) amounts to a funding ratio of approximately 120%. If the funding ratio falls below the required capital threshold, the pension fund enters a ten-year recovery period to meet the capital requirement. If the funding ratio remains below the minimum requirement for five years, a pension benefit cut may be necessary if recovery of the fund is not possible otherwise.¹¹

Since the valuation of the liabilities is nominal and market-based, the most important risk factor for Dutch pension funds is interest rate risk. This is especially true for green pension funds, i.e. pension funds with relatively young participants and therefore long-term pension liabilities. Pension funds can substantially reduce their capital requirements by hedging interest rate risk.

The regulatory framework does not force pension funds to hold capital depending on their portfolio liquidity, although the prudent-person rule requires pension funds to carefully consider liquidity in their investment strategies. Moreover, the capital requirements for liquid and illiquid assets may vary. For instance, non-listed equities, such as private equity, require a larger amount of capital than listed equities. On the other hand, direct real estate investments have lower capital requirements than listed equities. The differences in riskbased capital requirements depend on the overall perceived risk profile of the asset class.

¹⁰http://www.toezicht.dnb.nl/2/50-202138.jsp

 $^{^{11}\}mathrm{http://www.toezicht.dnb.nl/2/50-232915.jsp}$

5.3.2 Regulations in Canada

The Canadian pension system also consists of three pillars. The first pillar is a PAYG system and consists of state pensions, called Old Age Security (OAS). Canadians whose income is below a certain threshold receive a Guaranteed Income Supplement (GIS) in addition to OAS. The second pillar is the Canadian Pension Plan (CPP), except for Quebec, which has its own Quebec Pension Plan (QPP). CPP and QPP provide contributors and their families with partial replacement of earnings in case of retirement. The contribution rate is set each year and equals 10.2% in 2019.¹² The intention is to replace up to 25% of full-time income upon retirement. The contribution is equally split between employer and employee. The third pillar consists of workplace pensions and private savings plans. The Canadian pension funds participating in the interviews focus on the third pillar.

Although prior to 2010 restrictions applied for investments in foreign assets, real estate, and Canadian resource property, no investment restrictions exist for Canadian pension funds today. Canadian pension funds are also subject to a prudent-person rule.¹³ The valuations of assets and liabilities to determine the funding ratio are typically market-based. However, the Canadian Institute of Actuaries argues that valuation of assets producing a value different from market value may be suitable in some cases. For instance, smooth valuation of assets may be appropriate to moderate the volatility of contributions.¹⁴ In accordance with the Pension Benefits Act (PBA) and the Income Tax Act (ITA), each pension fund must file an actuarial valuation at least every three years in order to estimate the fund's surplus or deficit and to determine the fund's minimum funding requirements.¹⁵

The reported liabilities and funding ratios are based on a going-concern assumption as well as on a solvency basis. The going-concern basis implies that all assumptions made in estimating total pension fund liabilities are based on the principle that the pension fund will continue to operate. The liabilities under the going-concern assumption depend on expected returns of investment portfolios, inflation expectations, and the future growth rate of employee wages. The solvency basis focuses on the ability of the plan to meet its liabilities

¹²https://www.canada.ca/en/services/benefits/publicpensions/cpp/contributions.html

 $^{^{13}}$ For instance, such as set out in Section 22 of the Ontario Pension Benefits Act, R.S.O. 1990, c. P.8. https://www.ontario.ca/laws/statute/90p08

¹⁴https://www.cia-ica.ca/docs/default-source/2014/214100e.pdf

¹⁵For further details we refer to the certified general accountants association of Canada (Chartered Professional Accountants Canada): https://www.cpacanada.ca.

assuming that the plan is terminated on the date on which the report is filed. Liabilities are computed based on market interest rates, where in some provinces pension funds are allowed to average interest rates over the preceding five years.¹⁶

Unlike the minimum funding requirements in the Netherlands, the Canadian minimum funding requirements are not risk-based. The recovery period for pension funds that do not meet the minimum funding ratio differs across provinces. Federal plans and provincial plans in Alberta and Ontario have a maximum recovery period of ten years. Other provinces typically set this at five years (Boon et al., 2018).

5.4 Survey outcomes

In this section we provide the outcomes of our survey. We start by discussing the outcomes for The Netherlands and Canada separately, and we conclude this section with a comparison of the Dutch and Canadian outcomes.

5.4.1 Dutch fund outcomes

We start with the summary statistics of the Dutch pension funds in our sample (see Exhibit 5.1). The average AUM equals 57 billion euros, with a standard deviation of 73 billion euros. This shows that our sample contains both large and small funds. The average allocation to illiquid assets is 15%, close to the average 14% illiquid asset allocation reported in the Willis Towers Watson Global Pension Assets Study 2018 for the Netherlands. The ratio of active members to retirees is on average 1.2, indicating that the fraction of active members is only slightly higher than that of retirees. The average (nominal) funding ratio equals 115% with a standard deviation equal to 13%, revealing substantial variation in the funding status across pension funds. As noted in the introduction, the fiduciary managers are not included in the summary statistics as they manage assets for investors other than pension funds as well.

¹⁶For further details we refer to the certified general accountants association of Canada (Chartered Professional Accountants Canada): https://www.cpacanada.ca.

Table 5.1. Summary Statistics Dutch Pension Funds

This table provides summary statistics for six Dutch pension funds (fiduciary managers are excluded). AUM is in millions of euros. Source: annual reports (ultimo 2017) of participating pension funds.

	mean	std.dev.
AUM	57,030	72,510
Percentage illiquid assets	15%	5%
Actives to retirees ratio	1.2	0.9
Funding ratio	115%	13%

Strategic asset allocation to illiquid assets for Dutch funds

From the full set of interviews, including both Dutch and Canadian survey participants, five main reasons for investing in illiquid assets emerge: diversification benefits, liability hedging purposes, risk-return trade-off, smooth valuation of assets, and steady cash flows. Smooth valuation of assets results from the infrequent valuation of illiquid assets, and valuations may also lag relative to those of more frequently traded assets. From a balance sheet perspective these aspects may reduce the volatility of illiquid asset valuations. Several illiquid asset classes provide steady cash flows, such as rental income for real estate, which may help in ensuring that sufficient cash is available to meet cash requirements in the future.

The main reported reasons for the nine Dutch survey participants to invest in illiquid assets are summarized in Table 5.2. The risk-return trade-off is one of the main reasons for investing in illiquid assets, which fits with the search for yield phenomenon described by Rajan (2006) and Yellen (2011). Some survey participants specifically mentioned the natural comparative advantage that they have as long-term investors to capture liquidity premiums. As a second important reason for investing in illiquid assets, six out of nine also mentioned diversification. For the other three funds diversification is a positive side effect, but not the main reason to invest in illiquid assets. Only a few pension funds indicated liability hedging as a principal reason to invest in illiquid assets. The funds mentioning liability hedging as a main reason particularly see benefits for real estate and infrastructure but not so much for private equity. However, not all survey participants are convinced that real estate and infrastructure are good hedges against inflation risk. A fourth reason for investing in illiquid assets, given by one of the survey participants, is the ability to reduce balance sheet volatility through the smooth valuation of illiquid assets (see e.g. Hoesli et al., 2004, for a further discussion). Representatives of the Dutch pension funds did not mention steady cash flows as a principal reason for investing in illiquid assets.

Table 5.2. Reasons for Investing in Illiquid Assets

This table summarizes the number of Dutch survey participants who mentioned one of the following main reasons to invest in illiquid assets: steady cash flows, smooth valuation, risk-return trade-off, liability hedging, and diversification. The total number of survey participants is nine, and survey participants could mention more than one reason.

	number of participants
Smooth valuation	1
Steady cash flow	0
Risk-return trade-off	9
Liability hedging	2
Diversification	6

Table 5.3 summarizes how the survey participants determine their illiquid asset allocation. Six of the nine survey participants determine their illiquid asset allocation using an asset liability management (ALM) study. However, two survey participants mentioned that, if the optimal allocation resulting from the ALM study is too high, the pension fund board might decide to invest less in illiquid assets. Three of the nine survey participants stipulate a target return for illiquid assets. These pension funds invest only in illiquid assets that they expect to achieve at least the specified target return. Pension funds may therefore deviate from initial strategic allocations depending on the amount of assets available for investment that meet the specified target return.

The third question concerns whether pension funds apply an upper limit to the percentage of assets allocated to illiquid assets. Most pension funds do have a maximum illiquid asset allocation, of approximately 20-25%. The maximum allocations mentioned are relatively close to each other. The prudent-person rule described in the Pension Act is interpreted by many Dutch pension funds as a maximum of 25% in illiquid assets. Although some pension funds have not explicitly specified their maximum allocation to illiquid assets, some note that the maximum is implicitly set by the size of their portfolio components.

Dutch pension funds typically divide their total portfolio in a liability matching portfolio

Table 5.3. Determination of Illiquid Asset Allocation

This table summarizes the number of Dutch survey participants who mention one of the following ways to determine the allocation to illiquid assets: a target return, board decision, and/or ALM study. The total number of survey participants is nine, and survey participants could mention more than one approach.

	number of participants
Target return	3
Board	2
ALM study	6

and a return portfolio. As mentioned in our discussion of the regulatory framework, the liabilities of Dutch pension funds are computed using a market-based yield curve. For this reason, Dutch pension funds invest part of their portfolio in liquid fixed income assets to reduce the interest rate exposure mismatch between their assets and liabilities. These fixed income assets used for liability hedging purposes reside in the matching portfolio. With potential exceptions such as mortgages, illiquid assets typically reside in the return portfolio. If the liability matching portfolio is relatively large (say 60% of total assets), then even with a relatively modest allocation the illiquid assets (say 10% of total assets) forms a sizable fraction (in the example 25%) of the return portfolio. This may limit the allocation to illiquid assets as a fraction of the total portfolio, since these would otherwise crowd out return portfolio investments in more liquid assets such as listed equities.

Lastly, we asked pension funds which other aspects are taken into account when investing in illiquid assets. We mention a few of these aspects here. Pension funds also take into account the sustainability aspect of illiquid investments such as wind parks. The cost aspects of illiquid assets also play a major role for Dutch pension funds. They mentioned that the costs to invest in illiquid assets are generally much higher than the costs for liquid assets. Finally, pension funds set strategic asset allocation bounds on the target weights for all asset classes that they invest in. Since illiquid assets do not trade frequently, remaining within asset class bounds is more challenging for illiquid assets than for liquid assets.

Asset valuation for Dutch funds

The second set of questions focuses on the asset valuation of illiquid assets. We first asked how pension funds value their illiquid assets. The Dutch pension funds that we interviewed only invest in illiquid assets via fiduciary managers. Most pension funds rely on the valuations of their external fiduciary managers. A large fraction of pension funds analyzes past performance of illiquid assets to evaluate their performance. Some pension funds monitor whether the valuation method used by fiduciary managers is compliant with standards such as the ILPA Private Equity Principles. The fiduciary managers usually use discounted cash flow (DCF) methods for real estate and infrastructure, whereas for private equity the EBITDA multiple and the public market equivalent (PME) approach are more commonly used valuation methods.

Second, we asked how expected returns on illiquid assets are characterized, and in particular whether pension funds make assumptions for liquidity premiums. Most pension funds use a benchmark-plus-spread approach to determine expected returns. The benchmark is generally used as a public market equivalent, such as the S&P500 for private equity and REIT returns for real estate. On top of the benchmark, pension funds assign a spread, which they report varies between 100 to 300 basis points depending on the asset class and the beliefs of the pension funds about the asset classes. The survey participants report that this spread represents different aspects such as skill of the investment manager and compensation for illiquidity. This spread is relatively conservative compared to the respective liquidity premium estimates of 300 bps and 370 bps found by Franzoni et al. (2012) and Anson (2017) for private equity investments.¹⁷ The difference could be driven by differences in beliefs, the asset class, or both. Three of the nine pension funds aim to make explicit various components of the spread and thus have an explicit number for the liquidity premium of illiquid asset classes, at least where possible. Only one participant mentioned direct valuation of illiquid assets as a way to determine expected returns.

The final question was about the certainty of contract periods for illiquid assets. For private equity, all pension funds are aware that the scenario of an extension of the contract period beyond ten years is quite likely. Pension funds indicated that the announcement of contract extensions is more towards the end of the contract, and usually equals a two or three year period. As pension funds already take into account the possibility of a contract extension when entering the contract, pension funds do not consider contract extensions as a potential risk.

 $^{^{17}}$ Franzoni et al. (2012) use the Pastor and Stambaugh (2003) liquidity risk measure, whereas Anson (2017) considers the liquidity level.

Table 5.4. Assumption of Expected Returns on Illiquid Assets

This figure summarizes the number of Dutch survey participants who mention one of the following ways to determine expected returns of illiquid assets: direct valuation or benchmark plus a spread. We also asked whether survey participants specifically assume a liquidity premium for illiquid assets. The total number of survey participants is nine, and survey participants could mention more than one approach.

	number of participants
Separate liquidity premium	3
Direct valuation Benchmark $+$ spread	$\frac{1}{7}$

Liquidity management and stress scenarios for Dutch funds

Many of the survey participants indicated that they have put either implicit or explicit policies in place to ensure that sufficient resources are available to fulfill immediate obligations. For instance, a cash buffer of about 0% to 7.5% of AUM may be maintained to cover immediate liquidity needs. For the survey participants the cash buffer mainly exists to cover margin payments for derivatives positions. This need may grow as central clearing becomes the standard in the near future (EMIR regulation). Some survey participants have outsourced liquidity management to their fiduciary manager.

The survey participants indicated that they perform stress tests (for instance, a negative shock of a certain magnitude to financial markets) to analyze whether they have sufficient liquid instruments to cover immediate obligations. This may be done through an analysis that indicates after how many days they need to liquidate existing positions to free up cash.

In case of a stress event, the repo market may provide initial liquidity for some survey participants. A few survey participants reported that they may consider securities lending. Others indicated that they would sell positions in their most liquid instruments first (e.g. equities and highly rated government bonds). This may be done either in the order of liquidity, or according to the respective portfolio weights. In some cases the assets which are most above strategic portfolio weights are sold first.

Although all survey participants indicated that there is an implicit policy, not all of them have formally and explicitly formulated policies regarding which assets would be sold first and in what quantities in case of a stress event. In spite of these policies being in place, many survey participants mentioned that they do not expect to have to sell securities to free up liquidity. According to their stress analyses, the cash buffer should cover immediate liquidity needs.

5.4.2 Canadian fund outcomes

For the Canadian pension funds in our sample, we also start with the summary statistics (Exhibit 5.5). Average AUM of Canadian funds is 53 billion euros, with a standard deviation equal to 37 million euros. Also for Canadian pension funds, our sample represents a mix of relatively large and small funds. The ratio of active members relative to retirees is on average 1.8, showing that the average group of active members is larger than the group of retirees. As for the Dutch funds, the standard deviation of the ratio of active members to retirees reveals substantial variation across Canadian funds. Canadian funds invest on average 34% in illiquid assets, in line with findings of the Willis Towers Watson Global Pension Asset Study 2018.¹⁸ The average funding ratio (on a going-concern assumption) equals 109% and the standard deviation 14%, which again shows that pension funds substantially differ in their funding status. As noted in the introduction, the fiduciary managers are not included in the summary statistics as they manage assets for investors other than pension funds as well.

Table 5.5. Summary Statistics Canadian Pension Funds

This table provides summary statistics for four Canadian pension funds (fiduciary managers are excluded). AUM is in millions of Canadian dollars. Source: annual reports (year-end 2017) of participating pension funds.

	mean	std.dev.
AUM	50,748	41,977
Illiquid assets percentage	34%	12%
Actives to retirees ratio	1.8	0.5
Funding ratio	109%	14%

 $^{^{18}\}mathrm{The}$ average illiquid asset allocation in Canada equaled 31% in 2017.

Strategic asset allocation to illiquid assets for Canadian funds

The main reported reasons for the five Canadian survey participants to invest in illiquid assets are summarized in Table 5.6. For all survey participants the risk-return trade-off and diversification benefits are the most important reasons for investing in illiquid assets, again fitting with the search for yield phenomenon described by Rajan (2006) and Yellen (2011). Some survey participants specifically mentioned their comparative advantage as long-term investors to capture liquidity premiums. Four out of five also mentioned liability hedging as an important determinant to invest in illiquid assets. Liability hedging is a relevant reason mainly for real estate and infrastructure, and less so for private equity. Three out of five survey participants also mentioned a fourth reason to invest in illiquid assets: cash flow matching of liabilities. Canadian survey participants consider real estate and infrastructure most suitable to obtain steady cash flow streams. Smooth valuation of illiquid assets was not mentioned by Canadian survey participants as a principal reason for investing in illiquid assets.

Table 5.6. Reasons for Investing in Illiquid Assets

This table summarizes the number of Canadian survey participants who mentioned one of the following main reasons to invest in illiquid assets: steady cash flows, smooth valuation, risk-return trade-off, liability hedging, and diversification. The total number of survey participants is five, and survey participants could mention more than one reason.

	number of participants
Smooth valuation	0
Steady cash flow	3
Risk-return trade-off	5
Liability hedging	4
Diversification	5

One participant mentioned that they see an important advantage of investing in illiquid assets in the degree of control that pension funds have when investing in these asset classes. Canadian pension funds manage a substantial fraction of their illiquid assets internally. This implies that they invest directly in, for instance, real estate investments and unlisted equity. The participant noted that this gives pension funds a high degree of active ownership, incentive alignment, and superior information compared to listed asset classes.

Table 5.7 summarizes the responses to our second question, which concerns the most
important ways to determine the allocation to illiquid assets. The answers vary, and pension funds may take more than just one factor into account. Three out of five funds may deviate from target allocations, which could be arrived at via e.g. an ALM study, depending on a specified target return for illiquid assets. For instance, one survey participant stressed that the fund would like to implement a (much) larger allocation to illiquid assets, but insufficient investment opportunities that meet the target return were available at the time. Other funds mainly arrive at their illiquid asset allocation based on an ALM study. Two out of five funds mention the importance of the pension board in explicitly setting targets for the allocation to illiquid assets.

Table 5.7. Determination of Illiquid Asset Allocation

This table summarizes the number of Canadian survey participants who mention one of the following ways to determine the allocation to illiquid assets: a target return, board decision, and/or ALM study. The total number of survey participants is five, and survey participants could mention more than one approach.

	number of participants
Target return	3
Board	2
ALM study	2

Third, we asked whether funds have an upper limit for how much they invest in illiquid assets. In general, Canadian survey participants do have an upper limit for their illiquid asset allocation. For those who do have such an upper limit, it varies significantly across survey participants, from 15% to 80%. In most cases the board decides on the upper limit.

Finally, we asked if there are any other aspects that are considered when investing in illiquid assets. The survey participants indicated that there might be limits on how much the pension fund can invest in domestic versus foreign investments, that there may be limits for the amount invested in individual investments, and that there may be limits for the amount of embedded leverage that illiquid assets create for the portfolio.

Asset valuation for Canadian funds

The second set of questions addresses the valuation of illiquid assets. The valuation method for illiquid assets varies mainly across asset classes. For real estate and infrastructure the valuation is often done by an external appraiser using the DCF method. For private equity a mix of valuation methods is used by pension funds: DCF, EBITDA multiples, and PME multiples are often applied.

Table 5.8 summarizes the responses regarding the return assumptions on illiquid assets. As for Dutch survey participants, most Canadian survey participants formulate expected returns on illiquid assets through a benchmark-plus-spread approach, where the spread consists of different components including skill of the manager and liquidity. Two funds also aim to quantify the liquidity premium explicitly. As for Dutch survey participants, only one mentioned direct valuation of illiquid assets as a way to determine expected returns.

Table 5.8. Assumption of Expected Returns on Illiquid Assets

This figure summarizes the number of Canadian survey participants who mention one of the following ways to determine expected returns of illiquid assets: direct valuation or benchmark plus a spread. We also asked whether survey participants specifically assume a liquidity premium for illiquid assets. The total number of survey participants is five, and survey participants could mention more than one approach.

	number of participants
Separate liquidity premium	2
Direct valuation	1
Benchmark + spread	4

The final survey question asked how certain the contract periods of illiquid assets are. The answers are comparable to those given by Dutch survey participants. For private equity, all pension funds are aware that the scenario of an extension of the contract period beyond ten years is quite likely. Most survey participants indicated that a two to three year extension of the contract period is expected and taken into consideration when entering the contract.

Liquidity management and stress scenarios for Canadian funds

Similar to the Dutch survey participants, their Canadian counterparts also have put either implicit or explicit policies in place for liquidity management. Cash is held mainly to cover the liquidity needs of derivatives rather than to account for illiquid assets, although in certain cases it is necessary to plan for large capital calls generated by illiquid investments. The repo market was mentioned by one participant as a serious avenue for managing liquidity needs. One participant remarked that the attitude towards liquidity may depend on the ratio of payouts to contributions. For funds with higher payout amounts, liquidity may be more important since cash needs are larger.

Survey participants have projections for their cash inflows and outflows, so they explicitly plan sufficient liquidity to meet immediate obligations, also under adverse market conditions. When such obligations arise, many survey participants report the preference to first use cash, then the repo market, then selling liquid public securities, and as a last resort selling illiquid private assets.

We found that, when making their investment decision, some survey participants take into account the possibility to liquidate the investment in order to free up cash for immediate obligations. They tailor the investment type of the illiquid assets that they choose to invest in (e.g. open-end vs. closed-end real estate funds) such that they do not have their full position in a relatively illiquid type.

5.4.3 Comparison of Dutch and Canadian outcomes

In this section we describe the main differences in survey results between the Dutch and the Canadian survey participants and we provide a comprehensive summary of these differences in Exhibit 5.9. From the summary statistics we immediately note the biggest difference: the Canadian pension funds invest substantially more in illiquid assets, on average 34% of AUM, than the Dutch pension funds, for which the average is 15% of AUM. The variation in the fraction allocated to illiquid assets across pension funds is also substantially larger in Canada compared to the Netherlands. These differences are also reflected in the maximum allocations to illiquid assets reported by the survey participants. Dutch survey participants often indicated a maximum allocation to illiquid assets of approximately 20-25%, whereas this upper limit varies from 15 to 80% across Canadian survey participants.

The Willis Towers Watson Global Pension Assets Study 2018 shows that the difference in average allocations to illiquid assets between Dutch and Canadian pension funds was relatively small in 2007 and has increased strongly since then. The average allocation of Canadian pension funds to assets other than equities, bonds, and cash has increased from 14% in 2007 to 22% in 2017. For the eight largest Canadian pension funds, Bédard-Pagé et al. (2016) report that the allocation to less-liquid alternative assets has even increased from 21% in 2007 to 29% in 2015, in line with the finding of Broeders et al. (2020) that large pension funds invest more in illiquid assets than small pension funds. In contrast, for Dutch pension funds the allocation to assets other than equities, bonds, and cash has remained

Table 5.9. Comparison Dutch and Canadian outcomes

This table summarizes the main differences in the survey results between the Dutch and they Canadian survey participants. The statistics refer to six Dutch and four Canadian pension funds (fiduciary managers are excluded).

	Netherlands	Canada
Average AUM in millions of EUR	57,030	33,608
Average allocation illiquid assets	15%	34%
Cross-sectional variation allocation illiquid assets	5%	12%
Maximum allocation illiquid assets	20-25%	15-80%
Management illiquid assets	external	internal
Regulatory framework	nominal	real

stable going from 18% in 2007 to 17% in 2017.

Interestingly, the average AUM of the Canadian pension funds is slightly more than half the average AUM of the Dutch pension funds in the sample and yet they have a much larger allocation to illiquid assets on average. Consequently, pension fund size does not seem to be the driver of the stark difference in allocations. We put forward three other reasons to explain the allocation gap between the two countries.

First, the cost aspect of illiquid assets plays a major role for Dutch pension funds. One of the underlying reasons is that Dutch pension funds have to mandatorily report their investment costs to the supervisor (DNB), who subsequently publishes the investment costs for each individual pension fund on their website.¹⁹ The supervisor holds the position that pension funds should be able to justify the investment costs to the public, which creates the potential for public scrutiny and reputational risk. During the interviews, Dutch survey participants mentioned that the costs to invest in illiquid assets are generally much higher than the costs for liquid assets and that this plays an important role in their asset allocation decision. On the other hand, competitive pays are the standard in Canada. Commonwealth Partners (2017) reports that Canadian pension funds pay competitively such that they can build internal investment teams.²⁰ Canadian compensation costs are therefore generally

¹⁹The pension fund specific investment costs are available at https://statistiek.dnb.nl/downloads/.

²⁰See also PwC (2016), page 17, which reports on insourcing portfolio management that "In order to retain high-level talent, pension funds must pay competitive salaries to people who might otherwise choose to work in the commercial fund management sector."

higher than those of pension funds in some other countries. However, competitive pays also give pension funds the ability to bring investment management and other functions internally, which they believe tend to result in lower costs and better net performance, in line with the findings of Andonov et al. (2015).

The second explanation relates to both the supervisory framework as well as the difference in how illiquid assets are managed. The Dutch supervisor imposes stricter requirements on pension funds when they invest in alternative investments and other non-listed assets, and also when assets are managed internally.²¹ The Dutch supervisor may examine whether it finds that a pension fund sufficiently understands its investments and knows the associated risks.²² When pension funds do not manage illiquid assets directly, which is typically the case for Dutch pension funds, understanding illiquid assets is a much harder task. On the other hand, Canadian survey participants manage a large fraction of their illiquid assets internally. As mentioned by one of the Canadian survey participants, in-house management of illiquid assets has the advantage that there is a high degree of control. This participant finds that this gives a high degree of active ownership, incentive alignment, and superior information compared to listed asset classes.

Third, there is a division of Dutch pension fund assets into a liability matching portfolio and a return portfolio, where illiquid assets typically reside in the return portfolio. If the liability matching portfolio, which contains mainly fixed income assets, is relatively large (say 60% of total assets), then even with a relatively modest allocation the illiquid assets (say 10% of total assets) forms a sizable fraction of the return portfolio (in the example 25%). This may limit the allocation to illiquid assets as a fraction of the total portfolio, since these would otherwise crowd out return portfolio investments in more liquid assets such as listed equities. One of the Dutch survey participants indicated explicitly that the division in a matching and return portfolio limits their allocation towards illiquid assets.

Regarding the allocation decision, Dutch survey participants tend to arrive at their

²¹See e.g. the DNB policy rule regarding risk management for alternative investments, https://www.toezicht.dnb.nl/binaries/50-211807.pdf, and the DNB letter dated 3 July 2019 on their investment review findings, https://www.toezicht.dnb.nl/binaries/50-237819.pdf.

²²That the supervisory pressure is restrictive for Dutch funds is supported by the IPE article dated 20 June 2016 "APG: Dutch pension funds should follow Canadian schemes' example", https://www.ipe.com/apg-dutch-pension-funds-should-follow-canadian-schemes-example/10013847.article. In this article, the CIO of the Dutch pension fund APG is quoted saying "Pension funds over here are apprehensive because of regulatory pressure" regarding investments in infrastructure, private equity, and real estate in a specific comparison of the Netherlands and Canada.

illiquid asset allocation via an ALM study. Canadian survey participants indicate that they may deviate from target allocations, which could be arrived at via e.g. an ALM study, depending on how many investment opportunities are available that meet a specified required target return on illiquid assets.

In terms of liquidity management, the main difference is that Dutch survey participants are more concerned about larger future cash needs due to the central clearing of derivatives, while we see a large similarity between Dutch and Canadian survey participants in the approaches to liquidity stress testing and which assets to sell first to free up cash when necessary. The preference to sell liquid assets first to cover their liquidity needs is in line with Ben-David et al. (2012), Manconi et al. (2012), and Driessen and Xing (2017).

5.5 Best practices

We have formulated four best practices based on the results of our survey of nine Dutch and five Canadian pension funds and fiduciary managers on their investment and management decisions regarding illiquid assets. These best practices focus mainly on the actions and processes of the fiduciary managers and investors.

Rationale for investing in illiquid assets

Investors should have a clear rationale for investing in illiquid assets. Based on this rationale, informed decisions can be made regarding the role of specific illiquid assets in the investment portfolio. For instance, inclusion of illiquid assets to ensure an attractive risk-return tradeoff may lead to different decisions than inclusion for liability hedging purposes (this is illustrated by the framework of Sharpe and Tint, 1990b).

Awareness of valuation method used for illiquid assets

Investors should be aware of the valuation methods used for the illiquid assets they invest in. For example, the valuation method has important consequences for the extent to which the risk and return statistics reflect transaction-based prices. Appraisal-based valuations may lead to a lower volatility assessment than transaction-based valuations (Fisher et al., 1994; Hoesli et al., 2004). Depending on the intended method, inadvertently using a different method could lead to volatility measures which understate the actual risk. Investors should therefore understand how valuation methods affect their overall portfolio value, their risk measures, and in particular their solvency position. Anson (2016), for instance, demonstrates how to un-smooth illiquid asset returns for asset allocation and risk management purposes.

Analysis of liquidity needs and liquidity stress testing

Investors should regularly analyze their current and future liquidity needs. The available cash and liquid instruments should be sufficient to cover the net cash outflows arising from the various inflows and outflows (see, for instance, the framework of Ang et al., 2014). Cash inflows could, for example, originate from pension contributions and from distributions obtained from investments. Cash outflows could occur due to margin payments on derivatives, payouts to retirees, and capital calls related to investments.

Stress tests are needed to analyze the impact of an adverse shock to market conditions on the ability to satisfy immediate cash requirements since the latter may be substantial (Brunnermeier, 2009). Such stress tests should take into account that the size of the cash requirements themselves may also depend on these market conditions.

Liquidity stress testing is particularly relevant for investors in illiquid assets. Even when sufficient cash and liquid instruments are available to cover immediate cash requirements, the impact of an adverse shock as well as the sale of liquid instruments may lead to deviations from target asset allocations. Rebalancing towards the desired allocation may then involve the forced sale of illiquid assets, potentially with large losses (e.g. Diamond and Rajan, 2011), even though it was initially possible to satisfy the cash requirements without such forced sale.

Policy for liquidation of assets during stress events

Investors in illiquid assets should have an explicit policy regarding the methods to free up cash for immediate liquidity needs (see the considerations in Duffie and Ziegler, 2003). Such policy should stipulate the preferred order in which these methods should be used, as well as which assets to sell, and in which sequence and amount, should other sources to fulfill immediate obligations prove insufficient.

By establishing explicit policies to free up cash for immediate liquidity needs, it will be possible to investigate the impact of different policy choices under adverse shocks to market conditions. Such analyses are highly relevant to investors in illiquid assets for the same reasons as outlined in the best practice regarding the analysis of liquidity needs and liquidity stress testing.

5.6 Conclusions

In this paper we report the results of a survey of nine Dutch and five Canadian pension funds and fiduciary managers on their investment and management decisions towards illiquid assets. This is of particular relevance in a low interest rate environment where a search for yield may arise (e.g. Rajan, 2006; Yellen, 2011). The Dutch pension funds in our sample invest 15% of their portfolio in illiquid assets, while for the Canadian pension funds this fraction is 34%.

We put forward three reasons to explain the allocation difference between the two countries. First, the cost aspect of illiquid assets plays a major role for Dutch pension funds. During the interviews, Dutch survey participants mentioned that the costs to invest in illiquid assets are generally much higher than the costs for liquid assets and that this plays an important role in their asset allocation decision. The supervisor holds the position that pension funds should be able to justify the investment costs to the public, which creates the potential for public scrutiny and reputational risk. The second explanation relates to both the supervisory framework as well as the difference in how illiquid assets are managed. The Dutch supervisor imposes stricter requirements on pension funds when they invest in alternative investments and other non-listed assets, and also when assets are managed internally. Third, there is a division of Dutch pension fund assets into a liability matching portfolio and a return portfolio, where illiquid assets typically reside in the return portfolio. Even with a relatively modest allocation to illiquid assets, these potentially crowd out other more liquid investments from the return portfolio such as listed equities.

Both Dutch and Canadian survey participants most often indicated the risk-return tradeoff as their main reason to invest in illiquid assets, with diversification benefits as the second most often provided reason. The Canadian survey participants indicated steady cash flow and liability hedging as main reason more often than Dutch survey participants. In terms of the implementation, Canadian survey participants manage a large fraction of their illiquid assets internally, whereas Dutch survey participants typically manage illiquid assets externally.

Dutch survey participants often mentioned ALM studies as their main basis for determining how much to invest in illiquid assets, while Canadian survey participants often mentioned that they may deviate from target allocations, which could be arrived at via e.g. an ALM study, depending on how many investment opportunities are available that meet a specified target return on illiquid assets. One Canadian survey participant indicated that the fund would like to achieve a (much) larger allocation to illiquid assets, but that insufficient investment opportunities that meet the fund's target return were available at the time.

Regarding the valuation methods for illiquid assets, we found that most survey participants use discounted cash flow methods for real estate and infrastructure, and EBITDA multiples or the public market equivalent for private equity. Survey participants indicated that they determine expected returns on illiquid assets through a benchmark-plus-spread approach. The spread may include components such as liquidity and asset manager skill, although these components are not necessarily made explicit.

Most participants in the survey have either explicit or implicit liquidity policies in place, such as maintaining a cash buffer, making use of the repo market or securities lending, and an implicit or explicit sequence in which to liquidate positions to free up cash if necessary. Many of the survey participants perform liquidity stress tests under a given adverse market shock. They find that their cash buffer should be sufficient in most cases, and they do not expect to have to sell assets to free up cash under adverse market conditions.

The survey participants reported that the main cause of immediate cash requirements is margin payments on derivatives. They expect that this cause may become even more relevant in the near future as central clearing becomes the standard. The survey participants indicated that they either apply cash flow planning or specifically set cash aside for other immediate cash requirements, such as capital calls by private equity funds and payments to retirees.

We close with four best practices based on the results of the survey. First, investors should have a clear motivation for investing in illiquid assets, as this may depend on the role that illiquid assets play in the overall portfolio. Second, investors should be aware of the valuation methods used for illiquid assets that they invest in, so as to correctly interpret their portfolio value, risk measures, and solvency position. Third, to assess the adequacy of liquid resources in the portfolio, investors should regularly analyze their current and future liquidity needs. Fourth, investors should have an explicit policy regarding which methods to use and in which sequence to free up cash when immediate liquidity needs arise.

5.7 Appendix

A Survey questions illiquid assets

This appendix provides the list of questions answered by the survey participants.

Role in the organization

1. What is your role in your organization and how does it relate to the investment process?

Strategic asset allocation to illiquid assets

- 1. What are the main reasons to invest in illiquid assets (e.g. private equity, private debt, direct real estate, infrastructure, hedge funds)?
- 2. What determines the amount of total assets invested in illiquid assets? In other words, how is the optimal allocation to illiquid assets defined?
- 3. Is there an upper limit to the percentage of total assets invested in illiquid investments? If so, what is this upper limit based on?
- 4. Which aspects are taken into account in the decision to invest in illiquid assets?

Valuation illiquid assets

- 1. How is the value of an illiquid asset determined? Can you provide an example, i.e. how does it work in the case of private equity?
- 2. How are the expected returns of these assets characterized? In particular, what is the assumption about the liquidity premium for these assets?
- 3. How certain is the contract period of illiquid assets? For instance, private equity contracts generally last for ten years, but sometimes the contract is extended by one or two years. Is this known well in advance or does it remain uncertain for a long time?

Liquidity and fund management

1. What are the policies put in place to ensure that sufficient resources are available to fulfill immediate obligations at any point in time? For instance, does the fund hold a cash buffer to specifically compensate for the illiquidity of the portfolio?

2. Suppose a stress scenario occurs and market liquidity dries up. Is there a particular strategy to deal with this? Are there certain securities that would be sold first if the cash at hand is not sufficient to fulfill immediate obligations?

Other

1. Do you have any additional information that you think would be relevant in the context of the discussion?

B Survey participants

This appendix provides the list of survey participants.

Netherlands

- BlackRock
- Pensioenfonds ING
- Pensioenfonds Metaal en Techniek (PMT)
- Pensioenfonds Zorg en Welzijn (PFZW)
- PGB Pensioendiensten
- Philips Pensioenfonds
- SPF Beheer B.V.
- Stichting Pensioenfonds Vopak
- TKP Investments / Aegon Asset Management

Canada

- Healthcare Of Ontario Pension Plan (HOOPP)
- IMCO
- OPTrust
- UTAM
- One anonymous survey participant

Bibliography

- ACHARYA, V. AND L. PEDERSEN (2005): "Asset Pricing with Liquidity Risk," Journal of Financial Economics, 77, 375–410.
- ALESTALO, N. AND V. PUTTONEN (2006): "Asset allocation in Finnish pension funds," Journal of Pension Economics and Finance, 5, 27–44.
- AMIHUD, Y. AND H. MENDELSON (1986): "Asset Pricing and the Bid-Ask Spread," *Journal* of Financial Economics, 17, 223–249.
- AN, H., Z. HUANG, AND T. ZHANG (2013): "What determines corporate pension fund risk-taking strategy?" Journal of Banking and Finance, 37, 597–613.
- ANANTHARAMAN, D. AND Y. G. LEE (2014): "Managerial risk taking incentives and corporate pension policy," *Journal of Financial Economics*, 111, 328–351.
- ANDONOV, A. (2014): "Delegated Investment Management in Alternative Assets," Netspar Discussion Paper No. 03/2014-085.
- ANDONOV, A., R. BAUER, AND M. CREMERS (2017): "Pension fund asset allocation and liability discount rates," *Review of Financial Studies*, 30, 2555–2595.
- ANDONOV, A., P. EICHHOLTZ, AND N. KOK (2015): "Intermediated Investment Management in Private Markets: Evidence from Pension Fund Investments in Real Estate," *Journal of Financial Markets*, 22, 73–103.
- ANG, A., B. CHEN, W. GOETZMANN, AND L. PHALIPPOU (2013a): "Estimating Private Equity Returns from Limited Partner Cash Flows," Columbia Business School.
- ANG, A., N. NABAR, AND S. WALD (2013b): "Searching for a Common Factor in Public and Private Real Estate Returns," *The Journal of Portfolio Management*, 39, 120–133.

- ANG, A., D. PAPANIKOLAOU, AND M. WESTERFIELD (2014): "Portfolio choice with illiquid assets," *Management Science*, 60, 2737–2761.
- ANSON, M. (2016): "Asset Allocation with Private Equity," The Journal of Investment Consulting, 17.
- (2017): "Measuring Liquidity Premiums for Illiquid Asset Classes," *The Journal of Alternative Investments*, 20, 39–50.
- ASNESS, C. S., T. J. MOSKOWITZ, AND L. H. PEDERSEN (2013): "Value and Momentum Everywhere," *The Journal of Finance*, 68, 929–985.
- ATKINS, A. AND E. DYL (1997): "Transactions costs and holding periods for common stocks," *Journal of Finance*, 52, 309–325.
- ATTIG, N., S. CLEARY, S. EL GHOUL, AND O. GUEDHAMI (2012): "Institutional investment horizon and investment cash flow sensitivity," *Journal of Banking and Finance*, 36, 1164–1180.
- BAO, J., J. PAN, AND J. WANG (2011): "The illiquidity of corporate bonds," *Journal of Finance*, 66, 911–946.
- BASAK, S. AND A. SHAPIRO (2001): "Value-at-Risk-Based Risk Management: Optimal Policies and Asset Prices," *Review of Financial Studies*, 14, 371–405.
- BEBER, A., J. DRIESSEN, AND P. TUIJP (2020): "Pricing Liquidity Risk with Heterogeneous Investment Horizons," *Journal of Financial and Quantitative Analysis*, forthcoming.
- BÉDARD-PAGÉ, G., A. DEMERS, E. TUER, AND M. TREMBLAY (2016): "Large Canadian Public Pension Funds: A Financial System Perspective," Bank of Canada Financial System Review, June, 33–38.
- BEN-DAVID, I., F. FRANZONI, AND R. MOUSSAWI (2012): "Hedge Fund Stock Trading in the Financial Crisis of 2007–2009," *Review of Financial Studies*, 25, 1–54.
- BEN-REPHAEL, A., O. KADAN, AND A. WOHL (2015): "The Diminishing Liquidity Premium," Journal of Financial and Quantitative Analysis, 50, 197–229.

- BENVENISTE, L., D. CAPOZZA, AND P. SEGUIN (2001): "The value of liquidity," *Real* estate economics, 29, 633–660.
- BERK, J. AND J. VAN BINSBERGEN (2015): "Measuring Skill in the Mutual Fund Industry," Journal of Financial Economics, 118, 1–20.
- BIANCHI, R., G. BORNHOLT, M. DREW, AND M. HOWARD (2014): "Long-term U.S. infrastructure returns and portfolio selection," *Journal of Banking and Finance*, 42, 314–325.
- BIKKER, J., D. BROEDERS, AND J. DE DREU (2010): "Stock market performance and pension fund investment policy: Rebalancing, free float, or market timing?" *International Journal of Central Banking*, 53–79.
- BINSBERGEN, J., M. BRANDT, AND R. KOIJEN (2008): "Optimal Decentralized Investment Management," *Journal of Finance*, 63, 1849–1895.
- BLAKE, D., B. LEHMANN, AND A. TIMMERMANN (1999): "Asset Allocation Dynamics and Pension Fund Performance," *The Journal of Business*, 72, 429–461.
- BLAKE, D., A. ROSSI, A. TIMMERMANN, I. TONKS, AND R. WERMERS (2013): "Decentralized Investment Management: Evidence from the Pension Fund Industry," *The Journal of Finance*, 68, 1133–1178.
- BODIE, Z., R. MERTON, AND W. SAMUELSON (1992): "Labor Supply Flexibility and Portfolio Choice in a Life Cycle Model," *Journal of Economic Dynamics and Control*, 16, 427–449.
- BONGAERTS, D., F. DE JONG, AND J. DRIESSEN (2017): "An asset pricing approach to liquidity effects in corporate bond markets," *Review of Financial Studies*, 30, 1229–1269.
- BOON, L., M. BRIÈRE, AND S. RIGOT (2018): "Regulation and Pension Fund Risk Taking," Journal of International Money and Finance, forthcoming.
- BOUBAKER, S., D. GOUNOPOULOS, D. NGUYEN, AND N. PALTALIDIS (2017): "Assessing the effects of unconventional monetary policy and low interest rates on pension fund risk incentives," *Journal of Banking and Finance*, 77, 35–52.

- BRENNAN, M. AND A. SUBRAHMANYAM (1996): "Market microstructure and asset pricing: On the compensation for illiquidity in stock returns," *Journal of Financial Economics*, 41, 441–464.
- BROEDERS, D. (2010): "Valuation of contingent pension liabilities and guarantees under sponsor default risk," *Journal of Risk and Insurance*, 77, 911–934.
- BROEDERS, D. AND A. CHEN (2012): "Pension Benefit Security: A Comparison of Solvency Requirements, a Pension Guarantee Fund, and Sponsor Support," *Journal of Risk and Insurance*, 80, 239–272.
- BROEDERS, D., K. JANSEN, AND B. WERKER (2020): "Pension Fund's Illiquid Assets Allocation under Liquidity and Capital Constraints," *Journal of Pension Economics and Finance*, forthcoming.
- BROEDERS, D., A. OORD, AND D. RIJSBERGEN (2016): "Scale economies in pension fund investments: A dissection of investment costs across asset classes," *Journal of International Money and Finance*, 67, 147–171.
- BROEDERS, D. AND M. PRÖPPER (2010): *Pension Fund Risk Management*, Chapman and Hall/CRC, chap. 19.
- BROWN, J. R. AND D. W. WILCOX (2009): "Discounting State and Local Pension Liabilities," *American Economic Review*, 99, 538–542.
- BRUNNERMEIER, M. K. (2009): "Deciphering the Liquidity and Credit Crunch 2007-2008," Journal of Economic Perspectives, 23, 77–100.
- CAMPBELL, J., S. GIGLIO, AND P. PATHAK (2011): "Forced Sales and House Prices," American Economic Review, 101, 2108–2131.
- CAMPBELL, J. AND R. SHILLER (1991): "Yield Spreads and Interest Rate Movements: A Bird's Eye View," *The Review of Economic Studies*, 58, 495–514.
- CAMPBELL, J. AND L. VICEIRA (2002): Strategic Asset Allocation: Portfolio Choice for Long-Term Investors, Oxford University Press.
- CHANG, Y., H. HONG, AND I. LISKOVICH (2015): "Regression Discontinuity and the Price Effects of Stock Market Indexing," *Review of Financial Studies*, 28, 212–246.

- CHEN, L., D. LESMOND, AND J. WEI (2007): "Corporate Yield Spreads and Bond Liquidity," *Journal of Finance*, 62, 119–149.
- CHEN, X., J. HUANG, Z. SUN, T. YAO, AND T. YU (2020): "Liquidity Premium in the Eye of the Beholder: An Analysis of the Clientele Effect in the Corporate Bond Market," *Management Science*, 66, 932–957.
- COCHRANE, J. AND M. PIAZZESI (2005): "Bond Risk Premia," American Economic Review, 95, 138–160.
- COMMONWEALTH PARTNERS (2017): The Evolution of the Canadian Pension Model: Practical Lessons for Building World-class Pension Organizations, World Bank, Washington, DC.
- CONSTANTINIDES, G. (1986): "Capital Market Equilibrium with Transaction Costs," Journal of Political Economy, 94, 842–862.
- COSEMANS, M., R. FREHEN, P. SCHOTMAN, AND R. BAUER (2016): "Estimating Security Betas using Prior Information Based on Firm Fundamentals," *The Review of Financial Studies*, 29, 1072–1112.
- COVAL, J. AND E. STAFFORD (2007): "Asset Fire Sales (and Purchases) in Equity Markets," *Journal of Financial Economics*, 86, 479–512.
- CULBERTSON, J. (1957): "The Term Structure of Interest Rates," *The Quarterly Journal* of *Economics*, 71, 485–517.
- DASGUPTA, A., A. PRAT, AND M. VERARDO (2011): "Institutional Trade Persistence and Long-Term Equity Returns," *The Journal of Finance*, 66, 635–653.
- DE BONDT, W. AND R. THALER (1985): "Does the Stock Market Overreact?" *The Journal* of Finance, 40, 793–805.
- DE HAAN, L. (2018): "Recovery measures of underfunded pension funds: higher contributions, no indexation, or pension cuts?" Journal of Pension Economics and Finance, 17, 437–468.
- DECHERT AND PREQIN (2011): "Transaction and Monitoring Fees: On the Rebound?" https://docs.preqin.com/reports/Dechert_Preqin_Transaction_and_ Monitoring_Fees.pdf.

- DEL GUERCIO, D. AND P. TKAC (2002): "The Determinants of the Flow of Funds of Managed Portfolios: Mutual Funds vs. Pension Funds," *The Journal of Financial and Quantitative Analysis*, 37, 523–557.
- DIAMOND, D. W. AND R. G. RAJAN (2011): "Fear of Fire Sales, Illiquidity Seeking, and Credit Freezes," *The Quarterly Journal of Economics*, 126, 557–591.
- DIMMOCK, S., N. WANG, AND J. YANG (2019): "The endowment model and modern portfolio theory," NBER Working Paper No. w25559.
- DOMANSKI, D., H. SHIN, AND V. SUSHKO (2017): "The Hunt for Duration: Not Waving but Drowning?" *IMF Economic Review*, 65, 113–153.
- DRIESSEN, J. AND F. DE JONG (2015): "Can Large Long-term Investors Capture Illiquidity Premiums?" *Bankers, Markets and Investors*, 134.
- DRIESSEN, J. AND R. XING (2017): "Myopic or Dynamic Liquidity Management? A Study of Hedge Funds around the 2008 Financial Crisis," Available at SSRN: https://ssrn.com/abstract=2688567 or http://dx.doi.org/10.2139/ssrn.2688567.
- DUFFIE, D. AND A. ZIEGLER (2003): "Liquidation Risk," *Financial Analysts Journal*, 59, 42–51.
- DYCK, A. AND L. POMORSKI (2016): "Investor Scale and Performance in Private Equity Investments," *Review of Finance*, 20, 1081–1106.
- EASLEY, D. AND M. O'HARA (1987): "Price, Trade Size, and Information in Securities Markets," *Journal of Financial Economics*, 19, 69–90.
- ECB (2010): "The "Centralised Securities Database" in Brief," https://www.ecb.europa. eu/pub/pdf/other/centralisedsecuritiesdatabase201002en.pdf.
- EDELEN, R., O. INCE, AND G. KADLEC (2016): "Institutional Investors and Stock Return Anomalies," *Journal of Financial Economics*, 199, 472–488.
- ELLUL, A., C. JOTIKASTHIRA, AND C. LUNDBLAD (2011): "Regulatory pressure and fire sales in the corporate bond market," *Journal of Financial Economics*, 101, 596–620.
- ELTON, E., M. GRUBER, S. BROWN, AND W. GOETZMANN (2003): Modern Portfolio Theory and Investment Analysis, Wiley, Hoboken, NJ.

216

- FED (2020): "Z1. Financial Accounts of the United States Fourth Quarter 2019," Flow of Funds, Balance Sheets, and Integrated Macroeconomic Accounts.
- FISHER, J. D., D. M. GELTNER, AND R. B. WEBB (1994): "Value indices of commercial real estate: A comparison of index construction methods," *The Journal of Real Estate Finance and Economics*, 9, 137–164.
- FRANZONI, F., E. NOWAK, AND L. PHALIPPOU (2012): "Private Equity Performance and Liquidity Risk," *Journal of Finance*, 67, 2341–2373.
- FRAZZINI, A. AND L. H. PEDERSEN (2014): "Betting Against Beta," Journal of Financial Economics, 111, 1–25.
- GÂRLEANU, N. (2009): "Portfolio choice and pricing in illiquid markets," *Journal of Economic Theory*, 144, 532–564.
- GERARD, M. (2019): "Reform Options for Mature Defined Benefit Pension Plans: The Case of the Netherlands," IMF Working Paper No. 19/22.
- GOYAL, A. AND S. WAHAL (2008): "The Selection and Termination of Investment Management Firms by Plan Sponsors," *The Journal of Finance*, 63, 1805–1847.
- GREENWOOD, R., S. HANSON, AND G. LIAO (2018): "Asset Price Dynamics in Partially Segmented Markets," *The Review of Financial Studies*, 31, 3307–3343.
- GREENWOOD, R. AND D. VAYANOS (2010): "Price Pressure in the Government Bond Market," American Economic Review: Papers and Proceedings, 100, 585–590.
- (2014): "Bond Supply and Excess Bond Returns," *Review of Financial Studies*, 27, 663–713.
- GREENWOOD, R. AND A. VISSING-JORGENSEN (2018): "The Impact of Pensions and Insurance on Global Yield Curves," Harvard Business School - Working Paper 18-109.
- GUIBAUD, S., Y. NOSBUSCH, AND D. VAYANOS (2013): "Bond Market Clienteles, the Yield Curve, and the Optimal Maturity Structure of Government Debt," *The Review of Financial Studies*, 26, 1914–1961.
- GUTIERREZ, R. AND E. KELLEY (2009): "Institutional Herding and Future Stock Returns," Unpublished working paper. University of Oregon and University of Arizona.

- HANSEN, K. (1998): "Seasonality of moves and duration of residence." Tech. rep., Bureau of Census, U.S. Department of Commerce.
- HANSON, S. (2014): "Mortgage Convexity," Journal of Financial Economics, 113, 270–299.
- HARDIN III, W., K. LIANO, AND G. HUANG (2002): "The Ex-Dividend Pricing of REITs," *Real Estate Economics*, 30, 533–549.
- HARRIS, R., T. JENKINSON, AND S. KAPLAN (2014): "Private equity performance: What do we know?" *Journal of Finance*, 69, 1851–1882.
- HE, Z. AND A. KRISHNAMURTHY (2013): "Intermediary Asset Pricing," *American Economic Review*, 103, 732–770.
- HOESLI, M., J. LEKANDER, AND W. WITKIEWICZ (2004): "International Evidence on Real Estate as a Portfolio Diversifier." *Journal of Real Estate Research*, 26, 161–206.
- HOEVENAARS, R., R. MOLENAAR, P. SCHOTMAN, AND T. STEENKAMP (2008): "Strategic asset allocation with liabilities: Beyond stocks and bonds," *Journal of Economic Dynamics and Control*, 32, 2939–2970.
- HUANG, M. (2003): "Liquidity shocks and equilibrium liquidity premia," *Journal of Economic Theory*, 109, 104–129.
- JACOBS, H., S. MÜLLER, AND M. WEBER (2014): "How should individual investors diversify? An empirical evaluation of alternative asset allocation policies," *Journal of Financial Markets*, 19, 62–85.
- JANG, B., H. KOO, H. LIU, AND M. LOEWENSTEIN (2007): "Liquidity Premia and Transaction Costs," *Journal of Finance*, 62, 2329–2366.
- JEGADEESH, N. AND S. TITMAN (1993): "Returns to Buying Winners and Selling Losers: Implications for Stock Market Efficiency," *The Journal of Finance*, 48, 65–91.
- JIN, L., R. MERTON, AND Z. BODIE (2006): "Do a Firm's Equity Returns Reflect the Risk of Its Pension Plans?" *Journal of Financial Economics*, 81, 1–26.
- KAHL, M., J. LIU, AND F. LONGSTAFF (2003): "Paper millionaires: how valuable is stock to a stockholder who is restricted from selling it?" *Journal of Financial Economics*, 67, 385–410.

- KALLBERG, J. G., C. H. LIU, AND D. W. GREIG (1996): "The Role of Real Estate in the Portfolio Allocation Process." *Real Estate Economics*, 24, 359–377.
- KHAN, M., H. SCHEULE, AND E. WU (2017): "Funding liquidity and bank risk taking," Journal of Banking and Finance, 82, 203–216.
- KLINGER, S. AND S. SUNDARESAN (2018): "An Explanation of Negative Swap Spreads: Demand for Duration from Underfunded Pension Plans," *The Journal of Finance*, forthcoming.
- KOIJEN, R. AND M. YOGO (2015): "The Cost of Financial Frictions for Life Insurers," American Economic Review, 105, 445–475.
- (2019): "A Demand System Approach to Asset Pricing," Journal of Political Economy, 127, 1475–1515.
- KOIJEN, R. S., F. KOULISCHER, B. NGUYEN, AND M. YOGO (2017): "Euro-Area Quantitative Easing and Portfolio Rebalancing," *American Economic Review*, 107, 621– 627.
- (2020): "Inspecting the Mechanism of Quantitative Easing in the Euro Area," *Journal of Financial Economics*.
- KOIJEN, R. S., T. J. MOSKOWITZ, L. H. PEDERSEN, AND E. B. VRUGT (2018): "Carry," Journal of Financial Economics, 127, 197–225.
- LAKONISHOK, J., A. SHLEIFER, AND R. VISHNY (1992): "The Structure and Performance of the Money Management Industry," *Brookings Papers on Economic Activity*, 339–391.
- LONGSTAFF, F. (2003): "Portfolio Claustrophobia: Asset Pricing in Markets with Illiquid Assets," *American Economic Review*, 99, 1119–1144.
- LU, L., M. PRITSKER, A. ZLATE, K. ANADU, AND J. BOHN (2019): "Reach for Yield by U.S. Public Pension Funds," FRB Boston Risk and Policy Analysis Unit Paper No. RPA 19-2.
- LUCAS, D. J. AND S. P. ZELDES (2009): "How Should Public Pension Plans Invest?" American Economic Review, 99, 527–32.

- LUO, C., E. RAVINA, AND L. VICEIRA (2020): "Retail Investors' Contrarian Behavior Around News and the Momentum Effect," Available at SSRN: https://ssrn.com/abstract=3544949.
- LYNCH, A. AND S. TAN (2011): "Explaining the Magnitude of Liquidity Premia: The Roles of Return Predictability, Wealth Shocks, and State-Dependent Transaction Costs," *Journal of Finance*, 66, 1329–1368.
- MAGGIORI, M., B. NEIMAN, AND J. SCHREGER (2020): "International Currencies and Capital Allocation," *Journal of Political Economy*, 128, 2019–2066.
- MANCONI, A., M. MASSA, AND A. YASUDA (2012): "The role of institutional investors in propagating the crisis of 2007-2008," *Journal of Financial Economics*, 104, 491–518.
- MARKOWITZ, H. (1952): "Portfolio Selection," The Journal of Finance, 7, 77-91.
- MERTON, R. (1969): "Lifetime Portfolio Selection under Uncertainty: The Continuous-Time Case," The Review of Economics and Statistics, 51, 247–257.
- (1980): "On Estimating the Expected Return on the Market: An Exploratory Investigation," *Journal of Financial Economics*, 8, 323–361.
- METRICK, A. AND A. YASUDA (2010): "The economics of private equity funds," *Review* of Financial Studies, 23, 2303–2341.
- MILLER, N., L. PENG, AND M. SKLARZ (2011): "The economic impact of anticipated house price changes Evidence from home sales," *Real Estate Economics*, 39, 345–378.
- MITCHELL, M., L. PEDERSEN, AND T. PULVINO (2007): "Slow moving capital," American Economic Review, 97, 215–220.
- MODIGLIANI, F. AND R. SUTCH (1966): "Innovations in Interest Rate Policy," *The American Economic Review*, 56, 178–197.
- NEWEY, W. AND K. WEST (1987): "A Simple, Positive Semi-Definite, Heteroskedasticity and Autocorrelation Consistent Covariance Matrix," *Econometrica*, 55, 703–708.
- NIEDRIG, T. (2015): "Optimal asset allocation for interconnected life insurers in the low interest rate environment under solvency regulation," *Journal of Insurance Issues*, 38, 31–71.

- NOVY-MARX, R. AND J. RAUH (2009): "The Liabilities and Risks of State-Sponsored Pension Plans," *Journal of Economic Perspectives*, 23, 191–210.
- OECD (2014): "Institutional Investors and Long-term Investment," Project Report May 2014: http://www.oecd.org/finance/OECD-LTI-project.pdf.
- (2019): "Pension Markets in Focus," https://www.oecd.org/daf/fin/ private-pensions/Pension-Markets-in-Focus-2019.pdf.
- OMMEREN, J. (2008): "Transaction Costs in Housing Markets," Tinbergen Institute Discussion Paper: TI 2008-099/3.
- PALHARES, D. AND S. RICHARDSON (2019): "(II)liquidity premium in credit markets: A myth?" Journal of Fixed Income, 28, 5–23.
- PASTOR, L. AND R. STAMBAUGH (2003): "Liquidity Risk and Expected Stock Returns," Journal of Political Economy, 111, 642–685.
- PENG, J. AND Q. WANG (2019): "Alternative investments: Is it a solution to the funding shortage of US public pension plans?" *Journal of Pension Economics and Finance*, forthcoming.
- PHALIPPOU, L. AND O. GOTTSCHALG (2009): "The Performance of Private Equity Funds," *The Review of Financial Studies*, 22, 1747–1776.
- PULVINO, T. (1998): "Do asset fire sales exist? An empirical investigation of commercial aircraft transactions," *Journal of Finance*, 53, 939–978.
- PwC (2016): "Global Pension Funds: Best practices in the pension funds investment process," .
- QIAN, W. AND P. LIU (2012): "Does (and What) Illiquidity Matter for Real Estate Prices? Measure and Evidence," SSRN Electronic Journal.
- RAJAN, R. G. (2006): "Has Finance Made the World Riskier?" European Financial Management, 12, 499–533.
- RAUH, J. (2009): "Risk shifting versus risk management: investment policy in corporate pension plans," *Review of Financial Studies*, 22, 2687–2733.

- Ross, S. A. (1976): "The Arbitrage Theory of Capital Asset Pricing," Journal of Economic Theory, 13, 341–360.
- SADKA, R. (2010): "Liquidity risk and the cross-section of hedge-fund returns," *Journal of Financial Economics*, 98, 54–71.
- SCHOLES, M. (2000): "Crisis and Risk Management," *American Economic Review*, 90, 17–21, 12633, National Bureau of Economic Research, Cambridge, MA.
- SEN, I. (2019): "Regulatory Limits to Risk Management," Job Market Paper.
- SHARPE, W. (1976): "Corporate pension funding policy," *Journal of Financial Economics*, 3, 183–193.
- SHARPE, W. F. AND L. G. TINT (1990a): "Liabilities— A New Approach," *The Journal* of Portfolio Management, 16, 5–10.
- (1990b): "Liabilities— A New Approach," *The Journal of Portfolio Management*, 16, 5–10.
- SHUMWAY, T., M. SZEFLER, AND K. YUAN (2011): "The Information Content of Revealed Beliefs in Portfolio Holdings," Working paper.
- SIAS, R. (2004): "Institutional Herding," Review of Financial Studies, 17, 165–206.
- STAMBAUGH, R. (2014): "Presidential Address: Investment Noise and Trends," The Journal of Finance, 69, 1415–1453.
- STOCK, J. H. AND M. YOGO (2005): "Testing for Weak Instruments in Linear IV Regression," in *Identification and Inference for Econometric Models: Essays in Honor* of Thomas Rothenberg, Cambridge: Cambridge Univ. Press., chap. 5.
- STOUGHTON, N. AND J. ZECHNER (2011): "Intermediated investment management," *The Journal of Finance*, 66, 779–805.
- TIMMER, Y. (2018): "Cyclical investment behavior across financial institutions," *Journal* of Financial Economics, 129, 268–286.
- TONKS, I. (2005): "Performance Persistence of Pension-Fund Managers," Journal of Business, 78, 1917–1942.

- VASICEK, O. (1973): "A Note on Using Cross-Sectional Information in Bayesian Estimation of Security Betas," *The Journal of Finance*, 28, 1233–1239.
- VAYANOS, D. (1998): "Transaction Costs and Asset Prices: A Dynamic Equilibrium Model," *Review of Financial Studies*, 11, 1–58.
- VAYANOS, D. AND J. VILA (2009): "A Preferred-Habitat Model of the Term Structure of Interest Rates," Working Paper London School of Economics.
- VAYANOS, D. AND J. WANG (2013): "Chapter 19 Market Liquidity Theory and Empirical Evidence," Elsevier, vol. 2 of Handbook of the Economics of Finance, 1289– 1361.
- VERONESI, P. (2010): *Fixed Income Securities*, Valuation, Risk, and Risk Management, John Wiley and Sons, INC.
- WATSON, W. T. (2019): "The case for illiquid investments," https://rdweb.campus.uvt. nl/RDWeb/webclient/.
- WILLIS TOWERS WATSON (2019): "Global Pension Assets Study 2018," https://www.willistowerswatson.com/en-CA/insights/2019/02/global-pension-assetsstudy-2019.
- YELLEN, J. (2011): "Remarks the International Conference: Real at Policy," and Financial Linkage and Monetary Bank of Japan, http://www.federalreserve.gov/newsevents/speech/yellen20110601a.htm.

CENTER DISSERTATION SERIES

CentER for Economic Research, Tilburg University, the Netherlands

No.	Author	Title	ISBN	Published
579	Julius Rüschenpöhler	Behavioural Perspectives on Subsistence Entrepreneurship in Emerging Markets	978 90 5668 580 5	January 2019
580	Khulan Altangerel	Essays on Immigration Policy	978 90 5668 581 2	January 2019
581	Kun Zheng	Essays on Duration Analysis and Labour Economics	978 90 5668 582 9	January 2019
582	Tatiana Zabara	Evolution of Entrepreneurial Teams in Technology-Based New Ventures	978 90 5668 583 6	February 2019
583	Yifan Yu	Essays on Mixed Hitting-Time Models	978 90 5668 584 3	April 2019
584	Daniel Martinez Martin	Unpacking Product Modularity, Innovation in Distributed Innovation Teams	978 90 5668 585 0	April 2019
585	Katalin Katona	Managed Competition in Practice Lessons for Healthcare Policy	978 90 5668 586 7	April 2019
586	Serhan Sadikoglu	Essays in Econometric Theory	978 90 5668 587 4	May 2019
587	Hoang Yen Nguyen	Emotions and Strategic Interactions	978 90 5668 588 1	May 2019
588	Ties de Kok	Essays on reporting and information processing	978 90 5668 589 8	May 2019
589	Yusiyu Wang	Regulation, Protest, and Spatial Economics	978 90 5668 590 4	June 2019
590	Ekaterina Neretina	Essays in Corporate Finance, Political Economy, and Competition	978 90 5668 591 1	June 2019
591	Ruth Wandhöfer	Technology innovation in Financial Markets: Implications for Money, Payments and Settlement Finality	978 90 5668 592 8	June 2019

No.	Author	Title	ISBN	Published
592	Andinet Worku Gebreselassie	On communicating about taboo social issues in least developed countries: The case of Ethiopia	978 90 5668 593 5	June 2019
593	Filip Bekjarovski	Active Investing	978 90 5668 594 2	June 2019
594	Miguel Sarmiento	Essays on Banking, Financial Intermediation and Financial Markets	978 90 5668 595 9	June 2019
595	Xiaoyin Ma	Essays on Alternative Investements	978 90 5668 596 6	June 2019
596	Victor van Pelt	A Dynamic View of Management Accounting Systems	978 90 5668 597 3	June 2019
597	Shuai Chen	Marriage, Minorities, and Mass Movements	978 90 5668 598 0	July 2019
598	Ben Gans	Stabilisation operations as complex systems: order and chaos in the interoperability continuum	978 90 5668 599 7	July 2019
599	Mulu Hundera	Role Conflict, Coping Strategies and Female Entrepreneurial Success in Sub- Saharan Africa	978 90 5668 600 0	August 2019
600	Hao Hu	The Quadratic Shortest Path Problem – Theory and Computations	978 90 5668 601 7	September 2019
601	Emerson Erik Schmitz	Essays on Banking and International Trade	978 90 5668 602 4	September 2019
602	Olga Kuryatnikova	The many faces of positivity to approximate structured optimization problems	978 90 5668 603 1	September 2019
603	Sander Gribling	Applications of optimization to factorization ranks and quantum information theory	978 90 5668 604 8	September 2019
604	Camille Hebert	Essays on Corporate Ownership and Human Capital	978 90 5668 605 5	October 2019
605	Gabor Neszveda	Essays on Behavioral Finance	978 90 5668 606 2	October 2019

No.	Author	Title	ISBN	Published
606	Ad van Geesbergen	Duurzame schaarste - Een kritische analyse van twee economische duurzaamheids-paradigma's geïnspireerd door de filosofie van Dooyeweerd	978 90 5668 607 9	October 2019
607	Richard T. Mason	Digital Enrollment Architecture and Retirement Savings Decisions: Evidence from the Field	978 90 5668 608 6	November 2019
608	Ron Triepels	Anomaly Detection in the Shipping and Banking Industry	978 90 5668 609 3	November 2019
609	Feng Fang	When performance shortfall arises, contract or trust? A multi-method study of the impact of contractual and relation governances on performance in Public- Private Partnerships	978 90 5668 610 9	November 2019
610	Yasir Dewan	Corporate Crime and Punishment: The Role of Status and Ideology	978 90 5668 611 6	November 2019
611	Mart van Hulten	Aiming for Well-Being through Taxation: A Framework of Caution and Restraint for States	978 90 5668 612 3	December 2019
612	Carlos Sandoval Moreno	Three essays on poverty measurement and risk protection	978 90 5668 613 0	December 2019
613	Harmke de Groot	Core strength or Achilles' heel: Organizational competencies and the performance of R&D collaborations	978 90 5668 614 7	December 2019
614	Peter Brok	Essays in Corporate Finance and Corporate Taxation	978 90 5668 615 4	December 2019
615	Pascal Böni	On the Pricing, Wealth Effects and Return of Private Market Debt	978 90 5668 616 1	December 2019
616	Ana Martinovici	Revealing Attention: How Eye Movements Predict Brand Choice and Moment of Choice	978 90 5668 617 8	December 2019
617	Matjaz Maletic	Essays on international finance and empirical asset pricing	978 90 5668 618 5	January 2020
618	Zilong Niu	Essays on Asset Pricing and International Finance	978 90 5668 619 2	January 2020

No.	Author	Title	ISBN	Published
619	Bjorn Lous	On free markets, income inequality, happiness and trust	978 90 5668 620 8	January 2020
620	Clemens Fiedler	Innovation in the Digital Age: Competition, Cooperation, and Standardization	978 90 5668 621 5	October 2020
621	Andreea Popescu	Essays in Asset Pricing and Auctions	978 90 5668 622 2	June 2020
622	Miranda Stienstra	The Determinants and Performance Implications of Alliance Partner Acquisition	978 90 5668 623 9	June 2020
623	Lei Lei	Essays on Labor and Family Economics in China	978 90 5668 624 6	May 2020
624	Farah Arshad	Performance Management Systems in Modern Organizations	978 90 5668 625 3	June 2020
625	Yi Zhang	Topics in Economics of Labor, Health, and Education	978 90 5668 626 0	June 2020
626	Emiel Jerphanion	Essays in Economic and Financial decisions of Households	978 90 5668 627 7	July 2020
627	Richard Heuver	Applications of liquidity risk discovery using financial market infrastructures transaction archives	978 90 5668 628 4	September 2020
628	Mohammad Nasir Nasiri	Essays on the Impact of Different Forms of Collaborative R&D on Innovation and Technological Change	978 90 5668 629 1	August 2020
629	Dorothee Hillrichs	On inequality and international trade	978 90 5668 630 7	September 2020
630	Roland van de Kerkhof	It's about time: Managing implementation dynamics of condition- based maintenance	978 90 5668 631 4	October 2020
631	Constant Pieters	Process Analysis for Marketing Research	978 90 5668 632 1	December 2020
632	Richard Jaimes	Essays in Macroeconomic Theory and Natural Resources	978 90 5668 633 8	November 2020

No.	Author	Title	ISBN	Published
633	Olivier David Armand Zerbib	Asset pricing and impact investing with pro-environmental preferences	978 90 5668 634 5	November 2020
634	Laura Capera Romero	Essays on Competition, Regulation and Innovation in the Banking Industry	978 90 5668 635 2	December 2020
635	Elisabeth Beusch	Essays on the Self-Employed in the Netherlands and Europe	978 90 5668 636 9	December 2020
636	Sophie Zhou	Essays on the Self-Employed in the Netherlands and Europe	978 90 5668 637 6	November 2020
637	Vincent Peters	Turning modularity upside down: Patient-centered Down syndrome care from a service modularity perspective	978 90 5668 638 3	December 2020
638	Pranav Desai	Essays in Corporate Finance and Innovation	978 90 5668 639 0	January 2021
639	Kristy Jansen	Essays on Institutional Investors, Asset Allocation Decisions, and Asset Prices	978 90 5668 640 6	January 2021

This dissertation is a collection of five independent chapters that aim to better understand the investment decisions of institutional investors. Summarizing, this dissertation has three main findings. First, regulation drives a large part of the investment decisions of institutional investors. Second, their investment decisions matter for asset prices. And third, their investment decisions have important welfare implications for the investors where we ultimately care about, namely, the households.

KRISTY A.E. JANSEN obtained her Bachelor degree in Econometrics, Master degree in Quantitative Finance and Actuarial Science, and Research Master degree in Finance from Tilburg University. After obtaining her Master degrees in 2016, Kristy started her PhD under the supervision of prof. dr. Bas J.M. Werker, prof. dr. Theo E. Nijman, and prof. dr. Dirk W.G.A. Broeders at the Finance department of Tilburg University. Throughout her PhD she was a research affiliate of De Nederlandsche Bank. Since September 2020, she is a research fellow of Tilburg University and a data scientist at De Nederlandsche Bank.

ISBN: 978 90 5668 640 6 DOI: 10.26116/center-lis-2019

